MOLECULAR DIVERSITY IN THE LEISHMANIA SUBGENUS VIANNIA

A thesis submitted for the degree of Doctor of Philosophy of the University of London

June 2000

Deborah Nolder BSc, MSc

Department of Infectious and Tropical Diseases, London School of Hygiene and Tropical Medicine, London, United Kingdom.



VARIABLE PRINT QUALITY

on some figures



NUMEROUS ORIGINALS IN COLOUR



ABSTRACT

The aim of this work was to provide further insight into the relationships between species of the *Leishmania* subgenus *Viannia* by examining inter- and intra-specific genetic diversity.

Diversity among *Viannia* strains and stocks was investigated using biochemical and molecular techniques. Isoenzyme and microsatellite analyses were found to be the most discriminatory and generated results which could be interpreted genotypically. These techniques were used to study diversity in *Viannia* populations consisting of i) Brazilian *L. V. braziliensis* stocks, ii) Nicaraguan stocks belonging to different *Viannia* species complexes, including putative species-complex hybrids, and iii) uncharacterized stocks from a new epidemic focus in Huanuco, Peru.

IEA identified all stocks to at least the species level. Microsatellite analyses using Genescan Φ / Genotyper Φ and direct sequencing were found to be more discriminatory than IEA for all populations. The application of Genescan Φ / Genotyper Φ to the identification of alleles at these microsatellite loci has not been described previously. Phylogenetic analysis was carried out for each population using enzyme and microsatellite sequence data: phylogenies constructed from multilocus enzyme data were most accurate.

Putative species heterozygotes between L. V. braziliensis / L. V. panamensis and L. V. braziliensis / L. V. peruviana were identified in the Nicaraguan and Huánuco populations, respectively, using IEA. Microsatellite analysis identified heterozygous stocks in all 3 populations. This approach also supported the hybrid status of the Nicaraguan and Huánuco stocks. Population genetic analysis of stocks from Huánuco provided statistical evidence for a limited degree of genetic recombination between stocks in this population. Results indicated, however, that clonal expansion was the predominant mode of replication.

To explore the possibility of the occurrence of genetic recombination between species, genetic transformation experiments were initiated using putative parental stocks from the Nicaraguan population

ACKNOWLEDGEMENTS

At LSHTM, I would like to thank Michael Miles for his support and supervision, Russell Stothard, Iain Frame, Isabel Mauricio, Clive Davies, Richard Reithinger, Roslin Russell, Martin Taylor, Cally Roper, David Conway and Michael Gaunt for informed discussions, help and friendship during my PhD studies. I would also like to thank Rachel Gregory for her help in pouring numerous Genescan® gels and for troubleshooting the sequencing machine out of hours! Thanks are also due to Laila Al-Rustamani, David Evans, Patricia Mayrink and Sueli Vieira. I would also like to thank Jim Ajioka and his group at Cambridge for allowing me to use the *Viannia* microsatellite protocols, and for access to their data, before publication.

I would like to thank Clive Davies, Alejandro Llanos-Cuentas, Norma Roncal, Alex Belli and César Cuba-Cuba for generously providing stocks and also the many couriers who brought the Huánuco stocks from Peru. For help and friendship during my visit to Brazil I would particularly like to thank Diarmid & Leisa Campbell-Lendrum, César Cuba-Cuba (and family), Marinete Povoa (and family), Toby Barrett, Roberto and Marie-Cleide Naiff (and family).

I am indebted to the Sir Halley Stewart Trust and Michael Miles for funding my PhD and to the Wellcome Foundation, Clive Davies and Michael Miles for supporting my visit to Brazil.

Finally, extra-special thanks are due to my family and many friends who kept me going through the bad (and not so bad) times! Thanks are due especially to Jan, Mum and Dad, Sue and Gary, Pete & Tess, Anne, Libby & Milan, Helen, Dawn, Khai, Rhonda, Kathi, Steph, Ros and Steve.

This thesis is dedicated to the memory of my grandparents, in particular to that of "Big Grandad", Stan Henderson, who died while I was writing-up.

TABLE OF CONTENTS

Title		i
Abstr	act	ii
Ackne	owledgments	iii
Table	of contents	iv
List o	f tables	xi
List o	f figures	diii
Listo	fahhreviations	cvi
2154 0		
Chan	ter 1: Introduction	1
11:	Biology and life cycle of Leishmania	1
1.2	Basic molecular biology of Leishmania	1
1.3.	Classification of the Leishmaniae	3
1.5.	131: Subgenus Leishmania	4
	132: Subgenus Viannia	4
1 4	Vectors and reservoir hosts of Leishmania	6
A	1 4 1: Vectors of Leishmania	6
	142. Reservoir bosts of Leishmania	7
	1 4 2 1: Domestic and peridomestic reservoir hosts	8
	1 4 2 2: Sulvetic reservoir hosts	8
	1 4 3 Transmission cycles	11
15.	Leichmaniasis	12
1.5.	1.5.1. Cutaneous leishmaniasis	12
	1.5.2: Mucocutaneous leishmaniasis	12
	1.5.2. Viccord leichmaniasis	15
	1.5.4. Other clinical presentation	16
	1.5.4. 1. Diffuse outaneous leichmaniasis	16
	1.5.4.7: Dort Kala agar dermal leichmaniasis	16
	1.5.4.2. FOST Kala-azal ucilitat leisinnainasis	16
16.	Dethogenesis and Immunology	17
1.0.	1.6.1. Constin completion of Leichmanic infection	17
	1.6.1: Generic regulation of Leisnmania infection	17
	1.6.2: Parasite factors involved in disease expression	10
	1.6.5: Establishment of infection	10
	1.0.4: Immune response of the nost	19
	1.0.5: Infection and clinical disease	20
	1.6.5.1: Asymptomatic infection	20
	1.6.5.2. CL and MCL	21
	1.6.5.3: DCL	21
		21
	1.0.0. Resolution of Disease	22
	1.0.0.1. Acquired immunity	22
	1.0.0.2: Parasite persistence and metastasis	23
	1.6.6.3. Recurrent infection	24
1.7:	Diagnosis of Leishmania infection	25
	1.7.1: Clinical diagnosis	25

	1.7.2: Parasitological diagnosis	25
	1.7.3: Serological diagnosis	26
	1.7.3.1: Skin testing	26
	1732: Antibody detection methods	27
	174 Molecular diagnostic methods	28
1 8.	Treatment of leichmaniasis	30
1.0.	1 8 1: Anti-leichmanial druge	30
	1.9.2. Treatment of CI	21
	1.8.2. Treatment of CL	22
	1.8.5: Treatment of MCL	32
	1.8.4: I featment of VL	33
1.9:	Control of leishmaniasis	33
	1.9.1: Case detection and treatment	34
	1.9.2: Environmental management	34
	1.9.3: Vector control	34
	1.9.4: Reservoir control	35
	1.9.5: Vaccination	36
	1.9.5.1: Leishmanization	37
	1.9.5.2: Killed whole promastigote vaccines	37
	1.9.5.3; Attenuated vaccines	37
	1.9.5.4: Recombinant vaccines	38
	1.9.5.5: DNA vaccines	38
1.10:	Molecular analysis of genetic variation	39
	1 10 1. Diversity in genome organization	40
	1 10 1 1. Molecular karvotyne analysis	40
	1.10.1.2: Restriction fragment length polymorphism analysis	42
	1.10.1.2. Kinetoplast DNA and sobizodeme analysis	43
	1.10.2. DNA accurace diversity	43
	1.10.2. DNA sequence diversity	44
	1.10.2.1: Isoenzyme analysis and isoelectric focusing	44
	1.10.2.2: Monocional antibody typing	40
	1.10.2.3: PCR-RFLP analysis	40
	1.10.2.4: Random amplified polymorphic DNA analysis	47
	1.10.2.5: Microsatellite analysis	49
	1.10.2.6: Mutation detection methods	54
	1.10.2.7: DNA sequencing	55
1.11:	Genetic transformation	56
1.12:	Population structure and genetic exchange	56
	1.12.1: Genetic exchange in trypanosomatids	56
	1.12.2: Implications of genetic exchange	59
1.13:	Aims and objectives of the project	60
Chapt	ter 2: Materials & Methods	63
2.1:	Leishmania strains and stocks	63
	2.1.1: Reference strains	63
	212. Leishmania stocks	63
	2121: Stocks from Brazil	63
	2122 Stocks from Nicaragua	63
	2.1.2.2. Stocks from Peru	64
	a.t.a.a. Juora humi tuu	~

2.2:	Cultivation of Leishmania	64
	2.2.1: Solid media for cultivation of Leishmania promastigotes	64
	2.2.2: Liquid media for cultivation of Leishmania promastigotes	64
2.3:	Preparation of Leishmania enzyme lysates	64
2.4:	Preparation of genomic DNA	65
2.5:	Isoenzyme analysis (IEA)	65
	2.5.1: Thin-layer starch-gel electrophoresis (TSGE)	65
	2.5.2: Cellulose acetate electrophoresis (CAE)	66
2.6:	Isoelectric focusing (IEF)	67
2.7:	Random amplified polymorphic DNA (RAPD) analysis	67
	2.7.1: Primers	67
	2.7.2: PCR amplification	67
	2.7.3: Electrophoresis of RAPD PCR products	68
	2.7.4: Interpretation of RAPD results	68
2.8:	PCR-RFLP analyses: riboprinting and PCR-RFLP of the internal	
	transcribed spacer (ITS)	68
	2.8.1: Primers	69
	2.8.2: PCR reactions	69
	2.8.3: Electrophoretic visualization of the PCR product	70
	2.8.4: Preparation of the PCR product for RFLP analysis	70
	2.8.5: RFLP analysis	70
	2.8.6: Visualization of products from the digestion mix	70
	2.8.7: Interpretation of PCR-RFLP PAGE gels	70
2.9:	Denaturing gradient gel electrophoresis (DGGE)	71
2.10:	Microsatellite analysis	72
	2.10.1: Microsatellite analysis – background	72
	2.10.2: PCR of microsatellite loci for PAGE analysis	72
	2.10.3: Visualization of microsatellite PCR products	72
	2.10.4: PCR of microsatellite loci for Genescan@ and Genotyper@	
	analyses	73
	2.10.5: Genescan@ and Genotyper@ analyses	73
	2.10.6: Sequencing reactions	74
	2.10.7: Sample preparation for automated sequencing	74
	2.10.8: Automated sequencing	74
2.11:	Population genetic analyses	74
	2.11.1: Analyses of isoenzyme data	75
	2.11.1.1: Hardy-Weinberg test	75
	2.11.1.2: Tests for detection of linkage disequilibrium	75
	2.11.2: Analyses of microsatellite data	75
2.12:	Phylogenetic analyses	75
	2.12.1: Phylogenetic analysis of isoenzyme data	75
	2.12.2: Phylogenetic analysis of microsatellite data	76
2.13:	Vector construction	77
	2 13.1: Strategy	77
	2 13.2: Vectors	77
	2.13.2.1: pT3T719U-133P	77
	2.13.2.2: pTEX	78
	2.13.2.3: pTEX-ble	79
	2.13.2.4: Construction of the neomycin resistance vector	79
	2.13.2.5: Construction of the phleomycin resistance vector	79

	2.13.3: Transformation of competent E. coli	79
	2.13.4: Selection of E. coli transformants	79
	2.13.5: Bulk culture of plasmid vector DNA	80
2.14:	Transformation of Leishmania	80
	2.14.1: Electroporation	80
	2.14.2: Selection of Leishmania transformants	80
	2.14.3: Southern analysis of G418-resistant Leishmania	80
Chapt	ter 3: Comparison of reference strains	82
3.0:	Introduction	82
3.1:	Leishmania reference strains	82
3.2:	Isoenzyme analysis	83
3.3:	Isoelectric focusing	91
3.4:	RAPD analysis	92
	3.4.1: Discrimination between Leishmania subgenera	92
	3.4.2: Discrimination within the subgenus Viannia	92
	3.4.3: Subsequent RAPD analyses	95
	3.4.4: Discussion of RAPD	95
3.5:	Riboprinting	96
3.6:	PCR-RFLP analysis of rRNA gene ITS region	99
	3.6.1: PCR amplification of the rRNA gene ITS region	99
	3.6.2: PCR-RFLP analysis	100
	3.6.3: Discussion of PCR-RFLP analysis of the rRNA ITS region	102
3.7:	Denaturing gradient gel electrophoresis (DGGE)	104
	3.7.1: Discussion of DGGE	107
3.8:	Microsatellite analysis	108
	3.8.1: Specificity of the microsatellite loci for the subgenus Viannia	108
	3.8.2: Amplification of microsatellite loci	108
	3.8.3: Genescan@ and Genotyper@ 2.0 analyses	109
	3.8.4: Sequencing of the AC01 microsatellite locus	114
	3.8.5: Microsatellite analysis discussion	119
3.9:	Comparative assessment of techniques used for discriminating	
	between Leishmania reference strains	123
Chan	ter A. Brezil. I. V. braziliansis stocks from different geographical	
<u></u>	areas including Três Bracos / Corte de Pedra	126
41.	Leishmaniasis in Brazil - general introduction	126
4 2.	Leishmaniasis in Três Braços / Corte de Pedra, Brazil	128
43.	Leishmania stocks	131
	4.3.1: Brazilian stocks	131
	4.3.2. Três Bracos / Corte de Pedra stocks	131
44.	Isoenzyme analysis	134
•••••	441: Zymodemes	134
4.5	Microsatellite analysis	139
	4.5.1: PCR of microsatellite loci	139
	4.5.1.1: Stocks with "null alleles"	139
	4.5.1.2: PCR amplification of microsatellite loci	140

	4.5.2:	Non-denaturing PAGE analysis	141
		4.5.2.1: AC01	141
		4.5.2.2: AC16	144
		4.5.2.3: AC52	144
	4.5.3:	Genescan® and Genotyner® 2.0 analyses	146
		453 1: AC01	149
		4532 AC16	150
		4 5 3 3 AC52	152
		453 4' Multilocus analysis of microsatellite data	153
	454.	AC01 Sequencing	156
	4.5.4.	454 1: AC01 sequence variation among Brazilian stocks	158
		4.5.4.1: Comparison between data from sequencing and	150
		Genotymer® analyzes	150
16.	Dhuloa	enetic analysis	161
4.0.	A 6 1	Dhulogenetic analysis	161
	4.0.1	Phylogenetic analysis of isoenzyme data from microsotallito	101
	4.0.2:	Phylogenetic analysis of sequence data from microsatellite	166
	D .	locus ACUI	105
4.7:	Discus	sion of diversity among stocks from Brazil	109
Chant	6. N	licenseries company in the section of L. M. Assertitionals and	
	<u>er 5</u> : N	Icaragua: sympatric transmission of L. V. orazinensis and	171
£ 1.	L. Tanàna da	. V. panamensis	171
5.1;	hirodu		171
5.2:	Nicara	guan Leisnmania stocks	172
5.3:	Isoenzy	me analysis	173
<i>c</i> .	5.5.1:	Zymodemes	1/9
5.4:	RAPD	analysis	182
5.5:	PCR-R	FLP analysis of rRNA gene ITS region	184
5.6:	Micros	atellite analysis	186
	5.6.1:	PCR of microsatellite loci	186
	5.6.2:	Non-denaturing PAGE analysis	186
		5.6.2.1: AC01	186
		5.6.2.2: AC16	188
		5.6.2.3: AC52	189
	5.6.3:	Genescan@ and Genotyper@ 2.0 analyses	190
		5.6.3.1: AC01	190
		5.6.3.2: AC16	193
		5.6.3.3: AC52	193
		5.6.3.4: Multilocus analysis of microsatellite data	194
	5.6.4:	AC01 Sequencing	197
		5.6.4.1: AC01 sequence variation among Nicaraguan stocks	197
		5.6.4.2: Comparison of AC01 data from sequencing and	
		Genotyper®	200
5.7:	Phylos	genetic analysis of Nicaraguan stocks	201
	5.7.1:	Phylogenetic analysis of isoenzyme data	201
	5.7.2:	Phylogenetic analysis of sequence data from microsatellite	202
E 0	D		203
J.8 :	Discus	SSION	207

<u>Chapt</u>	<u>ter 6</u> : B	Iuanuco, Peru: sympatric transmission of L. V. braziliensis	
60.	a Taisha	nd L. V. peruviana	208
0.0:	Leisnmaniasis in Peru		
0.1:	Study	site	209
0.2:	Leisnn	ania reference strains and Huanuco stocks	211
	0.2.1:	Leisnmania reference strains	211
67.	0.2.2:	Huanuco stocks	211
0.3:	1soenz	Delementario enterna	214
	0.3.1:	C 2 1 1 1 M	214
			214
		0.3.1.2; NFIQ	215
		0.3.1.3: Nnl	210
	(22)	0.3.1.4: ES and PEPD	210
	6.3.2:	Zymodemes	218
		6.3.2.1: Zymodemes: associations with nost species, clinical	224
		presentation and geographical origin	224
0.4:	MICTO	satellite analysis	220
	6.4.1:	PCR of microsatellite loci	220
		6.4.1.1: Null alleles	226
		6.4.1.2: Agarose gel electrophoresis	220
	6.4.2:	Non-denaturing PAGE analysis	227
		6.4.2.1: AC01	227
		6.4.2.2: AC16	228
		6.4.2.3: AC52	228
	6.4.3:	Genescan® and Genotyper® 2.0 analyses	230
		6.4.3.1: AC01	230
		6.4.3.2: AC16	234
		6.4.3.3: AC52	237
		6.4.3.4: Multilocus analysis of microsatellite data	239
	6.4.4:	AC01 sequencing	241
		6.4.4.1: Interpretation of sequence data from Huanuco stocks	241
		6.4.4.2: AC01 sequence variation among Huánuco stocks	244
		6.4.4.3: Comparison of AC01 data from sequencing and	
		Genotyper®	250
		6.4.4.4: Comparison of microsatellite and isoenzyme data	251
6.5:	Phylo	genetic analysis	252
	6.5.1:	Phylogenetic analysis of isoenzyme data	252
	6.5.2:	Phylogenetic analysis of sequence data from AC01	257
6.6:	Popul	ation genetic analyses	261
	6.6.1:	Tests for segregation - Hardy-Weinberg equilibrium	261
	6.6.2:	Tests for recombination - linkage disequilibrium	264
		6.6.2.1: Index of Association (I _A)	264
		6.6.2.2: D' and r ² indices	. 267
	6.6.3:	Summary of population genetics analysis	270
6.7:	Discu	ssion of Huánuco population	. 271

Chapt	er 7: Experimental invest	tigation of genetic exchange	277
7.1:	Introduction		277
7.2:	Strategy		278
7.3:	Selection of Leishmania st	trains for transformation	279
7.4:	Construction of transformation	ation vectors	280
	7.4.1: Construction of th	e vector conferring neomycin resistance	280
	7.4.2: Construction of th	e vector conferring phleomycin resistance	283
7.5:	Discussion of genetic tran	sformation and genetic exchange	284
	7.5.1: Vectors: integratio	on by homologous recombination	284
	7.5.2: Crossing experime	ents and potential site of genetic exchange	285
	7.5.3: Models for genetic	c exchange, ploidy	286
	7.5.4: Detection of heter	ozygotes	287
<u>Chap</u>	<u>er 8</u> : Concluding remark	S	288
Refer	ences		294
Anne	dices		321
Anner	dix I: Culture media	*** ** *** *** *** *** *** *** *** *** *** *** *** *** *** *** *** *** *** *** *** *** *** ***	321
Apper	dix II. General reagents		323
	dix III. IFA reagents and	conditions	327
	div IV: PCR primers and	cycling parameters	333
Apper	dix V: Population genetic	c analyses	336

LIST OF TABLES

1.4.1:	Reservoir Host and Vector Species of OW Leishmania	9
1.4.2:	Reservoir Host and Vector Species of NW Leishmania	10
1.5.1:	Clinical manifestations of Leishmania infection	13
1.5.2:	Characteristic lesion type and geographical range of the major	
	parasite species causing cutaneous leishmaniasis	14
2.5.1:	Enzyme systems investigated for discrimination between	
	Leishmania reference strains using IEA (TSGE and CAE) and IEF	66
2.8.1:	Enzymes used for riboprinting and PCR-RFLP analyses	71
3.1:	Leishmania reference strains	82
3.2.1:	Enzyme systems discriminating between Viannia reference strains	83
3.2.2:	Reference strain allele and zymodeme designations	90
3.8.3.1:	Summary of the product (allele) sizes of the Viannia reference	
	strains for each microsatellite locus	114
3.8.4.1:	Summary of information from AC01 sequence data	118
3.8.5.1:	Comparison of data obtained for the AC16 and AC52 loci	121
4.1:	Cutaneous leishmaniasis in Brazil: new cases reported and cases /	
	100,000 inhabitants for 1985 and 1995, by region and by all regions	
	combined	127
4.3.1:	Leishmania Viannia stocks from Brazil	132
4.3.2:	Leishmania Viannia stocks from Três Bracos / Corte de Pedra	133
4.4.1:	Allele and zymodeme designations for I. Brazilian stocks; and	136
	II. Três Bracos / Corte de Pedra stocks	137
4.4.2:	Summary of L. V. braziliensis stock zymodemes, by geographical	
	origin host / vector source and clinical presentation	138
4511:	Summary of microsatellite locus PCR failures for stocks with null	
	alleles	140
4.5.3.1:	Summary of microsatellite allele sizes found in stocks from Brazil.	
	by geographical area	147
4.5.3.2:	Summary of microsatellite allele sizes of L. V. braziliensis stocks	
	from Brazil by host and by clinical type	148
4.5.3.3:	Summary of microsatellite allele sizes of L. V. braziliensis stocks	
	isolated from sand flies from Brazil	149
4.5.3.4:	AC01 genotypes found in stocks from Brazil	150
4535	AC16 genotypes found in stocks from Brazil	151
4.5 3.6	AC52 genotypes found in stocks from Brazil	152
4537	Summary of microdemes observed in Brazilian stocks	154
4538	Microdemes observed in <i>L. V. braziliensis</i> stocks from Brazil	
1.5.5.0.	summarized by geographical region	155
4539	Microdemes observed in <i>LV braziliansis</i> stocks from Brazil	155
1.0.0.9.	summarized by host / vector group	155
52.	Leishmania stocks from Nicaragua	172
5311.	Allele and zumodeme designations for Nicaraguan stocks	180
5312	Summary of Nicaraguan stock zymodemes, by geographical origin	100
J.J.1.2.	and clinical presentation	181

5.6.3:	Microsatellite allele sizes of stocks from Nicaragua by species type,	
	geographical origin and clinical presentation	192
5.6.3.4:	Comparison of microsatellite (Genotyper®) and isoenzyme data, by	
	genotypes observed at each microsatellite locus	195
5.6.3.5:	Summary of microdemes observed in Nicaraguan stocks	196
6.2.2:	Leishmania stocks from Huanuco, Peru	212
6.3.2.1:	Allele and zymodeme designations for Huanuco stocks	223
6.3.2.2:	Summary of Huanuco stock zymodemes by host species, clinical	
	presentation and by geographical region	225
6.4.3.1a:	Comparison of AC01 genotypes observed in Huánuco stocks using	
	Genotyper [®] , by species	231
6.4.3.1b:	AC01 alleles scored in Peruvian stocks and strains	234
6.4.3.2:	Comparison of AC16 genotypes observed in Huánuco stocks using	
	Genotyner® by species	235
643.3	Comparison of AC52 genotypes observed in Huanuco stocks using	
	Genotyper® by species	237
6434.	Summary of microdemes observed in stocks from Huanuco and	
0.1.2.1.	additional / V nerviana reference strains	240
6443	Comparison of the number of dinucleotide repeats scored from	200
0. 1. 1.0.	sequence data with Genotyner® results	251
661	Summary of Hardy-Weinberg exact test results by locus for IFA	
0.0.1.	and microsatellite data from Huánuco stocks	262
6621.	L. values for Huánuco L eishmania stocks and other trupanosomatids	265
6622.1	Summary of linkage disequilibrium results by pairwise locus	205
0.0.2.2.	combination for IEA and microsstellite data from Unanuco stocks	260
671.	Summary of minodemes, microdemes and ACO1 sequence times	20)
0.7.1.	sound from Unionenes, includences and ACOT sequence types	272
672.	Scored from Fluanuco stocks, by species group	215
0.7.2.	summary of nost and chinical origin of the ruanuco stocks, by	274
672.	species group	274
0.7.3:	Summary of genotypes scored from putative L. V. braziliensis	276
17.1.	L. V. peruviana hybrid stocks	270
V-I:	Allele and genotype frequency results based on IEA data from	224
17.3	Huanuco stocks	330
V-2:	Allele and genotype frequency results based on ACUI and ACIO	227
** •	microsatellite data from Huanuco stocks	331
V-3:	Allele frequencies of the whole Huanuco population and of	240
	subgroups therein, by locus	340
V-4:	IA values calculated from isoenzyme and microsatellite data for the	~ ~ ~
	Huanuco stocks	341

LIST OF FIGURES

1.1:	The life cycle of Leishmania	2
1.3:	Classification of the Trypanosomatidae	5
1.10.2.5a	"Identity by state"	53
1.10.2.5b	PCR amplification of a microsatellite locus	54
2.8.1:	Organization of the trypanosomatid rRNA locus	69
2.13.2:	pT3T719U-133P	78
3.2.1:	Isoenzyme banding patterns of Leishmania reference strains	84
3.2.2:	Diagrammatic representation of isoenzyme discrimination between	
	Leishmania reference strains	86
3.3.1:	IEF banding patterns of Leishmania reference strains – Nhi	91
3.4.1:	RAPD profiles generated from reference strains after PCR	
	amplification with primer H4	93
3.4.2:	RAPD profiles generated from reference strains after PCR	
	amplification with primer L3	94
3.4.3:	RAPD profiles generated from reference strains after PCR	
	amplification with primer H2	94
3.5.1:	Riboprint profiles generated from Leishmania reference strains after	
	digestions with TaqI and HpaII	97
3.5.2:	Riboprint profiles generated from Leishmania reference strains after	
	digestions with Sau3AI and RsaI	98
3.6.1:	Amplification products of the rRNA ITS region from Leishmania	
	reference strains	99
3.6.2:	RFLP analysis of rRNA ITS amplification products from	
	Leishmania reference strains	101
3.7.1:	Melting profile of the AluI-digested L. V. braziliensis 18S (SSU)	
	rRNA PCR amplification product generated using perpendicular	
	gradient DGGE	105
3.7.2:	DGGE analysis of AluI-digested 18S (SSU) rRNA gene	
	amplification products from Viannia reference strains	106
3.8.2.1:	Amplification products of the AC01 microsatellite locus from	
	Leishmania reference strains	108
3.8.3.1:	Genescan@ gel image showing microsatellite loci and size standard	
	markers	110
3.8.3.2:	Genotyper® output for the AC01 locus for Viannia reference	
	strains	111
3.8.4.1;	A typical sequencing electropherogram showing data from the	
	AC01 microsatellite locus for L. V. braziliensis, LTB300	116
3.8.4.2:	Sequence alignment of the AC01 locus from reference strains of the	
	subgenus Viannia	117
3.8.5.1:	Genotypers output for the AC16 locus for 4 Viannia reference	
	strains	122
4.2:	Map of region surrounding Três Braços, Bahia State, Brazil	129
4.4.1:	Isoenzyme banding patterns of Leishmania stocks from Brazil	135

4.5.2.1:	AC01 products from Brazilian stocks (PAGE)	142
4.5.2.2:	Formation of heteroduplex molecules	143
4.5.2.3:	AC16 products from Brazilian stocks (PAGE)	145
4.5.2.4:	AC52 products from Brazilian stocks (PAGE)	145
4.5.4.1:	Sequence alignment of the AC01 locus from Brazilian stocks and	
	the L. V. guyanensis reference strain	157
4.6.1.1:	Cladogram showing the results of quartet puzzling for Brazilian	
	stocks and reference strains based on zymodemes	162
4.6.1.2:	UPGMA dendrogram constructed for Brazilian stocks and reference	
	strains based on isoenzyme data	164
4.6.2.1:	Phylogram based on sequence data from the microsatellite locus	
	AC01 for the Brazilian stocks and reference strains	166
5.2:	Map of Nicaragua showing the geographical distribution of stocks	173
5.3.1.1:	Isoenzyme banding patterns of Nicaraguan stocks	174
5.3.1.2:	Diagrammatic representation of IEA banding patterns observed in	
	reference strains and Nicaraguan stocks	176
5.4.1:	RAPD profiles generated from parental and hybrid type Nicaraguan	
	stocks using primers A3, L1 and H2	183
5.5.1:	RFLP analysis of rRNA ITS amplification products from	
	Nicaraguan stocks using BstUI	184
5.6.2.1:	AC01 products from Nicaraguan stocks (PAGE)	187
5.6.2.2:	AC16 products from Nicaraguan stocks (PAGE)	188
5.6.2.3:	AC52 products from Nicaraguan stocks (PAGE)	189
5.6.3.1:	Genotyper® output from the AC01 locus for selected Nicaraguan	
	stocks, illustrating the peaks observed in putative parental and	
	hybrid profiles	191
5.6.4.1:	Sequence alignment of the AC01 locus from Nicaraguan stocks and	
	Viannia reference strains	198
5.7.1.1:	Cladogram showing the results of quartet puzzling for Nicaraguan	
	stocks and reference strains based on zymodemes	202
5.7.1.2:	UPGMA dendrogram constructed for Nicaraguan stocks and	
	reference strains based on isoenzyme data	204
5.7.2.1:	Phylogram based on sequence data from the microsatellite locus	
	AC01 for the Nicaraguan stocks and reference strains	205
6.1:	Map of the Huánuco study site, Huánuco, Peru.	210
6.3.1.1:	MPI phenotypes observed in the Huánuco stocks	215
6.3.1.2:	NHd phenotypes observed among Huánuco stocks	217
6.3.1.3:	NHi 1 / 2 phenotypes observed among Huánuco stocks	217
6.3.1:	Diagrammatic representation of banding patterns observed in	
	reference strains and Huánuco stocks	219
6.4.2.1:	AC01 products from Huánuco stocks (PAGE)	227
6.4.2.2:	AC16 products from Huánuco stocks (PAGE)	229
6.4.2.3;	AC52 products from Huánuco stocks (PAGE)	229
6.4.3.2:	Genotyper® output from the AC01 locus for selected Huánuco	
	stocks, illustrating the peaks observed in putative parental and	
	hybrid profiles	232
6.4.4.1:	Sequence alignment of the AC01 locus from Huánuco stocks and	
	Viannia reference strains	242

List of figures

6.4.4.2:	Dinucleotide repeat region illustrating the overlapping sequence reads in stocks with 2 distinct alleles	246
6.4.4.3:	Expected sequences of stocks heterozygous at the AC01 locus for	
	alleles encoding 8 and 10 dinucleotide repeats	248
6.4.4.4:	Expected sequences of stocks heterozygous at the AC01 locus for	
	alleles encoding 8 and 13 dinucleotide repeats	249
6.5.1.1:	Cladogram derived from quartet puzzling of isoenzyme data for	
	Huanuco stocks and reference strains	254
6.5.1.2:	UPGMA dendrogram constructed for Huanuco stocks and reference	
	strains based on isoenzyme data	256
6.5.2:	Phylogram based on sequence data from the microsatellite locus	
	AC01 for Huanuco stocks and reference strains	258
7.2:	Proposed experimental scheme for demonstrating genetic exchange	279
7.4.1:	Construction of <i>pTEX-neo-133P</i>	281

LIST OF ABBREVIATIONS

ε-ACA	ε-amino-caproic acid
ADP	adenosine diphosphate
ALAT	alanine aminotransferase
ASAT	aspartate aminotransferase
a.s.l.	above sea level
ble	bleomycin gene
CAE	cellulose acetate electrophoresis
CL	cutaneous leishmaniasis
CMIR	cell-mediated immune response
DAT	direct agglutination test
DCL	diffuse cutaneous leishmaniasis
ddH ₂ O	deionised distilled water
DGGE	denaturing gradient gel electrophoresis
DHFR	dihydrofolate reductase
diH	di-hydrogen
diNa	di-sodium
DNA	deoxyribonucleic acid
dNTP	deoxynucleotide triphosphate
DTT	dithiothreitol
EBr	ethidium bromide
EC	Enzyme Commission
EDTA	ethylenediaminetetraacetic acid
ELISA	enzyme-linked immunosorbent assay
ES	esterase
EST	expressed sequence tag
ETS	external transcribed spacer
FCS	foetal calf serum
F1 6 diP	fructose-1, 6-diphosphate
F6P	fructose-6-phosphate
en63	63 kDa glycoprotein gene
GGPD	glucose-6-phosphate dehydrogenase
G1P + G1 6diP	glucose-1-phosphate + glucose-1.6-diphosphate
GPI	glucose phosphate isomerase
HCI	hydrochloric acid
HIV	human immunodeficiency virus
hsp70	70 kDa heat shock protein gene
ICD	isocitrate dehydrogenase
IEA	isoenzyme analysis
IEF	isoelectric focusing
IFAT	immunofluorescence antibody test
IRT	intergenic repeat typing
ITS	internal transcribed spacer (ribosomal RNA gene)
kb(n)	kilobase (pair)
KCL	potassium chloride
kDa	kiloDalton
kDNA	kinetoplast DNA

LB	Luria-Bertani
LDH	lactate dehydrogenase
LSU	large subunit (ribosomal)
MCL	mucocutaneous leishmaniasis
MDH	malate dehydrogenase
ME	malic enzyme
MEM	minimal essential medium
MgCl ₂	magnesium chloride
MnCl ₂	manganese chloride
M6P	mannose-6-phosphate
MPI	mannose phosphate isomerase
MTT	3-(4,5-dimethyl triazolyl,1-2)-2,5-diphenyl tetrazolium bromid
4-MITR	4-methylumbilliferyl butyrate
mw	molecular weight
NaCl	sodium chloride
NAD	nicotinamide-adenine dinucleotide
NADU	nicotinamide-adenine dinucleotide, reduced form
NADD	nicotinamide-adenine dinucleotide phosphate
NADE	sodium hydroxide
NaOn	neomycin phosphotransferase II gene
NULA	nucleoside hydrolase - deoxyinosine substrate
	nucleoside hydrolase - inosine substrate
NTC	non-transcribed spacer
NUN	New World
IN WY	New World leishmaniasis
OTU	operational taxonomic unit
	Old World
OW	Old World leishmaniasis
DACE	polyacrylamide gel electrophoresis
PAGE	nhosphate-huffered saline
PBS	proline-balanced salt solution
P822	polymerase chain reaction
PCR	phosphoenol manyate
PEP	prosphoenor pyruvate
PEPD	6 phosphogluconste
6PG	6 phosphogluconate
6PGD	o-phosphogluconale denydrogenase
PGI	phosphate glucose isomerase
PGM	pnospnoglucomutase
PKDL	post kala-azar dermai leisnmaniasis
pI	isoelectric point
PK	pyruvate kinase
PMS	phenazine methosulphate
PO ₄	phosphate
RAPD	random amplified polymorphic DNA
RFLP	restriction fragment length polymorphism
RNase	ribonuclease
rRNA	ribosomal RNA
SDS	sodium dodecyl sulphate
SDW	sterile distilled water
SL	spliced leader

SSCP	single-stranded conformational polymorphism
SOD	superoxide dismutase
SSU	small subunit (ribosomal)
TAE	tris-acetate-EDTA
TBE	tris-borate-EDTA
TE	tris-EDTA
Tm	melting temperature
TS	thymidylate synthetase
TSGE	thin-layer starch gel electrophoresis
υ v	ultraviolet
VL	visceral leishmaniasis
WHO	World Health Organization
UMS	universal minicircle sequence

CHAPTER 1 - INTRODUCTION

Biology and life cycle of Leishmania

1.1

Organisms within the genus Leishmania (Kinetoplastida: Trypanosomatidae) are digenetic unicellular parasites which have life cycles alternating between various mammalian hosts and blood-sucking phlebotomine sand fly vectors (Diptera: Psychodidae, Phlebotominae). Figure 1.1 illustrates the life cycle of the parasite. Promastigotes are transmitted to the mammalian host during the bite of an infected female sand fly. The parasite enters macrophage cells of the host by complement receptor-mediated phagocytosis and transforms into the non-flagellated amastigote stage. The parasite resides inside the phagolysosome by evading the lysosomal degradative pathway and begins to multiply. Multiplication continues until the cell ruptures, liberating amastigotes which are taken up by other macrophages. The cycle is completed when parasites are taken up from an infected host during the blood meal of a sand fly. Amastigotes are released in the fly's stomach and transform into elongated promastigotes. These flagellated forms attach to the hindgut or midgut epithelium (dependent on Leishmania species), where they begin to divide rapidly. Promastigotes migrate through the lumen of the midgut into the foregut and pharynx; some transform into highly infective, elongated metacyclic forms. Parasites form a "plug" in the pharynx which is thought to be regurgitated when the fly is probing for blood, thus transmitting the parasite into the next mammalian host (Killick-Kendrick, 1990). Multiplication of the parasite occurs by binary fission in both host and vector: there is, as yet, no known sexual cycle.

1.2 Basic molecular biology of Leishmania

Leishmania are kinetoplastid organisms and, as such, are distinguishable from other protozoa by the presence of kinetoplast DNA (kDNA, see section 1.10.1.3), an unusual form of mitochondrial DNA found in the kinetoplast, a structure located near the basal body of the flagellum. Kinetoplastid organisms exhibit many unusual molecular phenomena: a lack of introns, multicistronic transcription, discontinuous transcription and *trans*-splicing, and RNA editing (Borst, 1986; Van der Ploeg, 1986). These phenomena have been extensively studied to provide an insight into the mechanisms of gene expression in these organisms. Several aspects of kinetoplastid biochemistry are also unusual: they carry out glycolysis from glucose to glycerol-3-phosphate



in a unique membrane-bound structure termed a glycosome (Opperdoes, 1987), are unable to synthesize purines *de novo*, and have several unique enzymes e.g. trypanothione reductase and pathways e.g. glycolipid biosynthesis. These features have been identified as potential targets for rational drug design since the chemotherapy of diseases caused by these organisms is lacking or unsatisfactory.

1.3 Classification of the Leishmaniae

Parasite morphology within the genus *Leishmania* is very similar hence early classifications were based mostly on clinical and geographical differences (Lainson & Shaw, 1987). The application of new molecular techniques (section 1.10) has revealed complex genetic relationships within the genus, adding to the controversial taxonomic status of the group (reviewed by Cupolillo *et al.*, 2000) and to doubts about the reproductive mechanisms involved (see Gibson & Stevens, 1999).

By definition, a species comprises a reproductively isolated population sharing a common gene pool. In organisms such as Leishmania, which are considered to exhibit essentially clonal reproduction (Tibayrenc et al., 1990), this definition is hard to demonstrate and other criteria have to be applied. Classifications of *Leishmania* have included criteria such as behaviour in the vector, hamster model and in vitro culture, species of sand fly vector and biochemical characteristics such as DNA and isoenzyme analysis (Lainson & Shaw, 1987). Some workers have preferred to use subspecific names when describing different populations in the same area (Lainson & Shaw, 1987). Subspecies are defined as "...local, geographically-isolated populations which show some taxonomic differences from other geographically separated populations of the same species ... " (Anonymous, 1985). Members of the subgenus Viannia were initially designated as subspecies of L. braziliensis, i.e. L. b. panamensis, L. b. guyanensis and L. b. peruviana. Whilst L. V. braziliensis and L. V. peruviana are very similar in many biological, biochemical and serological characteristics, L. V. panamensis and L. V. guyanensis, although sharing several features, are less similar. To clarify such discrepancies another phrase, that of the "species complex", was coined (Lainson & Shaw, 1972). Although not recognized as formal taxonomic units, species complexes have now been used in both subgenera to group similar Leishmania whose species status may be questionable (see figure 1.3). The classification of Leishmania remains controversial despite the introduction of modern techniques and many Leishmania researchers disagree over what constitutes a

species. For example, there is very little to differentiate between L. L. infantum (OW) and L. L. chagasi (NW) and many workers believe that they are the same species (Grimaldi et al., 1987; Rioux et al., 1990). To date there are 30 named species of Leishmania (Shaw, 1994): their current classification is shown in figure 1.3.

The genus is divided into two subgenera based on the location of parasite development within the sand fly vector. Species in which development is restricted to the midgut and foregut (suprapylarian) belong to the subgenus *Leishmania*, whilst those species in which an additional hindgut developmental phase occurs (peripylarian) belong to the subgenus *Viannia*.

1.3.1 Subgenus Leishmania

The subgenus *Leishmania* (Saf'janova, 1982) contains 21 species, 13 of which are known to infect humans; 10 occur in the Old World (OW), and 11 occur in the New World (NW). Important members of this subgenus include: (OW) *L. L. donovani, L. L. infantum, L. L. tropica, L. L. major, L. L. aethiopica*; (NW) *L. L. chagasi, L. L. amazonensis and L. L. mexicana.*

1.3.2 Subgenus Viannia

The subgenus Viannia (Lainson & Shaw, 1987), also known by its former name, the L. braziliensis complex, contains 7 named species, which are found only in the New World. [Two additional species, L. (V) equatorensis (Grimaldi et al., 1992) and L. (V) colombiensis (Kreutzer et al., 1991), were originally classified as outlying Viannia species but it has recently been proposed that they be reclassified as "paraleishmania" (Cupolillo et al., 2000)]. All of the 7 "true" Viannia species infect man causing cutaneous and/or mucocutaneous forms of leishmaniasis. The subgenus is divided into 3 species-complexes: the L. V. braziliensis complex (2 species), the L. V. guyanensis complex (3 species) and the L. V. naiffi complex (1 species). L. V. lainsoni remains unassigned at present (Shaw, 1994). L. V. braziliensis is the most geographically widespread species, frequently existing sympatrically with other Leishmania species. This is unfortunate since L. V. braziliensis is also the most important member of this subgenus in terms of clinical disease severity because it is largely responsible for the highly disfiguring and potentially fatal mucocutaneous leishmaniasis (MCL). The species of this subgenus form the basis for this thesis and are discussed further in chapters 3-6.

Figure 1.3: Classification of the Trypanosomatidae (WHO, 1990; Rioux et al., 1990; Lainson & Shaw, 1994)



S

Chapter 1 - Introduction

1.4 Vectors and reservoir hosts of Leishmania

1.4.1 Vectors of Leishmania

Leishmania are transmitted by phlebotomine sand flies of the genera Lutzomyia (Lu., NW) and Phlebotomus (P., OW). Vector species are listed for the different OW and NW Leishmania species in tables 1.4.1 and 1.4.2 respectively. The main criteria for incriminating a particular sand fly species as a Leishmania vector are anthropophilic behaviour and repeated identification and isolation of the same parasite from both human cases and the sand fly species in question (Killick-Kendrick, 1990).

Most vector species have definite habitat requirements and exhibit different behavioural characteristics: they are consequently limited in geographical range according to the flora and fauna of, and conditions in, a particular local habitat. Some vector species have distinct host preferences: *Lu. wellcomei*, for example, a vector of *L. V. braziliensis* in Brazil, is highly anthropophilic. Other vector species have a specific ecological niche outside which they are only rarely found. *Lu. umbratilis*, for example, the principal vector of *L. V. guyanensis*, has an arboreal ecology and lives in the canopy of primary forest. Human infections with *L. V. guyanensis* occur primarily in people who have had close association with the forest e.g. workers in deforestation projects. Destruction of this habitat can reduce the incidence of disease caused by *L. V. guyanensis* infection by two related mechanisms. Firstly, deforestation eliminates the reservoir host (sloths) of both parasite and vector as sloths are unable to adapt to other habitats (Shaw, 1994). Secondly, deforestation removes the vector from the transmission cycle by eliminating both its breeding and resting sites (the trunks of large native trees) and its primary food-source, the sloth.

Parasite species may be transmitted by several vector species however, most vector species are usually only responsible for, or capable of, transmitting a single parasite species e.g. *P. papatasi* and *L. L. major*. Some vector species however, e.g. *P. argentipes* and *Lu. whitmani*, can support infection by, and development of, various *Leishmania* species (Pimenta *et al.*, 1994; Rangel *et al.*, 1996). *Lu. whitmani* is a vector of three NW *Leishmania species*: *L. V. braziliensis; L. V. guyanensis* and *L. V. shawi*. *Lu. whitmani* transmits *L. V. guyanensis* and *L. V. shawi* in the primary forest of Amazonian Brazil: the vector in this region is poorly anthropophilic and transmission occurs in a sylvatic habitat. *Lu. whitmani* transmits *L. V. braziliensis* in S. and NE Brazil: the vector in this region is highly anthropophilic and transmission

occurs in a peridomestic / intra-domicilliary habitat. Considering the parasite specificity exhibited by other vector species, it was thought surprising that a single vector species was capable of transmitting different parasites from such disparate habitats. It has recently been proposed that *Lu. whitmani* comprises a species complex where geographically isolated populations have evolved acquiring sufficient differences to be considered separate species (Rangel *et al.*, 1996; Ready *et al.*, 1997, 1998). It is thought that *Lu. whitmani* adapted from a sylvatic niche to that of the peridomestic / intradomicilliary setting (Rangel *et al.*, 1996). Another recent paper reports the coexistence of two *Leishmania* species (one *L. Viannia* sp. and one *L. L. mexicana* complex sp.) in a naturally infected sand fly species, *Lu. ovallesi* (Barrios *et al.*, 1994), a finding that is important to the proposed theory of genetic exchange occurring between *Leishmania* species (see section 1.12).

Considerable sand fly - Leishmania species specificity exists in nature. Vector competence is thought to be mediated largely by the ability of Leishmania promastigotes to attach to the vector's midgut (Pimenta et al., 1994). The molecule responsible for parasite attachment is lipophosphoglycan, LPG, a surface molecule found almost exclusively on "procyclic" promastigotes. Structural polymorphisms observed in the terminal sugar residues of LPG between Leishmania species, in conjunction with variation in vector receptors, are thought to mediate vector-parasite compatibilities (Pimenta et al., 1994; Sacks et al. 1995).

1.4.2 Reservoir hosts of Leishmania

The reservoir hosts of most *Leishmania* species are non-human mammals which may be domestic, peridomestic or sylvatic. Reservoir hosts may be further subdivided into those which are responsible for maintaining the transmission cycle (primary hosts) and those hosts, in which *Leishmania* infection is also found, thought to be incapable of maintaining the cycle (secondary hosts). Man, generally considered an accidental host of *Leishmania*, can be the only reservoir of infection for some species in a given locale (see below). Reservoir hosts are listed for the different OW and NW *Leishmania* species in *tables 1.4.1* and *1.4.2* respectively.

Reservoir hosts may or may not show signs of infection: sylvatic mammals show few, if any, signs of infection whereas domestic animals such as dogs and equines often have lesions or signs of visceral infection (Marsden, 1994a,b). Additionally, the

localization of Leishmania in the reservoir may not correspond to that seen in human disease. For example L. V. guyanensis, which causes cutaneous disease in humans, is found in the viscera of the primary host, the two-toed sloth Choloepus hofmanni (Lainson et. al, 1981). Proving that an infected mammal is a reservoir of a human parasite demands extensive ecological studies including showing that the animal is in close contact with the vector population and sufficiently long-lived to permit transmission of the same parasite species / strain.

1.4.2.1 Domestic and peridomestic reservoir hosts

It is well known that the primary domestic/peridomestic reservoir of VL caused by infection with L. L. infantum /L. L. chagası is the dog (Canis familiaris) (Rees & Kager, 1987; Lainson et al., 1994). For other Leishmania species however, including L. V. peruviana and L. V. braziliensis, the dogs' status as a reservoir, although highly suspected, remains to be proven conclusively (Yoshida et al., 1990; Lainson et al., 1994; Davies et al., 1995; Llanos-Cuentas et al., 1999; Reithinger & Davies, 1999). Other domesticated animals e.g. cats (Felis domesticus), equines (horses - Equus caballus and donkeys - E. asinus) may also be infected (Aguilar et al., 1989; Yoshida et al., 1990; Bonfante-Garrido et al., 1991; Passos et al., 1996). Rodents may act as peridomestic or sylvatic reservoirs. The common rat, Rattus rattus, has been found to be infected with various Leishmania species in both OW and NW (WHO, 1990).

1.4.2.2 Svlvatic reservoir hosts

Known sylvatic reservoirs of Leishmania include many species of rodents (including Heteromys, Nyctomys, Ototylomys, Sigmodon, Akodon, Oryzomys, Proechimys, Rattus, Rhombomys, Meriones, Psammomys, Gerbillus, Arvicanthis). In addition, many larger mammals are known or implicated as reservoirs: foxes (Cerdocyon, Vulpes), the hyrax, jackals, badgers (Meles), serval (Felis serval), genet (Genetta), mongoose (Helogale), sloths (edentates: Choloepus and Bradypus spp.), anteaters (Tamandua), armadillos (edentates: Dasypus spp.), opossums (marsupials: Didelphis), procyonids and monkeys (WHO, 1990).

Clinical type	Species	Reservoir hosts	Vector species include
OWCL	Sympatric L. L. major & L. L. tropica	Dogs (Eg), Rodents (Iq, Pk, FU), Dogs & rodents (Af, In*, Ir, Is, Mo, SA*); Man (SA)	P. sergenti (Af, In, Ir, Iq, Is, Mo, SA, FU); P. papatasi (Af, In, Ir*, Iq, Is, Mo, Pk, SA, FU); P. salehi (In*, Pk); P. caucasicus (FU)
	L. L. major	Rodents (Al, Eg, Et*, Ga, Jo*, Ke, Se*, Su, Tu*); Dogs (Se*)	P. papatasi (Al, Eg, Jo [*] , Su, Tu), P. duboscqi (Et [*] , Ga, Ke [*] , Se [*] , Su); P. chabaudi (Tu), P. sergenti (Tu
	L. L. tropica	Man (Af, Gr, In, Ir, Iq, Ke, Pk, SA); Hyrax (Na)	P. saevus (Ke); P. rossi (Na); P. sergenti (Af, Gr, In, Ir, Iq, Ke, Pk, SA)
	L. L. aethiopica	Hyrax (Et*, Ke)	P. longipes (Et*); P. pedifer (Et*, Ke*)
	L L infantum	Dogs (Sp*); Rodents (Sp*)	P. ariasi (Fr*, Sp); P. perfiliewi (It); P. perniciosus (Sp)
OWVL	L. L. donovani ss	Man (Ch, In, SA, Su)	P.chinensis* (Ch); P. argentipes* (In); e.g. P. orientalis (SA, Su)
	L. L. donovani sl	Dogs (Iq*); Jackal (Iq)	inc. P. alexandri (Iq)
	Sympatric L. L. donovani sl & L. L. infantum	Dogs (Et, It*, Ke); Fox (It*), Mongoose (Ke); Genet (Ke, Su); Serval (Su); Jackal (Su); Rodents (It*, Su)	P. orientalis (Et), P. perniciosus (It); P. martini* (Ke); P. alexandri (Pk)
	L. L. infantum	Dogs (Al*, Fr*, Ga, Ch*, Eg*, Gr*, Ir, Is*, Jo, Mo, Po, SA, Se*, Sp*, Tu*, FU*) Fox (Fr*, Po*); Jackal (Ir, FU*); Badger (FU*); Rodents	P. perniciosus (Al*, Fr*, Po, Sp, Tu); P. ariasi (Fr*, Mo, Po, Sp); P. alexandri (Ch*); P. langeroni (Eg), P. balcanicus (Gr*); P. neglectus (Gr*); P. syriacus (Is, Jo); P. longiductus (FU*); P. halepensis (Ir), P. arabicus (SA); P. langeroni (Eg)

Table 1.4.1: Reservoir Host and Vector Species of OW Leishmania (WHO, 1990; Desjeux, 1992a)

9

Key: Af: Afghanistan; Al: Algeria; Ch: China; Eg: Egypt; Et: Ethiopia; Fr: France; Ga: Gambia; Gr: Greece; In: India; Ir: Iran; Iq: Iraq; Is: Israel; It: Italy; Jo: Jordan; Ke: Kenya; Mo: Morocco; Na: Namibia; Pk: Pakistan; Po: Portugal; SA: Saudi Arabia; Se: Senegal; Sp: Spain; Su: Sudan; Tu: Tunisia; FU: Former USSR. *: proven vector species or reservoir host.

Clinical type	Species	Reservoir hosts	Vector species include Lu.: Lutzomyia; Ps.: Psychopydogus.
NWCL	L. V. braziliensis	dogs (Ar, Br*, Pe*); sloths (Br, Pa*); equines (Ar, Br); rodents (Be*, Bo, Br*) opposums (Br*)	Ps. wellcomei (Br*); Lu. whitmani ss (Br*); Lu. intermedia (Ar, Br); Lu. carrerai (Bo*, Bt); Ps. complexus (Br*); Lu. yucumensis (Bo*); Lu. amazonensis (Br); Lu. ayrozai (Br); Lu. migonei (Br); Lu. paraensis (Br); Lu. pessoai (Br); Lu. spinicrassa (Co*); Lu. gomezi (Ve); Lu. ovallesi (Ve); Lu. panamensis (Ve)
	L. V. peruviana	Dogs (Pe*)	Lu. ayacuchensis (Pe); Lu. peruensis (Pe); Lu. verrucarum (Pe); Lu.
	L. V. guyanensis	1°: Sloths (Br*, FG*, Su); Antester (Br*, FG); 2°: Opposum (Br*, FG*); Rodents (Br, FG)	Lu. umbratilis (Br ⁺ , Co ⁺ , FG ⁺ , Gy,Su); Lu. anduzei (Br ⁺); Lu. whitmani (Br);
	L. V. panamensis	Dogs (CR); Rodents (CR*); Sloths (CR*)	Lu. trapidoi (Co [*] , CR [*] , Ec, Ni, Pa [*]); Lu. ylephiletor (CR [*] , Pa, Ni, Pa [*]); Lu. gomezi (Ec, Pa [*]); Lu. hartmanni (Ec); Lu. panamensis (Pa [*])
	L. V. shawi	Monkeys (Br*); Sloths (Br*); Coatimundi (Br*)	Lu. whitmani sl (Br*)
	L. V. lainsoni	Rodents (Co, Br*)	Lu. ubiquitalis (Br*)
	L. V. naiffi	Armadillo (Br*)	Ps. squamiventris (Br*); Lu. paraensis (Br*); Lu. ayrozai (Br*)
	Other Viannia species		L. V. colombiensis: Lu. hartmanni (Co); Lu. gomezi (Pa); Lu. panamensis (Pa)
	L.L. amazonensis	Rodents (Br ⁺ , FG ⁺ , Ve); Marsupials (Br ⁺); Primates (Sciuris) (Ec)	Lu. flaviscutellata (Br [*] , Co [*] , Ec, FG [*] , Su, Ve); Lu. olmeca nociva (Br); Lu. olmeca olmeca (Be [*])
	L. L. mexicana	Rodents (Be, Gu, Me)	Lu. olmeca olmeca (Be, CR, Gu, Ho, Me*); Lu. ylephiletor (Gu)
	L. L. chagasi	Dogs (Ho)	Lu. longipalpis (Ho)
NWVL	L. L. chagasi	Dogs (Bo [*] , Br [*] , Co [*] , ES, Gu, Ho, Me, Ni, Pa, Ve); Fox (Br [*]); Opossum (Br, Co)	Lu. longipalpis (Ar, Bo*, Br*, Co*, CR, ES, Gu, Ho, Me, Ni, Pa, Pr, Ve); Lu. evansi. (Br*)

Table 1.4.2: Reservoir Host and Vector Species of NW Leishmania (WHO, 1990; Desjeux, 1992a)

Key: Ar. Argentina, Be: Belize; Bo: Bolivia; Br: Brazil; Co: Colombia; CR: Costa Rica, Ec: Ecuador; ES: El Salvador; FG: French Guiana; Gy: Guyana, Ho: Honduras; Me: Mexico, Ni: Nicaragua; Pa: Panama; Pe: Peru; Pr: Paraguay; Su: Suriname, Ve: Venezuela. *: proven vector species or reservoir host.

1.4.3 Transmission cycles

Leishmania may be transmitted to humans via a zoonotic or anthroponotic cycle. Which cycle operates depends on the species of parasite, vector and reservoir hosts present in a given ecological niche. Leishmaniasis is generally considered a zoonosis, with man being an accidental host. Anthroponotic transmission of L. L. donovani sensu stricto occurs in India (vector P. argentipes) and in foci in the Sudan, Saudi Arabia and China (WHO, 1990). Anthroponotic transmission has also been documented for other Leishmania species including L. L. tropica (WHO, 1990). For this type of transmission cycle, active detection and treatment of cases is essential for control of the disease (see section 1.8). In the NW, despite rare reports to the contrary (WHO, 1990), leishmaniasis is thought to be entirely zoonotic.

Some species of Leishmania may exist in either a zoonotic or anthroponotic cycle depending on local conditions. CL due to L. L. tropica, for example, is an anthroponosis in parts of Kenya (vector P. saevus) whereas in other areas the same disease is strictly a zoonosis involving animal reservoirs such as dogs or rodents. VL due to infection with L. L. donovani can also be maintained in a zoonotic or anthroponotic cycle however, in this case, two different forms of this species, sensu stricto (anthroponotic) and sensu lato (zoonotic), are responsible (WHO, 1990). Leishmania species with different types of transmission cycle can overlap in a given area if ecological conditions permit. For example, L. L. donovani ss. and L. L. infantum, both of which cause VL, occur sympatrically in Saudi Arabia: L. L. donovani is maintained in a human-vector cycle whereas L. L. infantum is maintained in a dog-vector cycle with man as a secondary host.

The maintenance of *Leishmania*, for a given geographical area and ecological niche, is usually dependent on a small number of vector and reservoir host species. For example, *L. V. guyanensis* is transmitted by *Lu. umbratilis*, *Lu anduzei* and *Lu. whitmani* with sloths, anteaters, opossums and rodents acting as reservoirs. In Kenya, transmission of *L. L. aethiopica* involves just one vector (*P. pedifer*) and reservoir host (hyrax) species. Some *Leishmania* species however, notably *L. V. braziliensis*, seem capable of infecting a wide range of both vector and reservoir host species. *L. V. braziliensis* occurs in almost all central and South American countries and is the most widespread NW cutaneous species. Transmission of *L. V. braziliensis* involves at least 14 vector species (see *table 1.4.2*) and, although the definitive host is not known, this species has been found in many mammals including dogs, sloths, foxes, equines, rodents and opossums (WHO, 1990). L. V. braziliensis appears capable of adapting to different ecological situations using the local fauna as host and vector species. In Huanuco Department, Peru, for example, L. V. braziliensis exists sympatrically with L. V. peruviana (Dujardin et al., 1995b). The principal vector species (thought to be Lu. tejadai) of L. V. peruviana in this area has also been strongly implicated in the transmission of L. V. braziliensis, which has been recently introduced causing both CL and MCL (Clive Davies, personal comm.).

1.5 <u>Leishmaniasis</u>

Leishmaniasis, the term given to disease caused by *Leishmania* infection, is a collective term covering a wide range of clinical forms. Leishmaniasis is endemic throughout most of tropical and sub-tropical America (New World leishmaniasis - NWL), and across Africa, India, central Asia, Russia, parts of eastern Asia, the Mediterranean basin including several southern European countries (Old World leishmaniasis - OWL). Three hundred and fifty million people are at risk of acquiring *Leishmania* infection. The worldwide prevalence of leishmaniasis is 12 million cases with an annual incidence estimated to be in the region of 1.5-2 million clinical cases, although this is likely to be a gross underestimate as many cases go unreported (WHO, 1990; Liew & O'Donnell, 1993).

Infection with *Leishmania* can result in a spectrum of disease depending upon both parasite species involved and the efficiency of the host's immune response to the parasite (Pearson & Wilson, 1989). The main clinical forms of disease are summarized in *table 1.5.1* for the major parasite species involved.

1.5.1 Cutaneous leishmaniasis

Cutaneous leishmaniasis (CL) may present in a number of different clinical forms depending upon the infecting species of *Leishmania* and the genetic background and immune status of the patient. There is, unfortunately, no absolute separation of clinical forms for the different species however, empirical diagnoses have previously been made based on presentation of the lesion(s) coupled with knowledge of *Leishmania* species present in the geographical region in which the infection was acquired (Lainson & Shaw, 1987).

Clinical type	Clinical	Major parasite species		
	Presentation	involved		
Cutaneous (CL)	Single to multiple lesions which may be wet or dry, crusty or nodular, dependent on infecting species. Secondary infection frequently occurs.	L. L. mexicana, L. L. amazonensis L. V. braziliensis, L. V. guyanensis L. V. peruviana, L. V. panamensis L. L. major, L. L. aethiopica, L. L. tropica, L. L. infantum (adults)		
Diffuse	Many diffuse lesions, papular, non-	L. L. amazonensis, L. L. aethiopica		
cutaneous (DCL)	ulcerating.			
Mucocutaneous	Metastatic spread of parasite from CL ulcer, an early sign is nasal septum	Usually L. V. braziliensis; rarely: L. V. panamensis, L. V. guyanensis,		
(MCL)	perforation. Gross disfiguration occurs later if untreated.	L. L. amazonensis, L. L. aethiopica, L. L. donovani s.l.		
Visceral	General involvement of the reticulo- endothelial system (spleen liver, etc.).	L. L. donovani , L. L. infantum , L. L. chagasi		
(VL)	often with fever, diarrhoea, weight-loss and anaemia			
Post kala-azar	Nodular lesions and depigmented scarring	L. L. donovani		
dermal (PKDL)				

Table 1.5.1:	Clinical	l mani	festatio	ons of	Leis	hmani	a int	fecti	OD
---------------------	----------	--------	----------	--------	------	-------	-------	-------	----

The "classical" presentations of the major species, which cause CL, along with their known geographical range, are shown in *table 1.5.2*. Lesions ordinarily start as a papule at the site of the bite, which grows, crusts and then ulcerates. The incubation period, from time of bite to evolution of the lesion, ranges from 2 weeks up to two years. Histopathological observation shows intense chronic granulomatous inflammation. Secondary bacterial and fungal infections at the site of the ulcer are often observed. Healing of CL lesions usually, but not always, signifies a life-long immunity against reinfection with the same species.

1.5.2 Mucocutaneous leishmaniasis

The presentation of mucocutaneous leishmaniasis (MCL) ranges with time from simple nasal perforation to highly disfiguring leprosy-like facial tissue destruction. MCL presents months to years after a primary cutaneous episode (Marsden, 1986). Early lesions commonly involve the nasal septum; the disease then spreads through the palate and may also involve other tissues including the lips, pharynx and trachea. As the disease progresses, devastating consequences including malnutrition and respiratory distress ensue as the result of throat and nasal obstruction. Death, usually due to suffocation caused by obstructed airways or pneumonia, is inevitable without treatment, and this often fails.

Table 1.5.2:

Characteristic lesion type and geographical range of the major parasite species causing cutaneous leishmaniasis.

Species	Lesion characteristics	Known Geographical				
		Range				
L. L. tropica (OW)	Classical "oriental sore". Painless dry ulcer, which heals spontaneously, often leading to disfiguring scars. Incubation period 2-8 months.	Greece, Turkey, Algeria, Tunisia, Egypt, Ethiopia, Kenya, Morocco Namibia, India, Pakistan, Afghanistan, Kuwait, Lebanon, Israel, Jordan, Iraq, Iran, Saudi Arabia, Syria, Turkmenia, Uzbekistan, Yemen				
L. L. major (OW)	Painless wet ulcer with multiple lesions and secondary infection common. Slow to heal, leaving large scars. Incubation period <4 months.	Turkey, Algeria, Burkina Faso, Cameroon, Chad, Cote d'Ivoire, Djibouti, Egypt, Ethiopia, Gambia, Guinea, Kenya, Mali, Mauritania, Morocco, Niger, Nigeria, Senegal, Somalia, Sudan, Tunisia, Afghanistan, China, India, Pakistan, Iran, Iraq, Israel, Jordan, Kuwait, Lebanon, Libya, Pakistan, Saudi Arabia, Syria, Turkmenia, Uzbekistan, Yemen.				
L. L. aethiopica (OW)	Simple lesions with late or absent ulceration. Slow healing. Also MCL and DCL.	Ethiopia, Kenya				
L. L. infantum (OW)	Simple CL in adults only	France, Greece, Italy, Malta, Portugal, Romania, Spain, Turkey, Azerbaijan Khazakhstan, China, Afghanistan, China, Iran, Iraq, Israel, Jordan, Lebanon, Libya, Pakistan, Saudi Arabia, Syria, U.A.E, Algeria, Burkina Faso, Cameroon, Central African Republic, Chad, Cote d'Ivoire, Egypt, Ethiopia, Gambia, Guinea, Kenya, Morocco, Niger, Nigeria, Senegal, Sudan, Tunisia				
L. L. mexicana (NW)	"Chiclero's ulcer". Single benign, self-limiting papule, nodule or ulcer, usually on face and ears. Incubation period <6 months.	Colombia, Belize, Costa Rica, El Salvador, Guatemala, Honduras, Mexico, Panama, Trinidad & Tobago, Texas (USA).				
L. L. amazonensis (NW)	Also DCL	The Amazon Basin, Bolivia, Brazil, Colombia, Ecuador, Peru, Venezuela, French Guyana, Panama, Paraguay, Suriname.				
L. V. braziliensis (NW)	Single or multiple lesions, generally more severe and long lasting. Lymphatic spread common. Spontaneous healing very rare. MCL	Argentina, Bolivia, Brazil, Colombia, Ecuador, Paraguay, Peru, Venezuela, Belize, Costa Rica, French Guyana, Guatemala, Honduras, Mexico, Panama, Nicaragua, Suriname.				
L. V. peruviana (NW)	"Uta". One or a few lesions, usually self-limiting, no metastasis. Mostly in children.	Peruvian Andes (Peru, Ecuador)				
L. V. guyanensis (NW)	"Pian-bois" or "bush yaws". Multiple ulcerative lesions, frequent metastasis. Spontaneous healing very rare, frequent relapses.	Brazil, Colombia, Peru, Guyana, Surinam, French Guyana.				
L. V. panamensis (NW)	"Ulcera de Bejuco". Single or few shallow lesions. Frequent metastasis. Also MCL.	Colombia, Ecuador, Venezuela, Costa Rica, Honduras, Panama, Nicaragua.				

It is thought that MCL is due to the reactivation of parasites which metastasized to the oronasal and pharyngeal mucosa via the blood from the site of a primary lesion (Costa *et al.*, 1986; Martinez *et al.*, 1992, Almeida *et al.*, 1996). Metastasis may occur from an active lesion or after a latent period from a deep visceral site (Almeida *et al.*, 1996). Between 8 and 16% of MCL cases have no evidence of a previous clinical infection: it is thought that the initial infection was subclinical in these cases (Marsden *et al.*, 1986; Jones *et al.*, 1987). Early-onset MCL is associated with primary lesions located on the face: it has been suggested that parasites spread directly via the lymphatic system in these cases (Weigle *et al.*, 1986). Predisposing factors for MCL are not clear but may include parasite factors, local trauma or immunosuppression (Marsden *et al.*, 1986; Saravia *et al.*, 1990).

MCL is most commonly associated with *L. V. braziliensis*. In Três Braços, Brazil, during an epidemic involving *L. V. braziliensis*, MCL occurred in around 5% of infected individuals (Marsden, 1994a, b). In Colombia, *L. V. panamensis* commonly causes MCL (Saravia *et al.*, 1985) and cases involving *L. V. guyanensis* and *L. L. amazonensis* have also been reported (Sampaio *et al.*, 1985; Santrich *et al.*, 1990). There have also been sporadic reports of MCL due to *L. L. donovani*, *L. L. infantum* and *L. L. major* (Al-Gindan *et al.*, 1983; Alvar *et al.*, 1990; Ghalib *et al.*, 1992).

1.5.3 Visceral leishmaniasis

Visceral leishmaniasis (VL) is caused by species of the viscerotropic L. L. donovani complex: L. L. donovani; L. L. infantum (OW) and L. L. chagasi (NW). These parasites are not restricted to the skin or mucous membranes but migrate to the spleen, bone marrow and lymph nodes. Common symptoms include abdominal swelling, fever, malaise, weight loss, and diarrhoea accompanied by anaemia and hepatosplenomegaly. Enlargement of the liver is due to hyperplasia of Küpffer cells which are packed with amastigotes: extensive hyperplasia eventually impairs the functioning of the organ. The failure of the host to mount an effective immune response leads ultimately to death, usually as a result of secondary infection, if VL is left untreated. There have been sporadic reports of visceralization of other Leishmania species including L. L. amazonensis (Barral et al., 1991) and L. V. braziliensis (Hernandez et al., 1993; Almeida et al., 1996), but in these cases, the patient was usually immunocompromised.

1.5.4 Other Clinical Presentations

1.5.4.1 Diffuse cutaneous leishmaniasis (DCL)

DCL is a rare, anergic, non-ulcerative form of CL seen in individuals infected with L. L. amazonensis (NW) and L. L. aethiopica (OW) who have a defective cellular immune response. Such individuals lack a T-cell proliferative response and present a negative intradermal skin test, delayed-type hypersensitivity (DTH), reaction to leishmanial antigens (Bryceson, 1970a; Convit *et al.*, 1972; Barral *et al.*, 1991). The condition starts as a single, non-ulcerative lesion around which satellite lesions develop. This is followed by parasite metastasis to distant areas of the skin with resulting lesions all over the body. DCL progresses slowly and persists for a prolonged period, often life and can easily be confused with those of lepromatous leprosy. DCL lesions do not heal spontaneously and respond poorly to treatment (Bryceson, 1970b; Convit *et al.*, 1972). The characteristic histopathological picture of DCL lesions is of a massive infiltration of vacuolated macrophages filled with amastigotes and a striking lack of lymphocytic infiltration (Bryceson, 1969).

1.5.4.2 Post kala-azar dermal leishmaniasis (PKDL)

PKDL is a condition that generally occurs 6 months to several years following treatment for VL, although in some cases there may be no previous history of leishmaniasis (Rees & Kager, 1987). As visceral infection clears, parasites in the skin become activated, resulting in chronic, non-ulcerative, macular, papular or nodular lesions. Lesions may appear all over the body and face, a presentation which, like DCL, may be confused with leprosy. Lesions appear up to 2 years after treatment; most heal within a year or so leaving depigmented scars, but some may persist for as long as 20 years (Jeronimo & Pearson, 1989). PKDL is most common in the Indian subcontinent and is rare in the New World (WHO, 1990).

1.5.4.3 HIV and Leishmania co-infection

There have been many reports of VL associated with HIV infection (Clauvel *et al.*, 1986; Berenguer *et al.*, 1989; Altes *et al.*, 1991). Leishmaniasis is now considered a serious opportunistic complication of HIV infection, especially in Mediterranean countries where up to 75% of VL cases are seen in HIV-infected individuals (Desjeux, 1992b). The clinical presentation of VL in these cases is often unusual with several reports noting the presence of cutaneous lesions (Sanz *et al.*, 1991;
Postigo et al., 1997; Ara et al., 1998). The condition is associated with a high relapse rate and is often refractory to therapy (Alvar et al., 1989). CL has also been associated with HIV infection, although to a lesser extent than VL (Machado et al., 1992). CL in HIV-infected individuals is often aggressive with dissemination and mucosal involvement common (Coura et al., 1987; Machado et al., 1992; Echevarria et al., 1993). It has also been reported that L. V. braziliensis infection can present as VL in HIV-infected individuals (Hernandez et al., 1993); presumably the reduction in CD4 count resulting from the progression of HIV infection permits the parasite to visceralize.

1.6 Pathogenesis and immunology

Our understanding of the host response to leishmanial infection stems mostly from studies in experimental models using genetically-defined strains of mice: the most studied *Leishmania* species in this respect are *L. L. major* and *L. L. donovani* (for an overview, see Leiby *et al.*, 1994). Unfortunately, the diseases caused by these two species in experimental systems are not the same as those in humans; hence it is not usually possible to transpose results directly from experimental models to human disease. More recently, the hamster model has been used to study *Leishmania* from the subgenus *Viannia*, particularly with regard to their metastatic capabilities (Martinez *et al.*, 1991; Travi *et al.*, 1996). Despite their limitations, experimental models have provided a growing insight into the immune response and mechanisms of pathogenesis involved in leishmaniasis.

The clinical outcome of a *Leishmania* infection is largely dependent on the response of the host's immune system which itself is largely determined by the host's genetic composition (see section 1.6.1). Other factors are also thought to be important including parasite species, site of inoculation, infecting dose, the presence of acquired immunosuppression (discussed in section 1.6.5) and the sex of the host (Bogdan *et al.*, 1996). The following sections deal mainly with cutaneous and mucocutaneous disease since these forms can result from infection with *Leishmania Viannia* parasites (the subjects of the thesis); visceral disease will be mentioned only briefly.

1.6.1 Genetic regulation of Leishmania infection

A number of host genetic factors controlling resistance or susceptibility to Leishmania infection have been identified in the mouse model including the *lsh* gene (Bradley, 1977) and the major histocompatibility complex (MHC)-linked *H-11* gene (Blackwell *et al.*, 1985). Relatively little is known about genetic control of infection in humans: it is highly likely however, that genetically controlled factors play a major role considering the wide spectrum of clinical disease observed. An association between MCL and host HLA-DR type has been shown in Brazilian patients; populations of African ancestry were found to be far more likely to develop MCL than those of indigenous roots (Walton & Valverde, 1979; Petzl-Erler *et al.*, 1991). MCL is known to be accompanied by high circulating levels of tumour necrosis factor alpha (TNF α): recent research has shown that an increased risk of developing MCL is associated with the presence of a specific polymorphism in the promoter region of TNF genes (Cabrera *et al.*, 1995).

1.6.2 Parasite factors involved in disease expression

The importance of parasite determinants on the outcome of Leishmania infection has been demonstrated effectively by the ability of various species to cause distinct forms of disease within genetically defined mice. For example, BALB/c mice develop progressive lesions when infected with L. L. mexicana, but not with L. V. braziliensis (Childs et al., 1984). Similarly in human leishmaniasis, certain disease forms are associated with particular species or species complexes. For example, VL is primarily associated with species of the L. L. donovani complex, and MCL is caused by species of the subgenus Viannia, notably L. V. braziliensis. Parasite factors can be expressed as differences in infectivity, pathogenicity and virulence. The factors involved in parasite virulence and pathogenesis are relatively unknown (Weigle & Saravia, 1996); most work to date has been concerned with infectivity. Factors involved in infectivity include lipophosphoglycan (LPG) (Sacks & da Silva, 1987) and possibly gp63, the major surface glycoprotein (Kweider et al., 1987). The structure and composition of LPG varies amongst Leishmania species and between life cycle stages within a species: LPG structure is of paramount importance in the relationship between species, host and vector (Pimenta et al, 1994; see section 1.4).

1.6.3 Establishment of infection

Parasites injected during sand fly feeding must be taken up and begin replicating inside a macrophage cell to establish an infection. There is an intense inflammatory response to the sand fly bite and invading parasites: the site becomes infiltrated with large numbers of neutrophils (minutes) and inflammatory macrophages (hours). The

majority of injected promastigotes are killed by lytic serum components and oxidative metabolites produced during the respiratory burst from these inflammatory cells. Some promastigotes are able to survive this initial extracellular attack and are actively phagocytosed by dermal macrophages and Langerhans cells. The surviving parasites are thought to be metacyclic forms which have an LPG-thickened glycocalyx which can withstand complement-mediated lysis (Sacks *et al.*, 1995). It has recently been shown that components of sand fly saliva also interfere with extracellular parasite killing (Titus & Ribeiro, 1988; Hall & Titus, 1995).

Once inside the phagocytic cell, the parasite transforms into the amastigote stage and rapidly begins to divide. The parasite resides inside the phagolysosome and is able to resist destruction by the antimicrobial arsenal of the cell (i.e. oxygen and nitrogen metabolites, low pH and cationic proteins). There are several suggested mechanisms for intracellular survival including the modulation of cytokine production (Barral-Netto *et al.*, 1992; Proudfoot *et al.*, 1995), down-regulation of the macrophage response to external cytokines (e.g. Jardim *et al.*, 1995) and selection of immunologically quiescent macrophage subpopulations e.g. Langerhans cells, Küpffer cells (Locksley *et al.*, 1988; Leiby *et al.* 1994).

The vigorous response to infection recruits additional inflammatory macrophages to the site which become new host cells for amastigotes released from ruptured cells. Infection of inflammatory macrophages can result in localization, which ensures a continuous supply of new host cells to the site (dermatotropic species, e.g. *L. L. major*), or transportation to regional lymph nodes and visceral organs (viscerotropic species, e.g. *L. L. donovani*). The infection is thus established: what the immune system does next is crucial to the outcome of infection.

1.6.4 Immune response of the host

The humoral response is not thought to perform a crucial role in leishmaniasis (Bogdan *et al.*, 1996). The resistance to clinical disease and recovery from infection are dependent on protective T cell responses, particularly through the actions of cytokines produced by $CD4^+$ T cells (Liew *et al.*, 1982; Russo *et al.*, 1993). There are two types of $CD4^+$ T cell response, Th1 and Th2, resulting in the production of two different cytokine profiles. The Th1 or Th2 T cell response correlates with resistant or susceptible phenotype in genetically defined mice (Scott, 1989). The Th1

response is characterized by the production of gamma-interferon (IFN γ), interleukin (IL)-2 and IL-12. These cytokines control parasite multiplication non-specifically by activating NK and γ/δ T cells and increase the expression of MHC class II molecules to augment antigen presentation. Production of IFN γ early in the course of infection favours the production of a Th1 cell response and protection against disease (Scott, 1991). The Th2 response is characterized by the production of IL-4, IL-5, IL-6 and IL-10, cytokines which are involved in the regulation of humoral immune responses. The Th2 response is correlated with disease severity in humans (Ghalib *et al.*, 1993); IL-4, for example, which has a role in the inhibition of IFN γ -mediated macrophage activation, has been shown to predominate in MCL (Pirmez *et al.* 1993). The cytokines secreted as a result of each Th cell response positively select for the continued expansion of their Th cell population via feedback loops. For example, the Th2 cytokine IL-10 is responsible for decreased antigen presentation by macrophages and for the inhibition of cytokine production by Th1 cells and macrophages (Fiorentino *et al.*, 1991).

The general consensus is that T cells conferring protective immunity primarily belong to the CD4⁺ subset however, there is recent evidence to suggest that cytotoxic CD8⁺ T cells also play an important role (Barral-Netto *et al.*, 1995). CD8⁺ cells have been shown to be important in conferring protective immunity against *Leishmania* infection in BALB/c mice lacking CD4⁺ T cells (Hill *et al.*, 1989) and increased numbers of *Leishmania*-specific CD8⁺ T cells are associated with clinical cure in human CL (Smith *et al.*, 1991).

1.6.5 Infection and clinical disease

1.6.5.1 Asymptomatic infection

The occurrence of inapparent infections has been increasingly recognized in recent years (Weigle *et al.*, 1993; Davies *et al.*, 1995). In a Venezuelan study of individuals living in an endemic area, approximately 12% of individuals without lesions or scars demonstrated a positive delayed type hypersensitivity (DTH) response to leishmanial antigen, i.e. showed evidence of a prior infection (Ramirez & Guevara, 1997). Asymptomatic infections are, presumably, associated with a strong, appropriate immune response such that the host is able to control and kill invading parasites before an infection can become established. The presence of asymptomatic infection

is revealed in patients who have never had leishmaniasis but who develop the disease after becoming immunosuppressed (see section 1.5.4.3). The clinical manifestation in these patients is often more severe than in immunocompetent individuals.

1.6.5.2 <u>CL and MCL</u>

Uncomplicated CL in humans is characterized by a strong cell-mediated immune response which is detectable *in vivo* as a strong DTH response to leishmanial antigens (Ridley *et al.*, 1980). Chronic CL and MCL are also characterized by profound DTH responses (Saravia *et al.*, 1989), again detectable *in vivo* as an elevated antigenspecific DTH response. In MCL, there is evidence of previous CL in the majority of patients (Marsden, 1986; Jones *et al.*, 1987). The immunopathology seen in MCL results from chronic DTH, an overactive cellular response to leishmanial antigens. Simple, self-healing CL is associated with a Th1 type response whereas a Th2 type response is observed in chronic, non-healing CL and MCL (Pirmez *et al.*, 1993; Melby *et al.*, 1994).

1.6.5.3 <u>DCL</u>

DCL presents the other end of the immunological spectrum to MCL in that it is characterized by antigen-specific anergy. Patients with DCL are DTH negative and their T cells respond poorly to leishmanial antigens (Bryceson, 1970a; Convit *et al.*, 1972; Barral *et al.*, 1991). Successful treatment of DCL results in the recovery of the cellular immune response (Convit *et al.* 1993).

1.6.5.4 <u>VL</u>

Pathological changes in VL are dominated by the specific suppression of cellmediated immunity that permits the dissemination and expansion of the parasite population. There is also a marked humoral response in the form of a proliferative polyclonal or monoclonal B-cell response to parasite and non-specific antigens. VL is characterized by a negative DTH skin response during infection and hypergammaglobulinaemia. VL is associated with a depressed Th1 immune response: there are decreased numbers of CD4+ T cells, an impairment of gamma-interferon (IFN γ) production (Carvalho *et al.*, 1985) and high levels of the Th2 cytokines, IL-4 and IL-10 are observed (Ho *et al.*, 1992).

1.6.6 Resolution of disease

Healing of CL, characterized by re-epithelialization accompanied by a clear reduction in inflammatory cells, normally results in a characteristic scar. CL may heal spontaneously or be cured by therapeutic intervention. Spontaneous healing is associated with less virulent *Leishmania* species such as *L. L. major, L. L. tropica*, and *L. L. mexicana* (Herwaldt *et al.*, 1992; Convit *et al.*, 1993). Clinical disease caused by infection with the more pathogenic and virulent species, e.g. *L. V. braziliensis*, heals spontaneously only rarely (Marsden *et al.*, 1986). The resolution of clinical disease has several outcomes dependent on whether or not the patient has acquired a protective immune response and whether they harbour persistent parasites (Weigle & Saravia, 1996). Only those individuals who are parasite-free and have acquired immunity will resist subsequent infection.

1.6.6.1 <u>Acquired immunity</u>

The majority of new Leishmania cases in endemic areas (NW) are seen in children or young adolescents (Weigle et al., 1993; Davies et al., 1995) and most people who present with CL endure only one episode of the disease (Weigle & Saravia, 1996). These observations suggest that immunity against disease is acquired following infection, and that this immunity develops with time. Resolution of infection with some species, notably L. L. major, is associated with lifelong immunity to reinfection, an observation which has been exploited by using this species for vaccination purposes (Modabber, 1990). Clinical resolution does not, however, always result in the acquisition of protective immunity against reinfection or recurrent disease as demonstrated by the high frequency of relapse in some individuals. Immune status following NWL is often more difficult to assess than in OWL due to several factors: there are commonly several species and strains of Leishmania circulating in an epidemic area; reinfection and/or recurrent disease are often observed and metastasis is a common feature of many NW species (Weigle & Saravia, 1996). Experiments in the hamster model have shown that protective immunity elicited against Leishmania Viannia species is only partial and against homologous strains (Osorio et al., 1998).

Cell-mediated immune responses are more important in acquiring immunity to *Leishmania* infection than humoral antibody responses. Acquired immunity against infection can be transferred by T cells but not B cells (Scott *et al.*, 1986) and resistant mice rendered T cell-deficient by thymectomy and irradiation are virtually unable to

control L. L. major infection (Preston et al., 1972). In mice, the immune response to infection with some species of OW Leishmania can be shown to be lost progressively with time (Neal et al., 1990). It has been suggested that the continued presence of antigen is necessary for maintaining T cell memory (Moll et al., 1995). This has implications for vaccine design, inferring that a live attenuated vaccine, which allows persistence of the parasite and therefore antigen, may be the best approach.

1.6.6.2 Parasite persistence and metastasis

Persistence refers to the continued presence of parasites after clinical cure. Metastasis is the dissemination of parasites to body sites other than the inoculation / lesion site. The capacity for metastasis and parasite persistence are closely related: those parasites which are known to persist e.g. *L. V. braziliensis* and other species of the subgenus *Viannia*, are also capable of metastasizing (Almeida *et al.*, 1996) and are highly associated with recurrent disease.

Presumably, persistent parasites evade the host immune response by the choice of host cell and by interfering with its activity. Mononuclear phagocytic cells, including macrophages, are most likely the only host cells able to support the replication of *Leishmania*. These cells are also the effectors of dissemination which may occur via the lymphatic or vascular circulatory systems. Sites to which parasite metastasis occur are also potential sites of parasite persistence. Dermatotropic *Leishmania* have been observed in lymph nodes; it is thought that Langerhans cells which acquire *Leishmania* in the skin migrate to these sites in response to inflammatory cytokines (Moll *et al.*, 1993). Macrophages and blood monocytes are responsible for disseminating parasites via the blood system: *Leishmania* have been demonstrated in the peripheral blood (Martinez *et al.*, 1992), spleen (Magill *et al.*, 1993; Almeida *et al.*, 1996) and bone marrow (Barral *et al.*, 1991). *Leishmania* have also been isolated from the scars of healed CL lesions (Aebischer, 1994; Osorio *et al.*, 1998) and parasite DNA has been demonstrated in blood samples from and scars of patients cured many years previously (Guevara *et al.*, 1993; Schubach *et al.*, 1998).

Individuals harbouring persistent parasites may remain asymptomatic however the continued presence of parasites render the host disposed to relapse or parasite metastasis (Saravia *et al.*, 1990; Aebischer, 1994; Osorio *et al.*, 1998). L. V. *braziliensis* and other members of the Viannia subgenus are often associated with

relapsing infection (Saravia et al., 1990); L. L. donovani and L. L. tropica are also implicated (Aebischer, 1994). Leishmania are able to persist, after spontaneous resolution or successful chemotherapeutic treatment, despite the presence of an intact cellular immune response of the host. This is evident from reports of individuals with a previous history of leishmaniasis who develop the disease after moving to a nonendemic area (Guevara et al., 1994), and from the increasing number of reports of leishmaniasis occurring in patients, many of whom were unaware of their infection, who subsequently become immunosuppressed by concomitant HIV infection (Coura et al., 1987; Altes et al., 1991; Machado et al., 1992; Echevarria et al., 1993).

1.6.6.3 Recurrent infection

Recurrent leishmaniasis may occur as a result of reinfection or due to the reactivation of persistent parasites. It is difficult to discriminate between reinfection or reactivation unless it is possible to separate the parasites causing primary and secondary infections. It would be relatively easy, for example, to distinguish an infection caused by *L. L. mexicana* from a previous infection caused by *L. V. braziliensis* since these two species can easily be separated by a number of phenotyping and genotyping methods. This is not the case, however, for the *Viannia* species *L. V. peruviana* and *L. V. braziliensis* which are very difficult to separate. It has been noted that relapsing lesions usually occur close to the site of the original infection (Saravia *et al.*, 1990), are smaller and contain fewer parasites than primary lesions (Gutierrez *et al.*, 1990; Davies *et al.*, 1995).

In a Colombian study of 24 patients with recurrent disease caused by *Leishmania Viannia* species, genotypic and phenotypic analyses of parasites recovered from primary and secondary lesions showed that 50% of infections were due to reactivated parasites and that 50% were reinfections (Saravia *et al.* 1990). Population-based studies in Colombia and Peru have shown that the frequency of recurrent lesions in individuals with evidence of past CL were 2.0 and 2.9 per 100 person years respectively (Weigle *et al.*, 1993; Davies *et al.*, 1995). The frequency of reactivation is determined by several factors including *Leishmania* species, immune status of the host and any drug therapy given (Davies *et al.*, 1995). Reactivation of persistent *Leishmania* may be precipitated by local trauma (Rab *et al.*, 1992; Travi *et al.*, 1996) and is also observed in patients whose cellular immune system becomes suppressed as a result of i) concomitant infections e.g. tuberculosis (Escobar *et al.*, 1996) and HIV (e.g. Machado *et al.*, 1992; Berhe *et al.*, 1995), ii) treatment for cancer or transplantation (Shaw *et al.*, 1976; Golino *et al.* 1992) and iii) coincidental conditions e.g. malnutrition (Walton & Valverde, 1979; Badaro *et al.*, 1996) and diabetes (Rocha *et al.*, 1980).

1.7 Diagnosis of Leishmania infection

1.7.1 Clinical diagnosis

Leishmaniasis may be suspected in individuals living in, or with a history of having traveled to, known endemic areas who present with cutaneous or mucosal lesions or with signs and symptoms of visceral infection (see section 1.5). *Leishmania* infection may present in a wide spectrum of clinical forms and symptoms often overlap with those of other diseases. For example, differential diagnoses of NWCL include tropical ulcer, eczema, impetigo, sporotrichosis, blastomycosis, syphilis, leprosy, yaws and cutaneous tuberculosis. Other diagnoses must be ruled out. A definitive diagnosis of *Leishmania* infection requires demonstration of the parasite by microscopical and/or culture techniques. This parasitological diagnosis may be combined with the use of serological or molecular-based tests, or both.

1.7.2 Parasitological diagnosis

A parasitological diagnosis of leishmaniasis involves demonstrating amastigotes in clinical material by preparing smears or touch preparations, or by isolation of parasites from this material using *in vitro* or *in vivo* cultivation methods. For CL and MCL, clinical material is usually obtained by punch biopsy, slit-skin or needle aspiration of the lesion. Samples must be taken from the inflamed, swollen edge of the lesion as this region contains most parasites. The lesion and surrounding area must be meticulously prepared by cleaning with 70% ethanol and, if secondary infection is evident, with 20% by volume hydrogen peroxide. For VL, splenic, liver, bone marrow or lymph node aspirates are taken. Whichever method of sampling is used, smears or touch preparations are then made, air dried and stained with Giemsa. Amastigotes may be found singularly or in groups inside or around macrophages. Amastigotes are 2-3 μ m in diameter, contain a characteristic kinetoplast and appear identical microscopically for all species of *Leishmania*. In some instances it is important to identify the parasite species and so isolation must also be attempted.

This is usually when infection with L. V. braziliensis is suspected: identification is important owing to L. V. braziliensis⁴ potential for causing subsequent MCL. Isolation is usually attempted in any case, since microscopy often fails to reveal parasites, especially in MCL and VL cases where amastigotes numbers are usually insignificant.

For parasite isolation, material is inoculated into medium (*in vitro* culture) and / or into animals, usually hamsters (*in vivo* culture). Cutaneous parasites of the subgenus *Viannia* are very fastidious and require complex media including blood or foetal calf serum. Antibiotics must also be included in the medium to prevent parasites being overgrown with contaminating bacteria. The *in vitro* isolation procedure is very demanding and some laboratories, mostly in the NW, also routinely inoculate hamsters for *in vivo* culture of *Viannia* parasites. There are many limitations to isolation procedures: cultivation is usually slow (weeks), *in vitro* culture is highly prone to contamination and not all strains are cultivable.

1.7.3 Serological diagnosis

Serological assays provide a means of diagnosing leishmaniasis by characterizing and measuring the specific immune responses produced as the result of infection. Serological assays include skin testing (Leishmanin or Montenegro test), which detects the cell-mediated immune reaction (CMIR) to the parasite and other assays, i.e. the indirect immunofluorescent antibody test (IFAT), enzyme-linked immuno-sorbent assay (ELISA) and the direct agglutination test (DAT) which detect the humoral antibody response to infection.

1.7.3.1 Skin testing

The skin test (Leishmanin or Montenegro) demonstrates past or present exposure to leishmanial antigen (not species-specific) by testing the immune response of an individual to an intradermal injection of cultured promastigotes. The test is read, by measuring the presence and size of any resulting induration, after 48-72 hours. The skin test becomes positive 1-3 months after natural infection and remains positive for life. Hence a positive skin test from an individual living in an endemic area does not necessarily indicate an active infection. Conversely, some individuals (especially those progressing to DCL) present false negative results (Marsden, 1994b). Cross-reactions have been reported with leprosy, tuberculosis and some fungal infections.

Skin tests are used to aid the diagnosis of CL and MCL, but are of little use in the diagnosis of VL because the test is normally negative during infection, becoming positive only after the development of cellular responses, which occur after recovery.

1.7.3.2 Antibody detection methods

Antibody detection methods, including IFAT, ELISA and DAT, are of most use in the diagnosis of VL. No single test is without problems and which test is used depends on the laboratory. Wide variations in test specificities and sensitivities occur depending on the area's disease prevalence, species of parasite involved and immune status of the individual. A major problem for all of these assays is antigen production which requires bulk culture of L. L. donovani promastigotes and rigorous standardization of the prepared antigen against reference sera (both positive and negative, of known titre).

IFAT is the quickest of the three assays to perform and detects active infection earlier than DAT or ELISA. The disadvantages of the IFAT technique are that interpretation can be difficult and cross-reactivity with malaria and trypanosomiasis. IFAT is, however, a useful assay for monitoring relapses since the antibody response which it detects becomes negative 6-9 months after cure.

The ELISA (Hommel et al., 1978) is performed in antigen-coated microtitre plates and can be carried out for serum, plasma or blood spots. Dot-ELISA (Pappas et al., 1983), a modified protocol of ELISA, employs small quantities of parasite antigen dotted onto nitrocellulose paper and is suitable for whole blood, serum or plasma samples. In both of these assays, antibody is detected following the addition of sample using enzyme-conjugated anti-IgG and then substrate. The dot-ELISA has been found to be more sensitive than the standard ELISA (Suman et al., 1993). In addition, it is considerably cheaper since it does not require a spectrophotometer. Both techniques are very useful for epidemiological applications. Cross-reactivity with *Trypanosoma cruzi* has been reported for both formats however, the assays remain valuable for the diagnosis of OWVL since *T. cruzi* infection is unlikely in OW areas (Suman et al., 1993).

The main advantages of DAT (Harith *et al.*, 1986) are its simplicity, ease of interpretation and the stability and cost of equipment and reagents. The test can be performed on serum, plasma, blood spots and whole blood. The assay has been

shown to be highly specific however it has shown low sensitivity in some studies (Zijlstra et al., 1992).

Many alternative serological approaches are being developed and tested to circumvent the limitations of the current diagnostic methods. These include the detection of specific IgE which has recently been shown to be an indicator of active VL (Atta et al., 1998), and an ELISA based on the gene B protein (GBP) from L. L. major for diagnosis of OWCL (Jensen et al., 1996). The most promising advances however, have come from the use of defined recombinant antigens, in particular, rK39 (Badaro et al., 1996). rK39 is an amastigote antigen specific to organisms of the L. L. donovani complex. A recent study of Sudanese VL found that an ELISA using rK39 as antigen (rK39-ELISA) was more sensitive than DAT in detecting infection; the rK39-ELISA was positive 6 months earlier than DAT in 40% of cases (Zijlstra et al., 1998). Another recent study in Turkey found the rK39-ELISA to be more sensitive than a combination of microscopy and culture methods for the diagnosis of VL (Ozensoy et al., 1998). In addition, the rK39-ELISA also identified 6/83 cases of parasitologically-proven CL from the region. The rK39-ELISA has also been shown to be very useful as a rapid screening assay for canine leishmaniasis (Ozensov et al., 1998).

1.7.4 Molecular diagnostic methods

The potential of molecular-based assays offering rapid diagnosis of *Leishmania* infection with exquisite sensitivity and specificity has long been recognized. Despite more than a decade of research, however, molecular assays remain, for reasons some of which are outlined below, tools of the research or reference laboratory. Molecular techniques are discussed further in section 1.10.

Early molecular assays focused on the development of DNA probes capable of detecting *Leishmania*-specific kinetoplast DNA (kDNA) sequences (Wirth & Pratt, 1982; Barker & Butcher, 1983; Lopes & Wirth, 1986). Molecular probes based on repetitive genomic DNA sequences including the mini-exon gene and ribosomal RNA (rRNA) spacer region have also been developed (Uliana et al., 1991; Guevara et al., 1992; Van Eys et al., 1992; Hassan et al., 1993; Guizani et al., 1994). Although molecular probes have been shown to be capable of detecting and typing *Leishmania*, the hybridization technique suffers from low sensitivity and hence probes are not

generally useful for diagnostic purposes (Degrave *et al.*, 1994a). In addition, early assays required the use of radioactivity which precluded their use in many laboratories. The problems of low sensitivity and the use of radioactive detection methods have been addressed for the probe Lmet2 (Howard *et al.*, 1991) which is specific for the *L. L. donovani* complex. Initially used in an assay involving radioactivity (Howard *et al.*, 1991), Lmet2 was subsequently incorporated into a non-radioactive chemiluminescent system (Wilson *et al.*, 1992) and a more sensitive colorimetric PCR assay (Qiao *et al.*, 1995).

With the advent of the polymerase chain reaction (PCR) and the availability of more sequence data, highly promising systems have been developed for both detection and subsequent typing of Leishmania parasites. Assays have been developed for amplification of Leishmania genus-, subgenus-, complex- and species-specific sequences from kinetoplast or genomic targets including, for example, kDNA minicircles, mini-exon, gp63 and ribosomal RNA genes (de Bruijn. & Barker, 1992; Van Eys et al., 1992; Uliana et al., 1991; Guevara et al., 1992; Hassan et al., 1993; Guizani et al., 1994). PCR technology has eliminated the former problems of lack of sensitivity, permitting rapid detection of parasites in clinical samples. The exceptional sensitivity of this approach also permits the use of less invasive sampling methods. Despite the advantages of speed, sensitivity and specificity, PCR-based assays suffer several major disadvantages. PCR requires dedicated facilities, trained staff and a reliable power supply. The technique is expensive both in terms of equipment and reagents (although possibly less expensive than in vitro culture methods which are necessary for conventional typing methods, see section 1.7.2). Cost-aside, perhaps the biggest drawback to PCR-based assays is the false positive result. Changes in PCR conditions (for example reagent or sample quality, or even the ambient temperature of the laboratory) may result in lower sensitivity and specificity. False-positive results may occur due to contamination from other samples or "amplicons" in the laboratory. PCR-based assays can also be over-sensitive for the purpose of leishmaniasis diagnosis: there have been several recent reports of Leishmania DNA being detected from "cured" cases or inapparent infections (Guevara et al., 1993, 1994; Davies et al., 1995; Schubach et al., 1998). False negatives have also been reported: these may be due to the presence of PCR inhibitors in clinical samples (e.g. haemaglobin in blood) or due to the lack of parasites for example in a badly taken or unsuitable clinical sample (Degrave et al., 1994a). Many of these drawbacks can be overcome by standardizing protocols, separating the preparation and analysis of samples and by employing positive and negative controls. False positives can also be eliminated using probe hybridization detection on the PCR products, however, this significantly increases the time and cost of the assay.

Considering the important potential advantages of PCR-based diagnostic methods for leishmaniasis, research is continuing into the development and evaluation of PCR-based assays involving alternative target sequences and novel methodologies such as nested PCR, multiplex PCR and random amplified polymorphic DNA (RAPD) (Hassan et al., 1993; Lopez et al., 1993; Piarroux et al., 1994; Degrave et al., 1994b; Rodriguez et al., 1994; Cupolillo et al., 1995; Noyes et al., 1996; Harris et al., 1998).

1.8 Treatment of leishmaniasis

1.8.1 Antileishmanial drugs

The mainstays of chemotherapy for all forms of leishmaniasis (except DCL) are pentavalent antimonial (SbV) compounds. Two similar compounds are available: meglumine antimoniate (Glucantime[®]: Rhone-Poulenc, France) and sodium stibogluconate (Pentostam[®]: Wellcome Foundation, U.K.). Glucantime[®] contains 8.5% SbV and is used in French-, Spanish- and Portuguese-speaking countries; Pentostam[®] contains 10% SbV and is used in English-speaking countries. Both Glucantime[®] and Pentostam[®] are highly toxic: common minor side-effects include anorexia, nausea, vomiting, malaise, myalgia, and lethargy. Cardiac arrhythmias and renal damage are serious side effects of SbV treatment and indications for stopping therapy. Treatment failure and clinically resistant isolates are common. For recent reviews on chemotherapy of the leishmaniases, see Berman (1997) and Croft *et al.* (1997).

Second-line drugs for leishmaniasis include the anti-fungal polyene antibiotic, amphotericin B (Fungizone), the aromatic diamidine, pentamidine isothiocyanate and the broad-spectrum antiparasitic aminoglycoside paromomycin (aminosidine). Amphotericin B is currently recommended as a second line of treatment for VL and MCL (WHO, 1990). The use of early amphotericin B formulations was restrained due to a low therapeutic index, acute toxicity and infusion-associated side effects. New formulations however, notably liposomal amphotericin B (e.g. AmBisome), have been shown to be effective in the treatment of VL, with lower toxicity than earlier formulations (Coukell & Brogdan, 1998). Liposomal formulations have been used for the treatment of both CL and MCL (Llanos-Cuentas et al., 1991; Marsden 1994b) however their efficacy has not been proven conclusively (Yardley & Croft, 1997). In some areas, pentamidine administered in a low-dose / short course regimen has been proven to be highly efficacious in the treatment of CL, and can be used as an alternative first-line drug or for SbV treatment failure (Berman, 1997). The pentamidine regimen for VL, however, consists of a long course at high-dosages; at these levels, toxicity is thought to be greater than for SbV regimens hence pentamidine is used only when SbV treatment failure is common and where pentamidine resistance has not been reported (Berman, 1997). Paromomycin has some efficacy in the treatment of VL, but appears more efficacious when combined with short-course SbV regimens (Chunge et al., 1990; Seaman et al., 1993). This combination has also been used successfully in the treatment of OW DCL (Teklemariam et al., 1994). Other drugs are being tested in trials including ketoconazole and itraconazole. Immunotherapeutic approaches are also being tried using recombinant human interferon-gamma (IFN_Y) and interleukin-12 (IL-12) (for review, see Berman, 1997).

1.8.2 Treatment of CL

There is no established treatment for CL that applies to all species and all geographical areas. The "standard regimen" for the treatment of CL (WHO, 1990) is intramuscular (im) or intravenous (iv) SbV (20mg/kg/day X20 days). This regimen is not always applicable for reasons of toxicity and cost i.e. in areas or for species where lesions have a tendency to heal spontaneously. For example the relatively benign, self-healing lesions caused by *L. L. major* do not usually require treatment (WHO, 1990). A low-dose SbV regimen (im or iv, 5mg/kg/day X20 days), which produces fewer side effects without an increase in relapse rate, is gaining widespread use for NWCL (Oliveira-Neto *et al.*, 1997a). Many other therapeutic approaches are under continuous evaluation with varying degrees of success. These include: intralesional injection with SbV; topical ointments including paromomycin-methyl benzochloride (MBCl) and glyceryl trinitrate (GTN); and new drugs including itraconazole (Berman, 1996; Dogra & Saxena, 1996; Oliveira-Neto *et al.*, 1997b; Zeina *et al.*, 1997).

OW regimens include intralesional SbV (repeated 1-2X at 1-2 day intervals) for simple lesions (L. L. major, if necessary, and L. L. tropica) or im SbV 10-20mg/

kg/day until 1 week after parasitological cure for more severe lesions (*L. L. tropica*). *L. L. aethiopica* is unresponsive to SbV therapy; for CL either leave untreated (not oronasal lesions) or use pentamidine (3-4mg/kg 1-2X/week for one month). For DCL, pentamidine is used (3-4mg/kg once/week until 4 months after parasitological cure) or, alternatively, amphotericin B (10mg/kg/day up to 8 weeks), administered by slow iv infusion (WHO, 1990; Berman, 1997). Physical treatments are also employed, mostly for simple CL (i.e. excluding infections with *L. L. aethiopica, L. L. amazonensis* and *L. V. braziliensis*), including localized heating and freezing, curettage and radiation (Grenz rays - long wavelength X-rays) (WHO, 1990).

For NWCL, CL resulting from L. V. braziliensis infection is treated with SbV (im or iv; 5-20mg/kg/day for a minimum of 4 weeks). In South American countries, if species typing is not available, all cases of CL should be treated in areas where L. V. braziliensis is known to occur owing to the possible development of MCL. In Central America, MCL is very rare (Weigle & Saravia, 1996) and lesions resulting from L. V. braziliensis infection often self-heal, hence simple CL is commonly left untreated. For lesions caused by other Leishmania Viannia species, which regimen is used depends on the area and clinical disease presentation therein. For L. L. amazonensis infections, treatment is as for L. V. braziliensis, to prevent the development of DCL associated with infection by this species (Berman, 1997). L. L. mexicana lesions do not generally require treatment except if they occur on the ear (Chiclero's ulcer), where they tend to be persistent (WHO, 1990).

1.8.3 Treatment of MCL

Marsden (1994b) recommends SbV for MCL (iv, 20mg/kg/day for at least 30 days often up to 90+ days). Parasitological cure is very difficult to monitor due the paucity of parasites in the lesion and treatment tends to be monitored serologically (Marsden, 1994b; Berman, 1997). Failure to respond to SbV is an indication for using amphotericin B (1mg/kg for 20-40 doses or AmBisome), pentamidine (4mg/kg /48hours for 10-30 days) or paromomycin. Relapse and recurrence are documented for all regimens (Marsden, 1994b). Treatment can cause severe localized tissue inflammation which can be life-threatening in itself, especially in more advanced MCL, so treatment must be carried out in hospital.

1.8.4 Treatment of VL

WHO (1990) recommends SbV (im or iv, 20mg/kg/day -max 850mg/day) for VL in Africa, Brazil or Europe, treating for a minimum of 20 days or for 2 weeks after demonstration of apparent parasitological cure. The second-line drug of choice is amphotericin B, administered daily or 3x/week by slow iv infusion of 5-10mg (initial dose) rising by 5-10mg at each administration to 0.5-1mg/kg, continued until a maximum total dose of 1-3g is given. In India, where there have been many cases of SbV-resistance, amphotericin B has become the first-line drug of choice, with pentamidine (4mg/kg 3X/week; 15-25 injections) the second-line drug of choice (Berman, 1997). Aminosidine is also used. A new drug, miltefosine, a phosphocholine analogue which can be administered orally, is also showing promise in clinical trials for Indian VL (Jha *et al.*, 1999)

1.9 Control of leishmaniasis

The incidence of the leishmaniases has increased sharply since the early eighties: it is estimated that only 3 out of 10 of cases are actually reported (WHO, 1993), hence the current figure of 2 million new cases annually is likely to be a gross underestimate. Eighty-eight countries are endemic for leishmaniasis, 76 of which are developing countries. Up to 1993, 90% of all VL cases occurred in Brazil, Bangladesh, India and the Sudan; 90% of all MCL cases occurred in Brazil, Peru and Bolivia and 90% of all CL cases occurred in Brazil, Peru and Bolivia and 90% of all CL cases occurred in Brazil, Peru and Bolivia and Syria (WHO, 1993). The disease hampers social and economic growth of these countries and their peoples. Leishmaniasis is usually a disease of the poor for whom working practices, concomitant malnutrition and an inability to pay for treatment often lead to severe disease. Development programmes for resettlement (involving deforestation, the building of new roads and urban centres) have been associated with new outbreaks of the disease by creating conditions in which unexposed, and therefore unprotected, people are brought into contact with the parasite and its vectors (WHO, 1993).

The ultimate goal for controlling leishmaniasis is the development of a vaccine, which will prevent disease or infection (see section 1.9.5). At present, in the absence of such a vaccine, other measures continue to play a role in controlling and containing the spread of disease. Control strategies include: case detection and treatment; ecological management; vector-specific and reservoir host-specific measures. To be successful in interrupting the transmission cycle, control measures must be integrated and involve community participation. Implementation of control measures in a given area requires knowledge of the transmission cycle operating, i.e. which parasite, vector species and behaviour and reservoir hosts are involved.

1.9.1 Case detection and treatment

For anthroponotic leishmaniasis i.e. where man is the reservoir of *Leishmania*, the active detection and treatment of human cases is essential for controlling the spread of disease. The treatment of infected individuals not only cures the patient of disease, but also reduces the supply of *Leishmania* available to the vector for subsequent transmission. Medical surveillance is ideally carried out continuously in such areas to identify new infections and enable the implementation of early treatment (WHO, 1990). The treatment of passively-detected cases also helps to control transmission of zoonotic leishmaniases when combined with other control measures.

1.9.2 Environmental management

Land clearance is used extensively in central and South American countries to control transmission of leishmaniases. Deforestation to at least 300 metres has been shown to be highly effective in isolating villages from vectors and reservoirs thus reducing transmission and disease (WHO, 1993). However, some sand fly species have proven highly capable of adapting to new environments and different host blood sources after interference with, or destruction of, their natural habitat (Killick-Kendrick, 1990).

1.9.3 Vector control

Anti-sand fly measures include personal protection, chemical control and environmental management methods. Personal protection methods include the application of repellents such as diethyltoluamide to the skin and clothing, the use of fine-mesh screens on windows and doors of houses and the use of permethrin- or deltamethrin-impregnated bednets (WHO, 1990; Marzochi & Marzochi, 1994). Vector control for domestic dogs, by way of dog collars impregnated with deltamethrin (Killick-Kendrick *et al*, 1997), is also used although the efficacy of this approach remains to be determined. These general anti-vector measures can be implemented, without knowledge of the vector's behaviour, to prevent sand flyhuman contact. Spraying the inside and outside of houses with residual insecticides such as DDT, malathion and deltamethrin may reduce transmission of *Leishmania* by anthropophilic vector species (WHO, 1990; Marzochi & Marzochi, 1994). The efficacy of residual insecticides in reducing *Leishmania* transmission has often been demonstrated as a side-effect of malaria control campaigns (WHO, 1990).

The implementation of targeted vector control usually requires a detailed knowledge of a vector's behaviour and ecology, gained from entomological surveys. Sand fly vectors are often difficult to target owing to the lack of basic information regarding, for example, the whereabouts of breeding and resting sites. Potential vector breeding and resting sites in domestic / peridomestic situations, including animal shelters close to human habitation, should also be sprayed and, where possible, destroyed. Vector control methods are not usually possible for sylvatic species.

1.9.4 Reservoir control

As with vector control, the implementation of reservoir control methods also requires knowledge of which species are involved along with knowledge of their ecology and behaviour. The application of reservoir control methods also depends on local conditions and customs. There are three groups of reservoirs: man, domestic / peridomestic animals and sylvatic animals, control strategies vary for each group and in different foci. When the reservoir is man, control is achieved through active case detection and treatment (section 1.9.1) plus the use of personal protection measures (section 1.9.3).

Dogs are the main reservoirs of VL (WHO, 1990) and are increasingly implicated as reservoirs in peridomestic CL foci (Falqueto *et al.*, 1991; Passos *et al.*, 1993; Rojas *et al.*, 1994; Davies *et al.*, 1995, Reithinger & Davies, 1999). Control strategies include the elimination of stray and feral dogs. Mass-screening of domestic dogs is carried out in many countries. In several European countries, pet dogs have a high sentimental value and treatment of infected dogs may be carried out, however, relapse is common (Vexenat, 1998). In other less affluent countries, treatment is not generally an option and dogs found to be infected are usually destroyed (WHO, 1990; Vexenat, 1998). The efficacy of current methods of dog-control in containing transmission is questionable since in Brazil, where large numbers of dogs from endemic areas are culled, the incidence of VL is rising (Vexenat, 1998). For a comprehensive review of the dog as a potential target for CL control, see Reithinger & Davies (1999). Research is also ongoing into vaccination of dogs (Mayrink *et al.*, 1996).

Chapter 1 - Introduction

For many *Leishmania* species, the reservoir host(s) remains to be determined definitively: this is especially true of sylvatic *Leishmania* species causing NWCL. Methods to control a suspected reservoir may fail to reduce transmission if other reservoirs are present. In some instances, however, the reservoir is known and specific control methods may be highly effective in reducing transmission. For example, in some rural foci of CL in Jordan and Tunisia, the fat sand rat (*Psammomys obesus*) is the sole reservoir. Targeted control of this rodent is effective because its burrows are easily found and, in addition, the rat has a single food source, goosefoot. Control is achieved by ploughing burrows and destroying the goosefoot (WHO, 1993). In contrast, the use of reservoir control in reducing the transmission of sylvatic *L. V. braziliensis* is usually impossible because this parasite species infects a wide range of mammals and, in most foci, the definitive reservoir host is either unknown or not amenable to control methods.

1.9.5 Vaccination

The variable ecology and geographic distribution of the different Leishmania species precludes the use of the control methods described above when definitive vectors and reservoirs are unknown or not amenable to control. A vaccine would therefore be a practical and effective control measure for some leishmaniases. The acquired immunity observed following natural infection with Leishmania has provided the stimulus for vaccine development aimed at preventing infection and / or disease. Protective immunity induced by vaccination is dependent on the ability of the vaccine to evoke an immune response capable of controlling or eliminating an infection. In leishmaniasis, the generation of protective immunity depends on a cell-mediated immune response (section 1.6.6.1), more specifically on the generation of a T-cell response in which Th1 cytokines such as IL-12 prevail. Moreover, immunity is thought to be maintained after clinical cure by the continual stimulation of the immune system by persisting parasites (section 1.6.6.2) which, if correct, has serious implications for vaccine design. The outcome of any vaccination system depends on choice of antigens, route of administration and the use and type of adjuvant used: all these factors will influence the immune response mounted.

Since "leishmanization" (section 1.9.5.1), there have been two major approaches to the development of anti-Leishmania vaccines: the first involves the use of poorlydefined crude leishmanial antigen preparations and the second involves the use of molecularly-defined antigens. "First generation" vaccines consist of crude antigen preparations; these have been used for vaccination purposes in both animals and humans and have shown highly variable efficacy. "Second generation" vaccines include attenuated, recombinant and DNA vaccines. Molecularly-defined antigens or DNA, which require the use of sophisticated technologies, present the ultimate prospect for a vaccine but the majority of research into these vaccine candidates is still at the preclinical stage.

1.9.5.1 Leishmanization

Leishmanization, a crude immunization programme in which live L. L. major promastigotes were inoculated into a site on the body not normally visible, was carried out in epidemic situations in Israel, Iran and the former USSR. Unfortunately, the use of this live "vaccine" led to severe, persistent lesions in around 2-5 % of "immunized" individuals who subsequently required treatment (WHO, 1990). Leishmanization has been abandoned owing to issues of ethics, persistent parasites and other problems associated with using a live pathogen for vaccination (Modabber, 1990).

1.9.5.2 Killed whole promastigote vaccines

Killed whole promastigote vaccines against cutaneous leishmaniasis, with or without adjuvants, are at various stages of Phase I (safety), II (reactivity) and III (efficacy) trials in humans (Modabber, 1995; Grimaldi, 1995; Armijos *et al.*, 1998; Marzochi *et al.*, 1998). Some "polyvalent" vaccines incorporate several different strains of *Leishmania* (Armijos *et al.*, 1998) whereas others are monovalent, i.e. based on a single species (Marzochi *et al.*, 1998). A vaccine composed of killed promastigotes of several *Leishmania* strains, "Leishvacin", has been developed and is undergoing trials in Brazil with encouraging results (Mayrink *et al.*, 1985). Killed promastigote vaccines (*L. L. major* + BCG) have also been shown to elicit immunity against visceral leishmaniasis in monkeys in India (Dube *et al.*, 1998).

1.9.5.3 Attenuated vaccines

Avirulent and low virulence *Leishmania* clones have been isolated (da Fonseca *et al.*, 1997), however, a lack of knowledge regarding the mechanisms rendering them avirulent plus the possibility of reversion to virulent phenotype has made the use of these strains unacceptable (Handman, 1997). Genetically manipulated strains in

which a defined, controlled mutation has taken place are more acceptable. An attenuated L. L. major vaccine strain lacking the essential dihydrofolate reductase / thymidylate synthetase (DHFR/TS) gene has been highly effective in mice (Titus *et al.*, 1995), and is now being developed for use in human trials (Handman, 1997). In addition to the potential for reversion, another problem impeding the use of attenuated vaccines is the difficulty of ensuring that strains grown on the large scale that would be required for vaccination programmes are fully-attenuated.

1.9.5.4 Recombinant vaccines

Most novel vaccine research involves the use of genetically-defined Leishmania antigens including the surface antigens gp63, PSA-2, gp46/M-2 and LPG. PSA-2 (promastigote surface antigen-2 from L. L. major) has been shown to induce a beneficial Th1 cytokine response in mice (Handman et al., 1995). Defined antigens are also being incorporated into novel delivery systems such as immunostimulating complexes (ISCOMs; Papadopoulou et al., 1998) and live vectors (attenuated Salmonella typhimurium and Vaccinia virus). Salmonella vaccine vectors incorporating gp63 from L. L. major have been shown to protect mice from infection with L. L. major (Xu et al., 1995). Vaccinia vaccines have also been developed incorporating a membrane glycoprotein, gp46/M-2, from L. L. amazonensis. This vaccine has been shown to generate significant protection against challenge in mice (McMahon-Pratt et al., 1993) however gp46/M-2 is not found in Leishmania Viannia species (McMahon-Pratt et al., 1992).

In general, defined antigen or "subunit" vaccines elicit immunity which is highly protective but short-lived. It is thought that occasional "boosters", by way of natural infections, would maintain levels of immunity in endemic areas. Another potential problem with using subunit vaccines is that of non-responders, i.e. people who fail to mount an immune response to a particular antigen; this may be overcome by using polyvalent vaccines, those incorporating several antigens.

1.9.5.5 DNA vaccines

DNA vaccines consist of parasite antigen genes integrated into eukaryotic expression vectors, usually plasmids. After intramuscular or intradermal injection, DNA is taken up by cells and, following transcription and translation, antigen is produced and processed for presentation to the immune system. DNA vaccines for leishmaniasis are particularly attractive because antigens may be produced in their native conformation throughout the course of "infection", mimicking what happens during natural infection (Handman, 1997). Candidate genes for anti-*Leishmania* DNA vaccines, including gp63 and PSA-2, have been tested in mice and shown to be effective in generating immunity to reinfection (Xu & Liew, 1995; Handman, 1997). Other advantages of DNA vaccines are that they are stable, easy and cheap to produce in large quantities and, furthermore, administration does not require the use of adjuvants (Handman, 1997). Potential problems of DNA vaccines include integration of vaccine DNA into the host genome (which may result in insertional mutagenesis) and the induction of autoimmune or immune complex disorders; there has, however, been no evidence of these problems in animal models so far (Wahren, 1996).

1.10 Molecular analysis of genetic variation

Genetic variation forms the basis for studying the relationships between groups or individuals, whether for epidemiological or taxonomic purposes. Two processes are responsible for the variation observed between individuals of the same species: mutation and genetic exchange. Genetic exchange, discussed in section 1.12, is constrained by speciation, hence mutation alone is responsible for the diversity observed between different species.

Mutations are spontaneous or induced alterations in an individual organism's genetic material and thus may be inherited. Mutations may be deleterious, advantageous or neutral to the survival of the individual: deleterious mutations are usually removed by natural selection; advantageous mutations are selected for and will usually become fixed; chance dictates whether neutral mutations, i.e. those which have no phenotypic effect, become established. The vast majority of mutations are neutral and occur in Mutations may affect the number or morphology of non-coding regions. chromosomes (e.g. translocations, inversions or deletions), or may result in small changes in DNA sequence (e.g. point mutations, deletions and insertions). Both types may affect the survival of the individual. At the gene level, mutations may affect gene function by altering expression of the gene (regulatory mutation) or by altering the amino acid sequence of the gene product (structural mutation). Single base mutations in a codon may result in the same amino acid being encoded (synonymous or silent mutation) or the generation of a different amino acid (non-synonymous mutation). Non-synonymous mutations give rise to abnormal gene products: a change

of amino acid ("mis-sense" mutation) may alter the form of the protein whereas a mutation which results in the formation of a termination codon ("nonsense" mutation) results in a truncated product.

Data indicate that kinetoplastid organisms are diploid at most loci: ploidy has been determined indirectly by comparisons of total nuclear DNA (Wells *et al.*, 1987; Iovannisci & Beverley, 1989) estimates of nuclear genome size (Leon *et al.*, 1978), RFLPs (Gibson *et al.*, 1985; Wells *et al.*, 1987) and, with the advent of PFGE, by karyotype analysis (e.g. Gibson *et al.*, 1992; Dujardin *et al.*, 1993a). Aneuploids and tetraploids have also been identified from the analysis of experimental crosses of *Trypanosoma brucei* (see section 1.12.1) and have been suggested to occur in natural populations of *Leishmania* (Bastien *et al.*, 1992; Cruz *et al.*, 1993). The *Leishmania* genome, as with other kinetoplastids, comprises both chromosomal or nuclear DNA and independently-replicating kinetoplast DNA (kDNA). *Leishmania* exhibit considerable inter- and intra-species genetic diversity which has complicated their taxonomic classification. This diversity and variation can be defined at the level of both the genome and individual genes, using numerous techniques.

1.10.1 Diversity in genome organization

1.10.1.1 Molecular karvotype analysis

Classical karyotype analysis is impossible for *Leishmania* since chromosomes do not condense during the cell cycle, however, the development of pulse-field gel electrophoresis (PFGE) has allowed chromosome-sized DNA to be resolved (Schwartz & Cantor, 1984). The haploid genome of *Leishmania* consists of 34-36 chromosomes, which are resolved into 20-25 chromosomal bands by PFGE (Wincker *et al.* 1996; Britto *et al.*, 1998). Bands ranged in size from 0.35--3Mb, different sized homologues of the same chromosome were identified and some bands comprised more than one chromosome.

Molecular karyotype analysis couples PFGE with Southern blotting and the hybridization of gene- and/or chromosome-specific probes. Molecular karyotypes are defined by the assignment of genes to DNA "chromosome" bands after PFGE: strains and isolates sharing a molecular karyotype can be grouped into "karyodemes". Many *Leishmania* genes have been mapped to chromosomes including those for ribosomal DNA, the 70kb heat shock protein (*hsp*70:), α - and β -tubulin, trypanothione

reductase, gp63, gp46/M2 and mini-exon (Leon et al., 1978; Landfear et al., 1983; Samaras & Spithill, 1985, 1987; Lee et al., 1988; Giannini et al., 1990; Hanekamp & Langer, 1991; Dujardin et al., 1993a; Belli et al., 1994; Espinoza et al., 1995; Wincker et al., 1996). Wincker et al., (1996) used 41 known genes, 66 expressed sequence tags (ESTs) and 137 anonymous DNA sequences for molecular karyotype analysis which enabled them to compile physical linkage group maps constituting the genome of L. L. infantum. Linkage groups were shown to be conserved in other Leishmania species, despite being separated by large genetic differences.

Chromosomal plasticity is extremely common in Leishmania: molecular karyotyping has revealed considerable inter- and intra-species diversity, in both chromosome number and size (Spithill & Samaras, 1985; Scholler et al., 1986; Bishop & Miles, 1987; Pages et al., 1989; Bishop & Akinsehinwa, 1989; Hanekamp & Langer, 1991; Bastien et al., 1992; Lighthall & Giannini, 1992; Cruz et al, 1993; Dujardin et al., 1993a, 1994; Wincker et al., 1996; Britto et al., 1998). In terms of size, two types of chromosomes have been identified in different isolates of the same species: sizeconserved (consensus) and size-variable (hypervariable) chromosomes (Samaras & Spithill, 1987). Lighthall & Giannini (1992) identified three levels of karyotype conservation in Leishmania species. At the first level the karyotype is highly conserved (e.g. L. L. major): strains collected from a wide geographical area are remarkably similar to a single consensus karyotype. At the second level more than one consensus karyotype can be identified within a small geographical range but strains show little variation from the consensus karyotypes present (e.g. L. V. panamensis). At the third level molecular karyotypes are highly variable, even from geographically-restricted isolates: few size-conserved chromosomes are identifiable, even from patients with the same clinical presentation (e.g. L. L. amazonensis). In some Leishmania species, polymorphisms occur to such an extent that each strain or clone has a unique karyotype (Bastien et al., 1990; Lighthall & Giannini 1992; Dujardin et al., 1993a, 1995a).

The chromosomal structure in *Leishmania* has been shown to be similar to that of other kinetoplastids: genomic DNA consists of conserved low copy number genes, multi-copy genes and many repetitive sequences (Blaineau *et al.*, 1991; Eresh *et al.*, 1993). Small changes in chromosomal size can be due to intra-chromosomal gene amplification or deletion (Cruz & Beverley, 1990). Variation in chromosomal size

may also result from homologous recombination between repetitive sequences (Iovannisci & Beverley, 1989; Dujardin *et al.*, 1994). Chromosomal translocation, as a means of generating variation, is thought to be of little significance in *Leishmania* owing to the high level of linkage group conservation exhibited (Beverley *et al.*, 1987; Wincker *et al.*, 1996). Gene amplification (often as a result of drug selection) has been noted for metabolic genes such as the bifunctional dihydrofolate reductase-thymidylate synthetase (DHFR-TS) gene (Beverley *et al.*, 1988). Amplification of the mini-exon array has been shown to account for a 50kb size variation between homologues of chromosome 2 in *L. L. major* (Iovannisci & Beverley, 1989). Rearrangements of gp63 and ribosomal DNA (rDNA) genes are associated with variation in the size of chromosomes bearing these genes in members of the subgenus *Viannia* (Dujardin *et al.*, 1994; Inga *et al.*, 1998).

1.10.1.2 Restriction fragment length polymorphism (RFLP) analysis

RFLP analysis is based on the hybridization of molecular probes to specific DNA fragments which have been generated by digesting genomic DNA with restriction endonucleases (REs). Fragments are separated by electrophoresis and bound to membranes by Southern hybridization before probing. Fragment lengths are determined by the positions of RE recognition sites in the DNA sequence. RFLPs usually arise as a result of single base substitutions in a sequence: new RE sites may be created (resulting in fragment partition) or existing RE sites may be deleted (resulting in fragment 'fusion'). RFLPs may also arise from larger mutational events such as chromosomal rearrangements. Combinations of different enzymes and probes are used for RFLP analysis allowing strains and isolates to be typed to various taxonomic levels. The discrimination of RFLP analysis relies on the degree of similarity between sequences from different strains and on the level of conservation of the specific probe used. In addition, RFLP pattern variation may be used as a crude measure of the overall sequence divergence between strains.

RFLP analysis provides information on the genomic organization of genes and can identify polymorphisms between isolates, strains and species of *Leishmania*. *Leishmania* DNA contains many genes which are present as large multigene families. Copies of genes from multigene families may vary in number and location between species and may have diverse structure and sequence. For example, the genes encoding the major surface glycoprotein, gp63, occur in tandem arrays and as dispersed single genes and are variably organized in different species. There is a tandem array of 5 genes plus a single dispersed gene in *L. L. major*, a tandem array of 7 genes plus at least 3 dispersed genes in *L. L. donovani*, a tandem array of 4 genes plus 14 dispersed genes in *L. L. chagasi*, 2 tandem arrays of 4 and 5 genes plus one dispersed gene in *L. L. mexicana* and 4 tandem arrays, one of which contains at least 22 gene copies in *L. V. guyanensis* (Button *et al.*, 1989; Webb *et al.*, 1991; Roberts *et al.*, 1993; Medina-Acosta *et al.* 1993; Steinkraus *et al.*, 1993). RFLP analysis, involving many of the same probes as used for molecular karyotyping, has been used in many studies to show variation in *Leishmania* genomic DNA (Beverley *et al.*, 1987; Reiner *et al.*, 1989; Ellis & Crampton, 1991; Kelly *et al.*, 1991; Macedo *et al.*, 1992; Belli *et al.*, 1994; Dujardin *et al.*, 1995; Espinoza *et al.*, 1995; Gomes *et al.*, 1995; Mendoza-Leon *et al.*, 1995).

1.10.1.3 Kinetoplast DNA (kDNA) and schizodeme analysis

All kinetoplastids possess a single unusual mitochondrion and a "kinetoplast", an organelle that contains a unique DNA network consisting of thousands of concatenated circular DNA molecules. This kinetoplast DNA (kDNA) network consists of two classes of circular DNAs, termed maxicircles and minicircles. Maxicircles range in size (from 19 - 40kb) and number (approximately 10 - 50), depending on the species. Maxicircles are inherited uniparentally and are homogeneous in sequence within an individual (Gibson, 1989; Gibson & Garside, 1990; Turner et al., 1995). Maxicircles carry genes encoding mitochondrial ribosomal RNAs and several protein complexes of the mitochondrial oxidative phosphorylation system (Shapiro & Englund, 1995). Minicircles also range in size (0.5 - 2.5 kb) and number (3000 - 30000) however, they are heterogeneous in sequence and inherited biparentally (Gibson & Garside, 1990). Minicircles encode the sequences for the guide RNAs involved in RNA editing (Pollard et al., 1990). Minicircle sequence analysis reveals that each minicircle of Leishmania species carries a single copy of a 100-200bp conserved region which contains the origin of replication. Within this conserved region, there are three highly conserved sequences; one of these, the universal minicircle sequence (UMS), is identical in all kinetoplastids. Outside of these highly conserved regions, there is considerable diversity: Leishmania species-specific kDNA sequences have been identified and exploited for the development molecular diagnostic probes (Barker & Butcher, 1983;

Lopes & Wirth, 1986; Gramiccia *et al.*, 1992). In addition to RFLP analysis of genomic DNA (gRFLP), RFLP analysis of kDNA ("schizodeme analysis") has also been used for typing isolates of *Leishmania*, where a schizodeme is a group of isolates sharing the same kDNA-RFLP pattern (Lopes *et al.*, 1984; Degrave *et al.*, 1994a).

1.10.2 DNA sequence diversity

In addition to the diversity shown at the level of genomic organization, Leishmania display great diversity at the DNA sequence level, both in coding and non-coding regions. Variation in DNA sequence arises as the result of mutational events which may consist of additions or deletions of genetic material, nucleotide substitutions or sequence rearrangements. Sequence diversity may be demonstrated by phenotypic or genotypic methods. Phenotypic methods analyze the gene product itself and hence detect genetic changes indirectly. Phenotyping techniques, including isoenzyme analysis (IEA), isoelectric focusing (IEF) and monoclonal antibody (serodeme) typing, are described in sections 1.10.2.1 and 1.10.2.2. Genotypic methods can demonstrate changes in DNA sequence directly, in both coding and non-coding Many methods of detecting genotypic variation involve the use of the regions. polymerase chain reaction (Mullis & Faloona, 1987; Saiki et al., 1988), including RFLP analysis of amplification products from PCR (PCR-RFLP), random amplified polymorphic DNA (RAPD) analysis, microsatellite analysis, DNA sequencing and mutation detection methods such as denaturing gradient gel electrophoresis (DGGE) and single-stranded conformational polymorphism (SSCP) analysis. Genotyping techniques are described in sections 1.10.2.3 through 1.10.2.7.

1.10.2.1 Isoenzyme analysis (IEA) and isoelectric focusing (IEF)

IEA and IEF are phenotypic methods which are based on the electrophoretic separation of enzymes coupled with their histochemical visualization. In IEA, enzymes are separated according to size and charge by electrophoresis though a suitable matrix which may be a starch, polyacrylamide or agarose gel (Miles, 1980a; Lanham *et al.*, 1981) or a cellulose acetate plate (Kreutzer & Christensen, 1980). In IEF, enzymes are separated by electrophoresis through a polyacrylamide gel containing a pH gradient; enzymes migrate to the position of their isoelectric point (pI) and thus are separated by charge only.

Chapter 1 - Introduction

Isoenzymes are phenotypic characters since they result from the expression of an enzyme gene locus: they are, however, informative statistically since their interpretation has a genetic basis. Variations in electrophoretic mobility are evidence of nucleotide substitutions causing changes in the numbers of charged amino acids in the enzyme. For both IEA and IEF, banding patterns can usually be interpreted in terms of Mendelian genotypes for a given enzyme, based on previous experience of assigning genetic loci and alleles to banding patterns and knowledge of the enzyme structure. For reviews, see Harris & Hopkinson, 1976; Miles et al., 1980b; May, 1992. The simplest interpretation of IEA patterns involve monomeric enzymes (e.g. phosphoglucomutase, PGM) which are composed of a single polypeptide subunit. Isolates homozygous for a monomeric gene locus produce a single band on IEA however, enzyme mobility may vary between species or isolates depending on the size and / or charge. An isolate heterozygous for a monomeric enzyme locus, as in a hybrid gaining one allele from each of two homozygous parents with enzymes of differing mobility, would show a double-banded pattern, one band resulting from each of the two parental alleles. Enzymes may also be dimeric (e.g. glucose phosphate isomerase - GPI), trimeric (e.g. nucleoside phosphorylase - NP) or tetrameric (e.g. lactate dehydrogenase - LDH), each having characteristic heterozygote banding patterns reflective of random associations between the polypeptide subunits. Enzymes may also be encoded by more than one locus (e.g. nucleoside hydrolase - NH); banding patterns are usually interpretable by treating each product in the same manner as if encoded by a single locus. "Missing" bands may be attributable to a particular polypeptide combination forming a non-functional enzyme. Interpretation of IEA / IEF data obviously requires a knowledge of basic genetics and experience.

IEA, also known as "zymodeme analysis", has been used extensively for the identification, classification, numerical taxonomy and epidemiological study of *Leishmania* species (Miles *et al.*, 1980b; 1981; Kreutzer & Christensen, 1980; Evans *et al.*, 1984; 1987; Cuba-Cuba *et al.*, 1985; Le Blancq *et al.*, 1986; Kreutzer *et al.*, 1987; Arana *et al.*, 1990; Rioux *et al.*, 1990; Grimaldi *et al.*, 1991; Kelly *et al.*, 1991; Revollo *et al.*, 1992; Felinto de Brito *et al.*, 1993; Cupolillo *et al.*, 1994, 1995; Dujardin *et al.*, 1995b; 1998; Chouicha *et al.*, 1997). *Leishmania* isolates are assigned to zymodemes based on accumulated data from 10 or more enzyme systems; identification is made by comparison with reference strains.

IEF has not gained widespread use despite its reported higher resolving power. IEF cannot be used with all enzymes; for some, migration to their pI results in the loss of activity making visualization impossible. In addition, the use of IEF in numerical taxonomy is precluded since insufficient enzyme systems can be studied. Nevertheless, IEF can be used to complement IEA data, providing further phenotypic evidence for inferred genotypic variance (Ebert, 1987; Piarroux *et al.*, 1994).

1.10.2.2 Monoclonal antibody typing

Monoclonal antibodies have been used for *Leishmania* species identification in reservoir hosts and sand flies, in diagnostic assays and for typing and epidemiological purposes (for review, see Grimaldi & Tesh, 1993). Many monoclonal antibodies have been raised against *Leishmania*: these have been used in a variety of immunoassay formats including radioimmunoassay (RIA), enzyme-linked immunosorbent assay (ELISA) and indirect fluorescent antibody (IFA) assays (Grimaldi *et al.*, 1987; Grimaldi & McMahon-Pratt, 1996). Monoclonal antibody typing ("serodeme analysis") has identified stage-, subgenus-, complex- and species-specific monoclonal antibodies in addition to others which have been shown to identify strains within a species (McMahon-Pratt & David, 1981; McMahon-Pratt *et al.*, 1982; Jaffe *et al.*, 1984; McMahon-Pratt *et al.*, 1985; Grimaldi *et al.*, 1987; Eperon & McMahon-Pratt, 1991). The use of monoclonal antibodies for serodeme analysis of New World *Leishmania* has been extensively studied (for review, see Grimaldi & McMahon-Pratt, 1996).

1.10.2.3 PCR-RFLP analysis

PCR-RFLP analysis combines amplification of a DNA sequence using PCR with the RFLP analysis of the PCR product. PCR-RFLP fragments are separated by electrophoresis and are visualized directly by staining, usually with ethidium bromide or silver. As with genomic RFLP analysis, PCR-RFLP analysis detects variability only in sequence which encodes restriction enzyme (RE) sites; fragment lengths are determined by the positions of RE recognition sites in the DNA sequence. Obviously, the number of fragments generated by digestion of a product with a particular RE depends on the number of relevant RE sites present. The frequency at which a RE cuts is dependent, in part, on the length of its recognition sequence: REs which recognize 4 nucleotides ("4-cutters") will generally cut more often than REs which recognize for example, 6 nucleotides ("6-cutters"). For PCR products, which are

small compared to genomic DNA, REs which recognize 4 nucleotides ("4-cutters") are used because their cutting sites occur more frequently and hence there is an increased chance of detecting an RFLP. The choice of REs used can be made from sequence restriction maps and, if maps are not available, knowledge of nucleotide bias (*Leishmania* have a GC-rich genome).

Riboprinting (Clark & Diamond, 1991) is a term coined specifically for PCR-RFLP analysis of the 18S (small subunit, SSU) ribosomal RNA (rRNA) gene. The term "ribodeme" describes populations of a species that share the same riboprint pattern (Clark & Pung, 1994). Riboprinting has been used to assess diversity in several protozoan genera, including *Trypanosoma cruzi* (Clark, 1992; Clark & Pung, 1994; Stothard *et al.*, 1998a).

PCR-RFLP analysis has been used to demonstrate variation in other *Leishmania* sequences including the internal transcribed spacer regions of the mini-exon and rRNA genes (Ramirez *et al.*, 1987; Fernandes *et al.*, 1994; Cupolillo *et al.*, 1995), the gp63 gene array (Victoir *et al.*, 1995; 1998) and a *Leishmania Viannia*-specific subtelomeric repeat sequence (Fu *et al.*, 1998).

1.10.2.4 Random amplified polymorphic DNA (RAPD) analysis

RAPD analysis (Welsh & McClelland, 1990; Williams et al., 1990), also known as arbitrarily primed PCR (AP-PCR), employs 10 - 30mer oligonucleotide primers of arbitrarily chosen sequence in low-stringency PCRs. The major advantage of RAPD analysis over other PCR-based assays is that it does not require prior knowledge of specific DNA sequence. The low stringency conditions employed allows DNA synthesis to be initiated from any site to which the primer is matched. A number of sites, flanked by complementary primer sequences and randomly distributed throughout the genome, are simultaneously amplified to produce DNA fragments of variable length which can be resolved by gel electrophoresis.

RAPD profiles (scored by the presence and absence of fragments) may reveal genusspecific and/or species-specific fragments which can be used as characters for identification purposes. In addition, isolate-specific fragments may be observed. The ease with which large amounts of RAPD data can be generated has led to the increased use of RAPD markers for population genetic and phylogenetic analyses. The interpretation of RAPD data has, however, many limitations, with serious implications for subsequent analyses:

1. It is proposed that RAPD markers are inherited in a Mendelian manner (Williams *et al.*, 1990; Apostol *et al.*, 1996; Kazan *et al.*, 1993; Tibayrenc, *et al.*, 1993; 1995). For *Leishmania*, and other organisms in which mating experiments have, as yet, proven impossible, the Mendelian inheritance of observed RAPD variation is difficult to establish (Tibayrenc, 1995).

2. The origins of RAPD fragments are unknown hence fragments of identical size may not be homologous (i.e. may exhibit homoplasy and have arisen from separate areas of the genome). This means that loci, and therefore alleles / allelic polymorphisms, cannot be identified accurately or conclusively. Polymorphisms of a given fragment ("locus") may result in alleles having different- or same-sized bands - how do you tell which? For organisms in which crossing experiments are possible, it may be possible to infer RAPD loci (Kazan *et al.*, 1993) and hence score locus polymorphism.

3. RAPD markers are essentially binary characters that can be considered to exhibit "dominance" because only one of two possible states, present or absent, can be distinguished at a "locus" (Lynch & Milligan, 1994).. The presence of a band may be presumed indicative of a dominant allele (in either homozygous or heterozygous state), whereas the absence of a band may be assumed to indicate an unamplified, recessive, null allele.

One basic approach for population genetic analysis using RAPD data has been to carry out recombination tests which do not require the identification of individual loci and alleles (Tibayrenc, 1995). For a given primer, each distinct RAPD pattern is taken as a genotype (loci and alleles remain unknown), with genetic relationships inferred by scoring the presence and absence of bands between pairs of different genotypes. Many population genetic analyses carried out on RAPD data make assumptions the validity of which are highly questionable. Statistical methods have recently been developed for the population genetic and phylogenetic analyses of RAPD data, in which the limitations of RAPD data are acknowledged (Lynch & Milligan, 1994; Zhivotovsky, 1999). There remain conflicting views as to whether RAPD markers can be reliably used for defensible population genetic and phylogenetic analyses (Apostol *et al.* 1996; Nadler *et al.*, 1995; Noyes *et al.*, 1996; Stothard & Rollinson, 1996).

Nevertheless, RAPD data has shown parity with other molecular markers confirming that RAPD markers can provide valuable information (Tibayrenc *et al.*, 1993; Dujardin *et al.*, 1995b). Data from RAPD analysis has been used both for phylogenetic and population genetic analyses of *Leishmania*: results have shown good correlation with isoenzyme, RFLP and karyotype data. (Tibayrenc *et al.*, 1993; Dujardin *et al.*, 1995b; Gomes *et al.*, 1995; Pogue *et al.*, 1995; Schönian *et al.*, 1996). In the author's (and others) opinion, however, RAPD markers should be employed for population genetic and phylogenetic analyses only if specific DNA markers for a particular organism are lacking. In such cases, given that (i) alleles and genotypes cannot be identified conclusively, (ii) RAPD markers are considered dominant and (iii) identity by state assumptions are highly questionable, results should be interpreted with extreme caution and correlated with those obtained using other genetic markers.

1.10.2.5 <u>Microsatellite analysis</u>

A large part of non-coding DNA is organized in repeated sequences which often exhibit pronounced polymorphism. One class of repetitive DNA, satellite DNA, consists of tandemly-repeated sequence motifs which are present throughout the genomes of both prokaryotic and eukaryotic organisms. Satellite DNAs include socalled mini- and microsatellites which consist of arrays of repeats nominally 10-100 bp and 1-9bp in length respectively. Satellite DNA is tandemly-repeated between areas of conserved DNA sequence, allowing primers to be designed which enable the satellite region to be amplified by PCR. Alleles are generally distinguished by different copy numbers of the repeat, although diversity may also occur in the sequence of the flanking regions. Satellite DNA markers were initially developed (Jeffreys, 1985) and exploited for use in forensic medicine, parentage and pedigree analyses and in the study of human genetic disease (for review see Jeffreys *et al.*, 1993).

Microsatellites are tandemly-repeated sequences, 1-9 base pairs in length, which are widely dispersed in eukaryotic genomes (Harnada *et al.*, 1982; Tautz & Renz, 1984). Microsatellites consisting of dinucleotide repeats, e.g. $(CA/GT)_n$ (where n is the number of repeat units) are the most common type (Ashley & Dow, 1994). Most microsatellite loci are situated in non-coding areas of the genome and, since they have no known function, are assumed to be neutral markers (Ashley & Dow, 1994;



Schlotterer, 1998). Microsatellite loci are discrete and have co-dominant alleles which are inherited in a Mendelian manner (Ashley & Dow, 1994; Schlotterer, 1998). Alleles at microsatellite loci can be scored by size or by sequence analysis: the majority of studies are based on analysis of size variation after PCR amplification. Primers for PCR are designed from conserved sequences flanking the microsatellite locus. Most alleles at microsatellite loci vary in size according to the numbers of repeat units (Tautz, 1989; Ashley & Dow, 1994).

Population genetic and phylogenetic analyses require accurate sizing of microsatellite alleles. Accurate sizing is highly dependent on being able to resolve small (1-2bp) changes in size. Microsatellite alleles are most commonly visualized by radioactive or fluorophore labeling or silver staining following electrophoresis though long-range, high percentage denaturing PAGE gels since these combinations offer high levels of sensitivity and resolution. PCR products visualized by radiolabeling or silver-staining can be sized by running a sequencing ladder next to the PCR-amplified microsatellites. Such methods usually rely on visual scoring of bands by comparison with the ladder marker.

Automated fluorescence-based instruments, initially developed for sequence analysis, have recently been applied to microsatellite analysis by coupling to microsatellite-specific computer software (for a review, see David & Menotti-Raymond, 1998). PCR products are labeled with a fluorescent dye (fluorophore) either by using fluorescently-labeled PCR primers or by incorporating fluorescently-labeled dUTPs during PCR. Fluorophore-labeled PCR product is electrophoresed through a PAGE gel past a laser set at a fixed distance. The laser beam impinges on the fluorophore which becomes excited and emits fluorescence: the emission is detected, measured and recorded. The computer generates a gel image showing the bands that were detected; this data is analyzed using software such as Genescan@ and Genotyper@.

Some automated instruments, i.e. those of Applied Biosystems (AB), can distinguish between multiple fluorophores. These systems enable PCR products from distinct microsatellite loci labeled with different fluorophores to be combined and electrophoresed in the same lane on a gel (multiplex analysis). To further increase throughput, loci of distinct sizes can be labeled with the same fluorophore and run in the same lane. In addition, a single gel can be used two or three times. A number of fluorophores are available one of which is reserved for labeling a set of molecular weight markers. Markers are included in every lane during electrophoresis to serve as an internal lane standard, allowing constant band sizing within and between lanes. Automated fluorescence-based systems are becoming increasing popular as they eliminate the problems associated with radioactivity and manual scoring, have a high throughput and allow accurate sizing of microsatellite products.

The mechanisms creating microsatellite polymorphisms are only partially understood. It is generally thought that mutations in the number of repeats at microsatellite loci arise as a result of DNA slippage (mispairing of complementary bases) during DNA replication (Levinson & Gutman, 1987; Schlotterer & Tautz, 1992). Slippage, followed by DNA mismatch repair, results in the insertion or deletion of repeat units. Other mutational mechanisms, such as unequal sister-chromatid exchange (exchange of markers within same chromosome during mitosis and meiosis) and unequal crossing-over (exchange of markers between 2 different chromosomes during meiosis) may also occur. These mechanisms are, however, not thought to be as important as DNA slippage at microsatellite loci because of the relatively small size of microsatellite arrays (Ashley & Dow, 1994; Di Rienzo *et al.*, 1994).

Mutations involving changes in the number of dinucleotide repeats are generally thought to follow the stepwise mutation model (SMM) of Ohta & Kimura (1973) or the two-phase model (TPM) of Di Rienzo *et al.* (1994). The SMM, also known as the one-step model, assumes that alleles can be defined as characters: mutation at an allele results in an alteration of one step in a positive or negative direction. In terms of microsatellite alleles, this can be interpreted such that a mutation occurs by the gain or loss of a repeat. The TPM, a modification of the SMM, assumes that whilst most mutations involve changes of a single repeat, multiple repeat unit mutations can also occur. A few microsatellite loci have, however, been shown to fit the Kimura-Crow infinite allele model (IAM) (Kimura & Ohta, 1978) which predicts that each new mutation occurring in a population will be different from the existing alleles (Estoup *et al.*, 1995a).

Statistical analyses of microsatellite data are dependent on certain assumptions including: (i) mutation follows the SMM or TPM; (ii) no mutational pressures are operating within or across loci; (iii) mutation rates are high and constant across loci;

and (iv) all alleles can be called / scored (Schlotterer et al., 1991). Whilst many loci studied appear to meet these criteria, there is also evidence that these assumptions do not always hold. For examples: some loci appear to follow the IAM model (Estoup et al., 1995a); some alleles have specific mutation rates (Schlotterer et al., 1998) with different mutation rates having been scored between alleles in a heterozygote (Amos et al., 1996); there appear to be constraints on allele size in some loci (Garza et al., 1995; Lehmann et al., 1996a) and directional mutation of microsatellite loci has been demonstrated (Amos & Rubinsztein, 1996). It is obviously important therefore, that the above assumptions are shown to hold for a given locus prior to its use for population genetic analyses otherwise incorrect conclusions are likely to be drawn.

With the increase in microsatellite sequencing, evidence of mutations in the flanking regions has also emerged (Valdes *et al.*, 1993; Orti *et al.*, 1997; Grimaldi & Cronau-Roy, 1997; Viard *et al.*, 1998). Point mutations in the flanking regions are inapparent if using size-scoring only as such alleles usually appear identical in size (i.e. exhibit homoplasy) by electrophoretic techniques (Garza *et al.*, 1996; Estoup *et al.*, 1995b; Viard *et al.*, 1998). In addition, insertions or deletions in flanking regions may result in an incorrect repeat number being scored if these are calculated by subtracting the flanking region sizes from the size of the PCR product. Null alleles may also be scored (evident by the absence of a PCR product): these alleles presumably have a mutation(s) in one of the primer-binding sites such that no amplification occurs (Callen *et al.*, 1993; Pemberton *et al.*, 1995; Lehmann *et al.*, 1996a).

Many studies base population genetic and phylogenetic analyses solely on sizedefined alleles. It is not possible to determine whether a particular allele arose from a longer or shorter allele (see *Figure 1.10.2.5a*). Nevertheless, alleles that are closely related in size are thought "more likely to share a more common recent ancestry than alleles widely different in size" (Di Rienzo *et al.*, 1994).

Phylogenetic analyses based solely on length differences between alleles are unlikely to reflect precise relationships between alleles since mutations outside the repeat region itself, which are likely to involve different mutational mechanisms, will be ignored. A more precise approach is to base analyses on mutations in the flanking regions, i.e. excluding the microsatellite repeat region from phylogenetic analysis (Ashley & Dow, 1994). Such analyses should, however, be interpreted with caution,
bearing in mind the assumptions and limitations of the analyses and, where possible comparing results with those obtained from the analysis of other molecular characters. For localized populations of closely-related individuals, however, estimates of microsatellite allele and genotype frequencies can be made and used for population genetic analyses. In addition, heterozygotes may be scored from the presence of different-sized alleles at microsatellite loci (see *Figure 1.10.2.5b*).





Figure 1.10.2.5a: "Identity by state". Two homoplasic microsatellite alleles, both with 8 repeat units, are shown. These alleles are derived from different ancestral alleles thus, despite identical lengths, they are less closely related to each other than, for example, the 8- and 9-repeat alleles. (Figure modified from Schlotterer, 1998).

Mini- and microsatellites are increasingly employed as molecular markers in analyses of genetic variation, population structure and phylogeny of parasites and parasite vectors. They have been employed for studies on *Trypanosoma brucei* (Barrett *et al.*, 1997), *T. cruzi* (Oliveira *et al.*, 1998), *Plasmodium* spp. (Van Belkum *et al.*, 1992; Su *et al.*, 1998), *Anopheles* spp. (Rongnoparut *et al.*, 1996; Lehmann *et al.*, 1996b, 1997) and *Rhodnius pallescens*, a vector of *T. cruzi*, (Harry *et al.*, 1998). Mini- and microsatellite sequences have been used for DNA fingerprinting of OW and NW *Leishmania* strains and species (Macedo *et al.*, 1992; Rossi *et al.*, 1994; Gomes *et al.*, 1995; Schonian *et al.*, 1996; Oliveira *et al.*, 1997). Macedo *et al.* (1992) constructed phylogenies for NW *Leishmania* species based on band-sharing observations from DNA fingerprints. *Leishmania*-specific microsatellite loci (markers) have only very recently been described (Russell, *et al.*, 1999) and have not, until now (work described in this thesis), been used for population genetic and phylogenetic analyses.





Figure 1.10.2.5b PCR amplification of a microsatellite locus. Heterozygous individuals A and B differ in repeat number at one allele. PCR products from A and B for this microsatellite locus, separated by gel electrophoresis, differ in size for this allele. (Adapted from Schlotterer, 1998).

1.10.2.6 Mutation detection methods

Many methods of detecting variation due to small mutational changes now exist including single-strand conformational polymorphism (SSCP), denaturing gradient gel electrophoresis (DGGE), RNase cleavage, chemical cleavage of mismatches (CCM) and heteroduplex analyses (for a comprehensive review, see Hawkins, 1997).

DGGE (Fischer & Lerman, 1983) and SSCP (Orita et al. 1989) are gel electrophoresis-based techniques capable of detecting polymorphisms and mutations in otherwise identical PCR products. DGGE is based on the fact that DNA heteroduplexes differing in a single base pair have slightly different melting characteristics. DNA molecules melt in discrete segments that are dependent on DNA sequence and base composition. When two similar heteroduplexes are electrophoresed through a denaturing gradient (chemical e.g. urea, or temperature), DNA melting will occur at slightly different positions within the gradient, migration is thus slowed down allowing resolution of the original similar heteroduplexes. SSCP analysis is based on the principle that the electrophoretic mobility of a molecule within a gel matrix varies with the size, charge and shape of the molecule. Native single stranded DNA has a folded structure dictated by its sequence: in SSCP analysis, a single nucleotide difference between two similar sequences is sufficient to alter the folded structure of one relative to the other. This conformational change is detected as a mobility difference upon gel electrophoresis. The technique differs from DGGE in that the PCR products are denatured prior to electrophoresis in a nondenaturing gel. Although DGGE and SSCP allow the determination of single base changes, DNA sequencing needs to be used to definitively characterize specific differences in gene sequence.

DGGE and SSCP have been used to identify sequence variation in rRNA gene sequences of *T. cruzi* (Stothard *et al.*, 1997; 1998b) and several other organisms including *Plasmodium falciparum* (Kain *et al.* 1996) and the flatworm, *Dugesia* (Carranza *et al.*, 1996).

1.10.2.7 DNA sequencing

DNA sequencing is the ultimate in molecular typing methods but is rarely required or used, except for comparisons of gene sequences or other sequences of interest for phylogenetic purposes or for designing probes aimed at identifying specific taxonomic groups. For example, DNA sequencing of *Leishmania* 18S rRNA genes has allowed species-specific DNA probes to be designed (Uliana *et al.*, 1991, 1994). With the advent of PCR-based cycle sequencing and automated fluorescent techniques however, high-throughput typing can be achieved with exquisite sensitivity allowing accurate phylogenetic and population genetic analyses to be carried out.

1.11 Genetic transformation

Genetic transformation is a powerful technique which has facilitated the study of gene function, expression and control mechanisms both in asexual organisms e.g. *Giardia* (Singer *et al.*, 1998; Sun *et al.*, 1998) and in organisms such as *Trypanosoma cruzi* and *Leishmania* in which sexual reproduction is thought possible but where genetic crossing experiments have not been successful. Genetic transformation of these organisms has allowed information to be gained on putative virulence factors and potential drug targets for vaccination and chemotherapeutic interventions respectively (for reviews, see Kelly, 1995a; 1997). Genetic transformation will be discussed in greater detail in Chapter 7.

1.12 Population structure and genetic exchange

For Leishmania and other trypanosomatids, a clonal theory of population structure has been proposed in which reproduction occurs predominantly by binary fission (Tibayrenc et al., 1990). The clonal reproduction theory is based on the pronounced genotype linkage disequilibrium observed in natural populations of these organisms, as demonstrated by the presence of predominant genotypes over-represented in particular localities, over extensive geographical areas and long periods of time (Tibayrenc et al., 1990; Tibayrenc & Avala, 1991). According to this theory, sexual reproduction and genetic exchange make only minor contributions to the genetic diversity observed within populations and do not occur frequently enough to break the prevailing pattern of a basically clonal population structure. The applicability of the clonal theory to trypanosomatids has, however, been questioned as a result of the demonstration of genetic exchange in Trypanosoma brucei (Jenni et al., 1986) and the finding of putative hybrid genotypes in three major trypanosomatid taxa: Leishmania, T. brucei and T. cruzi (Evans et al., 1987; Darce et al., 1991; Kelly et al., 1991; Bonfante-Garrido et al., 1992; Belli et al., 1994; Carrasco et al., 1996; Dujardin et al., 1995b; Bogliolo et al., 1996).

1.12.1 Genetic exchange in Trypanosomatids

The first proposal for the occurrence of genetic exchange in trypanosomatids came in 1980 (Gibson *et al.*, Tait): isoenzyme analysis (IEA) of *Trypanosoma brucei* obtained from wild-caught tsetse flies revealed classical Hardy-Weinberg equilibrium among alleles of polymorphic enzymes, leading to the suggestion that populations were randomly mating.

The first direct evidence that genetic exchange could occur in these organisms came from Jenni et al. (1986) In their experiments, two clones of T. brucei were mixed and passaged simultaneously through the tsetse fly vector. Analysis of the resulting progeny by IEA, RFLPs, molecular karvotyping and schizodeme analysis showed that, in addition to parental types, hybrid progeny had also been generated (Jenni et al., 1986; Wells et al., 1987; Sternberg et al., 1988). Similar parental and hybrid patterns had been observed previously in isolates from natural populations (Gibson et al., 1980: Tait, 1980). Initial suggestions were that exchange occurred at the procyclic stage in the tsetse midgut (Schweizer et al., 1991) however Gibson & Whittington (1993), using transformed lines, have showed that the salivary glands are the likely site of exchange. Analysis of progeny from crossing experiments has shown that, since both parental and hybrid forms are generated, genetic exchange is not obligatory. Crosses with both homozygous and heterozygous parental strains have now been carried out, producing the equivalent of heterozygous F1 and homozygous backcross progeny respectively (Jenni et al., 1986; Gibson, 1989; Turner et al., 1990; Gibson & Whittington, 1993; Scwhweizer et al., 1994).

The precise mechanism of genetic exchange remains unclear. Pandavoine et al. (1986) proposed a complicated model involving multiple fusions of trypanosomes with a subsequent loss of DNA to yield diploid organisms. Several groups have reported that some T. brucei hybrid clones were found to have a 3n DNA content (Wells et al., 1987; Gibson & Bailey, 1994); in addition, trisomy of several chromosomes was demonstrated (Gibson et al, 1992). In conflict with earlier reports (Paindavoine et al., 1986), all T. brucei hybrid clones generated and analyzed in Gibson's laboratory conformed to 2n or 3n status and clones with intermediate DNA contents were not found. Models based on the proposals of Gibson (1995) involve fusion of diploid and/or haploid nuclei and meiosis (Gibson & Stevens, 1999). Additional evidence for fusion and meiosis comes from the observed frequency of chromosomal recombination in hybrids (Gibson & Bailey, 1994) and the analysis of marker segregation (Turner et al., 1990; Schweizer et al., 1994). Indeed, most data now support the involvement of meiosis with exchange explainable in terms of classical Mendelian genetics (Sternberg et al., 1989; Gibson, 1989; Turner et al., 1990; Gibson & Whittington, 1993; Schweizer et al., 1994). In the absence of evidence for a haploid life cycle stage, it is proposed that exchange involves the fusion of haploid nuclei as opposed to haploid cells (Gibson, 1995). In this case, the

mechanism of genetic exchange starts with the fusion of diploid organisms followed by meiosis and fusion of a single pair of haploid nuclei, with the rest being destroyed (Gibson, 1995).

Elucidation of the precise mechanism of exchange has also been complicated by the inheritance of kDNA: maxicircles are inherited uniparentally whereas minicircles are inherited from both parents (Gibson, 1989; Gibson & Garside, 1990; Turner *et al.*, 1995). Two models have been proposed to explain these observations (Birky, 1983; Shapiro & Englund, 1995). In the first model (Birky, 1983) one maxicircle type is lost during mitotic division by a random segregation process. In the second model (Shapiro & Englund, 1995) minicircles are exchanged between parental kDNA networks. For a comprehensive review of genetic exchange in trypanosomatids, see Gibson & Stevens (1998).

Indirect evidence of genetic exchange in natural populations can be demonstrated by population genetic analysis of data gained using molecular markers. These analyses can provide an indirect measure of gene flow within and between populations thus giving an insight into the reproductive strategy of, and subdivisions within, the population. All statistical approaches assume the null hypothesis of panmixia (i.e. random mating). Significant departures from this hypothesis, evidenced by the application of segregation or recombination tests, are indicative of non-sexual reproduction and/or population substructuring (for reviews, see Tibayrenc, 1995; Gibson & Stevens, 1998).

Evidence of genetic recombination has been demonstrated in other trypanosomatids including *Trypanosoma cruzi* and *Leishmania*. For *T. cruzi*, indirect evidence of diploidy and genetic recombination in natural populations comes from data generated by IEA, RAPD and RFLP analyses (Gibson & Miles, 1986; Tibayrenc *et al.*, 1986; Carrasco *et al.*, 1995; Bogliolo *et al.*, 1996). Experimental crosses have not been successful in *Leishmania* (Panton *et al.*, 1991), although naturally-occurring putative hybrids have been reported (Evans *et al.*, 1987; Darce *et al.*, 1991; Kelly *et al.*, 1991; Bonfante-Garrido *et al.*, 1992; Belli *et al.*, 1994; Dujardin *et al.*, 1995b; Banuls *et al.*, 1997). Spontaneous fusion of promastigotes of several *Leishmania* species has been reported in axenic culture and in the sand fly gut (Lanotte & Rioux, 1990).

1.12.2 Implications of genetic exchange

Although recombination experiments in the laboratory have shown that genetic exchange can occur in trypanosomatids, its epidemiological importance remains unclear. Truc and Tibayrenc (1993) point out that the demonstration of genetic exchange in the laboratory shows only that the potential for exchange is not lost. They propose that the demonstration of a correlation between two independent sets of genetic markers (isoenzymes and schizodemes), as shown for *T. brucei*, is classical evidence of clonality. Gibson (1990; 1995) points out however, that non-random sampling of a natural population with too many human and not enough wild vector / reservoir host isolates can give rise to the occurrence of seemingly predominant zymodemes generating substantial deviations from Hardy-Weinberg equilibrium, which Tibayrenc *et al.* (1993) have taken as evidence of clonality. In her analysis of sympatric *T. brucei* subspecies, individual zymodemes are seen to be generated by relatively few patterns for each enzyme, reassorting in every possible combination (Gibson *et al.*, 1990). Gibson thus concludes that genetic exchange is not a rare occurrence in undisturbed habitats.

Whilst the amount of genetic exchange occurring in or between a population is a controversial subject, its theoretical impact is not. The implications of genetic exchange are that drug resistance or virulence factor genes i.e. those encoding greater vector/host range, different clinical presentation in humans, etc. may be mobilized such that the gene may be able to spread more rapidly through the population. Obviously, gene flow would be quicker through a sexual population than through one that is obligately asexual. The selection of a particularly successful phenotype carrying such a gene, however acquired, may be responsible for increased incidence and prevalence of disease (Maynard-Smith *et al.*, 1993).

Evolution is driven at different rates by the divergent selective pressures operating on the parasite by vector and reservoir host combinations. Since the transmission of different *Leishmania* species involves such diverse patterns, the evolution of each species or population is likely to be distinct. For example, Dujardin (1994) proposed that chromosomal plasticity may be limited by the prevalent mode of reproduction within the species. Sexual reproduction, if common, would favour the conservation of chromosomal linkage groups in the population, resulting in conserved consensus karyotypes. However, if reproduction was prevailingly clonal, the genotype would be replicated as a unit, and independently-propagating clonal lines could evolve highly divergent karyotypes. Bearing in mind the three levels of conservation observed by Lighthall & Giannini (1992), it may be the case that different species of *Leishmania* undergo varying amounts of genetic exchange within and between their populations.

1.13 <u>Aim and objectives of the project</u>

Tibayrenc *et al.* (1991) proposed, based on results from population genetic analyses, that the predominant mode of reproduction in *Leishmania* was asexual. Research into *Leishmania* variation initiated by Prof. Miles and Dr. Evans in the Pathogen Molecular Biology and Biochemistry Unit at LSHTM, provided evidence that genetic exchange could occur in these organisms (Evans *et al.*, 1987; Belli *et al.*, 1994).

The aim of the work described in this thesis is to expand on this initial research by investigating variation and relationships within and between *Viannia* species with a view to providing an insight into the mechanisms generating this variability.

Specific objectives of this work were to:

1. Select populations of parasites from the same, and separate, *Vlannia* species including putative hybrid stocks.

Leishmania reference strains and stocks were made available for the study from the WHO Leishmania cryobank at LSHTM and from other sources. In addition to reference strains, three populations were selected:

L. Viannia stocks from Brazil, including stocks from the Brazilian States of Pernambuco, Pará and Amazonas (courtesy of Sinval Brandão-Filho, Jeff Shaw and Roberto Naiff) and from Três Braços / Corte de Pedra, Bahia State (courtesy of Cesar Cuba-Cuba and Philip Marsden). This population consisted mostly of *L. V. braziliensis* stocks and was selected to study variation between stocks of this species isolated from a wide geographical area and from different host and vector species.

IL L. V. braziliensis, L. V. panamensis and putative hybrid stocks from Nicaragua (courtesy of Alex Belli). These stocks had previously been shown to include putative L. V. braziliensis / L. V. panamensis hybrid stocks (Belli et al., 1994) and were selected to i) examine variation in and between L. V. braziliensis and L. V. panamensis stocks, ii) re-examine the evidence of genetic recombination between these species using a wider range of techniques and iii) for transformation to drug resistance for use in crossing experiments aimed at demonstrating genetic exchange in the laboratory.

III. Stocks isolated during an epidemiological survey of a cutaneous (CL) and mucocutaneous leishmaniasis (MCL) epidemic site in the Department of Huanuco, Peru (courtesy of Clive Davies, LSHTM and A. E. Llanos-Cuentas and N. Roncal, Peru). These stocks were from an area previously associated with *L. V. peruviana* but where MCL had recently been described, suggesting the presence of *L. V. braziliensis* parasites. This uncharacterized set of stocks provided a unique sample collected over a short time from a localized area and therefore would support the requirements of population genetics analysis.

2. Assess diversity in *Leishmania* reference strains and in the selected populations using phenotypic and genotypic methods.

- Compare the level of discrimination between WHO-designated *Leishmania* reference strains revealed by isoenzyme analysis (IEA) and other techniques including: isoelectric focusing (IEF); random amplified polymorphic DNA (RAPD) analysis; polymerase chain reaction-restriction fragment length polymorphism (PCR-RFLP) analysis of ribosomal RNA small subunit (SSU) gene and internal transcribed spacer (ITS) region; denaturing gradient gel electrophoresis (DGGE).
- Apply techniques to stocks from defined parasite populations.
- Carry out phylogenetic and population genetics analyses of appropriate data obtained for defined parasite populations.
- Assess and implement any new methods or gene targets which become available.

- 3. Attempt to demonstrate genetic exchange in the laboratory.
- Construct transformation vectors from available trypanosomatid shuttle vectors, incorporating *L. V. braziliensis*-specific sequence for promoting integration.
- Transform putative parental *L. V. braziliensis* and *L. V. panamensis* clones from the Nicaraguan population to drug resistance using established transformation techniques.
- Perform crossing experiments between cloned transformed lines (in culture in vitro, in macrophage cell culture, experimental animals and sand flies) in an attempt to generate hybrid progeny, selectable by dual drug resistance.
 - Analyze offspring generated from crossing experiments using previously established techniques.

CHAPTER 2 - MATERIALS & METHODS

2.1 Leishmania strains and stocks

2.1.1 Reference strains

Leishmania reference strains, selected as representative of species present in the study areas, were obtained from the cryobank at LSHTM, the Instituto de Medicina Tropical "Alexander von Humboldt", Lima, Peru, the Instituto Nacional de Pesquisas da Amazônia (INPA), Arnazonas State, Brazil and the Instituto Evandro Chagas, Belém, Para State, Brazil. Reference strains are listed in *table 3.1* (Chapter 3).

2.1.2 Leishmania stocks

Leishmania subgenus Viannia stocks were selected from three South American countries: Brazil, Nicaragua and Peru. These stocks and their geographical origins will be discussed in greater detail in the relevant chapters (4-6).

2.1.2.1 <u>Stocks from Brazil</u> (see Chapter 4).

Stocks were selected from geographically separate areas and from Três Braços/Corte de Pedra, a localized region in Bahia State. Many of the Brazilian stocks have been analyzed previously (e.g. Cuba-Cuba *et al.*, 1985; 1991; Grimaldi *et al.*, 1991; Lainson *et al.*, 1981). Stocks from geographically separate areas of Brazil were selected mostly from sand fly vector and non-human reservoir sources. These stocks are listed in *Table 4.2* (Chapter 4). The Três Braços/Corte de Pedra stocks represent a population of *L. V. braziliensis* which has shown little variation using isoenzyme analysis (Cuba-Cuba *et al.*, 1985; 1991): stocks were selected from human, animal and sand fly vector sources, and are listed in *table 4.3* (Chapter 4).

2.1.2.2 Stocks from Nicaragua (see Chapter 5).

These human stocks, from geographically separate areas of Nicaragua, have also been partially analyzed (Noyes et al., 1996; Belli et al., 1994). The species L. V. braziliensis, L. V. panamensis and a putative L. V. braziliensis/L. V. panamensis hybrid form have been described (Belli et al., 1994). Stocks were selected to include several representatives of each type, and are listed in table 5.1 (Chapter 5).

2.1.2.3 Stocks from Peru (see Chapter 6).

These stocks, which represent a localized *Leishmania* population, were isolated from dogs with cutaneous lesions and from human cases of leishmaniasis in a new epidemic focus of cutaneous and mucocutaneous leishmaniasis in the Department of Huanuco, Peru. Seven stocks from this area have previously been characterized (Dujardin *et al.*, 1995b), however the stocks used in this investigation have not been studied previously. Huanuco stocks are listed in *table 6.1* (Chapter 6).

2.2 <u>Cultivation of Leishmania</u>

2.2.1 Solid Media for cultivation of *Leishmania* promastigotes

Parasites held in the LSHTM cryobank were retrieved from liquid nitrogen storage, rapidly thawed and transferred immediately to bi-phasic 4N blood slopes (Evans *et al.*, 1989) overlaid with 250μ L proline-balanced salt solution (PBSS). Recipes for culture media are given in *Appendix I*. Peruvian stocks, and Brazilian stocks not held in the LSHTM cryobank, were received on 4N slopes after a minimal number of passages and immediately transferred to fresh 4N blood slopes. Cultures were incubated at 23° C and observed twice weekly until rapidly dividing. Using standard methodology (Evans *et al.*, 1989), newly-received parasites were cryopreserved as soon as possible in order to minimize the number of passages thus retaining the original parasite type(s) present.

2.2.2 Liquid Media for cultivation of Leishmania promastigotes

Promastigotes growing well on solid medium were passaged in alpha-modified minimal essential medium (Sigma) supplemented with 10% heat-inactivated foetal bovine serum, 50µg/mL gentamycin, 30mM NaHCO₃, 40mM HEPES, 20mM D-glucose, 4mM L-glutamine, 10µM haemin, 30µM adenine, 10µM folic acid and 10µM D-biotin. (All supplements were purchased from Sigma).

2.3 <u>Preparation of Leishmania enzyme lysates</u>

Enzyme stabilates were prepared according to the method of Godfrey & Kilgour (1976). Promastigotes were harvested in exponential growth phase from liquid culture by centrifugation (3000g at 4°C for 10 minutes). The cells were washed once in ice-cold phosphate-buffered saline (PBS), pH 7.2 and centrifuged as previously. Excess PBS was removed from the pellet by inverting the tube onto tissue followed by

swabbing the inside of the tube using paper towel, carefully avoiding the pellet. The cell pellet was resuspended in a volume of enzyme stabilizer [2mM EDTA; 2mM ε -aminocaproic acid; 2mM dithiothreitol] approximately equal to that of the pellet. Suspensions were freeze-thawed three times in liquid nitrogen, transferred to microfuge tubes and centrifuged at top speed (13krpm) at 4°C for 30 mins. in an Eppendorf microfuge. The supernatant was "beaded" by dropping 15µL aliquots into liquid nitrogen. Frozen "beads" were stored under liquid nitrogen prior to use.

2.4 <u>Preparation of genomic DNA</u>

Genomic DNA was prepared according to the method of Kelly (1993). Promastigotes were harvested in exponential growth phase from liquid culture by centrifugation (3000g at 4°C for 10 minutes). The cells were washed once in ice-cold phosphatebuffered saline (PBS), pH 7.2, centrifuged as previously, then gently resuspended in cell lysis buffer [50 mM Tris, pH 8.0, containing 50mM NaCl, 50mM EDTA, 1% SDS] to a density of 10⁹ parasites/mL. Proteinase K (Sigma) was added to a final concentration of 100 μ g/mL, and the mixture was incubated at 37°C overnight. The DNA was purified by extraction with equal volumes of phenol, phenol:chloroform (1:1), and chloroform respectively, followed by precipitation with 2 volumes of absolute ethanol. The precipitate was washed once with 70% ethanol, dried and redissolved in TE buffer (pH 7.2) or ddH₂O both containing 10 μ g/mL heat-treated RNAse A (Boehringer). The integrity, purity and concentration of DNA samples was assessed on agarose gels, and by measuring the absorbance at 260nm and 280nm in a spectrophotometer (1 OD unit @ A₂₆₀ = 50mg/mL ds DNA).

2.5 <u>Isoenzvme analvsis</u> (IEA)

2.5.1 Thin-layer starch-gel electrophoresis (TSGE)

Thin-layer starch-gel electrophoresis (TSGE) was carried out essentially as described by Godfrey & Kilgour and Harris & Hopkinson, (1976). Conditions for electrophoresis and enzyme development were based on those of Miles *et al.* (1980a, 1980b, 1981) and Evans *et al.* (1984) [see *Appendix III*]. Sixteen enzyme systems were investigated for discriminating stocks within *Leishmania Viannia* populations: these are listed in *table 2.5.1*. Eleven systems (bold type in *table 2.5.1*) were selected for IEA of all stocks based on reproducibility and the ability to make a reliable genetic interpretation of the banding patterns observed.

Enzyme	Abbreviation	E.C. number [†]
mannose phosphate isomerase	MPI§	EC 5.3.1.8
nucleoside hydrolase [inosine substrate]	NHi*§	EC 3.2.2.1
nucleoside hydrolase [deoxyinosine substrate]	NHd	EC 3.2.2.x
esterase	ES	EC 3.1.1.1
proline dipeptidase	PEPD*§	EC 3.4.13.9
phosphoglucomutase	PGM§	EC 2.7.5.1
6-phosphogluconate dehydrogenase	6PGD	EC 1.1.1.44
glucose phosphate isomerase	GPI*§	EC 5.3.1.9
glucose-6-phosphate dehydrogenase	G6PD	EC 1.1.1.49
aspartate aminotransferase	ASAT	EC 2.6.1.1
alanine aminotransferase	ALAT	EC 2.6.1.2
malate dehydrogenase	MDH*	EC 1.1.1.37
isocitrate dehydrogenase [NADP+]	ICD	EC 1.1.1.42
superoxide dismutase	SOD*	EC 1.15.1.1
pyruvate kinase	PK	EC 2.7.1.40
malic enzyme	ME*	EC 1.1.1.40

Table 2.5.1:Enzyme systems investigated for discrimination betweenLeishmania reference strains using IEA (TSGE and CAE*) and IEF§.

Notes to table:

- 1. Enzyme abbreviation in **bold** type indicates those enzyme system which were used for IEA of all stocks.
- 2. * enzyme systems investigated using CAE
- 3. § enzyme systems investigated using IEF
- 4. [†] EC: Enzyme Commission
- 5. x full EC number not assigned

2.5.2 Cellulose acetate electrophoresis (CAE)

CAE was carried out according to the methods described by Lanham *et al.* (1981) and according to the manufacturer's handbook (Helena Laboratories, Beaumont, Texas, USA). Conditions for electrophoresis were based on those of Miles *et al.* (1980a, 1980b, 1981) [see *Appendix III*]. Six enzymes (MDH, NHi, PEPD, GPI, SOD and ME) were analyzed using CAE^{\bullet}, see *table 2.5.1*. Staining reactions used for CAE were the same as those used for TSGE gels.

2.6 <u>Isoelectric focusing</u> (IEF)

Isoelectric focusing was performed on $0.5 \times 120 \times 240$ mm polyacrylamide gels using a refrigerated horizontal electrophoresis apparatus (FBE-300, Pharmacia). Enzymes used for IEA were tested for suitability in IEF analysis on a small set of stock samples using broad-range (pH 3.5 - 10) Ampholine (Pharmacia)-based polyacrylamide gels. Gels were cast on GelbondTM (Pharmacia), prefocused at 2000V (15mA, 15W) for 20 mins; $1.5-5\mu$ L sample was applied to the gel either directly, or by using a casting applicator strip. Samples were focused at 2000V for 1 hour. Enzymes were detected using the same developers used for IEA. Five enzyme systems resolved well using IEF (see §, *table 2.5.1*). These systems were reassessed using narrow range gels (range ascertained from pI locations of enzyme bands on broad-range gels prior to use for the IEF of samples). For IEF, GPI, MPI, and PGM were analyzed on pH 5-8 gels; NHi and PEPD were analyzed on pH 4-6 gels.

2.7 Random amplified polymorphic DNA (RAPD) analysis

RAPD analysis was carried out on *Leishmania* reference strains and representative stocks of the putative parental and hybrid stocks from Nicaragua only (see chapters 3 and 5 for discussions). Subsequent RAPD analyses were found to be unreproducible (see section 3.4.4) and, with the development of microsatellite markers for *Leishmania Viannia* species (see section 2.10), this technique was not applied to other stocks.

2.7.1 Primers

Twenty-eight decameric (10-mer) oligonucleotides of arbitrary sequence were used to amplify genomic DNA from *Leishmania* reference strains and from representative stocks of the putative parental and hybrid stocks from Nicaragua. Oligonucleotide primers A1-A6, H1-H6 and L1-L6 were obtained from R&D products (Abingdon, UK), primers D1-D10 were obtained from D. Barker (University of Cambridge). RAPD primer sequences are listed in *Appendix IV*.

2.7.2 PCR amplification

Amplification reactions were performed to a protocol modified from that of Williams *et al.* (1990). PCR amplification reactions contained 10x NH₄ reaction buffer [160mM (NH₄)₂SO₄, 670mM Tris-HCl pH 8.8, 0.1% Tween-20 (Bioline)], 2.5mM MgCl₂, 0.2mM each of dATP, dCTP, dGTP, and dTTP (Pharmacia LKB), 20pmol primer, 1U *Taq* DNA polymerase (Bioline) and 25ng genomic DNA in a total reaction

volume of 20μ L. Double distilled water (used in the reactions and for preparation of reagents) was irradiated in a UV crosslinker to inactivate potentially-contaminating DNA. Reactions were overlaid with 30μ L mineral oil (Sigma) and amplified in a Hybaid Thermal Reactor (Hybaid Ltd, Middx., UK) using: 1 cycle of 94°C for 5 min., 37° C for 1 min., 72° C for 1 min., followed by 40 cycles of 94°C for 1 min., 37° C for 1 min., followed by a final extension of 10 min. at 72° C.

2.7.3 Electrophoresis of RAPD PCR products

Amplification products were electrophoresed through 2% agarose in 1x TAE buffer. After staining with ethidium bromide, products were visualized by UV illumination and documented via a video copy processor (Mitsubishi).

2.7.4 Interpretation of RAPD results

RAPD profiles were scored visually to identify primers which discriminated: between (i) the *Viannia* and *Leishmania* subgenera, (ii) species within the *Viannia* subgenus and (iii) putative parental and hybrid strains from Nicaragua.

2.8 <u>PCR-RFLP analyses: riboprinting and PCR-RFLP of the internal</u> transcribed spacer (ITS)

Two PCR-RFLP techniques were employed to study variation in the Leishmania Viannia strains and stocks. Riboprinting, a specific term coined by Clark & Diamond (1991), is the PCR-RFLP analysis of the 18S [also known as the small subunit (SSU)] ribosomal RNA (rRNA) gene. Riboprinting was carried out on Leishmania reference strains and representative stocks of the putative parental and hybrid stocks from Nicaragua only (see chapters 3 and 5 for discussions). PCR-RFLP analysis of the internal transcribed spacer (ITS) region of the rRNA locus (also known as Intergenic Repeat Typing [IRT], Cupolillo et al., 1995) was carried out on Leishmania reference strains and stocks from Nicaragua only. Both PCR-RFLP techniques were limited in terms of discriminatory power for Viannia species and, as for RAPD, with the development of microsatellite markers for Leishmania Viannia species (see section 2.10), these techniques were not applied to stocks from the Brazil or Huánuco populations. Since the methodologies employed are almost identical, and since the target sequence of each approach concerns the same gene locus, the techniques will be described together. The organization of the trypanosomatid rRNA locus, a consensus from studies of several genera, is shown in figure 2.8.1.



Figure 2.8.1: Organization of the trypanosomatid rRNA locus

Figure 2.8.1: Organization of the trypanosomatid rRNA locus. Figure adapted from Cupolillo et al., 1995, based on data from *Crithidia fasciculata, T. cruzi, T. brucei, & Leishmania* spp. ETS: external transcribed spacer, NTS: non-transcribed spacer. SSU1 & SSU2 are primers for ribotyping, ITS1 & ITS2 are primers for PCR-RFLP analysis of the ITS region (see Appendix IV). Not drawn to scale.

2.8.1 Primers

Primers for the amplification of the 18SrRNA gene were essentially those designed from multiple alignment of trypanosomatid 18S rDNA sequences by Uliana *et al.* (1991, 1994). Oligonucleotide primers SSU1 (S1) and SSU2 (S4) were obtained from R&D Products (UK). *Eco*RI and *Hind*III restriction endonuclease sites were added to the 5' ends of SSU1 and SSU2 respectively to facilitate cloning at a later date. Primers for amplification of the rRNA locus ITS were IR1 and IR2 as described by Cupolillo *et al.* (1995). Primer sequences are given in *Appendix IV*.

2.8.2 PCR reactions

For riboprinting analysis, PCR reactions contained 10x NH₄ reaction buffer [160mM (NH₄)₂SO₄, 670mM Tris-HCl pH 8.8, 0.1% Tween-20 (Bioline)], 1.5mM MgCl₂, 0.2mM each of dATP, dCTP, dGTP, and dTTP (Pharmacia), 20pmol each primer, 1U *Taq* DNA polymerase (Bioline) and 25ng genomic DNA in a total volume of 100 μ L. Reactions were overlaid with 60 μ L mineral oil (Sigma) and amplified in a Hybaid Thermal Reactor (Hybaid Ltd, Middx., UK) using 1 cycle of 95°C for 5 min., 58°C for 60s, 72°C for 2 min., followed by 25 cycles of 95°C for 60s, 58°C for 30s, 72°C for 90s, followed by a single cycle of 72°C for 10 min. For PCR-RFLP analysis of the ITS region between the 18S and 28S rRNA genes, PCR parameters used were those of Stothard *et al.* (1996).

2.8.3 Electrophoretic visualization of the PCR product

Amplification products were checked for size and purity on 1.2% agarose gels in 1x TAE buffer [0.04M Tris-acetate. 0.001M EDTA], stained with ethidium bromide, visualized under UV and documented as described previously.

2.8.4 Preparation of the PCR product for RFLP analysis

PCR products were prepared by adding ddH_2O to a final volume of 300μ L. 500μ L of chloroform was added, the tube was inverted several times and then spun in an Eppendorf microfuge at top speed (13k rpm) for 5 mins. The diluted product (upper aqueous phase), free of mineral oil, was transferred to a sephadex G-50 spin column to remove unincorporated primers, dNTPs, etc. The column was centrifuged at room temperature for 3 minutes at 1500 rpm and the eluate transferred to a fresh tube. Prepared products were stored at +4°C before digestion.

2.8.5 RFLP analysis

Enzymes were selected for digestion of the PCR-products based upon data from a restriction map of the *L. V. braziliensis* 18SrRNA sequence (Genbank accession no.: M80292), high G/C base content (*Leishmania* have a GC-rich genome) and short length (4-cutters cut more often than enzymes of greater length). Under conditions recommended by the enzyme supplier, 20μ L of each diluted amplification product (100-200µg DNA) was digested in 30μ L (total volume) with each of the enzymes listed in *table 2.8.1*. Not all enzymes were used for both techniques: combinations are listed in *table 2.8.1*. Digestion mixes were incubated at 37°C for 2 hours (except *Taq*I which was incubated at 65°C for 2 hours).

2.8.6 Visualization of products from the digestion mix

For PCR-RFLP of the ITS only, products from the digestion were run out on 1.2% agarose gels, as previously described, to confirm complete digestion. Products were analyzed on 6% (T), 3.5%(C) polyacrylamide gels (see Appendix II) in 1x TBE buffer. PAGE gels were silver-stained according to the method of Vidigal *et al.* (1994) and photographed onto Polaroid print film (type 667) using a Polaroid MP-4 land camera.

2.8.7 Interpretation of PCR-RFLP PAGE gels

Fragments were scored visually, as for RAPD analysis, to identify enzymes which discriminated: between (i) the *Viannia* and *Leishmania* subgenera, (ii) species within the *Viannia* subgenus and (iii) putative parental and hybrid strains from Nicaragua.

Enzyme	Recognition site	Technique	Supplier*
Rsal	GT AC	Riboprinting	Р
SacII / NarI	CCGC*GG	Riboprinting	В
Sau3AI	"GATC	Riboprinting	Р
AluI	AG [•] CT	Riboprinting / PCR-RFLP (ITS)	Р
Hpa∏ / MspI	C*CGG	Riboprinting / PCR-RFLP (ITS)	В
TaqI	T*CGA	Riboprinting / PCR-RFLP (ITS)	Р
HaeIII	GG*CC	Riboprinting / PCR-RFLP (ITS)	Р
BstUI	CG [*] CG	PCR-RFLP (ITS)	N
Cfol / Hhal	GC*GC	PCR-RFLP (ITS)	N
Csp6I	G [*] TAC	PCR-RFLP (ITS)	N
DdeI	C"TNAG	PCR-RFLP (ITS)	N
DpnI	GATC	PCR-RFLP (ITS)	N
EcoRI	G *AATTC	PCR-RFLP (ITS)	Р
FspI	TGC*GCA	PCR-RFLP (ITS)	N
ScrFI	CC'NGG	PCR-RFLP (ITS)	N
SphI	GCATG [•] C	PCR-RFLP (ITS)	N

<i>Table 2.8.1:</i> Enzy	mes used for	riboprinting a	and PCR-R	FLP	analyses.
--------------------------	--------------	----------------	-----------	-----	-----------

*Enzyme supplier: Promega [P], Boehringer [B], New England Biolabs [N]

2.9 Denaturing gradient gel electrophoresis (DGGE)

DGGE was performed according to the method of Stothard *et al.* (1998), based on the protocols of Myers *et al.* (1987, 1990) and Michaelides *et al.* (1995). Perpendicular gradient gels (polyacrylamide gel containing 0 - 40% gradient of denaturant, 8M urea and 40% formamide) were used to determine the melting profile of one of the 18SrRNA amplification products in order to select the denaturation range of the product for subsequent analysis on parallel gradient gels. 30μ L of *Alu*I-digested *L. V. braziliensis* PCR product was added to 60μ L of loading buffer (1mL 10% sucrose, 1μ L 1% bromophenol blue, 1% xylene cyanol). The sample was electrophoresed at 150V for 2.5 hours at 60°C, then silver stained. 10 - 25% parallel gradient gels were subsequently used to analyze 10μ L of each of the complete range of amplification products. Parallel gels were electrophoresed at 65V for 6 hours at 60°C. Gels were silver stained and photographed as described previously.

2.10 Microsatellite analysis

2.10.1 Microsatellite analysis - background

Microsatellite DNA sequences specific for Leishmania subgenus Viannia species were identified by screening a genomic library of L. V. peruviana strain LC1116 with a $(CA)_{11}$ oligonucleotide by Russell *et al.* (1999). Positive clones were sequenced to identify microsatellite sequences of which three (AC01, AC16, AC52) were selected for further characterization and study (GenbankTM accession numbers AF139110, AF139111 and AF139112, respectively). Primers were designed from the flanking regions of these three microsatellites using DNAstaro software. Primer sequences are listed in Appendix IV.

2.10.2 PCR of microsatellite loci for PAGE analysis

Amplification reactions were carried out according to a protocol modified from that of Russell *et al.* (1999). PCR amplification reactions contained 10X NH₄ reaction buffer [160mM (NH₄)₂SO₄, 670mM Tris-HCl pH 8.8, 0.1% Tween-20 (Bioline)], 1mM (AC01, AC52) or 2mM (AC16) MgCl₂, 0.2mM each of dATP, dCTP, dGTP, and dTTP (Pharmacia LKB), 5pmol each primer, 0.5% formamide (v/v), 1U *Taq* DNA polymerase (Bioline) and 25ng genomic DNA in a total reaction volume of 50 μ L. Reactions were overlaid with 30 μ L mineral oil (Sigma) and amplified in a Hybaid Thermal Reactor (Hybaid Ltd, Middx., UK) using 35 cycles of 95°C for 30s, 62°C (AC01 and AC52) or 60°C (AC16) for 30s, 72°C for 1 min, followed by a final extension period of 10 mins at 72°C. The lower annealing temperature was used for those stocks which gave poor yield at the higher temperature; in addition, for these stocks, the number of amplification cycles was increased to 38.

2.10.3 Visualization of microsatellite PCR products

PCR products were visualized using both agarose gel electrophoresis (AGE) and polyacrylamide gel electrophoresis (PAGE) as described previously. PCR product(s) were initially analyzed in 1.2% agarose gels after staining with ethidium bromide. This permitted sizing of the product and identified those stocks which failed to amplify. Stocks which failed to amplify under the standard PCR conditions were subjected to PCR again, using lower annealing temperatures and differing magnesium concentrations. DNA of stocks which consistently failed to amplify was tested for integrity by subjecting it to a PCR control using primers for the 18S ribosomal RNA gene (described in section 2.8). Stocks producing a product were subsequently analyzed by non-denaturing PAGE and silver staining as for PCR-RFLP analysis (section 2.8). These techniques have improved resolution and sensitivity over agarose gel electrophoresis / ethidium bromide staining. Stocks were scored by, and divided into, distinct pattern types visually.

2.10.4 PCR of microsatellite loci for Genescan® and Genotyper® 2.0 analyses.

PCR reactions and conditions were as above (section 2.10.2) but with several modifications. Fluorophore-labeled reverse primers were used in the reaction for labeling the PCR product (each primer was included at 5pmol concentration). PCR amplifications were carried out in microtitre plates with sealed lids and used the heated-lid option. Using microtitre plates enabled multichannel pipettes to be used for multiplexing as the PCR reaction for each stock was carried out in the same position in a separate microtitre plate. The use of the heated lid option obviated the need for an oil overlay: the latter complicates multiplexing and interferes with Genescan® analysis. Reaction volumes were reduced to 10μ L and PCR was carried out in an MJ Research PTC-200 Peltier thermocyler (Genetic Research Instrumentation Ltd., UK).

2.10.5 Genescan® and Genotyper® 2.0 analyses

To obtain an accurate size of the PCR product(s) for each stock at each microsatellite locus, stocks were subjected to Genescan® and Genotyper® 2.0 analyses. These methods, described in more detail in chapter 3, allow the detection and sizing of fluorescently-labeled PCR products from denaturing PAGE using the ABI 377 PRISMTM automated sequencing machine (Applied Biosystems, UK). PCR products are accurately sized by the inclusion of a fluorescently-labeled internal size marker.

Each reverse PCR primer was labeled with one of three fluorescent labels: AC01B-FAMTM (blue), AC16B-TETTM (green) or AC52B-HEXTM (yellow) (Perkin-Elmer, UK). As each different microsatellite locus PCR-product was labeled with a different fluorophore, all three PCR products for a given stock were run in the same lane of the gel, so called "multiplex analysis". PCR products were pooled in the ratio 1:50:10 (FAM:TET:HEX) for multiplex analysis. 2uL of the pooled product was added to 3.5uL of loading mix [containing ABI PRISMTM TAMRA^{TM-500} (red), Perkin-Elmer, UK]: 2uL of this mix was loaded for analysis. Each gel was used twice: in between

sample runs, the gel was pre-run for 30-60 minutes to ensure that all size marker bands had completely run out before the next loading. Sizing of the PCR products was carried out using Genescan® and Genotyper® version 2.0 programs.

2.10.6 Sequencing reactions

Sequencing reactions were carried out using a dye terminator cycle sequencing kit (ABI PRISMTM Dye Terminator Cycle Sequencing Ready Reaction Kit, Perkin-Elmer, UK), slightly modified from those given in the manufacturer's instructions. Sequencing reactions contained 4 μ L ABI sequencing reaction mix, 0.5pmol primer, 4 μ L ddH2O and 25ng genomic DNA in a total reaction volume of 10 μ L. Reactions were overlaid with 10 μ L mineral oil (Sigma) and amplified in a Hybaid Thermal Reactor (Hybaid Ltd, Middx., UK) using 35 cycles of 95°C for 30s, 50°C for 15s and 72°C for 4 min.

2.10.7 Sample preparation for automated sequencing

Sequencing reaction products were prepared for automated sequencing by adding 90μ L ddH₂O, followed by a pulse spin in a microfuge to separate the mineral oil and aqueous phase. The aqueous phase was transferred to a Sephadex G-50 spin column. The column was centrifuged at 1500 rpm for 3 minutes at room temperature and the eluate transferred to a fresh tube. The eluate, containing DNA from the sequencing reaction, was frozen in liquid nitrogen; DNA was precipitated from the eluate by freeze-drying overnight. DNA was stored dry at -20°C until ready for sequencing.

2.10.8 Automated sequencing

Dried DNA sequencing products were resuspended in 5μ L loading buffer containing dextran blue; 2μ L of this mix was loaded per lane. Automated sequencing was carried out on an ABI 377 PRISMTM automated sequencing machine (Applied Biosystems, UK). The sequences were edited and aligned using ABI PRISMTM Sequence Navigator version 1.01 (1984 -94, Applied Biosystems, UK).

2.11 <u>Population genetic analyses</u>

Population genetic analyses were carried out on the Huánuco stocks only. These stocks encompassed all those sampled, and hence represent an unbiased, indiscriminate sample. Furthermore, these stocks fulfill the criteria for population genetic tests (Tibayrenc & Ayala, 1991) in that they were collected from a small area over a discrete time period. The tests used for population genetic analyses will be described and discussed further in Chapter 6.

2.11.1 Analyses of isoenzyme data

Isoenzyme data is phenotypic but it is possible to infer genotypes from these electromorphs based on knowledge of an enzyme's structure (May, 1992).

2.11.1.1 Hardy-Weinberg test

For each enzyme locus, genotype frequencies were calculated and tested for departure from Hardy-Weinberg (HW) equilibrium using the Genepop package (Raymond & Rousset, 1995) version 3.1d (March 1999) [<u>http://www.cefe.cnrs-mop.fr/pub/</u><u>pc/msdos/genepop</u>]. Exact P-values associated with H₀ (i.e. null hypothesis of panmixia) were calculated using the probability test (exact HW test); no standard error value is computed for exact tests. In addition, the expected number of heterozygotes was calculated and used to determine the within-population fixation index (Fis, as described by Weir & Cockerham, 1984) for each locus.

2.11.1.2 Tests for detection of linkage disequilibrium

Three statistical tests for detecting linkage disequilibrium (LD) were performed: the Index of Association (I_A; Maynard Smith *et al.*, 1993); the D' index (Lewontin, 1964) and the r^2 (square of the correlation coefficient) index (Hill and Robertson, 1968).

2.11.2 Analyses of microsatellite data

Hardy-Weinberg and linkage disequilibrium analyses were carried out for the AC01 and AC16 loci only, since AC52 data was inconclusive for many of the Huanuco stocks. Tests were carried out as for enzyme data.

2.12 Phylogenetic analyses

2.12.1 Phylogenetic analysis of isoenzyme data

Isoenzyme phylogenies were constructed using quartet puzzling (Strimmer & von Haeseler, 1996), PUZZLE^O version 4.0.2 (1999) [ftp://ftp.ebi.ac.uk/pub/software (European Bioinformatics Institute, UK)]. This programme reconstructs phylogenies of nucleotide, amino acid and two-state data using a maximum-likelihood (ML) approach. Analyses were performed on "Kilburn" (an Origin 2000 public-access

Chapter 2 - Materials & Methods

supercomputer based at Manchester Computer Centre). ML estimations were calculated for 10,000 quartet puzzling steps. The ML distance matrix was reconstructed using the neighbour-joining method (Saitou & Nei, 1987). The number of invariant loci was automatically estimated from the data. All alleles were equally weighted, therefore highly polymorphic enzyme loci were assumed to be more informative than less polymorphic loci. Phylogenetic trees were viewed and printed from the TreeView programme [http://taxonomy.zoology.gla.ac.uk/rod/treeview.html. (Roderic Page, unpublished)]. It was not possible to calculate bootstrap values for these data since ML approaches are highly reiterative processes and as such are computationally too expensive. However, "quartet supports", measures similar to bootstraps (Strimmer & von Haeseler, 1996) were calculated. These values are the percentage support of each branch, representing the number of quartets supporting each given branching order.

For comparison, isoenzyme phylogenies were also constructed using a similarity approach (based on a similarity coefficient matrix, SCM, produced according to Jaccard's method, 1908). Using this approach, equal weighting is given to all alleles, as with the ML approach, but the "over-weighting" of polymorphic loci is removed. Dendrograms were constructed by UPGMA (the unweighted pair-group method using arithmetic averages) using the SYN-TAX-pc package (Podani, 1993).

2.12.2 Phylogenetic analysis of microsatellite data

Molecular phylogenies for sequence data from the AC01 microsatellite locus were constructed using the neighbor-joining method for 1000 bootstrap replications, using the Kimura 3-parameter model in PAUP* 4.0 beta version (Swofford, 1997). This distance-based method uses a comprehensive algorithm which takes into account reverse mutations, transition-transversion ratios and transversion type. A maximum likelihood approach was not taken for this data set as it was computationally too expensive to perform due to the added complexity (4 nucleotide states) of the data.

2.13 <u>Vector construction</u>

2.13.1 Strategy

The aim of this work was to attempt to demonstrate genetic exchange in *Leishmania*. Representatives of the Nicaraguan "parental" *L. V. braziliensis* and *L. V. panamensis* stocks (XD28 and ZF01 respectively) and the *L. V. braziliensis* stock, M2903, from which the hsp70 targeting fragment (see below) was derived, were selected for transformation to different drug resistance phenotypes. Hybrid offspring generated from crossing experiments would be selected by dual drug resistance phenotype.

2.13.2 Vectors

Kinetoplastids have unusual mechanisms for gene transcription. To ensure that these transformation vectors functioned in the *Leishmania* cell, these vectors carried, in addition to the resistance gene, processing elements involved in mRNA maturation (i.e. the 5'-spliced leader [SL] acceptor site and 3'-polyadenylation [pA] signal).

Vectors were designed so as to permit homologous integration of the vector DNA into the host cell's genome (as opposed to episomal vectors which replicate independently of the host cell's machinery). Integrative vectors were particularly required for the analysis of progeny generated from crossing experiments. The analysis of progeny is made easier with the use of integrative vectors because simple Mendelian inheritance is obeyed resulting in stable drug resistance phenotypes. With episomal vectors, the drug resistance phenotype of progeny may be unstable owing to the variable number of episome copies. To this end, *Leishmania*-specific gene targeting fragments were incorporated. Fragments targeting the *L. V. braziliensis hsp70* genes were isolated from the plasmid pT3T719U-133P (see below). The *hsp70* gene array is multicopy and constitutively-expressed to allow life-cycle-independent expression.

Two vectors, carrying genes conferring different drug resistance phenotypes, were required for transformation experiments: these were constructed from three existing plasmid vectors (*pT3T719U-133P*, *pTEX*, and *pTEX-ble*).

2.13.2.1 <u>*pT3T719U-133P*</u>

pT3T719U-133P is a pUC-based vector containing 133P, a 2472bp cDNA corresponding approximately to the C-terminal three-quarters of the coding sequence

and 3' untranslated region of the 70kDa heat shock protein gene (*hsp70*) from L. V. *braziliensis* strain M2903 (Amorim *et al.*, 1996). The coding region of 133-P contains a unique *Bg/II* site. The structure of *pT3T719U-133P*, is shown in *figure 2.13.2*.

Figure 2.13.2: pT3T719U-133P



Figure 2.13.2: structure of pT37719U-133P. MCS – multiple cloning site; UTR – untranslated region. pT37719U-133P was a gift from M. Carrington (Cambridge)

2.13.2.2 <u>*pTEX*</u>

pTEX is a *pBluescript*-based vector containing the gene encoding neomycin phosphotransferase II and has been described previously (Kelly *et al.*, 1992). The neomycin resistance gene (*neo^r*) and multiple cloning site (MCS) are both flanked by the 5'upstream and the 3'-downstream regions of the *T. cruzi* genes encoding glyceraldehyde phosphate dehydrogenase (*gapdh*). *pTEX* has been shown to confer resistance to neomycin in *T. cruzi* and *Leishmania* spp. (Kelly *et al.*, 1992). The vector also contains the ampicillin resistance gene (*amp^r*) derived from *pBluescript*. *pTEX* is maintained episomally in transformed parasites.

2.13.2.3 <u>*pTEX-ble</u>*</u>

pTEX-ble was constructed from *pTEX* (Nozaki & Cross, 1994) by replacing the *neo^r* of pTEX with the phleomycin resistance gene (*ble^r*) from *Streptoalloteichus hindustanus. Ble^r* confers resistance to phleomycin derivatives (Drocourt *et al.*, 1990).

2.13.2.4 Construction of the vector conferring neomycin resistance

pT3T719U-133P was digested with EcoRI. The digest was run out on a 0.8% agarose gel in TBE buffer containing ethidium bromide. The 2.5kb EcoRI-cut 133P fragment was excised from the gel and purified using the Geneclean[®] kit (Bio101 Inc, CA, USA) according to the manufacturer's instructions. pTEX was linearized with EcoRIand then treated with shrimp alkaline phosphatase (Boehringer) to prevent subsequent religation. The linearized pTEX and 133P insert were ligated overnight at room temperature using T4 DNA ligase (Boehringer).

2.13.2.5 Construction of the vector conferring phleomycin resistance

pTEX-ble was digested with PvuII and BamHI to isolate the gene for phleomycin resistance (*ble^r*). The digest was run out on a 0.8% agarose gel in TBE buffer containing ethidium bromide. The 0.5kb fragment containing *ble^r* was excised from the gel and purified as described previously. For ligation, the fragment was blunt-ended by DNA polymerase Klenow fragment using standard protocols (Sambrook *et al.*, 1989). *pT3T719U-133P* was linearized with *BgIII* and blunt-ended using DNA polymerase Klenow fragment (Boehringer). This was then treated with shrimp alkaline phosphatase to prevent subsequent religation. The *pT3T719U-33P* vector and *ble^r* fragment were ligated overnight as previously described.

2.13.3 Transformation of competent E. coli

Ligated pTEX-133P DNA was transformed into competent *E. coli* (strain DH5 α) cells. After heat-shocking (3mins at 42°C), cells were plated out on LB agar plates containing ampicillin (250U/mL) and incubated at 37°C overnight.

2.13.4 Selection of *E. coll* transformants

Colonies were picked and inoculated into LB broth containing ampicillin (250U/mL) and were incubated at 37°C overnight. Plasmid DNA was harvested from each culture using a standard alkaline lysis method (Birnboim & Doly, 1979) and was analyzed by

electrophoresis in 0.8% agarose gel. The expected size for the Neo^r (*pTEX-133P*) vector was 7.9kbp and for the *Ble^r* (133P-*ble*) vector was 5.4kbp. Plasmid DNA of the required size was subjected to digestion with *Bg/II*, *NcoI* and *Sa/I* to confirm the presence and orientation of the inserted DNA.

2.13.5 Bulk culture of plasmid vector DNA

Cultures containing the required vector DNA were subcultured into 200mL of LB broth containing 250U/mL ampicillin and grown up overnight at 37°C. Plasmid DNA was harvested for transformation as previously.

2.14 Transformation of Leishmania

2.14.1 Electroporation

Electroporation of *L. V. braziliensis* stocks M2903 and XD28, and *L. V. panamensis* stock ZF01 was carried out according to the method of Kelly *et al.* (1995b), based on conditions used for electroporation of *Trypanosoma cruzi*. The *L. V. braziliensis* stock M903 was included as an integration targeting control as this stock was the source of the *hsp*70 targeting sequence. Briefly, cells were washed in PBS, resuspended to 10^{9} /mL in 2 x 1mL electroporation buffer [272mM sucrose, 7mM sodium phosphate, pH 7.2], and 1mL aliquots transferred to labeled wells of a 24-well microtitre plate. After a 5 minute incubation on ice, $25\mu g pTEX-133P$ vector DNA was added to the required wells (DNA+); cells were pulsed (400V/99ms) 9x using the Hoeffer Progenitor electroporator (model PG1). The plate was incubated at 23° C overnight.

2.14.2 Selection of Leishmania transformants

Approximately 24 hours after electroporation, electroporated cultures were checked microscopically for viability. Subcultures were made from each electroporated well into α MEM liquid culture medium containing 0, 25, 50, 75 and 100µg/mL of the neomycin analogue, G418. Cells were incubated at 23°C, and checked daily for growth.

2.14.3 Southern analysis of G418-resistant Leishmania

Genomic DNA was extracted from uncloned transformed cells grown in liquid medium as described in section 2.3.1. Approximately 5µg genomic DNA was digested

Chapter 2 - Materials & Methods

using SaII, BamHI and PstI separately, and SaII/NcoI combined. Products were run out at 50V for 6 hours on a 1.2% agarose gel containing ethidium bromide. After photographing the gel, DNA was transferred to nylon membrane (Hybond N, Amersham) using standard Southern blotting protocols (Sambrook *et al.*, 1989), and subsequently bound to the membrane by UV irradiation. Probes for the neomycin resistance gene (from pGEM-*neo* and pTEX) were radiolabelled with $[\alpha^{-32}P]dCTP$ using the Rediprime[®] random-primer DNA labeling kit (Amersham International plc) according to the manufacturer's instructions. The membrane was hybridized overnight at 42°C in a solution containing 6X SSC [0.3M NaCl, 0.03M sodium citrate, pH 7.0], 5X Denhardt's, 0.5% SDS, 40% formamide, 100µg/mL sonicated salmon sperm DNA. Blots were washed after hybridization in 2x SSC and 0.2x SSC at 55°C before autoradiography at -70°C.

CHAPTER 3 - COMPARISON OF REFERENCE STRAINS

3.0 Introduction

This work aimed to clarify relationships of species within the *Leishmania* subgenus *Viannia*. To this end, new and established techniques were assessed for their ability to discriminate within and between the major *Leishmania* species found in the New World. Techniques found able to discriminate between the species of the subgenus *Viannia* were subsequently used to study the diversity of parasites from three areas (chapters 4-6).

3.1 Leishmania reference strains

Reference strains, selected as representative of *Leishmania* species present in the areas to be studied, were obtained from the *Leishmania* cryobank at LSHTM and from R. Naiff[†] (Instituto Nacional de Pesquisas Amazonas, Manaus, Amazonas, Brazil). The strains are listed in *table 3.1*.

Table 3.1: Leishmania reference strains

ISOLATE CODE NO.	riation)	HOST	ORIGIN	<i>LEISHMANIA</i> SPECIES	LESION TYPE
MHOM/BR/84/LTB300*	(Lb)	Human	Bahia / Brazil	L. V. braziliensis	MCL
MHOM/PA/71/LS94*	(Lpa)	Human	Panama	L. V. panamensis	CL
MHOM/BR/75/M4147*	(Lg)	Human	Para / Brazil	L. V. guyanensis	CL
MHOM/PE/84/LC39	(Lpe)	Human	Ancash/Peruvian Andes	L. V. peruviana	CL
IWHI/BR/94/M15065	(Ls)	Sandfly	Parana / Brazil	L. V. shawi	-
MHOM/BR/81/M6426*	(LI)	Human	Para / Brazil	L. V. lainsoni	CL
ISQU/BR/86/IM2832†§	(IM)	Sandfly	Amazonas / Brazil	L.V. sp. n. gp III	-
MHOM/BZ/82/BEL21*	(Lm)	Human	Belize	L. L. mexicana	CL
MHOM/BR/73/M2269*	(La)	Human	Para / Brazil	L. L. amazonensis	CL/DCL
MHOM/BR/74/PP75*	(Lc)	Human	Brazil	L. L. chagasi	VL

• indicates WHO-designated reference strains; § Grimaldi et al., 1991

3.2 Isoenzyme analysis

Eleven enzyme systems (twelve loci) were employed which were found to discriminate between all of the reference strains. *Table 3.2.1* lists the enzyme systems found to discriminate between reference strains of species from the *Viannia* subgenus. No additional data was gained using CAE since the resolution of bands from *Viannia* stocks was greater using TSGE.

Table 3.2.1: Enzyme systems discriminating between Vlannia reference strains.

	Lb					
Lpe	1, 3	Lpe				
Lg	1, 2, 5, 6, 7, 9,	1, 2, 3, 5, 6, 7,	Lg			
	10, 11	9, 10, 11				
Lpa	1, 2, 5, 6, 7, 10,	1, 2, 3, 5, 6, 7,	5, 9	Lpa		
	11	10, 11				
Ls	1, 2, 5, 6, 7, 9,	1, 2, 3, 5, 6, 7,	5, 6	5, 6, 9	Ls	
	10, 11	9, 10, 11				
Ll	1, 2, 3, 4, 5, 6,	1, 2, 3, 4, 5, 6,	1, 2, 3, 4, 5, 6,	1, 2, 3, 4, 5, 6,	1, 2, 3, 4, 5, 6,	Ll
	7, 8, 9, 10, 11	7, 8, 9, 10, 11	7, 8, 9, 10, 11	7, 8, 9, 10, 11	7, 8, 9, 10, 11	
IM	1, 4, 5, 6, 7, 8,	1, 3, 4, 5, 6, 7,	1, 2, 4, 5, 6, 7,	1, 2, 4, 5, 6, 7,	1, 2, 4, 6, 7, 8,	1, 2, 3, 4, 5,
	9, 10, 11	8, 9, 10, 11	8, 9, 10, 11	8, 9, 10, 11	9, 10, 11	7, 8, 9, 10

9-G6PD; 10-ASAT; 11-ALAT

Banding patterns observed in the reference strains are shown for the enzymes NHi, PEPD, 6PGD and ES in *figure 3.2.1*. For each enzyme locus, bands were scored from 1 - n, where 1 was the fastest migrating band and n was the slowest. Banding patterns are shown schematically for all enzymes studied in *figure 3.2.2*.











Figure 3.2.1: Isoenzyme banding patterns of the reference strains. Lanes 1 & 2 - L. V. braziliensis, lanes 3 & 4 - L. V. peruviana; lane 5 - L. V. guyanensis, lane 6 - L. V. panamensis; lane 7 - L. V. shawi, lane 8 - L. V. lainsont; lane 9 - L. V. sp (see text); lane 10 - L. L. amazonensis; lane 11 - L. L. mexicana; lane 12 - L. L. chagasi. Arrow indicates direction of electrophoresis.







Lane: 1 2 3 4 5 6 7 8 9 10 11 12











86







87



intense band

weak band










A genetic interpretation, in terms of alleles present, was possible for all twelve loci studied. Some stocks presented heterozygous genotypes, i.e. presented more than one band. Profiles for non-monomeric enzymes included additional heterozygous bands which were not scored (e.g. the central band of a triple-banded pattern produced by a heterozygote at a dimeric enzyme locus). *Table 3.2.2* summarizes the allele(s) present at each enzyme locus for each reference strain and gives the zymodeme designation for each. Each strain belonged, not surprisingly, to a separate zymodeme.

Table 3.2.2:	Reference strain allele a	nd zymodeme designations

REFERENCE STRAIN	ENZYME	MPI	NHi	NHi	NHd	ES	PEPD	PGM	6PGD	GPI	G6PD	ASAT	ALAT	Z*
CODE NO.	SPECIES		1	2										
LTB300	L. V. braziliensis	6/6	2/2	7/7	4/4	4/4	2/2	3/3	5/5	4/4	2/2	5/5	4/4	LON200
LC39	L. V. peruviana	7/7	2/2	7/7	5/5	4/4	2/2	3/3	5/5	4/4	2/2	5/5	4/4	LON201
LS94	L. V. panamensis	8/8	3/3	5/5	4/4	4/4	7/7	4/4	7/7	4/4	2/2	2/2	3/3	LON202
M4147	L. V. guyanensis	8/8	3/3	5/5	4/4	4/4	7/7	4/4	4/4	4/4	1/1	2/2	3/3	LON123
M15065	L. V. shawi	8/8	3/3	5/5	4/4	4/4	7/7	5/5	6/6	4/4	1/1	2/2	3/3	LON203
M6426	L. V. lainsoni	4/4	1/1	6/6	3/3	2/2	5/5	2/2	8/8	3/3	5/5	1/1	5/5	LON204
IM2832	L. V. sp.†	5/6	2/2	7/7	4/4	5/5	4/4	2/2	6/6	2/2	3/3	4/4	5/5	LON205
BEL21	L. L. mexicana	1/1	4/4	8/8	2/2	1/1	1/1	1/2	2/2	6/6	6/6	7/7	1/1	LON206
M2269	L. L. amazonensis	3/3	4/4	9/9	1/1	2/3	3/3	5/5	1/1	5/5	3/5	7/7	2/2	LON207
PP75	L. L. chagasi	2/2	1/1	9/9	1/3	1/3	6/6	5/5	3/3	1/1	4/4	3/6	6/6	LON208

Z = zymodeme; * LON numbers above 200 have been reassigned by the author as previous records have been lost (zymodemes are assigned based on the IEA conditions used, i.e. for *Leishmania Viannia* species). For enzyme names, see chapter 2. See *figure 3.2.2* for allele scoring scheme. Data given assumes diploidy, see introduction.

3.3 <u>Isoelectric focusing (IEF)</u>

IEF generated enzyme banding patterns using 5 out of the 11 enzyme systems (PGM, MPI, GPI, NH and PEPD) tested. Presumably, those enzymes which failed to produce visible products did so because they were rendered inactive at their pl. *Figure 3.3.1* shows the IEF result for NHi for reference strains and a selection of stocks from the Nicaraguan and Huanuco populations which were known to show different IEA banding patterns. For the 5 enzymes which could be visualized, IEF generated patterns similar to those seen by IEA for the reference strains (e.g. PGM, lanes 1-9, *figure 3.3.1*). Some IEF reference strain bands were, however, not at the equivalent position to that observed using IEA, for example *L. V. lainsoni*, lane 5, with NHi. It is not surprising that some pattern differences were observed since IEF separates by charge only, in contrast to IEA which separates both by size and charge.



Figure 3.3.1: IEF banding patterns of Leishmania reference strains using NHi. Lane 1 - L. V. braziliensis; lane 2 - L. V. peruviana; lane 3 - L. V. guyamensis; lane 4 - L. V. panamensis; lane 5 - L. V. lainsoni; lane 6 - L. V. sp. (see text); lane 7 - L. L. chagasi.; lane 8 - L. L. mexicana; lane 9 - L. L. amazonensis. Lanes 10-12 - Nicaraguan stocks: 10: XD28 (L. V. braziliensis); 11: XD09 (putative L. V. braziliensis) / L. V. panamensis hybrid); 12: ZF01 (L. V. panamensis). Lanes A-I - Huanuco stocks: A: HR80; B: HR108; C: HR110; D: HR399; E: HR410; F: HR413; G: HR419; H: HR434; I: HR529. Arrow indicates direction of electrophoresis.

IEF generated less informative banding patterns than IEA for NH, PEPD and MPI and equivalent information for GPI and PGM. Since only five enzymes could be studied using IEF yielding no further information, IEF was discontinued.

3.4 RAPD analysis

Of the 28 primers screened for discrimination between the selected reference strains (see Appendix IV), 19 yielded amplification products which were polymorphic and scorable. Nine primers generated no or too few bands or smears. Scorable bands ranged in size from approximately 500 - 2500 bp. Band intensity within and between profiles for a given primer was variable. Figures 3.4.1-3 show the RAPD profiles generated using 3 of the primers used (H4, L3, and H2 respectively).

3.4.1 Discrimination between *Leishmania* subgenera

RAPD analysis clearly discriminated between reference strains of the Leishmania and Viannia subgenera. Eleven primers (A4, A5, D1, D5, D6, D9, D10, H1, H2, H4, L3) generated profiles which included bands specific to (at least) the four main species of the Viannia subgenus i.e. L. V. braziliensis, L. V. peruviana, L. V. guyanensis and L. V. panamensis. Figure 3.4.1 and figure 3.4.2 show the profiles generated by the reference strains with primers H4 and L3 respectively: Viannia subgenus-specific bands are indicated by arrows marked "V". In both figures, the L. L. mexicana complex species (L. L. mexicana and L. L. amazonensis, lanes 6 and 7) also generate distinct profiles from that of L. L. chagasi (L. L. donovani complex, lane 8).

3.4.2 Discrimination within the subgenus Vlannia

RAPD analysis was able to discriminate between species complexes of the Viannia subgenus. Fifteen primers (A1, A3, A4, L1, L3 - 5 H1, H2, D3, D4, D6, D7, D9 and D10) generated profiles which discriminated between the two most important groups of the subgenus i.e. the *L. V. braziliensis* and *L. V. guyanensis* species complexes. Figures 3.4.2 and 3.4.3 show profiles generated by the reference strains with primers A3 and H2 respectively. Bands specific to the *L. V. braziliensis* complex or the *L. V. guyanensis* complex are indicated by arrows marked with "B" or "G" respectively.

RAPD analysis was also able to discriminate between member species of the L. V. braziliensis and L. V. guyanensis complexes. Three primers (A1, H1, H2) could differentiate between the L. V. braziliensis and L. V. peruviana reference strains e.g.

band indicated by arrow marked "Pe" for primer H2 (figure 3.4.3). Ten primers (A2, A5, D10, H1, H2, H4, L1, L3, L4, L5) could differentiate between the *L. V. guyanensis* and *L. V. panamensis* reference strains e.g. bands indicated by arrow marked with "Pa" for primer L3 (figure 3.4.2).

Whilst several primers generated similar profiles for the four major Viannia species reference strains (lanes 1-4, figure 3.4.1, primer H4), L. V. lainsoni, a more disparate species of the subgenus, commonly generated a profile distinct from those of the other Viannia species, e.g. lane 5 in figures 3.4.1 and 3.4.2.

RAPD profiles generated from reference strains



Figure 3.4.1:

Figures 3.4.1-3: RAPD profiles generated from reference strains after PCR amplification with primers H4, L3 and H2 respectively, visualized after electrophoresis through 1.2% agarose gel and ethidium bromide staining. Lane 1 - L. V. braziliensis; lane 2 - L. V. panamensis; lane 3 - L. V. guyamensis; lane 4 - L. V. peruviana; lane 5 - L. V. lainsoni; lane 6 - L. L. mexicana; lane 7 - L. L. amazonensis; lane 8 - L. Chagasi; lane 9 - negative control; lane 10 - 1kb molecular weight (mw) marker (Gibco BRL, UK). Arrows (marked with V, B, G, Pe, Pa) indicate bands specific for the Viannia subgenus, braziliensis complex, guyamensis complex, L. V. peruviana and L. V. panamensis respectively.



RAPD profiles generated from reference strains after PCR amplification with primer L3



Figure 3.4.3:

RAPD profiles generated from reference strains after PCR amplification with primer H2





3.4.3 Subsequent RAPD analyses

Despite initial success with this technique, subsequent analyses were found to be unreproducible. Visualization of later PCRs using the same DNA from the same reference strains revealed missing bands at both the higher and lower size range. The DNA was shown to be intact by electrophoresis and by PCR amplification of other, specific loci. The reagents used for RAPD analyses were rigorously controlled throughout and the same thermocycler was used but the irreproducibility remained unresolved. With the availability of other molecular techniques, and with the advent of microsatellite analysis, RAPD analysis was not investigated further.

3.4.4 Discussion of RAPD

Not unsurprisingly, most published works in which the RAPD technique has been applied comment favourably on its reproducibility. However, it is my experience that the technique is extremely sensitive to experimental parameters to the extent that reproducibility is hindered. Other groups have also discontinued the use of RAPD analysis after encountering difficulties (Cupolillo *et al.*, 1995). Private communications with co-workers has revealed that this lack of reproducibility is by no means isolated. Even authors who report the technique's applicability acknowledge that reproducibility, in terms of the number and intensity of bands, is not absolute (Noyes *et al.*, 1996).

The generation of reproducible RAPD fragment profiles, as with all techniques based on PCR amplification, is highly dependent on the PCR reagents (including source and concentrations of polymerase, DNA, nucleotides and magnesium chloride) and conditions used (cycling parameters, in particular the annealing temperature, and electrophoretic techniques). Since, to the best of my ability, these factors were kept constant, the lack of reproducibility encountered cannot be attributed to variation therein. In addition, RAPD analysis also employs short arbitrary-sequence primers and low-stringency annealing temperatures. As a consequence, amplification can result from contaminating DNA sequences, resulting in variation of the number of bands produced. Amplification of contaminating DNA was not a problem here, as evidenced by the lack of bands in negative control lanes. Short arbitrary sequence primers and low-stringency annealing temperatures also permit amplification from only partially-matched priming sites, giving rise to less intensely-staining bands. It has been reported that the number of bands generated by RAPD is highly sensitive to fluctuations in the annealing temperature, with as little as 1°C change affecting the

reproducibility of the profile (Williams et al., 1993; Noyes et al., 1996). It has been recently been suggested that the *Taq* source may affect the outcome - the negative control lanes were generally negative so it is unlikely that the *Taq* source used here (Bioline) was contaminated, but others (Carrasco et al., 1996 and Yeo et al., unpublished data) have reported that superior results are achieved using *Taq* obtained from Stratagene.

It is widely recognized that RAPD fragments of the same size are not necessarily homologous in sequence. Their use for inferring relationships is, therefore, questionable. Despite the limitations discussed here and in chapter 1 (section 1.10.2.4), phylogenetic and population genetic analyses of *Leishmania* stocks based on data from RAPD analysis have shown parity with isoenzyme, RFLP and karyotype data (Tibayrenc *et al.*, 1993; Dujardin *et al.*, 1995b; Gomes *et al.*, 1995; Pogue *et al.*, 1995; Schönian *et al.*, 1996).

In summary, RAPD analysis is reported by workers in some laboratories to be a valid method for generating data for inferring relationships between *Leishmania* stocks (Tibayrenc *et al.*, 1993; Dujardin *et al.*, 1995b; Noyes *et al.*, 1996; Banuls *et al.*, 1997). It is my opinion, however, that the information which may be obtained from the analysis of RAPD data is more consistently obtained using data from other techniques (see sections 3.6 and 3.8 below). Reproducibility between laboratories, and even co-workers in the same laboratory, is fundamental to the adoption of a technique. I concede that RAPD analysis is a means of generating large amounts of molecular data but believe that the technique is most powerful for, and applicable to, the study of organisms for which no other specific molecular markers are available.

3.5 <u>Riboprinting</u>

PCR amplification of the 18S (SSU) rRNA gene generated a product of approximately 2.2kb, the expected size calculated from sequence data. Figure 3.5.1 shows digests of this product using the enzymes TaqI and HpaII, visualized after electrophoresis through 1.2% agarose gel and ethidium bromide staining. Analysis of the reference strains using 4 of the enzymes (AluI, TaqI, Sau3AI and RsaII) discriminated clearly between L. L. chagasi and L. Viannia spp. (e.g. L. L. chagasi band of >0.5kb, arrow marked "C", in lane 8 TaqI digests figure 3.5.1). However, all 8 enzymes failed to discriminate between species within the Viannia subgenus, (e.g. HpaII digests, lanes 2-7 in figure 3.5.1).



Taq1 ,



HpaII



Figure 3.51: Riboprint profiles generated from reference strains after digestion of the 18S (SSU) rRNA gene PCR amplification product with TaqI and HpaII, visualized after electrophoresis through 1.2% agarose gel and ethidium bromide staining. Lane 1 - 1kb molecular weight (mw) marker (Gibco BRL, UK); lane 2 - L. V. braziliensis; lane 3 - L. V. peruviana, lane 4 - L. V. guyanensis; lane 5 - L. V. panamensis; lane 6 - L. V. lainsoni; lane 7 - L. V. braziliensis; lane 8 - L. L. chagasi. Arrow (marked with C) indicates bands specific for L. L. chagasi.

To maximize resolution and sensitivity of detection, digested products were also analyzed on polyacrylamide gels (PAGE) stained with silver. *Figure 3.5.2* shows *Sau3AI* and *RsaII* digests analyzed by PAGE and silver staining. Although detection using PAGE followed by silver staining is more sensitive than by agarose electrophoresis and ethidium bromide staining, no further information was gained by analyzing riboprinting products using this approach. Riboprinting was unable to differentiate between species of the subgenus *Viannia* and its further use in this study was therefore precluded.

Figure 3.5.2:

Riboprint profiles generated from *Leishmania* reference strains after digestions with *Sau*3AI and *Rsa*I.



Figure 3.5.2: Riboprint profiles generated from reference strains after digestion of the 18S (SSU) rRNA gene PCR amplification product with Saw3AI and RsaI, visualized after electrophoresis through 6% polyacrylamide gel and silver staining. Lane 1 - 1kb molecular weight (mw) marker (Gibco BRL, UK); lane 2 - L. V. braziliensis; lane 3 - L. V. panamensis; lane 4 - L. V. guyanensis; lane 5 - L. V. peruviana; lane 6 - L. V. lainsont; lane 7 - L. V. braziliensis; lane 8 - L. L. chagasi

It is perhaps not surprising that riboprinting failed to discriminate within the Viannia subgenus since the major species of this subgenus are known to be very closely

related. Riboprinting was able, however, to differentiate between reference strains of the more distantly-related *Leishmania* and *Viannia* subgenera. For further discussion of riboprinting, see section 3.6.3.

3.6 PCR-RFLP analysis of rRNA gene ITS region

3.6.1 PCR amplification of the rRNA gene ITS region

ITS PCR amplification products of *Viannia* species were approximately 1kb, which is the expected size, and in agreement with previous reports (Cupolillo *et al.*, 1995). *Figure 3.6.1* shows the rRNA ITS PCR products obtained from *Leishmania* reference strains, visualized after agarose gel electrophoresis and ethidium bromide staining. PCR products from species of the subgenus *Leishmania* (i.e. *L. L. mexicana, L. L. amazonensis* and *L. L. chagasi*) were larger (approx. 1.1-1.2kb) and, as such, could be clearly distinguished from the products of *Viannia* species on product size alone, without RFLP analysis.

Figure 3.6.1: Amplification products of the rRNA ITS region from Leishmania reference strains.



Figure 3.61: Amplification products of the ribosomal RNA (rRNA) internal transcribed spacer (ITS) region from Leishmania reference strains, visualized after electrophoresis through 1.2% agarose gel and ethidium bromide staining. Lane 1 - 1kb molecular weight (mw) marker (Gibco BRL, UK); lane 2 - L. V. braxiliensis; lane 3 - L. V. peruviana; lane 4 - L. V. guyanensis; lane 5 - L. V. panamensis; lane 6 - L. V. lainsont; lane 7 - L. L. chagasi; lane 8 - L. L. amazonensis; lane 9 - L. L. mexicana; lane 10 - L. V. braxiliensis. Arrows (marked with V, C, M) indicate bands specific for the Viannia subgenus, L. L. chagasi or L. L. mexicana complex respectively.

3.6.2 PCR-RFLP analysis

PCR-RFLP products were visualized by electrophoresis through 6% polyacrylamide gels and silver staining to maximize both the fragment resolution and sensitivity of detection. RFLP fragments were scored visually to identify enzymes which discriminated between the *Viannia* and *Leishmania* subgenera and between species within the *Viannia* subgenus.

RFLP analysis revealed polymorphism among the *Leishmania* reference strains using all enzymes tested. *Figure 3.6.2* shows the RFLPs produced using two of these enzymes: *FspI* and *ScrFI*. All enzymes used differentiated between reference strains of the *Leishmania* and *Viannia* subgenera: fragments indicated by arrows marked with "M" and/or "C" in *figure 3.6.2* are unique to the *L. L. mexicana* complex and *L. L. chagasi* respectively. RFLPs of the *L. L. mexicana* complex species were similar but could always be differentiated; *L. L. mexicana* and *L. L. amazonensis* could clearly be separated for example, by the presence of a fragment of approx. 140bp in *L. L. amazonensis* using *FspI* (arrow marked "A", lane 8 of *FspI* digest, *figure 3.6.2*).

The overall RFLPs of the *Viannia* reference strains were very similar using all enzymes (see lanes 2-6, *figure 3.6.2*). Subgenus-specific bands were commonly observed, e.g. fragments of approx. 150bp (*FspI*) and 50bp (*ScrFI*) indicated by arrows marked with "V" (lanes 2-6, *figure 3.6.2*). Five enzymes (*FspI*, *Bst*UI, *ScrFI*, *TaqI*, *SphI*) also differentiated between the species complexes of the subgenus. For example, species of the guyanensis complex (*L. V. guyanensis* and *L. V. panamensis*, lanes 4 and 5) could be differentiated from other *Viannia* species (lanes 2, 3 and 6) by the absence of another fragment of approx. 170bp using *FspI* (*figure 3.6.2*). *L. V. lainsoni* could be separated from other members of the subgenus using 5 of the enzymes used (*SphI*, *ScrFI*, *Csp6I*, *DpnI*, *TaqI*), clearly indicating its more distantly related position within the group.

In addition to discrimination at the species complex level, several enzymes generated RFLPs which, using a combination of fragment presence and absence, allowed individual species to be identified. For example, *L. V. guyanensis* could be separated from *L. V. panamensis* by the presence of a *FspI* fragment of approximately 250bp (arrow marked "G", lane 4, *figure 3.6.2*). Members of the *guyanensis* complex could be separated using 3 enzymes (*FspI*, *TaqI*, *SphI*).





Figure 3.6.2 RFLP analysis of ribosomal RNA (rRNA) internal transcribed spacer (ITS) region amplification products from *Leishmania* reference strains, visualized after electrophoresis through 6% polyacrylamide gel and silver staining. Lane 1 - 100bp molecular weight (mw) marker (Gibco BRL); lane 2 - L. V. braziliensis; lane 3 - L. V. peruviana; lane 4 - L. V. guyamensis; lane 5 - L. V. panamensis; lane 6 - L. V. lainsoni; lane 7 - L. L. mexicana; lane 8 - L. L. amazonensis; lane 9 - L. L. chagasi Arrows (marked with V, C, M) indicate fragments specific for the Viannia subgenus, L. L. chagasi or L. L. mexicana complex respectively. The arrows marked with "G" and "A" indicate fragments unique to L. V. guyamensis and L. L. amazonensis respectively. The arrow marked with M/pe indicates a fragment shared by the L. V. peruviana reference strain and members of the L. L. mexicana complex.

Within the *braziliensis* complex, polymorphism was limited to 2 enzymes, *Scr*FI and *BstUI*. Using both of these enzymes, the RFLP patterns of the *L. V. peruviana* reference strain presented fragments not seen in any other *Viannia* reference strain. These fragments were not unique to *L. V. peruviana*, but appeared to be shared with members of the *Leishmania* subgenus. For example, the *L. V. peruviana* fragment of approx. 110bp indicated by the arrow marked "M/pe" generated using *Scr*FI (*figure 3.6.2*) is shared with members of the *L. L. mexicana* complex.

3.6.3 Discussion of PCR-RFLP analysis of the rRNA ITS region

PCR-RFLP analysis of the rRNA ITS region generated data that discriminated the *Leishmania* reference strains to the species and species-complex level, in agreement with previously published work for *Leishmania* (Cupolillo *et al.*, 1995). A great advantage of this technique is that it can be carried out on clinical material without recourse to isolation of the parasite as, in contrast to RAPD analysis, the PCR primers employed are specific for *Leishmania*. In the work presented here however, PCR amplification was carried out on DNA extracted from cultured *Leishmania*.

PCR amplification of the rRNA ITS region revealed single bands using agarose gel electrophoresis and ethidium bromide staining (*figure 3.6.1*), however, RFLP analysis revealed complex fragment patterns which were found to be not entirely reproducible using products from different PCR reactions (data not shown). Polyacrylamide gel electrophoresis and silver staining of the PCR amplification products revealed that additional bands were generated during PCR which had not been visible using agarose gel electrophoresis and ethidium bromide staining (data not shown). It was not possible to determine whether these additional bands were artefactual or due to ITS sequence variants, however these bands did not disappear after extensive work to reoptimize the PCR conditions. The problem of reproducibility was circumvented by extracting and "cleaning" the major PCR product prior to RFLP analysis. RFLPs generated by digestion of "cleaned" PCR amplification products were reproducible. Minor, less intensely-staining bands observed after RFLP analysis of "cleaned" products were also seen in some strains and possibly indicate sequence heterogeneity of ITS sequences within these individuals (section 3.5; Stothard *et al.*, 1996).

The requirement for PCR products to be "cleaned" in order to generate reproducible RFLP patterns entailed a great deal of work bearing in mind the number of stocks to be studied. Since the discriminatory power of this technique within the *Viannia* subgenus was greatest at the species complex level, PCR-RFLP analysis of the rRNA ITS region was only carried out on stocks from the Nicaraguan population. These stocks consisted of parasites belonging to both *braziliensis* and *guyanensis* complexes i.e. *L. V. braziliensis, L. V. panamensis* and putative hybrids thereof. Stocks from the Huånuco and Brazilian parasite populations, which mostly comprised parasites from the *braziliensis* complex, were not analyzed by this technique.

For genes such as the rRNA genes, variation in gene coding sequence is likely to be lacking in closely related taxa due to the conservation imposed by functional constraints on the gene product. Analysis of gene sequence variation between such closely related taxa by the restriction enzyme digestion of such a conserved gene is, therefore, somewhat limited in that enzyme recognition sequences cover only a fraction of the small number of potentially variable sites in the gene. The lack of discrimination between *Viannia* species by riboprinting is evidence of the conserved nature of the 18S rRNA gene. The conserved nature of this gene among *Viannia* species was also reported by Uliana *et al.* (1994). In their study, synthetic oligonucleotides corresponding to areas of known sequence diversity in the *Leishmania* SSUrRNA gene were hybridized to PCR products from different *Leishmania* species. Discrimination of all the major *Leishmania* species complexes was evident except those within the *Viannia* subgenus.

In contrast to riboprinting however, PCR-RFLP analysis of the rRNA ITS was found to discriminate among species of the subgenus *Viannia*. This discrimination is evidence that the spacer region is less conserved than the rRNA genes themselves, which are constrained due to their essential function. *L. V. braziliensis* is estimated to have in the region of 160 copies of the rRNA repeating unit making up this multigene family (Villalba & Ramirez, 1982). Sequence heterogeneity of the rRNA repeat unit is thought to arise from concerted evolution, involving the mechanisms of unequal crossing over and biased gene conversion, resulting in the rapid accumulation of mutations in such families. Macedo *et al.* (1992) point out that sequence heterogeneity could also be due to mixed parasite populations in an isolate. Mixed populations can be excluded here since all of the reference strain parasites used were derived from clones.

3.7 Denaturing gradient gel electrophoresis (DGGE)

The rRNA repeat unit is highly conserved among closely-related taxa. The 18S rRNA gene in particular has been widely used for constructing phylogenies with the assumption that its sequence, within an organism, is highly conserved. It has recently been reported however, that sequence heterogeneity can be found in rRNA genes of parasites using DGGE (Gasser *et al.*, 1996). Sequence heterogeneity amongst gene copies of the 18S rRNA multigene family in *Plasmodium falciparum* and the free-living flatworm *Dugesia mediterranea* has also been reported (Kain *et al.*, 1996).

DGGE is a relatively new technique which has been shown to be capable of distinguishing sequence variants among a DNA product containing sequence variants of identical size (Lessa *et al.*, 1992; Gasser *et al.*, 1996; Stothard *et al.*, 1998b). The melting profile of a DNA fragment (i.e. the point at which strand-separation occurs) is dependent upon the sequence of the molecule (Michaelides *et al.*, 1995). DNA strands of divergent sequence, when denatured through a suitable gradient, will melt successively according to their sequence: single strands of different sequence exhibit distinct mobilities which can be visualized. It has been reported that DGGE permits the detection, under optimal conditions, of single base changes (Myers *et al.*, 1987; 1990).

DGGE was used here to assess homogeneity within the 18S rRNA gene of *Leishmania* subgenus *Viannia* species. To establish the melting profile of the 18S (SSU) rRNA PCR product, *Alu*I-digested *L. V. braziliensis* (LTB300) product was electrophoresed through a perpendicular gradient gel in which the direction of electrophoresis is at right angles to the chemical gradient (urea - gradient range 0 - 7M). *Figure 3.7.1* shows the melting profile of this product using perpendicular gradient DGGE.

At lower concentrations of urea (left-hand side of the gel), the AluI-digested 18S rRNA gene PCR product fragments run as native, double-stranded (ds) DNA. Progressive strand separation (melting) occurred in the range 0.7 - 1.75M (10 - 25%) urea. The smaller digestion products (C and D) remained as single bands after denaturation, indicating the homogeneity of the sequences in both strands of these fragments. The largest fragment (A), which had 2 melting domains, resolved into 2 single strands after denaturation, indicating the presence of two distinct sequences in this fragment. The second largest fragment (B) denatured resolving 4 single stranded curves, indicating that four distinct sequences were present in the strands of this fragment. The 18S rRNA gene copies of *L. V. braziliensus* (LTB300) were thus shown to consist of at least 4 heterogeneous sequences.

Figure 3.7.1:

Melting profile of the *Alu*I-digested *L. V. braziliensis* 18S (SSU) rRNA PCR amplification product generated using perpendicular gradient DGGE.





To see if sequence heterogeneity in 18S rRNA gene copies was conserved across the *Viannia* subgenus, parallel DGGE analysis was carried out on the *Alul* digestions of 18S rRNA PCR products from six *Leishmania* stocks. In parallel DGGE, the denaturation gradient runs from top (low concentration) to bottom (higher concentration) in the same direction as electrophoresis. Parallel DGGE gels examine a narrow "window" of the original perpendicular gel and allow the comparative visualization of single-stranded product(s) from different strains or stocks. A parallel gradient gel, with the gradient spanning 10-25% 7M urea (the denaturation range identified in perpendicular DGGE), was used. *Figure 3.7.2* shows parallel DGGE analysis of the *Alul*-digested 18S rRNA PCR products from 6 *Leishmania* strains.



Figure 3.7.2:

DGGE analysis of *Alu* I-digested 18S (SSU) rRNA gene amplification products from *Viannia* reference strains.

Figure 3.7.2: DGGE (parallel gradient) analysis of Alul-digested 18S (SSU) rRNA PCR amplification products from Leishmania reference strains. Lane 1 - Ikbp moleclar weight marker; lane 2 - L. V. braziliensis (LTB300); lane 3 - L. V. peruviana (LC39); lane 4 - L. V. panamensis (LS94); lane 5 - L. V. guyanensis (M4147); lane 6 - L. V. shawi (M15065); lane 7 - L. L. chagasi (PP75). Arrows (A - C) correspond to single-stranded species arising from denaturation of native (double-stranded) Alul digestion products A-C in figure 3.7.1, numbers correspond to the number of single-stranded species determined after denaturation Faint / diffuse bands e.g. \leftarrow * arise from the denaturation curves of the digestion products The sequence variants resolved from fragments A (2) and B (4) are present in all species tested. Denatured fragment C (homogenous) is also visible; fragment D (also homogenous) is not evident because its position falls outside of the parallel gel "window". Other, more diffuse, bands can also be observed in this "window". Bands interspersed with the sequence variant bands resolved from fragments A and B are other portions of the fragment melting curves (stain reddish-brown not black, not reproducible in print); higher bands are likely to be denatured undigested DNA and are not relevant here. Hence, the number of sequence variants resolved for each fragment in parallel DGGE was identical for all *Viannia* species examined.

3.7.1 Discussion of DGGE

rRNA genes are amongst the slowest evolving (Olsen *et al.*, 1986) and hence many phylogenetic analyses of parasitic protozoa have been inferred from rRNA gene sequence data (Sogin *et al.*, 1986; Fernandes *et al.*, 1993; Maslov *et al.*, 1996; Vickerman, 1994). The majority of analyses have compared single or few sequences from a small number of species/strains to construct phylogenies (e.g. Fernandes *et al.*, 1993). Homogeneity in gene copies of a multi-gene family, such as the rRNA gene family, is maintained by concerted evolutionary mechanisms (Zimmer *et al.*, 1980). Here, DGGE analysis of PCR products from the 18S rRNA gene showed the presence of at least 4 sequence variants in gene copies from cloned strains of *Viannia* species. The existence of these 4 rRNA sequence types in all of the *Leishmania* species tested suggests that the variation in these sequences evolved prior to speciation within the genus; it would be interesting to see if Old World species also exhibit this variation.

Sequence variants have also been reported among 18S rDNA copies in other parasite taxa (Carranza et al., 1996; Stothard et al., 1996). Sequence diversity in such genes is most likely to occur at silent sites (in which case the function of the gene product remains unaffected). As has been suggested recently (Carranza et al., 1996; Gasser et al., 1996; Stothard et al., 1996) diversity in 18S rRNA gene copies from a single organism should lead to a more cautious use of the gene for inferring phylogeny.

3.8 Microsatellite analysis

Microsatellite analysis as a technique for discriminating between species of the *Leishmania* subgenus *Viannia* has only very recently been described (Russell *et al.*, 1999). Here, results are reported for reference strains of the *Viannia* subgenus for three microsatellite loci: AC01, AC16 and AC52. Microsatellite analysis of all stocks was carried out and will be discussed further in chapters 4-7.

3.8.1 Specificity of the microsatellite loci for the subgenus Vlannia

In agreement with Russell et al. (1999) PCR amplification products were not generated for any of the three microsatellite loci from reference species of the *Leishmania* subgenus (*L. L. mexicana* -BEL21, *L. L. amazonensis* - M2269 and *L. L. chagasi* - PP75). DNA from these stocks did however amplify the 18SrRNA gene indicating that the DNA itself was intact.

3.8.2 Amplification of microsatellite loci

All *Viannia* reference strains amplified the AC01 locus, generating a product of the expected size range (approx. 220 - 250bp). *Figure 3.8.2.1* shows the AC01 microsatellite products from reference strains visualized using agarose gel electrophoresis and ethidium bromide staining.

Figure 3.8.2.1 Amplification products of the AC01 microsatellite locus from Leishmania reference strains.



Figure 3.8.2.1: Amplification products of the microsatellite ACO1 from Leishmania reference strains, visualized after electrophoresis through 1.2% agarose gel and ethidium bromide staining. Lane 1 - L. V. braziliensis; lane 2 - L. V. peruviana; lane 3 - L. V. guyanensis; lane 4 - L. V. panamensis; lane 5 - L. L. chagasi; lane 6 - negative control; lane 7 - 100bp molecular weight (mw) marker (Gibco BRL, UK); Arrow indicates the ACO1 product.

Products from members of the guyanensis complex appeared slightly larger than those of the braziliensis complex (see figure 3.8.2.1). All Viannia reference strains, with the exception of IM2832, amplified the AC16 and AC52 loci, with product sizes ranging from approx. 240 - 270bp and 280 -310bp respectively (data not shown). The band sizes of all products were reproducible. The amplification products from the reference strains were also visualized using non-denaturing PAGE and silver staining as this approach allows higher resolution and greater sensitivity. However, in the case of the reference strains no further information was gained in terms of additional bands and the data is not shown

3.8.3 Genescan@ and Genotyper@ 2.0 analyses

PCR product(s) for each reference strain from each microsatellite locus were accurately sized using Genescan[®] and Genotyper[®] software. These programmes, allow the detection and sizing of fluorescently-labeled PCR products from denaturing PAGE using the ABI 377 PRISMTM automated sequencing machine (Applied Biosystems, UK). This approach has not been used previously for *Leishmania*.

Genescant is a software programme which detects fluorochrome emission from fluorescently-labeled DNA after laser beam-excitation. *Figure 3.8.3.1* shows a typical gel image generated by Genescant for these 3 microsatellite loci. Odd numbered lanes are loaded and run-in first followed by even-numbered lanes, giving rise to the staggered appearance of the bands. Size standard markers (ABI PRISMTM TAMRATM-500) are red and appear symmetrically distributed across the lanes on the gel. Lanes are indicated by blue triangles at the top of the gel. Manual manipulation of lane "tracker lines" was required to ensure that the software scored all bands in any given lane. This is necessary for every new run to overcome variations in the gel. The microsatellite products are represented as coloured bands: blue for the AC01 locus, green for the AC16 locus and yellow for the AC52 locus. These bands are not symmetrically distributed, reflecting the different allele sizes present.

Data generated by Genescant was imported into Genotypert for the analysis of loci within individual lanes. The output of each lane is visualized, by microsatellite locus, as a chromatogram in which peaks represent the labeled DNA. *Figure 3.8.3.2* shows the chromatogram of *Viannia* reference strains for the AC01 locus.

Figure 3.8.3.1: Genescan® gel image showing microsatellite loci and size standard markers.



Figure 3.8.3.1 Genescan gel image showing microsatellite loci (AC01 - blue, AC16 - green, AC52 - yellow) and size standard markers (ABI PRISMTM TAMRATM-500 - red). Lanes are shown by blue triangles at the top of the gel. Odd numbered lanes are loaded and run in first followed by even-numbered lanes, giving rise to the staggered appearance of the bands. The data generated by Genescan® is imported into Genotyper® for analyses of loci within individual lanes.



Figure 3.8.3.2: Genotyper \textcircled output for the AC01 locus for Viannia reference strains. Numbers across the top of the figure represent size, in bases. The exact sizes of individual peaks were scored automatically by the software. Exact peak sizes were assigned to "bins" which allows grouping of alleles, manually (see text). The scale to the right of each trace indicates the peak intensity. Intensities of less than 30 units were not scored (the *L. V. panamensis* peak was scored from another Genescan run). Where more than one peak was present, other criteria were taken into account. Criteria included: 1) peak height: all bands of equal intensity were scored; 2) pattern of the peaks - stutter bands (see text) were ignored; 3) information from other electrophoretic/staining methods was taken into account when deciding whether a peak was real (i.e. an allele) or artefactual. See text for further discussion.

Allele scoring was subject to visual interpretation which required a degree of knowledge regarding the locus concerned. The Genotyper® results for the AC01 locus were generally very clear-cut and could be corroborated using data from PAGE and sequencing analyses. Sequencing data was not available from the AC16 and AC52 loci: since PAGE analysis of these loci revealed complex banding patterns, allele scoring results for these two loci are preliminary and require confirmation. Several basic rules were applied for scoring peaks as alleles. The scale to the right of each trace (see figure 3.8.3.2) indicates the peak intensity: peaks with intensities of less than 30 units were not scored unless unambiguous. All peaks of equal intensity were scored: stocks showing two peaks were considered to be heterozygous at that locus. Single peaks were assumed to indicate homozygosity although it is possible that the second allele was "null" and therefore failed to amplify. Indeed, homozygous null alleles were identified among stocks of the three populations under study (see chapters 4-6). In cases where peaks of unequal intensity were observed, information from other electrophoretic/staining methods was taken into account when deciding whether a peak was real (i.e. an allele) or artefactual (see below). For example, many stocks from the Nicaraguan population sample generated 2 peaks for the AC01 locus: PAGE analysis suggested that these stocks were heterozygous at this locus, however the Genotyper peaks were of very unequal intensities (see chapter 5). Both peaks corresponded to allele sizes found in other stocks and hence, despite their unequal intensities, both peaks were scored giving rise to a heterozygous genotype. It has been suggested that in such cases, the allele generating the peak of low intensity has a polymorphism in the PCR priming site such that binding of the associated PCR primer is reduced (Wang et al., 1999).

Additional peaks were also observed in some stocks. "Stutter peaks", additional bands which are thought to result from DNA slippage during *in vitro* PCR amplification (Schlotterer, 1998), were commonly observed, especially for the AC52 locus which made assigning alleles difficult. The pattern of stutter bands is generally locus-specific: the pattern is usually observed in the traces of all samples at that locus (Schlotterer, 1998) and can also be visualized in PAGE analysis. Stutter bands for the loci studied here were smaller than the actual allelic band and were evident as a ladder, with peaks of increasing intensity up to the allelic peak (e.g. LC39 peak, *figure 3.8.3.2*). Stutter bands were not scored. In addition to stutter peaks, other peaks were also sometimes observed above the size range expected for the locus (data not shown);

these peaks were generally ignored. It is possible that these bands result from the terminal transferase activity of *Taq* polymerase, used for PCR amplification, as described by Schlötterer (1998).

Peak sizes were calculated automatically by the Genotyper software. Peaks were manually assigned to "allele bins": each bin contains groups of alleles likely to contain microsatellite repeats of the same size. This was done by using a well-resolved peak to determine whether the size of the allele was odd or even (in numbers of base pairs) and to establish a correction factor to apply to the other alleles. At the AC01 locus, for example, the LTB300 peak size was scored as 227.47: the allele size was rounded to the nearest whole number, i.e. 227. The correction factor to apply to all other peak sizes was thus -0.47 (227 - 227.47). All other AC01 peak sizes were then assigned to bins by subtracting 0.47 before rounding to the nearest number. For example, the *L*. *V*. sp. reference strain IM2832 generated 2 peaks of sizes 227.63 and 245.88: applying the correction factor (giving 227.16 and 245.41 respectively) and rounding to the nearest number gives allele sizes of 227 and 245 respectively. Since the microsatellite repeat unit consisted of dinucleotides, allele sizes generally varied by units of 2 base pairs (excluding insertions and deletions in the flanking region, see section 3.8.4.).

It has been reported that identical alleles usually migrate within 0.5bp of each other on a gel, with larger variations observed between identical alleles from different gels (David & Menotti-Raymond, 1998). In order to maintain consistent allele bin designation between gels, the L. V. braziliensis reference strain, LTB300, was included as a control on all gels. Samples producing no or poorly resolved peaks, or peaks of low intensity, were re-run and / or repeated. The reference strains of L. V. braziliensis and L. V. panamensis (LTB300 and LS94 respectively) were homozygous at all three loci. The L. V. peruviana reference strain (LC39) was homozygous at AC01; its status at AC16 and AC52 was not determined. The reference strain of L. V. guyanensis (M4147) was heterozygous at AC01, homozygous at AC16, with its status at AC52 undetermined. The reference strains of L. V. shawi (M15065) and L. V. lainsoni (M6426) were homozygous for AC01 and heterozygous for both AC16 and AC52. L. V. sp. (IM2832) was heterozygous at AC01 but had "null" alleles, i.e. failed to amplify, the AC16 and AC52 loci. The sizes of the PCR amplification products (alleles) of the Viannia reference strains are summarized, by microsatellite locus, in table 3.8.3.1. The peaks obtained reflect sizes of the alleles at each locus; single

peaks were assumed to indicate homozygous alleles. Each strain was assigned a distinct microsatellite multilocus genotype on the basis of combined locus genotypes observed.

			Microsatellite locus					
				201	AC	216	AC52	
<i>Viannia</i> species	Strain Code No.	Multilocus genotype	allele 1ª	allele 2 ^b	allele 1	aliele 2	allele 1	allele 2
braziliensis	LTB300	LON1	227	227	244	244	283	283
peruviana	LC39	LON2	231	231	ND	ND	ND	ND
guyanensis	M4147	LON3	225	241	242	242	ND	ND
shawi	M15065	LON4	241	241	246	248	297	299
panamensis	LS94	LON5	239	239	244	244	291	291
lainsoni	M6426	LON6	231	231	240	270	281	283
sp. n.	IM2832	LON7	227	245	x	х	x	х

Table 3.8.3.1Summary of the product (allele) sizes of the Vianniareference strains for each microsatellite locus.

Table 3.8.3.1: ^a allele bin size, in base pairs, of the smaller allele. Assignment of allele bins is explained in the text. ^b allele bin size, in base pairs, of the larger allele. Where alleles 1 and 2 are given as identical, it has been assumed that the sample is homozygous at that locus. It is also possible that a second allele was "null", i.e. failed to amplify. Scoring of peaks is described in the text. ND indicates than Genescan® analysis failed to determine a peak which could be called with any confidence. X indicates that no PCR amplification was achieved, probably due to mutation in the primer binding site such that alleles at this locus are "null".

3.8.4 Sequencing of the AC01 microsatellite locus

Sequencing was carried out for the AC01 locus only. Sequencing of products from the AC16 locus was attempted but failed to generate readable sequence data, probably due to overlapping of multiple DNA sequences (data not shown). These mixed sequences arose from the presence of multiple PCR amplification products, discussed in *section 3.8.2*. Isolation and purification of individual bands for this locus proved to be technically impossible in the short time available due to the lack of band resolution in agarose gel. Sequencing of the AC16 locus, and similarly the AC52 locus, was therefore not carried out.

All *Viannia* reference strains generated readable AC01 sequence except for *L. V.* sp. (IM2832). Each reference strain was sequenced at least twice; the sequence was reproducible. *Figure 3.8.4.1* shows a typical electropherogram from sequencing analysis of the AC01 microsatellite locus, shown here for the *L. V. braziliensis* reference strain LTB300. The microsatellite repeat sequence $(TG)_n$ can be seen between bases 50 - 70; for LTB300, n = 8.

Sequence at the extreme 5' end was often ambiguous hence for phylogenetic analyses, sequences were aligned commencing with the motif of bases "ATG" which was present in all samples sequenced. *Figure 3.8.4.2* shows the alignment of sequence from the AC01 microsatellite locus for reference strains of the *Viannia* subgenus. The sequences of the *L. V. braziliensis* and *L. V. peruviana* reference strains were identical before the microsatellite repeat motif and varied at only 5 heterologous peaks downstream of this. Complex-specific variation was observed: all *guyanensis* complex species had "C" not "T" at positions 28 and 135, the number of dinucleotide repeats was 14 or greater. Within this complex, each reference strain could be differentiated: *L. V. shawi* and *L. V. guyanensis* had "CG" insertions in the repeat motif. *L. V. shawi* also had a 4 base pair insertion immediately before the repeat motif.

Notes to figure 3.8.4.2 (see page 117) Alignment of sequence data from Viannia reference strains at the AC01 locus. L. V. b - L. V. braziliensis; L. V. pe - L. V. peruviana; L. V. g - L. V. guyanensis; L. V. s - L. V. shawi; L. V. pa - L. V. panamensis; L. V. I - L. V. lainsoni. Numbers above sequence data indicate the base position used for alignment purposes. "" represents an identical base to that seen in the LTB300 (L. V. braziliensis) sequence; "" represents a gap. Gaps between positions 88 - 91 have been introduced to account for additional repeats in stocks from other populations. The sequences were edited and aligned using ABI PRISMTM Sequence Navigator version 1.01 (1984 -94, Applied Biosystems, UK). Ambiguous bases were scored visually: nucleotide codes were assigned to peaks according to the standard coding system [see Appendix IV] which is the format required by phylogenetic packages. Sequence data. Sequence alignments were made using 165 bases of the total sequence generated since this region gave the most readable sequence data and contained the dinucleotide repeat region; approximately 60-90 bases at the 3' end of the product were not scored. Bases in blue type form the microsatellite repeat region, bases in red type illustrate the TCCGC end motif; bases in green type are invariant downstream in reference strains.



Figure 3.8.4.1 electropherogram illustrating the nucleotide sequence of the AC01 microsatellite locus, shown here for the L. V. braziliensis reference strain LTB300 Bases called are given above the individual peaks along with base numbers The microsatellite repeat sequence $(TG)_n$ can be seen between bases 50 - 70. For this sequence, n = 8 For phylogenetic analyses, sequences were aligned commencing with the motif of bases "ATG" (base numbers 1-3) since for numerous stocks the sequence at the 5' end was ambiguous

Figure 3.8.4.2: Alignment of sequence data from Viannia reference strains at the AC01 locus.

Species	Strain /stock*	10	20	30	40	50	60	70	80
\$	code no.	1	ł		1	T	F		1
L. V. b	LTB300	ATGTGCCTCT	CCCACCCTTA	GTGCTTGTCT	TCTTCCTGCT	TTGCCTCTCT	GTCTGT	GTGTGTGTGT	GTG
L. V. pe	LC39								TGTG
L. V. pa	LS94			cc					TGTGTGT
L. V. g	M4147			C		c			TGYGYGT
L. V. s	M15065		•••••	c			TGTC		TGTGCGT
L. V. I	M6426			AC	c	T.GC	C		TGTG

Strain /stock*	90	100	110	120) 130	0 140) 15() 160) 165
code no.	+	1	1	1	1	1	1	1	1
LTB300		-TCCGCTTCA	GTGGGCCGAT	CCGTTTCACT	TTTTGCCGGT	GACGTTTGTG	TGCTGACGTG	TCTGGTGGCC	TCWCA
LC39			• • • • • • • • • • •			K	W	K	W.A
LS94	GTGTG	YW	M	Y		CG	XY	KK	A
M4147	GTGYGYS	sca	.KSAY	scw	GY.SSA	scg	SAY	.SGK	W.A
M15065	GCG		•••••		K	ск	w		
M6426			T.		c		w	• • • • • • • • • • •	A
	Strain /stock* code no. LTB300 LC39 LS94 M4147 M15065 M6426	Strain /stock* 90 code no 1 LTB300 LC39 M4147 GTGTGGS M15065 GCG M6426	Strain /stock* 90 100 code no I I LTB300	Strain /stock* 90 100 110 code no 1 1 1 1 LTB300	Strain /stock* 90 100 110 120 code no -TCCGCTTCA GTGGGCCGAT CCGTTTCACT LC39 Strain /strain Million LS94 GTGTG Million Million <td< th=""><th>Strain /stock* 90 100 110 120 131 LTB300 </th><th>Strain /stock* 90 100 110 120 130 140 LTB300 </th><th>Strain /stock* 90 100 110 120 130 140 150 LTB300 </th><th>Strain /stock* 90 100 110 120 130 140 150 160 LTB300 </th></td<>	Strain /stock* 90 100 110 120 131 LTB300	Strain /stock* 90 100 110 120 130 140 LTB300	Strain /stock* 90 100 110 120 130 140 150 LTB300	Strain /stock* 90 100 110 120 130 140 150 160 LTB300

Table 3.8.4.1 compares information gained from sequence data for the Viannia reference strains. The length of the sequence varied depending on the size of the microsatellite repeat and the presence of any insertions in the flanking regions. Generally, sequence was more variable in the 3' flanking region (downstream) of the microsatellite repeat. This is likely due to the creation of the microsatellite itself, i.e. due to DNA slippage during replication and inaccurate subsequent DNA mismatch repair (Levinson & Gutman, 1987; Schlötterer & Tautz, 1992). Point mutation, insertions or deletions and other mutational mechanisms such as unequal sister-chromatid exchange and unequal crossing-over mechanisms may also operate (Levinson & Gutman, 1987; Schlötterer & Tautz, 1992; Ashley & Dow, 1994; Di Rienzo et al., 1994), adding to sequence variation at the locus.

Species	Strain Code No.	sequence length from	sequence length from	* n =	% identity to LTB300	% identity to LTB300
		Genotyper®	alignment [†]		(5' end)	(3' end)
L. V. braziliensis	LTB300	227	143	8*	100	100
L. V. peruviana	LC39	231	147	10•	100	93.2
L. V. shawi	M15065	241	157	13	98.1	93.2
L. V. panamensis	LS94	239	155	14*	96.2	79.7
L. V. guyanensis	M4147	225/241ª	157	15	96.2	64.9
L. V. lainsoni	M6426	231	147	10*	86.8	93.2

`able 3.8.4.1	Summary of	information i	from AC01	sequence data
---------------	------------	---------------	-----------	---------------

* perfect TG repeat: L. V. guyanensis and L. V. showi had "CG" insertions within the microsatellite repeat. [†] 84 bases at the 5' end were not used for alignment since this region was not clearly readable in all strains / stocks. The "ATG" motif was common to all samples sequenced and was chosen as an arbitrary place to start alignments. ^a Genotyper[®] indicated that L. V. guyanensis strain M4147 was heterozygous at the AC01 locus: the sequence length is concordant with the 241 allele having been sequenced however the ambiguity within and downstream of the microsatellite repeat motif may indicate that, in fact, both alleles were sequenced and that the sequence read contained the overlapping sequences from both.

Genotyper[®] data indicated that *L. V. guyanensis* strain M4147 was heterozygous at the AC01 locus, with allele sizes of 225 and 241 bp. The sequence read length obtained is compatible with the 241 allele having been sequenced, however, the variation observed in this strain both within and downstream of the microsatellite repeat motif

indicates that both alleles were, in fact, sequenced and that the electropherogram contained the overlapping sequences from both. Scoring of overlapping sequences would explain why *L. V. guyanensis* was so divergent from the other reference strains (64.9% identity) in the 3' flanking region: the 5' flanking region is highly conserved hence a difference in total sequence length caused by different numbers of dinucleotide repeats would cause ambiguity only after the repeat. It is possible also that the sequences of the two alleles of the reference strain *L. V.* sp. IM2832 were so divergent that, when co-sequenced, the overlapping sequence outputs rendered the output unreadable.

3.8.5 Microsatellite analysis discussion

Microsatellite analysis was found to be a highly informative technique for discriminating between the *Viannia* reference strains. This approach was capable of differentiating all reference species tested and, importantly, was able to discriminate between reference strains of *L. V. braziliensis* and *L. V. peruviana*.

Initial analysis of the microsatellite loci PCR products was carried out using nondenaturing PAGE analysis and silver staining. Results for the reference strains varied from single bands for the AC01 and AC16 loci to several bands for AC52: the banding patterns at all loci were reproducible. The resolution of individual bands using standard PAGE gels and silver staining was not generally sufficient to accurately differentiate or size products from different strains. In addition, an allelic interpretation of patterns consisting of multiple bands (i.e. AC52) was generally not possible using this approach. Bands additional to the microsatellite allele(s) were probably conformational variants, such as dimers, or stutter bands as described in section 3.8.3.

The availability of Genescan Φ / Genotyper Φ and the automated sequencing machine led to the described approach being taken. Genescan Φ and Genotyper Φ have not previously been used to study variation at these microsatellite loci. Genescan Φ analysis utilizes long-range denaturing PAGE gels: the long range of the gel allowed maximal resolution of bands, which were accurately sized by comparison with internal size standards. Conformational variant bands were eliminated under the denaturing conditions of Genescan Φ PAGE analysis. Stutter bands could generally be identified after Genotyper Φ analysis and were ignored. In some stocks, mostly at the AC52

locus, it was not possible to infer alleles (and hence genotypes) from Genotyper® since more than two peaks were present. In such cases, alleles/genotypes were scored as "not determined - ND". The analysis of these stocks was repeated but failed to resolve less than two peaks. Possible explanations for these additional peaks is that they were the result of non-specific amplification from another part of the genome, or that the stock in question was not diploid at that locus.

Unsurprisingly, sequencing analysis revealed the most diversity among reference strains. Each strain had a unique AC01 locus sequence. Sequencing strengthened the interpretation of AC01 results from Genescan Φ / Genotyper Φ in that it was generally possible to explain the size variation of the product simply in terms of variation in the numbers of dinucleotide repeats. The sequence of a few stocks did, however, contain insertions which would complicate the typing / population and phylogenetic analyses of stocks based on locus product size alone. The number of dinucleotide repeats resolved at the AC01 locus by sequence analysis is in agreement with that of Russell *et al.*, 1999, however, there are a few nucleotide discrepancies in the absolute AC01 sequence (Roslin Russell, personal comm.) of *L. V. guyanensis* (M4147) and *L. V. panamensis* (LS94). Since I provided the DNA from both of these strains for their analyses and we both sequenced the reference strains repeatedly using automated DNA sequencing machines, it is not clear why these sequences differed.

In addition to sequencing AC01, Russell *et al.* (1999) also reported sequencing results from the AC16 and AC52 loci. The microsatellite repeat in the AC16 locus was found to be composed of simple (TG)n repeats, where n ranged from 6 - 48 for the reference species used here. The microsatellite repeat in the AC52 locus was more complex and many species contained (TA) insertions. The AC52 repeat length varied among reference strains from 18 - 72 nucleotides. It is not clear how the authors managed to generate readable sequence data from the AC52 locus when they reported that multiple alleles were present (Russell *et al.*, 1999).

Table 3.8.5.1 compares the data obtained from sequence analysis of the AC16 and AC52 loci (Russell *et al.*, 1999; Roslin Russell, pers.comm.) with data obtained from Genescan Φ / Genotyper Φ analyses. Comparisons were made by calculating the expected product size assuming that allele sizes differ by changes in the microsatellite repeat number alone (see notes to table).

Table 3.8.5.1:

Comparison of data obtained for the AC16 and AC52 loci.

AC16 locus

Viannia species and code no.	Dinucleotide repeat [*]	Relative difference (in bases) [±]	Expected product size †	Genotyper® data§	Comment (see notes)
L. V. braziliensis LTB300	(TG)7	2	244	244	1
L. V. peruviana LC39	(TG)48	84	326	F	F
L. V. panamensis LS94	(TG)7	2	244	244	1
L. V. guyanensis M4147	(TG) ₆ ª	0	(242) ^b	242 ^b	b
L. V. shawi M15065	(TG)8	4	246	246 / 248	2
L. V. lainsoni M6426	(TG)20	28	270	240 / 270	2

AC52 Locus

Viannia species and code no.	Dinucleotide repeat	Relative difference (in bases) [±]	Expected product size †	Genotyper® data§	Comment (see notes)
L. V. braziliensis LTB300	(CA)9 ^a	0	(283) ^b	2 8 3 ^b	b
<i>L. V. peruviana</i> LC39	(CA) ₁₈ (TA) ₂ (CA) ₂	26	333	F	F
L. V. panamensis LS94	(CA) ₁₀ TA(CA) ₂	8	291	291	1
L. V. guyanensis M4147	(CA)9TA(CA)2	6	289	F	F
L. V. shawi M15065	(TG)17	16	299	297 / 299	2
L. V. lainsoni M6426	(CA)5TA(CA)3(TA)2	4	287	281 / 283	3

Notes to table 3.8.5.1:

[•] Dinucleotide repeat size, in bases, from Russell *et al.* (1999) and Roslin Russell (pers. comm.): data from sequence analysis. [±] Relative difference (in bases) expected in locus product size assuming no insertions or deletions in flanking regions. Calculated by subtracting the number of dinucleotide repeats observed in the smallest product⁸ from the total number of dinucleotide repeats (including TA repeats in AC52 locus) and multiplying by 2 to convert to number of bases. Expected product size calculated by adding the relative difference to the size of the smallest AC16 allele observed using Genotyper^(D). [§] Data from this thesis.

Comments:

- 1: Exact correlation observed between Genotyper@ results (this thesis) and the expected size calculated from sequencing data (Russell *et al.*, 1999, R. Russell, personal communication).
- 2: Unambiguous Genotyper@ results: 2 peaks of equal intensity scored, see *figure 3.8.5.1* for AC16 alleles (AC52 data not shown). One of these alleles correlated with the expected size calculated from sequence data of Russell *et al.* (1999).
- 3: Unambiguous Genotyper@ results: 2 peaks of equal intensity scored (data not shown). Neither peak correlated with the expected size calculated from sequence data of Russell *et al.* (1999).
- F: Genotyper® analysis failed repeatedly, no comparison possible.
- b: Genotyper@ size of smallest allele, used to calculate expected size of other alleles.

Several discrepancies were found for both loci, in both cases involving the reference strains of *L. V. shawi* and *L. V. lainsoni*. Genotyper® results for both of these strains show that 2 AC16 peaks are clearly present, see figure 3.8.5.1.

Figure 3.8.5.1 Genotyper® output for the AC16 locus for 4 Viannia reference strains.



Figure 3.8.5.1. Genotyper[®] output for the AC16 locus for 4 of the Viannia reference strains. Numbers across the top of the figure represent size, in bases. The exact sizes of individual peaks were scored automatically by the software. Exact peak sizes were assigned to "bins" which allows grouping of alleles, manually (see text). The scale to the right of each trace indicates the peak intensity. For L. V. braziliensis, L. V. guyanensis and L. V. panamensis (data not shown), an exact correlation was observed between Genotyper[®] results (this thesis) and the expected size calculated from sequencing data (Russell et al., 1999, Roslin Russell, personal communication). For L. V. showi and L. V. lainsoni however, Genotyper[®] results showed 2 peaks of equal intensity. Only one of these alleles correlated with the expected size calculated from sequence data of Russell et al. (1999).

Two peaks of equal intensity were unambiguously scored by Genotyper® (making it unlikely that one of the peaks is a stutter band), hence it is not clear why only one allele was observed by Russell *et al.* (1999) in these strains using sequence analysis. For both *L. V. shawi* and *L. V. lainsoni* at the AC16 locus, and for *L. V. shawi* only at the AC52 locus, one of the alleles identified by Genotyper® corresponded to the expected product size calculated from sequence data. Genotyper® data for *L. V. lainsoni* at the AC52 locus however also revealed 2 peaks, but in this case, neither peak (281 and 283) correlated with the expected size (287) from sequence data.

Microsatellite analysis of *Leishmania Viannia* has only very recently been described (Russell *et al.*, 1999) and until now has not yet been subjected to evaluation by other laboratories. Described in chapters 4 - 6, microsatellite analysis has been applied here to the examination of populations of *Leishmania* consisting of i) stocks mostly of a single species, *L. V. braziliensis* (chapter 4, Brazil / Trés Braços), ii) putative hybrids between parasites belonging to different *Viannia* species complexes (chapter 5, Nicaragua) and iii) putative hybrids between parasites belonging to the same *Viannia* species complex and which have been collected from a discrete geographical area (chapter 6, Huanuco, Peru). Microsatellite analysis will be discussed in greater detail in chapters 4-7.

3.9 <u>Comparative assessment of techniques used for discriminating</u> between *Leishmania* reference strains.

Of the techniques assessed for discrimination between *Viannia* reference strains, IEA, PCR-RFLP and microsatellite analysis revealed the most variability: in addition, the results from each of these methods could be interpreted genetically.

IEA is the "gold standard" method of classifying *Leishmania* to the species level. The major drawback of the technique is that parasite isolation and culture are required to provide sufficient material for analysis. Parasite isolation is prone to contamination and has a variable success rate. Some *Leishmania* species, notably *L. V. peruviana*, have been found particularly difficult to expand in *in vitro* culture. In addition, isolation and long-term culture, *in vivo* or *in vitro*, may result in strain selection (the loss of variant parasite types present during the original infection). Cultivation is expensive and time consuming. IEA itself is also expensive and time consuming to perform, although fairly simple, generating results which are easy to read and readily

interpretable in terms of genotype. IEF was assessed as an alternative to IEA because it is possible using IEF to run many more samples per gel and hence increase throughput. This technique however, proved to be less informative than IEA owing to the reduced number of examinable loci.

RAPD analysis was assessed because of its apparent popularity, as based on the literature at the time. Initially promising in terms of discrimination between reference strains, problems of reproducibility were encountered which led to its use being discontinued. The disadvantages of the RAPD technique were described at length in section 3.4. In addition, the use of non-parasite-specific PCR primers means that isolation and culture of the parasite is, like IEA, a necessity to eliminate contaminating DNA.

Most other molecular-based approaches are based on PCR using parasite-specific primers which eliminates the requirement for the parasite isolation and culture: strain selection is thus minimized and infections involving mixed species or strains may be detected.

The RFLP technique can be applied to PCR products from any locus. The variability shown depends on the conservation level of the locus being examined. Multigene families, such as rRNAs, are usually organized as arrays of tandem repeated units and, as such, provide a number of loci suitable for PCR-RFLP analysis. The rRNA genes of all organisms are functionally and evolutionarily homologous and are highly conserved both in structure and nucleotide sequence. The intergenic regions and flanking non-transcribed regions of the arrays are less conserved. Riboprinting was unable to discriminate between *Viannia* species and so was not used further. PCR-RFLP analysis of the rRNA ITS region, however, did differentiate between the various reference strains: the discriminatory power was found to be greatest at the species complex level. Hence PCR-RFLP analysis of the rRNA ITS region was used only to examine stocks of the Nicaraguan population which encompassed more than one species complex.

Microsatellite loci are ideal molecular markers because they are discrete and have codominant alleles which are inherited in a Mendelian manner (Ashley & Dow, 1994; Schlötterer, 1998). They have no known function and it is therefore assumed that there is no functional constraint limiting diversity and that they are selectively neutral
(Ashley & Dow, 1994; Schlötterer, 1998). Microsatellite analysis is an important new technique for differentiating species of the subgenus *Viannia*. This technique was shown, in agreement with Russell *et al.* (1999), to reveal a large amount of variation among *Viannia* strains and stocks. Combining the data from Genescan@ / Genotyper@ and sequencing allowed the unique identification of each reference strain.

The application of Genescan \oplus and Genotyper \oplus to the analysis of these loci has not been described previously. Genescan \oplus and Genotyper \oplus analyses were relatively easy to perform and proved superior to the PAGE / silver staining method. A great advantage of the Genescan \oplus approach was multiplex analysis which allowed the simultaneous analysis of multiple loci. Here, PCR products were pooled manually: even greater throughput could have been obtained by multiplexing the PCR reactions themselves. Another advantage is that peak sizes were scored automatically by the software in comparison with internal size standards. This meant that size scoring was accurate, consistent within and between gels and that any possible bias due to manual interpretation were eliminated.

The sequencing of any locus provides the ultimate data for typing, population genetic and phylogenetic analyses. For small numbers of samples, sequencing may be a viable approach, however, it has limited applicability to the analysis of large numbers of stocks or for long sequences. Here, sequencing of the AC01 locus was carried out to examine the variation between stocks and to see if variation was limited only to variable numbers of the dinucleotide repeat. Although size variation was generally found to correlate with variable numbers of repeats, mutations in the flanking regions were also demonstrated by sequencing. Phylogenetic and population genetic analyses are likely to be inaccurate (insertions / deletions) or underestimated (point mutations) if such mutations are present: phylogenetic analyses here were based on sequence data.

Another advantage of microsatellite analysis using Genescan Φ / Genotyper Φ is that heterozygotes at a single locus are readily detectable as double peaks. Russell *et al.* (1999) reported finding a single heterozygote among their stocks. Two of the populations examined in this thesis contained putative hybrids identified using IEA, RAPD and RFLP analyses. The analysis of heterozygous parasites are reported and discussed in detail in chapters 5 and 6.

CHAPTER 4 – BRAZIL: *L. V. BRAZILIENSIS* STOCKS FROM DIFFERENT GEOGRAPHICAL AREAS INCLUDING TRÊS BRAÇOS / CORTE DE PEDRA

4.1 Leishmaniasis in Brazil - general introduction

Brazil, the largest country in South America, stretches some 4350 kilometres from the Andes eastwards to the Atlantic Ocean, bordering on every country of the continent bar Chile and Ecuador. Brazil, as would be expected in such a vast country, has a diverse vegetation ranging from the tropical rain forest ('selva') of the Amazon region, through savannah or tropical grassland areas to low, scrub and thorn forests ('caatinga'). Brazil has rich agricultural (coffee, sugar, cacao, soybeans, timber) and mineral resources (iron ore and gold) however, despite the recent modernization of its industry, it has an economy that is hampered by a huge foreign debt. The rapid population growth rate, particularly in urban areas, has exacerbated unemployment, lack of housing etc. such that around 45% of the population (65 million people) live in poverty (J. Arias, personal comm.).

Cutaneous leishmaniasis is endemic in Brazil, with foci in virtually every state. The worst affected regions are those of the north, northeast and central/west, see table 4.1 (from Arias, 1996). The incidence of cutaneous leishmaniasis in Brazil has increased from 10.45 cases per 100, 000 inhabitants in 1985 to 22.81 in 1995. Over all regions, there has been a 261% increase in the number of cases reported during this time period. Whilst improved detection and reporting have been implemented, there is no doubt that CL is on the increase. Until relatively recently, CL transmission most commonly occurred in rural forested areas which enabled control measures to be centered around the treatment of infected individuals coupled with health education. CL epidemics and new endemic foci have traditionally been associated with rural workers colonizing sylvatic habitats (i.e. settlement in deforested areas etc., e.g. Barrett & Senra, 1989). More recently however, changes have been noted in transmission patterns. There are increasing reports of transmission occurring in peridomestic and domestic habitats in the absence of direct contact with the forest (Yoshida et al., 1990; Felinto de Brito et al., 1993; Brandão-Filho et al., 1999) and in peri-urban foci (Oliveira-Neto et al., 1988; Passos et al., 1993). These changes are

		1	1985	1	1995	% increase	
Region	States in region	cases. reported (new)	cases /100, 000 inhabitants	cases. reported (new)	cases /100, 000 inhabitants	in cases reported (1985-95)	
North	Pará, Rondõnia, Amazonas, Acre, Roraima, Amapá, Tocatins	5, 935	82.48	13, 017	112.13	219.3	
Northeast	Maranhão, Bahia , Ceará, Pernambuco , Piauí, Paraíba, Sergipe, Alagoas, Rio Grande do Norte	4, 402	11.56	13, 883	30.87	315.4	
Southeast	Minas Gerais, São Paulo, Espírito Santo, Rio de Janeiro,	844	1.50	2, 605	3.93	308.7	
South	Paraná, Rio Grande do Sul, Santa Catarina	420	2.06	796	3.44	189.5	
Central & West	Mato Grosso, Goiás, Mato Grosso do Sul, Distrito Federal	2, 038	23.41	5, 343	52.01	262.2	
Brazil	All Regions / States	13, 639	10.45	35, 644	22.81	261.3	

 Table 4.1:
 Cutaneous leishmaniasis in Brazil: new cases reported and cases / 100, 000 inhabitants for 1985 and 1995, by region and by all regions combined.

Table 4.1: new cases of cutaneous leishmaniasis and cases / 100, 000 inhabitants reported in Brazil for 1985 and 1995. Figures are given by region and for all regions combined. Data from Arias, 1996. Stocks examined in this thesis were sampled from States written in **bold** type: all Regions bar Central / West were represented. alarming as it likely involves the adaptation of parasite and/or vector species to nonsylvatic habitats and / or alternative reservoirs.

Leishmania species causing CL in Brazil include L. V. braziliensis, L. V. guyanensis, L. V. lainsoni, L. V. naiffi, L. V. shawi and L. L. amazonensis. VL (caused by infection with L. L. chagasi) is also widespread and common in Brazil and is spreading, especially through areas of greater poverty (Desjeux, 1992b).

4.2 Leishmaniasis in Três Braços and Corte de Pedra, Brazil

Três Braços and Corte de Pedra are situated 600-800m above sea level (a.s.l.) in the south west of the state of Bahia, (northeast region) Brazil, see *Figure 4.2*. The region is endemic for cutaneous and mucocutaneous leishmaniasis: Três Braços and Corte de Pedra have been the focus of research, both clinical and scientific, for over 25 years (Marsden, 1994b). A virtual monotransmission of *L. V. braziliensis* occurs in the area; both visceral disease and Chagas' disease are absent (Marsden, 1994a).

Três Braços ("three arms") is situated at the junction of three rivers in one of the few surviving stands of littoral forest in Bahia State and undergoes mini-epidemics of CL and MCL. A field clinic and study site was set up by the late Philip Marsden, Cesar Cuba-Cuba and others in 1973 to treat infected individuals and to study the epidemiology of the disease. The initial study site encompassed 13 farms and 2 hamlets, comprising 276 families and 1956 individuals, within a 20km radius of the village of Três Braços.

Jones *et al.* (1987) reported results from a 5 year (1980 - 1984) study of leishmaniasis in Três Braços: the incidence of leishmaniasis was found to be 8 cases/1000 individuals, with a prevalence of 160 cases/1000 individuals in 1984. CL was predominantly observed in males and in individuals between the ages of 10 to 30 years (Jones *et al.*, 1987). Cutaneous lesions were usually rapidly-evolving (often reaching several centimetres diameter within a month), consisting of deep, delineated ulcers with raised borders (Jones *et al.*, 1987). More than 25% of patients presented with two or more active lesions; over 70% of individuals with active lesions presented primary lesions on the leg; other sites included arms (10%), trunk (10%) neck (5%) and nose (5%) (Jones *et al.*, 1987).

MCL was initially reported in up to 30% of cases from the field-clinic at Três Braços (Barreto *et al.*, 1981), however, this figure reflected a biased population sample consisting of the more severe cases. Jones *et al.* (1987) reported that 2.7% of CL cases developed MCL in Três Braços: MCL was found to be more common in individuals with large and / or multiple primary lesions and usually developed between 2 and 10 years after an episode of CL. MCL has also been reported in individuals never having had CL (Marsden, 1994b).

Figure 4.2: Map of region surrounding Três Braços, Bahia State, Brazil. (from Cuba-Cuba et al., 1985)



Figure 4.2: Map of region surrounding Três Braços, Bahia State, Brazil (from Cuba-Cuba et al., 1985). Inset shows location of Bahia State (white) and Três Braços (dot) within a map of Brazil. Scale relates to regional map only.

Corte de Pedra is situated, 45km from Três Braços, in an area with virtually no virgin forest remaining, most having been cleared for planting with cocoa, banana and manioc. A major CL epidemic occurred between 1984 and 1986, with a recorded incidence of 83 cases /1000 individuals (Franca *et al.*, 1991).

Primary isolates were made from humans (both CL and MCL), dogs, equines and sand flies in the region using both *in vivo* (hamster inoculation) and *in vitro* (culture media inoculation) methods (Cuba-Cuba *et al.*, 1985; 1986). Almost all isolates were identified as *L. V. braziliensis*, two human cases have been reported in which parasites belonging to the *L. L. mexicana* complex were identified (Cuba-Cuba *et al.*, 1985).

There is evidence for both peridomestic and sylvatic transmission cycles of leishmaniasis in the region. Evidence for peri- and intradomestic transmission comes from observations of the disease both in very young and elderly individuals who do not enter forested areas (Jones *et al.*, 1987). Evidence for sylvatic transmission stems from observations that the disease is most common in male workers employed in plantation farming. Presumably, the predominance of the disease in these individuals is the result of greater exposure from the semi-forested and forested working environment.

Around 8% of individuals with no history of CL gave positive DTH responses, as revealed by skin-testing (Barreto *et al.*, 1981). Such inapparently-infected individuals may serve as disease reservoirs. Although humans and domestic animals are implicated as reservoirs for initiating peridomiciliary transmission in Três Braços / Corte de Pedra, the primary reservoir host is thought to be an as yet unidentified forest-dwelling mammal (Marsden, 1994a, b). More than 800 sylvatic mammals (sloths, anteaters, marsupials, rodents monkeys and rabbits) have been examined however no isolations have been made (Silva *et al.*, 1992).

The probable vector of leishmaniasis in Três Braços is *Lu. whitmani*, a species found in large numbers in and around houses in the region (Vexenat, 1986). Thirty-one other *Lutzomyia* species were recorded from the area, mostly from the remaining forest region. None of these however were found to be infected with *Leishmania* promastigotes (Vexenat, 1986). *Lu. flaviscutellata* is the probable vector of *L. L. amazonensis* in the Três Braços area: this vector species rarely bites man and has a

sylvatic cycle involving forest rodents, which may explain the rarity of infections caused by this species in humans and domestic animals in Três Braços.

In Três Braços today, transmission remains unchecked in the absence of control measures barring personal therapy (Vieira *et al.*, 1990). The cause of the mini epidemics occurring in Três Braços is still not clear but probably stem from an infected mammal (man, dog) initiating a peridomiciliary transmission cycle.

Três Braços stocks have been partially characterized using IEA and RAPD (Cuba-Cuba *et al.*, 1985, 1991; Larsen, 1994); one stock, LTB300, is a WHO-designated *L. V. braziliensis* reference strain.

4.3 Leishmania stocks

4.3.1 Brazilian stocks

Brazilian Leishmania stocks were selected from diverse geographical areas of Brazil, in seven States encompassing all Regions bar Central / West (see *table 4.1*). Stocks were isolates mostly from sand flies, but also from human and rodent sources. Stocks were obtained from the cryobank at the London School of Hygiene & Tropical Medicine (LSHTM)^a, from R. Naiff (Centro de Pesquisas Amazonas, Manaus, Amazonas, Brazil)^b, from Prof. J. Shaw (Instituto Evandro Chagas, Fundação Nacional de Saude, Belém, Pará, Brazil)^c and from S. Brandão-Filho (Centro de Pesquisas Aggeu Magalhães, Fundação Oswaldo Cruz, Recife, Pernambuco, Brazil)^d. The 19 stocks selected for study are listed in *table 4.3.1*.

4.3.2 Três Braços and Corte de Pedra stocks

Três Braços and Corte de Pedra *Leishmania* stocks (Bahia State) were selected to encompass those from human, sand fly, canine and equine sources. Stocks were obtained from the cryobank at the LSHTM and from Prof. J. Shaw (Instituto Evandro Chagas, Fundação Nacional de Saude, Belém, Pará, Brazil)^C. The 18 stocks selected for study are listed in *table 4.3.2*.

ISOLATE CODE NO.	ISOLATED FROM:	REGION / STATE	<i>LEISHMANIA</i> Species	LESION TYPE*
MHOM/BR/75/M2903 ⁸	Human	Carajas / Para	L. V. braziliensis	CL
MHOM/BR/75/M2904ª	Human	Carajas / Para	L. V. braziliensis	CL
MMES/BR/83/HM1BC	Rodent	Miracatu / Sao Paulo	L. V. braziliensis	-
MHOM/BR/81/IM403C	Human	Manaus / Amazonas	L. V. shawi	CL
ISQU/BR/86/IM2832b	Sand fly	Balbina / Amazonas	L. V. sp. n.†	-
IWEL/BR/84/M8522 ^c	Sand fly	Carajas / Para	L. V. braziliensis	-
IWHI/BR/87/M11367 ^C	Sand fly	Carajas / Para	L. V. shawi	-
MAKO/BR/87/BH325R ^c	Rodent	Caratinga / Minas Gerais	L. V. braziliensis	-
IWHI/BR/88/M12045°	Sand fly	Carajas / Para	L. V. shawi	-
ICOM/BR/89/M12613C	Sand fly	Paragominas / Para	L. V. braziliensis	-
ICOM/BR/89/M12620 ^c	Sand fly	Paragominas / Para	L. V. braziliensis	-
MHOM/BR/91/M14405 ^c	Human	Flora / Paraná	L. V. braziliensis	CL
MHOM/BR/93/709-LOAC	Human	Peabiru / Paraná	L. V. braziliensis	CL
MHOM/BR/93/769-VU ^C	Human	S. Jorge do Ivai / Parana	L. V. braziliensis	CL
IWHI/BR/94/M15065C	Sand fly	Paragonimas / Parana	L. V. shawi	- 1
MHOM/BR/96/"Amaro"d	Human	Amariji / Pernambuco	L. V. braziliensis	CL
MHOM/BR/96/"Manoel"d	Human	Amariji / Pernambuco	L. V. braziliensis	CL
MHOM/BR/96/"Paulino"d	Human	Amariji / Pernambuco	L. V. braziliensis	CL
MHOM/BR/96/"Fragoso"d	Human	Amariji / Pernambuco	L. V. braziliensis	CL

Table 4.3.1: Leishmania Viannia stocks from Brazil (excluding stocks from Três Braços and Corte de Pedra)

Table 4.3.1: Leishmania Viannia stocks from Brazil (excluding stocks from Tres Braços and Corte de Pedra). Stocks were obtained from the cryobank at the London School of Hygiene & Tropical Medicine (LSHTM)^a, from R. Naiff (Centro de Pesquisas Amazonas, Manaus, Amazonas, Brazil)^b, from Prof. J. Shaw (Instituto Evandro Chagas, Fundação Nacional de Saude, Belém, Para, Brazil)^c and from S. Brandão-Filho (Centro de Pesquisas Aggeu Magalhães, Fundação Oswaldo Cruz, Recife, Pernambuco, Brazil)^d. [†] Grimaldi *et al.*, 1991. ^{*} Lesion type: CL - cutaneous leishmaniasis, MCL - mucocutaneous leishmaniasis.

ISOLATE CODE NO.	ISOLATED FROM:	REGION / STATE	LEISHMANIA SPECIES	LESION TYPE*
MCAN/BR/77/Dog-Lamprea	Dog	Tr ês Braços / Bahia	L. V. braziliensis	CL
MCAN/BR/78/Dog-Rodrigo	Dog	Três Braços / Bahia	L. V. braziliensis	CL
MCAN/BR/81/Dog-Rico	Dog	Três Braços / Bahia	L. V. braziliensis	CL
MEQU/BR/86/JCP-184	Donkey	Corte de Pedra / Bahia	L. V. braziliensis	CL
MEQU/BR/86/JCP-225	Donkey	Corte de Pedra / Bahia	L. V. braziliensis	CL
vIWHI/BR/86/WHI-10187	Sand fly	Corte de Pedra / Bahia	L. V. braziliensis	-
IWHI/BR/86/WHI-CIPD	Sand fly	Corte de Pedra / Bahia	L. V. braziliensis	-
IWHI/BR/86/WHI-M11468C	Sand fly	Corte de Pedra / Bahia	L. V. braziliensis	-
IWHI/BR/86/WHI-EO1094	Sand fly	Três Braços / Bahia	L. V. braziliensis	-
IWHI/BR/86/WHI-EO1161	Sand fly	Três Braços / Bahia	L. V. braziliensis	-
ITUP/BR/86/TUP-EO252	Sand fly	Três Braços / Bahia	L. V. braziliensis	-
MHOM/BR/85/LTB12	Human	Três Braços / Bahia	L. V. braziliensis	MCL
MHOM/BR/76/LTB0014	Human	Três Braços / Bahia	L. V. braziliensis	MCL
MHOM/BR/00/LTB30	Human	Três Braços / Bahia	L. V. braziliensis	MCL
MHOM/BR/81/LTB179	Human	Três Braços / Bahia	L. V. braziliensis	CL
MHOM/BR/82/LTB206	Human	Três Braços / Bahia	L. V. braziliensis	CL
MHOM/BR/82/LTB253	Human	Três Braços / Bahia	L. V. braziliensis	CL
MHOM/BR/84/LTB300	Human	Três Braços / Bahia	L. V. braziliensis	MCL

Table 4.3.2: Leishmania Viannia stocks from Três Braços / Corte de Pedra

Table 4.3.2 Leishmania Viannia stocks from from Tres Braços and Corte de Pedra. Stocks were obtained from the cryobank at the London School of Hygiene & Tropical Medicine (LSHTM)^a and from Prof. J. Shaw (Instituto Evandro Chagas, Fundação Nacional de Saude, Belém, Para, Brazil)^c. [•] Lesion type: CL - cutaneous leishmaniasis; MCL - mucocutaneous leishmaniasis.

4.4 Isoenzyme analysis

Three distinct groups were identified among the 37 stocks from Brazil (all stocks, including those from Três Braços / Corte de Pedra): L. V. braziliensis (32 stocks), L. V. shawi (4 stocks) and a potential new Viannia species (1 stock, L. V. sp. - IM2832, Grimaldi et al., 1991, also used as a reference strain). L. V. shawi stocks (guyanensis species complex) could easily be differentiated from L. V. braziliensis stocks using 9 enzyme loci: MPI, NHi1, NHi2, PEPD, PGM, 6PGD, G6PD, ASAT and ALAT. IM2832 (L. V. sp. n.) could be differentiated from L. V. braziliensis stocks using 8 enzyme loci (MPI, ES, PGM, 6PGD, GPI, G6PD, ASAT and ALAT) and from L. V. shawi using 10 enzyme loci (MPI, NHi1, NHi2, ES, PEPD, PGM, GPI, G6PD, ASAT and ALAT). Examples of isoenzyme banding patterns observed in the stocks from Brazil are shown in figure 4.4.1 for the enzymes MPI (polymorphic) and NHd (monomorphic).

4.4.1 Zymodemes

Ten zymodemes were observed among the stocks from Brazil: stocks identified as L. *V. shawi* (LON203) and *L. V.* sp. n. (LON205) accounted for two of these. Stocks identified as *L. V. braziliensis* could be divided into 8 zymodemes. *Table 4.4.1* (I. and II.) summarizes the IEA allele and zymodeme designations for the Brazilian (I.) and Três Braços / Corte de Pedra stocks (II.). Allele designations are the same as those given in *figure 3.2.2* (chapter 3). "Missing alleles" were scored (-) when no enzyme activity was repeatedly detected.

Of the 32 L. V. braziliensis stocks, 4 enzyme loci (MPI, PEPD, PGM AND ES) were shown to be polymorphic: 8 zymodemes were scored among stocks from combinations of polymorphisms at these loci. LON200 accounted for 21/32 (65.6%) of stocks examined: this figure is biased by the inclusion of the 18 stocks from Três Braços / Corte de Pedra since all isolates bar one from this locality were this zymodeme. Three zymodemes (LON210, LON212, LON215) contained a single stock and 4 (LON209, LON211, LON213, LON214) contained 2 stocks each. L. V. braziliensis stock zymodemes are summarized by geographical origin, host / vector source and clinical presentation in *table 4.4.2*.











Figure 4.4.1: Isoenzyme variation in stocks from Brazil using MPI and NHd. Lane 1 - L. V. braziltensis reference strain LTB300; lane 2 - L. V. guyanensis reference strain M4147; lane 3 - M2903; lane 4 - M2904; lane 5 - HMIB; lane 6 - M11367; lane 7 - BH325R; lane 8 - M12613; lane 9 - M14916; lane 10 - M8522; lane 11 - M14913; lane 12 - M12045; lane 13 - L. V. shawi reference strain M15065. Using MPI, L. V. braziltensis variants are observed in lane 3 and 7. Stocks in lanes 6, 12 & 13 were identified (in combination with other enzyme loci) as L. V. shawi - using MPI they are identical to the L. V. guyanensis reference strain).

							EN2	YME]
STOCK CODE NO.	* HOST SPECIES	MPI	NHi 1	NHi 2	NHd	ES	PEPD	PGM	6PGD	GPI	G6PD	ASAT	ALAT	Z*
Amaro, 709-LOA, 769-VU	мном	6/6	2/2	7/7	4/4	4/4	2/2	3/3	5/5	4/4	2/2	5/5	4/4	LON200
M8522	IWEL													
M15065, M12045, M11367	IWHI	8/8	3/3	5/5	4/4	4/4	7/7	5/5	6/6	4/4	1/1	2/2	3/3	LON203
IM403	мном													
IM2832	ISQU	5/6	2/2	7/7	4/4	5/5	4/4	2/2	6/6	2/2	3/3	4/4	5/5	LON205
Fragoso	мном	6/6	2/2	7/7	4/4	4/4	2/2	-	5/5	4/4	2/2	5/5	4/4	LON209
HMIB	MMES													
M2903, Manoel	мном	6/7	2/2	7/7	4/4	4/4	2/2	3/3	5/5	4/4	2/2	5/5	4/4	LON211
BH325R	МАКО	6/9	2/2	7/7	4/4	4/4	2/2	3/3	5/5	4/4	2/2	5/5	4/4	LON212
Paulino, M14405	мном	6/6	2/2	7/7	4/4	4/4	2/2	4/4	5/5	4/4	2/2	5/5	4/4	LON213
M2904	мном	6/6	2/2	7/7	4/4	4/4	4/4	3/3	5/5	4/4	2/2	5/5	4/4	LON214
M12613	ІСОМ													
M12620	ІСОМ	6/6	2/2	7/7	4/4	4/4	2/4	3/3	5/5	4/4	2/2	5/5	4/4	LON215

 Table 4.4.1:
 I. Allele and zymodeme designations for Brazilian stocks.

Notes to table 4.4.1. Allele and zymodeme designations for Brazilian stocks. Code names given in **bold** are reference strains. * host / vector species code: MHOM mammal, human; MMES - mammal, rodent (*Mesocricetus* sp.); MAKO, mammal, rodent (*Akodon* sp.); IWHI - sand fly, *Lu. (Nys.) whitmant*, ICOM - sand fly, *Lu. complexus*, IWEL - sand fly, *Lu. (Psy.) wellcomet*; ISQU - sand fly, *Lu. (Psy.) squamiventris.*. Z = zymodeme, * LON numbers above 200 have been reassigned by the author as previous records have been lost (zymodemes are assigned based on the IEA conditions used, i.e. for *Leishmania Viannia* species). For full enzyme names, see chapter 2. See figure 3.2.2 for allele scoring scheme. Data given assumes diploidy, see introduction.

Table 4.4.1:	
--------------	--

II. Allele and zymodeme designations for Três Braços / Corte de Pedra stocks.

							ENZ	YME]
STOCK CODE NO.	* HOST SPECIES	MPI	NHi 1	NHi 2	NHd	ES	PEPD	PGM	6PGD	GPI	G6PD	ASAT	ALAT	Z*
LTB300, LTB12, LTB14, LTB30, LTB179, LTB206, LTB253	мном													
JCP-184, JCP-225	MEQQ	6/6	2/2	7/7	4/4	4/4	2/2	3/3	5/5	4/4	2/2	5/5	4/4	LON200
Dog-Lamprea, Dog-Rico,	MCAN													
Dog-Rodrigo,														1
WHI-M10187, M11468,	IWHI													
EO1094, EO1161														
TUP-EO252	ITUP													
WHI-CIPD	IWHI	6/6	2/2	7/7	4/4	-	2/2	•	5/5	4/4	2/2	5/5	4/4	LON210

Notes to table 4.4.1: Allele and zymodeme designations for Três Braços / Corte de Pedra stocks. Code names given in **bold** are reference strains. * host / vector species code: MHOM - mammal, human; MCAN - mammal, canid (dog); MEQU - mammal, equine (donkey); IWHI - sand fly, Lu. (Nys.) whitmani; ITUP - sand fly, Lu. (Nys.) whitmani; IT

STOCK	Z*	GEOGRAPHIC	AL ORIGIN	ISOLATED	CLINICAL
CODE NO.		REGION	STATE	FROM:	TYPE
M12613	LON214	Paragominas	Para	Sand fly	-
M12620	LON215	Paragominas	Pará	Sand fly	-
M8522	LON200	Carajas	Para	Sand fly	-
M2903	LON211	Carajas	Para	Human	CL
M2904	LON214	Carajas	Pará	Human	CL
HM1B	LON209	Miracatu	São Paulo	Rodent	-
BH325R	LON212	Caratinga	Minas Gerais	Rodent	
"Amaro"	LON200	Amariji	Pernambuco	Human	CL
"Manoel"	LON211	Amariji	Pernambuco	Human	CL
"Paulino"	LON213	Amariji	Pernambuco	Human	CL
"Fragoso"	LON209	Amariji	Pernambuco	Human	CL
Dog-Lamprea	LON200	Três Braços	Bahia	Dog	CL
Dog-Rodrigo	LON200	Três Braços	Bahia	Dog	CL
Dog-Rico	LON200	Três Braços	Bahia	Dog	CL
JCP-184	LON200	Corte de Pedra	Bahia	Donkey	CL
JCP-225	LON200	Corte de Pedra	Bahia	Donkey	CL
WHI-10187	LON200	Corte de Pedra	Bahia	Sand fly	-
WHI-CIPD	LON210	Corte de Pedra	Bahia	Sand fly	
WHI-M11468	LON200	Corte de Pedra	Bahia	Sand fly	-
WHI-E01094	LON200	Três Braços	Bahia	Sand fly	-
WHI-E01161	LON200	Três Braços	Bahia	Sand fly	-
TUP-EO252	LON200	Três Braços	Bahia	Sand fly	-
LTB12	LON200	Três Braços	Bahia	Human	MCL
LTB0014	LON200	Três Braços	Bahia	Human	MCL
LTB30	LON200	Três Braços	Bahia	Human	MCL
LTB179	LON200	Três Braços	Bahia	Human	CL
LTB206	LON200	Três Braços	Bahia	Human	CL
LTB253	LON200	Três Braços	Bahia	Human	CL
LTB300	LON200	Três Braços	Bahia	Human	MCL
M14405	LON213	Flora	Paraná	Human	CL
709-LOA	LON200	Peabiru	Paraná	Human	CL
769-VII	1.0N200	S. Jorge do Ivai	Paraná	Human	CI.

 Table 4.4.2:
 Summary of L. V. braziliensis stock zymodemes, by geographical origin, host / vector source and clinical presentation.

Notes to table 4.4.2 (previous page) summary of L. V. braziliensis stock zymodemes, by geographical origin, host / vector source and clinical presentation. Code names given in **bold** are reference strains. Z = zymodeme (assigned based on the IEA conditions used, i.e. for *Leishmania Vicannia* species). For full enzyme names, see chapter 2. See *figure 3.2.2* for allele scoring scheme. Data given assumes diploidy, see introduction.

The 18 L. V. braziliensis Três Braços / Corte de Pedra stocks were shown, in agreement with previous reports (Cuba-Cuba *et al.*, 1985; Rosa *et al.*, 1988), to be remarkably similar using IEA: all stocks bar one were zymodeme LON200, identical to the WHO reference strain LTB300 (see *figure 3.2.2*). The only stock among those from Três Braços / Corte de Pedra to show any variation was WHI-C1PD (LON210): this stock differed only in that it repeatedly showed a lack of detectable activity of the enzymes ES and PGM. Stocks from other areas of Brazil were more variable than those from Três Braços / Corte de Pedra, however too few stocks were examined from each locality to be able to associate zymodemes with geographical area. The four stocks from Amariji deserve further comment: these stocks were identified as L. V. braziliensis, however, each stock presented a different zymodeme (LON200, LON209, LON211 and LON213). Others groups have also reported L. V. braziliensis isoenzyme variants from the region (Felinto de Brito *et al.*, 1993).

Of the 17 stocks from humans, 13 were isolated from CL lesions and 4 from MCL lesions. Human stocks encompassed 5 zymodemes (LON200, 209, 211, 213, 214, see table 4.4.2). All stocks isolated from MCL lesions were LON200. No correlation was observed between zymodeme and clinical presentation. There was also no association between zymodeme and host / vector species.

4.5 Microsatellite analysis

4.5.1 PCR of microsatellite loci

4.5.1.1 Stocks with "null alleles"

Ten stocks from Brazil failed to generate PCR amplification products at one or more of the microsatellite loci. The amplification profiles of these stocks are summarized in *table 4.5.1.1.* PCR amplification failure (assumed to be due to mutation in the primer binding site) was observed most often for the AC01 and AC52 loci. Four stocks failed to amplify any of the three loci. The DNA from these stocks was shown to be intact

by electrophoretic techniques and by the successful PCR amplification of the 18S rRNA gene locus. Stocks failing to generate PCR products were scored as having "null alleles": mutation(s) in the binding site were not investigated further and it is therefore not possible to establish whether one or both priming sites were implicated. For reasons not determined, sand fly stocks from Três Braços / Corte de Pedra appeared to have the most null alleles (4/6 stocks, 66.6%) compared with stocks of non sand fly origin (1/14 stocks, 7.1%).

Stock code	Species*	G e ographical origin	Isolated from	AC01	AC16	AC52
IM2832	L. V. sp. n.	Balbina, AM	sand fly		x	x
IM403	L. V. shawi	Manaus, AM	human	x		x
M12613		Paragominas, PA	sand fly	x		x
709-LOA		Peabiru, PN	human	x		x
769-VU		S. Jorge do Ivai, PN	human	x	x	x
JCP-184	L. V. braziliensis	Corte de Pedra, BA	donkey	x		x
WHI-E01161		Três Braços, BA	sand fly	x		
WHI-M11468		Corte de Pedra, BA	sand fly	x	x	x
TUP-EO252		Três Braços, BA	sand fly	x	x	x
WHI-EO1094		Três Braços, BA	sand fly	x	x	x
	PCR failures			9	5	9

Table 4.5.1.1Summary of microsatellite locus PCR failures for stocks
with null alleles (failing to amplify one or more loci.)

Table 4.5.1.1: Summary of microsatellite locus PCR failures for stocks with null alleles (i.e. those failing to amplify one or more loci.). X indicates no amplification. * Species status was determined from isoenzyme analysis

4.5.1.2 PCR amplification of microsatellite loci

Analyzing PCR products using agarose gel electrophoresis and ethidium bromide staining, stocks from Brazil (excluding those with null alleles) generated PCR products for AC01 in the range 220 - 250 bp, in agreement with Russell *et al.* (1999). An additional faint band of approximately 450 bp was also observed in some stocks, which was probably a dimer since subsequent analyses (Genotyper and sequencing) of these stocks revealed a single peak / sequence.

PCR products generated from Brazilian stocks for AC16 were in the range 240 - 270 bp, also in agreement with Russell *et al.* (1999). PCR products generated from Brazilian stocks for AC52 were in the range 280 - 310 bp, also in agreement with Russell *et al.* (1999). Diffuse bands were observed for several stocks; this suggested that more than one band was generated in these stocks which were not resolved using agarose gel electrophoresis.

4.5.2 Non-denaturing PAGE analysis

Agarose gel electrophoresis was used initially to determine the success of PCR amplification. Products were subsequently analyzed using PAGE gels, which permit greater resolution, and silver staining, which is highly sensitive. PAGE analysis of microsatellite locus products from all stocks revealed reproducible pattern types.

4.5.2.1 <u>AC01</u>

Figure 4.5.2.1 shows AC01 products from Brazilian stocks analyzed using nondenaturing PAGE and silver staining. Two intensely-staining bands were observed for most stocks at this locus: the band corresponding to the expected allele size was generally more intense. Another band, usually fainter, was also present in most stocks and always at a set distance from the allelic band. This additional band was not large enough to be a dimeric form of the allelic band. Several observations indicate that this additional band was a conformational variant of the allelic band: i) this band was not observed using agarose and ethidium bromide although it was intense and of a sufficiently different size from the allelic band to be detected within the limits of agarose gel electrophoresis / ethidium bromide staining; ii) this band "disappeared" after denaturing PAGE (Genescan@ / Genotyper®) analysis (see section 4.5.3); iii) this band was too large to be considered as the second allele identified in some stocks by Genotyper® (see section 4.5.3); iv) sequencing analysis revealed that most products were composed of a single sequence type (see section 4.5.4). Similar additional bands have been described in reviews of microsatellite analysis as being PCR artefacts which are consistent between isolates giving a pattern of bands which is peculiar to the locus (Ashley & Dow, 1994; Schlötterer, 1998).

Figure 4.5.2.1





Figure 4.5.2.1: AC01 products from Brazilian stocks analyzed using non-denaturing 6% PAGE and silver staining. Lane 1 - 100 bp molecular weight marker (Gibco BRL, UK); lane 2 - L. V. braziliensis reference strain, LTB300; lane 3 - M8522; lane 4 - M11367; lane 5 - M11425; lane 6 - L. V. shawi reference strain, M15065; lane 7 - M12620; lane 8 - M14405; lane 9 - M12045; lane 10 - Amaro; lane 11 - Manoel; lane 12 - Fragoso; lane 13 - Paulino. The allelic band is the lowest intensely-staining band in each lane. Note the additional band (at set distance from the allelic band in each lane) and heteroduplex bands (indicated): see text for discussion. PCR products from L. V. shawi stocks are in lanes 4, 6 & 9.

In some stocks, heteroduplex bands were also observed. Heteroduplex molecules may arise from both homozygous and heterozygous individuals, and are formed during late PCR cycles (Schlötterer, 1998). Figure 4.5.2.2 illustrates the formation of heteroduplex molecules during PCR amplification. Heteroduplex molecules arise when complementary strands fail to base-pair precisely when reannealing. Failure to base-pair correctly may be due to the reannealing of complementary strands from heterozygous alleles or occasionally due to an error during PCR amplification such as slippage (especially with microsatellite alleles) or due to the insertion of an incorrect base into the sequence of one strand. An apparently homozygous locus may have alleles which differ slightly in sequence but not size: reannealing of complementary strands from such different alleles may, therefore, also result in heteroduplex formation. Whatever the mechanism generating the mismatch, when the strands reanneal secondary structures occur in the double-stranded molecule at sites of mismatched bases.



Formation of heteroduplex molecules



Figure 4.5.2.2: Formation of heteroduplex molecules. At a homozygous locus, assuming the sequence composition of alleles 1 and 2 are the same[†], two complementary single strands will arise from denaturation: A1 (X 2) and B1 (X 2). Reannealing regenerates homoduplex molecules (A1 / B1). If an error occurs during PCR amplification such that, for example, there is a base change in one strand (say A1, with mutation called A1^{*}), a heteroduplex molecule (A1^{*} / B1) may be evident if the error results in a mismatch which causes secondary structure in the double-stranded, reannealed molecule. [†](An apparently homozygous locus may also have alleles which differ slightly in sequence but not size: reannealing of complementary strands from different alleles at such a locus may also result in heteroduplex formation.) At a heterozygous locus, alleles 1 and 2 differ, for example by a dinucleotide insertion. Four distinct single strands will arise from denaturation: A1, A2, B1 and B2. Reannealing may regenerate both homoduplex molecules (A1 / B1 or A2 / B2) and heteroduplex molecules (A1 / B2 or A2 / B1). Further heteroduplex bands may also arise by the same processes described for homozygous alleles.

At homozygous loci, usually none or a single heteroduplex molecule is observed. At heterozygous loci, 2 or more bands are commonly present, resulting from combinations of strands from different alleles. Heteroduplex molecules appear much larger than homoduplex molecules when electrophoresed because the secondary structure retards their movement though the gel. The retardation of these molecules depends on the degree of secondary structure and the sequence composition of each molecule.

Among the Brazilian stocks, heteroduplex bands were particularly evident at the AC01 locus for the *L. V. shawi* stocks (lanes 4, 6 and 9 of *figure 4.5.2.1*). The heteroduplex bands for these stocks (2 in each) have slightly different mobilities, indicating the presence of different mismatches within the molecules (and therefore different secondary structures). Heteroduplex molecules (single) were also observed in *L. V. braziliensis* stocks (lanes 3, 7, 8, 11, 12): these exhibit an identical mobility suggesting that the sequence difference generating the mismatch is common to these stocks and was therefore unlikely to have been generated as a result of random error during PCR amplification.

4.5.2.2 <u>AC16</u>

Figure 4.5.2.3 shows the AC16 products from Brazilian stocks. At this locus, one or two bands were observed in each stock: a second band was less common than in AC01. Where a second band was present this band was, as for AC01, usually less intense and always at a set distance from the allelic band. Heteroduplex bands were rarely observed, even in stocks which were shown using Genotyper® to be heterozygous. Some size variation in the allelic bands was evident between stocks at this locus, e.g. lanes 4, 9, and 10. Heterozygote bands are evident in lanes 9 and 10 (weak) which reflect the Genotyper® results in these stocks (see section 4.5.3).

4.5.2.3 <u>AC52</u>

Figure 4.5.2.4 shows AC52 products from Brazilian stocks. The PCR products from this locus revealed complex banding patterns: even for stocks shown to be homozygous at AC52 using Genotyper[®] analysis (e.g. lanes 2-6, figure 4.5.2.4), several bands observed. As discussed previously, these bands were probably locus-specific PAGE artefacts as the larger bands did not correlate with allele peak sizes identified using Genotyper[®] analysis (see section 4.5.3). In addition, many less intense bands were observed which were most likely heteroduplex molecules.

Figure 4.5.2.3

Chapter 4 - Brazil including Três Bracos and Corte de Pedra





Figure 4.5.2.3: AC16 products from Brazilian stocks analyzed using non-denaturing 6% PAGE and silver staining. Lane 1 - 100 bp molecular weight marker (Gibco BRL, UK); Iane 2 - L. V. braziliensis reference strain, LTB300; Iane 3 - IM403; Iane 4 - M8522; Iane 5 - WHI-EO1611; Iane 6 - M11367; Iane 7 - BH325R; Iane 8 - M12045; Iane 9 - M12613; Iane 10 - M12620; Iane 11 - M14405; Iane 12 - 709-LOA; Iane 13 - L. V. shawi reference strain, M15065. Note the additional larger band (at set distance from the allelic band in several lanes) and heteroduplex bands (weak, indicated): see text for discussion.

Figure 4.5.2.4

AC52 products from Brazilian stocks



Figure 4.5.2.4: AC52 products from Brazilian stocks analyzed using non-denaturing 6% PAGE and silver staining. Lane 1 - 1kbp molecular weight marker (Gibco BRL, UK); lane 2 - LTB12; lane 3 - LTB14; lane 4 - LTB30; lane 5 - LTB179; lane 6 - LTB206; lane 7 - LTB253; lane 8 - L. V. braziliensis reference strain LTB300; lane 9 - WH1-10187; lane 10 - WH1-CIPD; lane 11 - JCP-225; lane 12 - Dog-Rico. Note the additional bands (at set distances from the allelic band(s) in each lane) and heteroduplex bands (indicated): see text for discussion.

In all stocks which were heterozygous for AC52 (e.g. lanes 7 and 12), 3 intense bands were observed as well as many heteroduplex bands. The origins of these additional bands are unclear, however the AC52 pattern in these stocks generally correlated with Genotyper® results.

Owing to the general lack of resolution and allele scoring power of non-denaturing PAGE and the availability of Genescan® and Genotyper® software, no further analyses were carried out from PAGE data, other than to note banding patterns and to attempt a correlation of these with Genotyper® data.

4.5.3 Genescan@ and Genotyper@ 2.0 analyses

The complexity of banding patterns observed using non-denaturing PAGE analysis was not easily interpreted: it was not always clear which band(s) to score and accurate sizing was not possible on the short PAGE gels used. To resolve these problems, Genescan® and Genotyper® analyses employing long-range, denaturing PAGE gels were used.

Using Genotyper® software, the PCR product(s) from each microsatellite locus were accurately sized for each stock. Heteroduplex bands and other additional bands "disappeared" such that accurate allele size could be assigned at all loci in almost all stocks.

Table 4.5.3.1 summarizes the allele sizes observed at each microsatellite locus in stocks from Brazil, listed by geographical origin. Table 4.5.3.2 summarizes the data from L. V. braziliensis stocks by host and by clinical type. Table 4.5.3.3 summarizes the data from L. V. braziliensis stocks isolated from sand flies. Results are discussed by each microsatellite locus and finally by multilocus genotype.

Table 4.5.3.1: (see next page) Summary of microsatellite allele sizes found in stocks from Brazil, listed by species then geographical area. Code names in **bold** type indicate stocks used as reference strains. * - isolate IM2832, an unnamed *Viannia* species (Grimaldi *et al.*, 1991). \dagger - stocks identified as *L. V. shawi*. Assignment and scoring of alleles are explained in chapter 3. X indicates that no PCR product was generated. ND indicates not determined: Genescan@/ Genotyper@ analysis failed to generate a scorable peak. Allele sizes given in **bold** type indicate alleles different to the most common allele observed.

Table 4.5.3.1

20

Summary of microsatellite allele sizes found in stocks from Brazil, by geographical area.

		Microsatellite locus					
		AC	201	AC	216	AC	52
Stock code	Region / State	allele	allele	allele	allele	allele	allele
no.*†		1ª	2 ⁶	1	2	1	2
IM2832*	Balbina / Amazonas	227	245	X	X	X	X
IM403+	Manaus / Amazonas	X	X	246	248	X	X
M12045†	Carajas / Pará	231	231	246	246	293	293
M11367†	Carajas / Para	231	231	244	248	291	291
M15065+	Paragominas / Para	241	241	246	248	297	299
M2903	Carajas / Pará	227	227	250	266	297	297
M2904	Carajas / Pará	227	227	246	254	293	295
M8522	Carajas / Pará	227	227	248	254	ND	ND
M12613	Paragominas / Pará	X	X	244	256	X	X
M12620	Paragominas / Pará	223	223	254	262	285	293
Manoel	Amariji / Pernambuco	227	231	244	252	283	283
Paulino	Amariji / Pernambuco	227	231	244	250	285	303
Fragoso	Amariji / Pernambuco	227	231	248	248	283	301
Amaro	Amariji / Pernambuco	231	231	244	244	283	301
HM1B	Miracatu / São Paulo	227	227	244	244	283	301
BH325R	Caratinga /Minas Gerais	227	227	244	246	283	283
Dog-Lamprea	Três Braços / Bahia	227	227	244	244	283	305
Dog-Rodrigo	Três Braços / Bahia	227	227	244	244	283	283
Dog-Rico	Três Braços / Bahia	227	227	244	244	283	305
WHI-EO1094	Três Braços / Bahia	X	X	X	X	X	X
WHI-E01161	Três Braços / Bahia	X	X	244	244	283	283
TUP-EO252	Três Braços / Bahia	X	X	X	X	X	X
LTB12	Três Braços / Bahia	227	227	244	244	283	283
LTB14	Três Braços / Bahia	227	227	244	244	283	283
LTB30	Três Braços / Bahia	227	227	244	244	283	283
LTB179	Três Braços / Bahia	227	227	244	244	283	283
LTB206	Três Braços / Bahia	227	227	244	244	283	283
LTB253	Três Braços / Bahia	227	227	244	244	283	303
LTB300	Três Braços / Bahia	227	227	244	244	283	283
JCP-184	Corte de Pedra / Bahia	X	X	244	246	X	X
JCP-225	Corte de Pedra / Bahia	227	227	244	246	283	283
WHI-10187	Corte de Pedra / Bahia	227	227	244	244	283	283
WHI-CIPD	Corte de Pedra / Bahia	227	227	244	244	283	283
M11468	Corte de Pedra / Bahia	X	X	X	X	X	X
M14405	Flora / Paraná	227	227	242	242	283	301
709-LOA	Peabiru / Paraná	X	X	244	244	X	X
769-VU	S. Jorge do Ivai / Parana	X	X	X	X	X	X
					-		

			Microsatellite locus					
			AC	:01	AC	:16	AC	52
Stock code no.	Host	Clinical type	allele 1 ^a	allele 2 ^b	allele 1	alleie 2	allele 1	allele 2
M2903	Human	luman CL		227	250	266	297	297
M2904	Human	CL	227	227	246	254	293	295
Manoel	Human	CL	227	231	244	252	283	283
Paulino	Human	CL	227	231	244	250	285	303
Fragoso	Human	CL	227	231	248	248	283	301
Amaro	Human	CL	231	231	244	244	283	301
M14405	Human	CL	227	227	242	242	283	301
709-LOA	Human	CL	x	x	244	244	x	х
769-VU	Human	CL	x	x	X	X	x	x
LTB30	Human	CL	227	227	244	244	283	283
LTB206	Human	CL	227	227	244	244	283	283
LTB253	Human	CL	227	227	244	244	283	303
LTB12	Human	MCL	227	227	244	244	283	283
LTB14	Human	MCL	227	227	244	244	283	283
LTB179	Human	MCL	227	227	244	244	283	283
LTB300	Human	MCL	227	227	244	244	283	283
Dog-Lamprea	Dog	CL	227	227	244	244	283	305
Dog-Rodrigo	Dog	CL	227	227	244	244	283	283
Dog-Rico	Dog	CL	227	227	244	244	283	305
JCP-184	Donkey	CL	x	X	244	246	x	x
JCP-225	Donkey	CL	227	227	244	246	283	283
HM1B	Rodent	-	227	227	244	244	283	301
BH325R	Rodent	-	227	227	244	246	283	283

Table 4.5.3.2Summary of microsatellite allele sizes of L. V. braziliensisstocks from Brazil, by host and by clinical type.

Table 4.5.3.2: Summary of microsatellite allele sizes of L. V. braziliensis stocks from Brazil, by host and by clinical type. Code names in **bold** type indicate stocks used as reference strains. Assignment and scoring of alleles are explained in chapter 3. X indicates that no PCR product was generated. Allele sizes given in **bold** type indicate alleles different to the most common allele observed.

			Mi	crosate	llite loo	cus	
		AC	201	AC	C16	AC52	
Stock code no.	Sand fly vector species	allele 1ª	alleie 2 ^b	allele 1	allele 2	allele 1	alleie 2
M8522	Lu. wellcomei	227	227	248	254	ND	ND
M12613	Lu. complexus	x	x	244	256	x	x
M12620	Lu. complexus	223	223	254	262	285	293
WHI-EO1094	Lu. whitmani	x	x	X	x	x	x
WHI-E01161	Lu. whitmani	x	x	244	244	283	283
TUP-EO252	Lu. tupynambai	x	x	x	x	x	x
WHO-10187	Lu. whitmani	227	227	244	244	283	283
WHI-CIPD	Lu. whitmani	227	227	244	244	283	283
WHI-M11468	Lu. whitmani	x	x	X	x	X	X

Table 4.5.3.3Summary of microsatellite allele sizes of L. V. braziliensisstocks isolated from sand flies from Brazil.

Table 4.5.3.3: Summary of microsatellite allele sizes of L. V. braziliensis stocks isolated from sand flies from Brazil. Assignment and scoring of alleles are explained in chapter 3. X indicates that no PCR product was generated. ND indicates not determined: Genescan Φ / Genotyper Φ analysis failed to generate a scorable peak. Allele sizes given in **bold** type indicate alleles different to the most common allele observed.

4.5.3.1 <u>AC01</u>

AC01 genotypes identified in Brazilian stocks are summarized in *table 4.5.3.4*. The most common allele size identified among *L. V. braziliensis* stocks was 227: 4 genotypes were scored. Most *L. V. braziliensis* stocks were homozygous (i.e. generated a single peak), with a presumed genotype of 227/227 (19/24 stocks).

All stocks from Três Braços / Corte de Pedra, excluding those having "null alleles", were genotyped as 227/227 (see *table 4.5.3.1*). Among *L. V. braziliensis* stocks from Pará, one (M12620) was genotype 223/223; all others were 227/227. In Brazilian *L. V. braziliensis* stocks, the 231 allele was found only in Amariji (Pernambuco State). Three Amariji stocks were heterozygous at AC01 (genotype 227/231) and a fourth stock was homozygous (genotype 231/231). The 231 allele was identified originally in the *L. V. peruviana* reference strain LC39: among these Brazilian stocks it has also been found to be associated with *L. V. braziliensis* and 2 *L. V. shawi* stocks. *L. V. shawi* stocks had genotypes of 231/231 or 241/241. The stock IM2832 was

heterozygous, as indicated by the presence of 2 peaks, with genotype 227/245. All stocks isolated from MCL lesions were AC01 genotype 227/227.

Two AC01 alleles, 227 and 231, were identified in *L. V. braziliensis* stocks from host species: 231 was found only in stocks isolated from human hosts. This allele was also observed in sand fly isolates, but these stocks were identified as *L. V. shawi*. Two genotypes, 227/227 and 223/223, were observed in *L. V. braziliensis* stocks from sand fly vectors: 223 was found only in 1 sand fly stock. Null alleles were commonly observed among sand fly stocks from Três Braços / Corte de Pedra (4/6 stocks).

Species§	Inferred genotype*	No. stocks	Stocks
L. V. sp. n.	227 / 245	1	IM2832*
L. V. shawi	231/231	2	M12045, M11367
	241/241	1	M15065
	223 / 223	1	M12620
L. V. braziliensis	227 / 227	19	M2903, M2904, M8522, HM1B, BH325R, Dog-Lamprea, Dog-Rodrigo, Dog-Rico, LTB12, LTB14, LTB30, LTB179, LTB206, LTB253, LTB300, JCP-225, WHI-10187, WHI-C1PD, M14405
	227/231	3	Manoel, Paulino, Fragoso
	231/231	1	Amaro

 Table 4.5.3.4
 AC01 genotypes found in stocks from Brazil

Table 4.5.3.4: AC01 genotypes found in stocks from Brazil. [§] Species status assigned from IEA. * Genotype inferred from scoring Genotyper® peaks. Code names in **bold** type indicate stocks used as reference strains. * - isolate IM2832, an unnamed *Vicarmia* species (Grimaldi *et al.*, 1991). Allele assignment is explained in chapter 3.

4.5.3.2 <u>AC16</u>

AC16 genotypes found in stocks from Brazil are summarized in *table 4.5.3.5*. Among *L. V. braziliensis* stocks, the most common genotype (of the 11 genotypes scored) at the AC16 locus was 244/244. Many more alleles (10) and genotypes (11) were scored among *L. V. braziliensis* stocks at this locus than at AC01, and many of the stocks (10/28) were heterozygous.

Species§	Inferred genotype*	No. stocks	Stocks
L. V. shawi	244 / 248	1	M11367
	246 / 246	1	M12045,
	246 / 248	2	M15065, IM403
	242 / 242	1	M14405
	244 / 244	16	Amaro, HM1B, Dog-Lamprea, Dog- Rodrigo, Dog-Rico, LTB12, LTB14, LTB30, LTB179, LTB206, LTB253, LTB300, WHI-EO1161, WHI-10187, WHI-C1PD, M14405, 709-LOA
	244 / 246	3	BH325R, JCP-184, JCP-225
	244 / 250	1	Paulino
L. V. braziliensis	244 / 252	1	Manoel
	244 / 256	1	M12613
	246 / 254	1	M2904
	248 / 248	1	Fragoso
	248 / 254	1	M8522
	250 / 266	1	M2903
	254 / 262	1	M12620

Table 4.5.3.5 AC16 genotypes found in stocks from Brazil

Table 4.5.3.5: AC16 genotypes found in stocks from Brazil. § Species status assigned from IEA. * Genotype inferred from scoring Genotyper[®] peaks. Code names in **bold** type indicate stocks used as reference strains. Allele assignment is explained in chapter 3.

Most stocks from Três Braços / Corte de Pedra were genotyped as 244/244: the 2 donkey stocks were genotyped as 244/246. The 246 allele was also observed in two other stocks: BH325R, a rodent isolate from Minas Gerais and M2904, a human stock from Carajas (Pará). Stocks from Pará State were highly variable at this locus: 5 stocks from 2 regions each had a distinct heterozygous genotype. L. V. braziliensis allele sizes in stocks from more northern Brazilian states (Para, Pernambuco) were commonly larger than those from other states. The four stocks from Amariji (Pernambuco) each had a different genotype at this locus, 2 of which were heterozygous. All alleles identified in the L. V. shawi stocks were also found in the L. V. braziliensis stocks, however in combinations giving rise to different genotypes. All stocks isolated from MCL lesions were AC16 genotype 244/244.

The AC16 genotype 244/246 was identified only in non-human host species: the 246 allele was also observed in 1 human stock, M2904, but in a different genotype (246/254). Among the AC16 genotypes observed from sand fly stocks, those from Para State were, similarly to host species stocks, more variable than those originating from Três Braços / Corte de Pedra. As with AC01, null alleles were commonly observed among sand fly stocks from Três Braços / Corte de Pedra (3/6 stocks).

4.5.3.3 <u>AC52</u>

AC52 genotypes found in stocks from Brazil are summarized in *table 4.5.3.6*. Among *L. V. braziliensis* stocks, the most common genotype (of the 8 genotypes scored) was 283/283 (13/24). As with AC16, many *L. V. braziliensis* stocks (11/24) were heterozygous.

Species§	Inferred genotype*	No. stocks	Stocks
L. V. shawi	293 / 293	1	M12045
	291 / 291	1	M11367
	297 / 299	1	M15065
	283 / 283	13	Manoel, BH325R, Dog-Rodrigo, LTB12, LTB14, LTB30, LTB179, LTB206, LTB300, JCP-225, WHI- EO1161, WHI-10187, WHI-C1PD
	283 / 301	4	Fragoso, Amaro, HM1B, M14405
	283 / 303	1	LTB253
L. V. braziliensis	283 / 305	2	Dog-Lamprea, Dog-Rico
	285 / 293	1	M12620
	285 / 303	1	Paulino
	293 / 295	1	M2904
	297 / 297	1	M2903
	ND	1	M8522

Table 4.5.3.6 AC52 genotypes found in stocks from Brazil

Table 4.5.3.6: AC52 genotypes found in stocks from Brazil. § Species status assigned from IEA. * Genotype inferred from scoring Genotyper® peaks. Code names in **bold** type indicate stocks used as reference strains. Allele assignment is explained in chapter 3.

Three genotypes were scored among the 14 stocks from Três Braços / Corte de Pedra (excluding stocks with null alleles). Most stocks were genotype 283/283, 2 dog stocks were 283/305 and a single human stock was 283/303. Three genotypes were scored from the 4 stocks from Amariji: 2 genotypes (3 stocks) were heterozygous. Of the 3 stocks scorable at AC52 from Para State, each had a distinct genotype. Of these, two stocks from Carajas had genotypes which were more similar to those of *L. V. shawi* stocks (also from Para). Each *L. V. shawi* stock had a different AC52 genotype: four alleles were observed in these stocks of which two (293 and 297) were also found in the *L. V. braziliensis* stocks from Para State. As with AC16, however, these alleles were observed in different genotypes. All stocks isolated from MCL lesions were AC52 genotype 283/283. The AC52 genotype 283/305 was identified only in 2/3 stocks isolated from dogs: the 305 allele was not found in any other stocks. As with AC01 and AC16, null alleles were commonly observed among sand fly *L. V. braziliensis* stocks (4/9 stocks).

4.5.3.4 <u>Multilocus analysis of microsatellite data</u>

I propose here to use the term "microdeme" to describe *Leishmania* sharing the same microsatellite multilocus genotype (i.e. combined genotype data from the 3 microsatellite loci: AC01, AC16 and AC52). "LMS" numbers ("London microsatellite") have been used to identify different microdemes (i.e. LMS1, 2, 3 etc.). Microdemes observed in stocks from Brazil are summarized in *table 4.5.3.7*.

Stocks which failed to generate PCR products at any of the three loci were not assigned a microdeme number because the genetic basis of failure to amplify could not be determined. Stocks which had null alleles at one or two loci were assigned LMS numbers: each of these stocks belonged to a different microdeme owing mostly to variation at the AC16 locus. Including these stocks, 23 microdemes were scored among the 33 Brazilian stocks: *L. V. braziliensis* stocks (28) were separated into 18 microdemes and *L. V. shawi* stocks (4) into 4 microdemes. LMS1 was the most common microdeme observed, primarily because of the relatively large number of stocks from Três Braços / Corte de Pedra, most of which were LMS1. Most microdemes contained a single stock, testimony to the discriminatory power of this technique.

Microdeme		Microsatellite locus genotype				e	Geogra	phical origin	Stock(s)	Viannia
ŧ	A	C01	01 A(A	C52	State/Dept	Region/Area		species
LMSI	227	227	227 244		283	283	Bahia	Três Braços / CDP	LTB300, LTB12, LTB14, LTB30, LTB179, LTB206, Dog-Rodrigo, WHI-10187, WHI-CIPD	b
LMS4	241	241	246	248	297	299	Para	Paragominas	M15065	8
LMS7	227	245	X	X	X	X	Amazonas	Balbina	IM2832*	*
LMS8	227	227	244	246	283	283	Bahia Minas Gerais	Três Braços / CDP Caratinga	JCP-225 BH325R	b
LMS9	227	227	244	244	283	301	São Paulo	Miracatu	HM1B	b
LMS10	227	227	244	244	283	303	Bahia	Três Braços / CDP	LTB253	b
LMS11	227	227	244	244	283	305	Bahia	Três Braços / CDP	Dog-Rico, Dog-Lamprea	b
LMS12	227	227	242	242	283	301	Paraná	Flora	M14405	b
LMS13	227	227	246	254	293	295	Pará	Carajas	M2904	b
LMS14	227	227	248	254	F	F	Para	Carajas	M8522	b
LMS15	227	227	250	266	297	297	Pará	Carajas	M2903	b
LMS16	227	231	244	250	285	303	Pernambuco	Amaraji	Paulino	b
LMS17	227	231	244	252	283	283	Pernambuco	Amaraji	Manoel	b
LMS18	227	231	248	248	283	301	Pernambuco	Amaraji	Fragoso	b
LMS19	231	231	244	244	283	301	Pernambuco	Amaraji	Amaro	b
LMS20	231	231	244	248	291	291	Pará	Carajas	M11367	S
LMS21	231	231	246	246	293	293	Pará	Carajas	M12045	S
LMS22	223	223	254	262	285	293	Pará	Paragominas	M12620	b
LMS23	X	X	244	244	283	283	Bahia	Três Braços / CDP	WHI-EO1161	b
LMS24	X	x	244	244	x	x	Paraná	Peabiru	709-LOA	b
LMS25	X	x	244	246	X	X	Bahia	Três Braços / CDP	JCP-184	b
LMS26	X	x	244	256	X	X	Pará	Paragominas	M12613	b
LMS27	X	X	246	248	X	X	Amazonas	Manaus	IM403	S
not assigned [±]	x	x	x	x	x	x	Bahia Parana	Três Braços / CDP S. Jorge do Ivai	WHI-EO1094, TUP-EO252, WHI-M11468 769-VU	b

Table 4.5.3.7: Summary of microdemes† (multilocus microsatellite genotypes) observed in Brazilian stocks.

154

Chapter 4 - Brazil including Três Bracos and Corte de Pedra

Notes to *table 4.5.3.7:* (see previous page) summary of microdemes observed in Brazilian stocks. \dagger microdeme is the term coined here to describe *Leishmania* sharing the same microsatellite multilocus genotype (i.e. combined genotype data from the 3 microsatellite loci: AC01, AC16 and AC52). "LMS" numbers have been used to identify different microdemes (i.e. LMS1, 2, 3 etc.). Stocks which failed to generate PCR products at any of the three loci were not assigned a microdeme number because the genetic basis of failure to amplify could not be determined. \diamond CdP - abbreviation for Corte de Pedra. $\S Viannia$ species codes: b - *braziliensis*; s - *shawi*; * - unnamed *Viannia* species (*L. V.* sp. n., see Grimaldi *et al.* 1991). Code names in **bold** type indicate reference strains. Stocks which are in *italics* are from MCL lesions. F - failed to generate scorable Genotyper peak.

Microdemes observed in L. V. braziliensis stocks from Brazil are summarized by geographical region in *table 4.5.3.8* and by host / vector group in *table 4.5.3.9*.

Geographical region	Number of stocks	Number of microdemes† observed	Microdemes† observed (LMS No.)
Três Braços / Corte de Pedra	15	6	1, 8, 10, 11, 23, 25
Para	5	5	13, 14, 15, 22, 26
Pernambuco	4	4	16, 17, 18, 19
Other	4	4	8, 9, 12, 24

Table 4.5.3.8:Microdemes [†] observed in L. V. braziliensis stocks from
Brazil, summarized by geographical region.

Table 4.5.3.9: Microdemes⁺ observed in L. V. braziliensis stocks from Brazil, summarized by host / vector group.

Host / vector group	Number of stocks	Number of microdemest observed	Microdemest observed (LMS No.)		
Human	15	10	1, 10, 12, 13, 15, 16, 17, 18, 19, 24, 27		
Dog	3	2	1, 11		
Donkey	2	2	8, 25		
Rodent	2	2	8,9		
Sand fly	6	5	1, 14, 22, 23, 26		

Tables 4.5.3.8 and 4.5.3.9: microdemes observed in L. V. braziliensis stocks from Brazil, summarized by geographical region (4.5.3.8) and by host / vector group (4.5.3.9). + "microdeme" is the term coined here to describe *Leishmania* sharing the same microsatellite multilocus genotype. LMS numbers have been used to identify different microdemes. Stocks which failed to generate PCR products at any of the three loci are not included in this multilocus analysis because the genetic basis of their null alleles could not be determined.

Of the 15 *L. V. braziliensis* stocks assigned from Três Braços / Corte de Pedra, 6 microdemes were scored. This is in contrast to the single zymodeme scored in these stocks using isoenzyme analysis. All stocks isolated from MCL lesions (all from Três Braços / Corte de Pedra) were LMS1. Every stock from Para (Carajas and Paragominas) and Pernambuco (Amariji) States belonged to a different microdeme. Only one microdeme (LMS8) contained stocks isolated from different geographical areas. Microdemes 1 and 8 encompassed stocks from different host / vector groups. LMS1 was observed in human, dog and sand fly stocks (all Três Braços / Corte de Pedra); LMS 8 was observed in 1 stock each from the dog and rodent groups.

4.5.4 AC01 Sequencing

This method of microsatellite analysis is the most sensitive of all the approaches used, allowing visualization of the microsatellite region itself and identification of any diversity in the flanking regions. Sequencing analysis was carried out for the AC01 locus only. Of the 28 Brazilian stocks which generated a PCR product for AC01, all presented readable AC01 sequence data except the reference strain IM2832 (L. V. sp., see section 3.8.4). Several stocks were sequenced at least twice and the sequence was found to be reproducible. *Figure 4.5.4.1* shows the alignment of sequences from the AC01 microsatellite locus for Brazilian stocks.

Notes to figure 4.5.4.1: (see next page) sequence alignment of the AC01 locus from Brazilian stocks and L. V. guyanensis. §Viannia species abbreviation: b - braziliensis; s - shawi; g - guyanensis. * Stock type: LTB300 type (LTB300, LTB12, Dog-Rico, Dog-Lamprea, WHI-10187, M2904); LTB14 type (LTB14, LTB30, LTB179, LTB206, LTB253, Dog-Rodrigo, JCP-225, WHI-C1PD, HM1B, BH325R, Fragoso, Paulino); M2903 type (M2903, M8522, M12620). Reference strains are indicated in **bold** type. Bases in blue type form the microsatellite repeat region; bases in red type illustrate the TCCGC end motif; bases in green type are invariant downstream in all stocks. Numbers above sequence data indicate the base position used for alignment purposes. Only 165 bases of the total AC01 product were aligned and used for phylogenetic analysis (see chapter 3). See Appendix IV for base coding system.

Of the 15 L. V. braziliensis stocks assigned from Três Braços / Corte de Pedra, 6 microdemes were scored. This is in contrast to the single zymodeme scored in these stocks using isoenzyme analysis. All stocks isolated from MCL lesions (all from Três Braços / Corte de Pedra) were LMS1. Every stock from Pará (Carajas and Paragominas) and Pernambuco (Amariji) States belonged to a different microdeme. Only one microdeme (LMS8) contained stocks isolated from different geographical areas. Microdemes 1 and 8 encompassed stocks from different host / vector groups. LMS1 was observed in human, dog and sand fly stocks (all Três Braços / Corte de Pedra); LMS 8 was observed in 1 stock each from the dog and rodent groups.

4.5.4 AC01 Sequencing

This method of microsatellite analysis is the most sensitive of all the approaches used, allowing visualization of the microsatellite region itself and identification of any diversity in the flanking regions. Sequencing analysis was carried out for the AC01 locus only. Of the 28 Brazilian stocks which generated a PCR product for AC01, all presented readable AC01 sequence data except the reference strain IM2832 (L. V. sp., see section 3.8.4). Several stocks were sequenced at least twice and the sequence was found to be reproducible. *Figure 4.5.4.1* shows the alignment of sequences from the AC01 microsatellite locus for Brazilian stocks.

Notes to figure 4.5.4.1: (see next page) sequence alignment of the AC01 locus from Brazilian stocks and L. V. guyanensis. §Viannia species abbreviation: b - braziliensis; s - shawi, g - guyanensis. * Stock type: LTB300 type (LTB300, LTB12, Dog-Rico, Dog-Lamprea, WHI-10187, M2904); LTB14 type (LTB14, LTB30, LTB179, LTB206, LTB253, Dog-Rodrigo, JCP-225, WHI-CIPD, HM1B, BH325R, Fragoso, Paulino); M2903 type (M2903, M8522, M12620). Reference strains are indicated in **bold** type. Bases in blue type form the microsatellite repeat region; bases in red type illustrate the TCCGC end motif, bases in green type are invariant downstream in all stocks. Numbers above sequence data indicate the base position used for alignment purposes. Only 165 bases of the total AC01 product were aligned and used for phylogenetic analysis (see chapter 3). See Appendix IV for base coding system.

Species	Strain /stock*	10	20	30	40	50	60	70	80
ş	code no	1	1	1	1	1	1	1	1
b	LTB300 type	ATGTGCCTCT	CCCACCCTTA	GTGCTTGTCT	TCTTCCTGCT	TTGCCTCTCT	GTCTGT	GTGTGTGTGT	GTG
b	LTB14 type								
b	M14405								
b	M2903 type						A		
b	Manoel						A		
b	Amaro						A	c	
\$	M15065			C			TGTC		TGTGCGT
5	M11367			CC					TGTGTGT
S	M12045			C			TGTC	• • • • • • • • • • •	TGTGTGY
g	M4147			C		C	· · · · · · · ·	• • • • • • • • • • • •	TGYGYGT

Figure 4.5.4.1	Sequence alignment of the ACU	l locus from Brazilian stocks and	the L.	V. guyanensis	reference strain.
----------------	-------------------------------	-----------------------------------	--------	---------------	-------------------

Species	Strain /stock*	90	100) 110	0 120) 130	140	J <u>15</u>	101	J 165
ş	code no.	1	1	1	1	1	.1	1	1	1
b	LTB300 type		-TCCGCTTCA	GTGGGCCGAT	CCGTTTCACT	TTTTGCCGGT	GACGTTTGTG	TGCTGACGTG	TCTGGTGGCC	TCWCA
b	LTB14 type									A
b	M14405							N		A
b	M2903 type									A
b	Manoel									
b	Amaro									
5	M15065	GCG				K	CK	₩		
s	M11367	GTGTG	c	A	CW	GA	CG	A	G	W.A
\$	M12045	GYGYG	YGC	CWSSG	AA	GW	ACGC	YASA	.s	A.A
8	M4147	GTGYGYS	SCA	.KSAY	SCW	GY.SSA	SCG	SAY	.SGK	W.A

4.5.4.1 AC01 sequence variation among Brazilian stocks

L. V. braziliensis and L. V. shawi stocks could easily be separated based on the size of the repeat (8 : 13/14 respectively) and the base at position 28 ([T : C] respectively). The sequence upstream of the repeat itself was highly conserved in all stocks: a single (different) base change was identified in stocks of both L. V. braziliensis ([G to A] at position 51 in 5 stocks) and L. V. shawi ([T to C] at position 25 in 1 stock).

Among L. V. braziliensis stocks, 6 sequence types were scored compared to the 4 AC01 genotypes scored using Genotyper®. Six stocks had AC01 sequences identical to that of the L. V. braziliensis reference strain LTB300: all bar one (M2904) came from Três Braços / Corte de Pedra. L. V. braziliensis stocks could be divided into 2 groups based on comparison of the upstream flanking region. Five stocks (M2903-type stocks [3], Amaro and Manoel) had [C] at position 51 instead of the [G] present in all other (19) stocks. The base change in the sequence of these 5 stocks (originating from 2 northern States, Pará and Pernambuco) was very clear. Stocks from Três Braços / Corte de Pedra could be split into 2 groups: LTB300-type stocks were heterozygous at position 163 [W, i.e. A or T] whereas all LTB14-type stocks had [A] at this position. Permutations of bases present at positions 51 and 163 generated 4 of the sequence types in L. V. braziliensis stocks from Brazil. Two further sequence types were observed involving a single base change ([T to C] at position 63, within the repeat) and a failed base (Amaro and M14405 stocks respectively).

The L. V. shawi stocks, all homozygous at AC01 using Genotyper®, each had unique AC01 sequence. In addition to the guyanensis complex-specific [C] at nucleotide position 28, two of the 3 L. V. shawi stocks (reference strain M15065 and M11367) had a 4 base pair insertion immediately before the repeat motif. The third L. V. shawi stock had a mutation [T to C] at position 25. All L. V. shawi stocks were found to have [CG] or [YG] insertions in the repeat motif ([Y] indicates [C] or [T]). The L. V. shawi reference strain M15065 had a dinucleotide repeat [(TG + CG)]n, where n = 13; both of the other L. V. shawi stocks had n = 14. The stock M12045 appeared to have heterozygous sites [Y] within the repeat suggesting that the AC01 alleles varied at these positions. The sequence downstream of the repeat for all L. V. shawi stocks was more diverse than those of L. V. braziliensis stocks. Many of these downstream sites in L. V. shawi stocks were scored as heterozygous, i.e. had split peaks. This variability is likely due to the fact that the repeat is longer in these stocks resulting in more

replication errors (i.e. DNA slippage and subsequent inaccurate DNA mismatch repair, see chapters 1 and 3). Overall, *L. V. braziliensis* stock repeats are shorter and the entire locus sequence appears to be much more stable than those of *L. V. shawi*.

4542 Comparison of AC01 data from sequencing and Genotyper® analyses Sequence data sometimes failed to correlate with Genotyper® data. Among L. V. braziliensis stocks, the majority of alleles identified as 227 using Genotyper® had virtually identical sequence lengths where the number of dinucleotide repeats (n) was 8. However, 2 other alleles (223 and 231) were scored in three genotypes: 223/223 (1 stock), 227/231 (3 stocks) and 231/231 (1 stock) in these stocks. AC01 sequence analysis did not account for these differences in allele size: the number of dinucleotide repeats scored in these stocks was identical to other L. V. braziliensis stocks (i.e. n =8) and no insertions or deletions were revealed. It is possible that a 4 base pair insertion (to give 231) or deletion (to give 223) occurred in the extreme 5' sequence which was not used for alignment. Stocks scored as 227/231 using Genotyper® generated a single sequence which was compatible, in length, to the L. V. braziliensis reference strain. If heterozygous, two sequences should have been produced which, if changes in repeat number were the cause of the allele size difference, should have resulted in the sequences overlapping after the repeat. It is possible that Genotyper® analysis was incorrect or that cycle sequencing greatly favoured the amplification of the shorter allele: the latter seems more likely of the two although both Genotyper® and sequence data for these stocks appeared unequivocal. Heteroduplex bands were observed in PAGE analysis of these stocks (see figure 4.5.2.1) but, as discussed previously, these are not always evidence of heterozygous loci as they can occur at both heterozygous and homozygous loci.

Sequence analysis of AC01 for the L. V. shawi reference strain M15065 reveals, in comparison with LTB300 (allele size 227), the presence of a 4 base insert plus 5 additional dinucleotide repeats. This gives a difference in bases of 14 (4 + 5x2): the expected allele size is therefore 241 (227 + 14). The Genotyper® data from M15065 (observed allele size 241) thus correlates with sequence data. The Genotyper® data from M11367 and M12045 (231/231) however, do not correlate with sequence data. M12045 has the 4 base insertion and 6 additional dinucleotides which should give a size difference of 16 bases (4 + 12), i.e. expected allele size 243. M11367 does not have the 4 base insert but has 6 additional dinucleotide repeats, giving a size
Chapter 4 - Brazil including Três Braços and Corte de Pedra

difference of 12 bases, i.e. expected allele size 239. Genotyper® analysis of both of these stocks reveals allele sizes of 231. The discrepancy between the two methods may be due to insertions (12 and 8 bases respectively) outside the sequence aligned for phylogenetic analysis, or due to an error in Genotyper® analysis. The ambiguities observed in some stocks between sequencing analysis and Genotyper® allele size scoring need to be resolved. Ambiguities were more commonly encountered in stocks which were identified as heterozygous by Genotyper® (i.e. double peaks scored) but which produced sequence data which appeared to be homozygous.

Sequencing analysis is not an option for the multilocus analysis of many stocks, or for the analysis of populations containing large numbers of heterozygote individuals. The use of microsatellite allele sizes for population and phylogenetic analyses is increasingly common (Rongnoparut et al., 1996; Lehmann et al., 1997; Oliveira et al., 1998), despite observations that allele size variation may also result from insertions and deletions in the flanking regions (here; Valdes et al., 1993; Orti et al., 1997; Grimaldi & Cronau-Roy, 1997; Viard et al., 1998). Over all stocks examined, a good correlation was observed between sequencing and Genotyper® results, suggesting that population and phylogenetic analyses based on Genotyper® data (i.e. allele size) alone is a valid approach, giving rise to only a few errors. Of course, without sequence analysis, any errors would remain unidentified and could seriously affect later analyses. Even for those stocks for which an exact correlation between sequence data and allele size was found, subsequent population and phylogenetic analyses based on Genotyper® data would underestimate the true level of diversity (as revealed by sequence analysis). As discussed in chapters 1 and 3, Genotyper® analysis lends itself very well for population genetic purposes because large amounts of data can be generated using a multiplex approach. Another advantage of Genotyper[®] analysis is that heterozygote stocks may be easily identified: heterozygote detection is paramount to stocks from Nicaragua and Huánuco, Peru, and is thus discussed in more detail in chapters 5 and 6.

Chapter 4 - Brazil including Tres Braços and Corte de Pedra

4.6 Phylogenetic analysis of Brazilian stocks

Phylogenetic analysis of isoenzyme data (multilocus) included data from all *Leishmania* reference strains. Phylogenetic analysis of sequences from the microsatellite locus AC01 included data from all reference strains of the *Viannia* subgenus, with the exception of L. V. sp. n. reference strain IM2832 which failed to produce readable sequence. The AC01 microsatellite locus is specific to the *Viannia* subgenus hence it was not possible to include reference strains from the subgenus *Leishmania* in this phylogenetic analysis.

4.6.1 Phylogenetic analysis of isoenzyme data

Isoenzyme phylogenies were constructed, as described in chapter 2, using a maximumlikelihood approach, quartet puzzling (Strimmer & von Haeseler, 1996). Figure 4.6.1.1 depicts a cladogram showing the results of quartet puzzling of zymodeme data for the Brazilian stocks and reference strains. The corresponding phylogram is not shown because the outgroup branch lengths (i.e. subgenus *Leishmania* species) therein were, by comparison, very long thus making the shorter branch lengths (i.e. between stocks of the subgenus *Viannia*) difficult to see.

The cladogram was generated from analysis of data spanning 8 of L. V. braziliensis zymodemes and 9 zymodemes from other Leishmania taxa. Two of the Leishmania subgenus reference strains (PP75, L. L. chagasi and M2269, L. L. amazonensis) were clearly distinct (55% quartet support) from species of the Viannia subgenus. The L. L. mexicana reference strain BEL21 however, clustered amongst the more distantly related species of the Viannia subgenus. The reasons for this disparity are not immediately evident, but most likely due to shared characters observed between these taxa using this particular set of enzyme loci. The phylogenetic positions of the reference strains IM2832 (L. V. sp. n.) and M6426 (L. V. lainsoni) as members of the subgenus Viannia were not clearly defined by this analysis as both of these strains fell outside of the 2 main Viannia species complexes.

Members of the guyanensis complex (cluster I) were distinct from outlying species. Strains and stocks examined were separated into species groups (L. V. panamensis, LS94, L. V. guyanensis, M4147 and L. V. shawi, M15065) which were highly supported (91%, 94% and 98% quartet support). All Brazilian L. V. shawi stocks clustered together (cluster Ia, figure 4.6.1.1) indicating the validity of this species group. Chapter 4 - Brazil including Tres Bracos and Corte de Pedra



Figure 4.6.1.1 Cladogram showing the results of quartet puzzling for Brazilian stocks and reference strains based on zymoderne data. Values on branches are "quartet supports", figures similar to bootstraps (Strimmer & von Hassier, 1996). Values below 50% are not shown. Quartet supports indicate the percentage number of quartets (differing in allele frequencies) supporting each given branching order. Branch lengths are arbitrary. Cluster I contains strains and stocks belonging to the *L. V. guyanensis* complex and includes all *L. V. shawi* stocks (cluster Ia, 1 zymodeme, LON203). Clusters II - VI contains strains and stocks of the *L. V. braziliensis* complex: II contains 2 zymodemes (LON202 and LON211), one of which is the *L. V. peruviana* reference strain LC39; IV contains 2 zymodemes (LON204 and LON215); V contains 1 zymodeme, LON203 and VI contains 2 zymodemes (LON200 and LON210).

Chapter 4 - Brazil including Três Braços and Corte de Pedra

Strains and stocks of the L. V. braziliensis complex were also separated from other taxa (52% quartet support). The heterogeneous nature of this taxon is illustrated by the division into clades ($\mathbf{II} - \mathbf{VI}$, figure 4.6.1.1) which were all highly supported by quartet supports (range 82 - 91% quartet support). L. V. braziliensis stocks were divided into 5 clades: the largest clade (\mathbf{II}) contained 22 stocks from 2 zymodemes including the L. V. braziliensis reference strain LTB300 and all bar one of the Três Braços / Corte de Pedra stocks. Clade **III** contains 2 L. V. braziliensis stocks (both LON211) and the L. V. peruviana reference strain LC39 (LON201), testimony to the close relationship of L. V. peruviana and L. V. braziliensis stocks. Clades **IV** and **VI** each contain 3 stocks from 2 zymodemes; clade V contains 2 stocks from a single zymodeme.

Owing to the difficulty of weighting binary data for maximum likelihood analysis, all alleles were equally weighted in this analysis. Equal weighting of alleles, as opposed to enzymes, results in highly polymorphic enzyme loci assuming more importance than less polymorphic loci because, by nature, more alleles can be scored. Equal weighting of enzymes may have been more appropriate since mutation rates of alleles at a given enzyme locus may differ.

Analysis of these stocks using a similarity approach (based on a similarity coefficient matrix, SCM, produced according to Jaccard's method, 1908) was also carried out. In this approach, equal weighting is also given to all alleles, but the "over-weighting" of polymorphic loci is removed. Dendrograms were constructed by UPGMA (the unweighted pair-group method using arithmetic averages) using the SYN-TAX-pc package (Podani, 1993). *Figure 4.6.1.2* shows the dendrogram constructed for these stocks from a similarity coefficient matrix. Stocks are classified according to similarity (range 0 - 1).

The clustering observed using this method is in overall agreement with results from quartet puzzling, however individual zymodemes are identified. The *Viannia* and *Leishmania* subgenera are clearly defined with the exception of the *L. V. lainsoni* reference strain M6426, which groups more closely with members of the *Leishmania* subgenus.



Figure 4.6.1.2: UPGMA dendrogram constructed from the similarity coefficient matrix for Brazilian stocks and reference strains based on isoenzyme data. Similarities between zymodemes (y-axis, range 0-1) were calculated, using SYN-TAX-pc (Podani, 1993), by averaging the Jaccard coefficients calculated from data from twelve enzyme loci. Stocks / strains in **bold** type are reference strains. Zymodemes (LON numbers) are indicated on branches.

Chapter 4 - Brazil including Três Bracos and Corte de Pedra

The L. L. mexicana reference strain BEL21 clearly groups in the Leishmania subgenus using this approach, however it is still separated from L. L. amazonensis (L. L. mexicana complex) which again shows more similarity to L. L. chagasi (L. L. donovani complex).

Using this approach, the L. V. sp. n. reference strain IM2832 lies between the *braziliensis* and *guyanensis* complexes, but clearly within the *Viannia* subgenus. This parasite, as a potential new *Viannia* species (Grimaldi *et al.*, 1991) will be discussed further in chapter 6.

Within the L. V. guyanensis complex, each species was identified, sharing 50-60% divergence. Stocks in the L. V. braziliensis complex were more closely related with only 25 – 40% divergence. The close relationship of strains and stocks in this complex has been previously reported based on analyses of isoenzyme data (Cupolillo *et al.*, 1994), IEA and RFLP analysis of the gp63 gene locus (Victoir *et al.*, 1998) and RAPD markers (Dujardin *et al.*, 1995b). As found in the previous analysis, the L. V. peruviana reference strain LC39 nested within the L. V. braziliensis complex; the L. V. braziliensis reference strain LTB300 was more similar to L. V. peruviana than to many other L. V. braziliensis stocks.

4.6.2 Phylogenetic analysis of sequence data from microsatellite locus AC01

Molecular phylogenies for sequence data from the AC01 microsatellite locus were constructed using the neighbor-joining method using the Kimura 3-parameter model in PAUP* 4.0 beta version (Swofford, 1997). Figure 4.6.2.1 shows a phylogram of the neighbour-joining consensus tree for the Brazilian stocks and reference strains based on sequence data from the microsatellite locus AC01. The reliability of tree topologies was assessed by the bootstrap method (n = 1000). Only branches with greater than 50% bootstrap support are indicated.

The authors who developed microsatellite analysis for *Viannia* subgenus parasites (Russell *et al.*, 1999) acknowledge that alleles identified using single microsatellite loci did not correlate with designated species and that multilocus data was required to separate stocks into correct species groups. These authors did not attempt phylogenetic analysis, despite having sequencing data from three microsatellite loci.



NJ

---- 0.005 changes

Figure 4.6.2.1: unrooted phylogram of the neighbour-joining consensus tree for the Brazilian stocks and reference strains based on sequence data from the microsatellite locus AC01. The reliability of tree topology was assessed by the bootstrap method (n = 1000). Numbers indicate the percentage of bootstrap support for each branch. Values are given only for branches with more than 50% support (50% majority rule). The scale bar represents 0.005 (0.5%) nucleotide changes per sequence position. Numbers in parentheses indicate the number of dinucleotide repeats (n) scored (for L. V. guyanensis and L. V. shawi stocks the repeats are not perfect). Strains / stocks in clusters I and II belong to the L. V. braziliensis complex. Stocks / strains in cluster III belong, with the exception of LC1152, to the L. V. guyanensis complex: *LC1152 is an additional L. V. peruviana reference strain. M6426 is the L. V. lainsoni reference strain. here used as the outgroup (see text).

Chapter 4 - Brazil including Três Bracos and Corte de Pedra

It proved difficult to select an outgroup for this data because only *Viannia* strains could be amplified. The *L. V. lainsoni* reference strain M6426 was selected as the outgroup on the basis that this species is agreed to be distantly related to other *Viannia* species. Depicted as such (*figure 4.6.2.1*), the bifurcation occurring from M6426 should not be taken as indicative of the ancestral status of this species. A trifurcating layout with 3 major branches separating *L. V. lainsoni*, the *L. V. braziliensis* complex and the *L. V. guyanensis* complex would be more correct here.

Less diversity was evident among L. V. braziliensis stocks based on AC01 sequence data compared to IEA data: this is not surprising since the latter comprises multilocus data. Nevertheless, the phylogram highlights the close relationship between these L. V. braziliensis stocks, in particular those stocks from Três Braços / Corte de Pedra. Among L. V. braziliensis stocks, two clusters were evident (I and II), supported by 63% and 62% bootstrap values, respectively. All stocks from Três Braços / Corte de Pedra, including the reference strain LTB300, grouped in cluster II. Since all L. V. braziliensis stocks had n = 8, the clusters were generated from sequence diversity in the flanking regions. Stocks in clusters I and II could be separated only by a single base difference in the upstream flanking region. Cluster II contained stocks from 3 AC01 sequence groups separated, using visual examination of the data, by single base differences.

As with analyses of isoenzyme data, the L. V. peruviana reference strain LC39 grouped with L. V. braziliensis stocks in cluster II: this strain differed from the L. V. braziliensis reference strain LTB300 by having 2 additional dinucleotide repeats (data not shown): the relationship between L. V. braziliensis and L. V. peruviana will be discussed further in Chapter 6.

Stocks in cluster III, with the exception of LC1152, belonged to the L. V. guyanensis complex. These stocks were highly variable in AC01 sequence (see figure 4.5.4.1), both in the microsatellite repeat and downstream flanking region. It is therefore not surprising that these stocks were separated by relatively long branches and supported by high bootstrap values. It is not clear why the L. V. peruviana stock LC1152 groups with the L. V. guyanensis complex members. This stock does not have the upstream sequence changes shown by all guyanensis complex members, nor the diverse downstream flanking sequence shown by L. V. guyanensis and L. V. shawi stocks. In

Chapter 4 - Brazil including Tres Bracos and Corte de Pedra

fact, LC1152 was identical to the *L. V. peruviana* reference strain LC39 in both upstream and downstream flanking regions: sequence variation between these stocks was limited to LC1152 having 3 additional dinucleotide repeats (data not shown, see chapter 6). The grouping of this stock in the *guyanensis* complex therefore appears to be solely the result of this longer repeat region.

Ambiguous clustering could be due to an incorrect sequence alignment. Aligning these sequences was subject to interpretation owing to insertions and deletions in and around the dinucleotide repeat region. The upstream flanking region was easily aligned because it was generally highly conserved. The repeat region was identified by a "TG/ATC" motif at the start and by a "TCCGC" motif at the end (see *figure 4.5.4.1*). In 2 *L. V. shawi* stocks (M15065 and M12045), however, the start motif appeared to have been duplicated (bases 54-57 in *figure 4.5.4.1*) and these 4 bases were not counted as part of the repeat. It is possible however, that these bases did form part of the repeat (giving 2 additional dinucleotide repeats for these stocks) which may have led to a different tree topology. The position of LC1152 is more likely the result of limitations of the phylogenetic analysis used.

Phylogenetic reconstruction programs specifically written for the analysis of microsatellite data are still in their infancy. The difficulty in writing these programs stems from the lack of knowledge regarding the creation of microsatellite arrays. As discussed in chapter 1, although it is not possible to determine whether a particular microsatellite allele arose from a longer or shorter allele, alleles closely related in size are thought to share a more common recent ancestry than alleles of much different size (Di Rienzo *et al.*, 1994).

The analysis used here was the best available at the time of analysis (Michael Gaunt, pers. comm). The neighbor-joining (NJ) method constructed topologies from pairwise comparisons of 4 character states using the Kimura 3-parameter model. The latter distinguishes between transitions and transversions, splits transversion types and takes into account reverse mutation. In this analysis, pairwise comparisons were made between 4 character states only and although ambiguity codes were scored, gaps (which in microsatellite arrays are highly informative) were ignored.

Chapter 4 - Brazil including Três Bracos and Corte de Pedra

Several authors have suggested that, in the absence of knowledge regarding the creation of the microsatellite array, more accurate phylogenies may be derived from microsatellite data with the repeat region excluded (Ashley & Dow, 1994; Orti *et al.*, 1997). Hence phylogenetic analyses would take into account mutations in the flanking regions only. Only very recently however, a 5-state model has been developed which scores gaps as informative characters (MATRIX, David Posado, <u>http://bioag.byu.edu/zoology/crandall-lab/programs.htm.</u>), and which may prove applicable to the analysis of microsatellite sequences including the repeat region.

4.7 Discussion of diversity among stocks from Brazil.

Microsatellite analysis revealed greater diversity than IEA which is probably not surprising considering that these markers appear not to be functionally constrained (Ashley & Dow, 1994; Schlötterer, 1998). Among Brazilian stocks, Genotyper® analysis revealed 23 microdemes from 33 stocks compared to 10 zymodemes from 37 stocks using IEA. L. V. braziliensis stocks were separated into 18 microdemes and 6 AC01 sequence types (28 stocks scored) compared to 8 zymodemes (32 stocks scored). L. V. shawi stocks were separated into 4 microdemes and 3 AC01 sequence types compared to a single zymodeme (4 stocks scored by both methods). Notwithstanding the fact that the multilocus Genotyper® results from these microsatellite loci need to be verified, this technique was highly discriminatory and has great potential.

In agreement with previous reports (Cuba-Cuba *et al.*, 1985; Rosa *et al.*, 1988), stocks from Três Braços / Corte de Pedra were shown to be remarkably similar using IEA. Microsatellite analysis, using both Genotyper® and sequencing analyses, has revealed previously undetected diversity among these stocks: 6 microdemes and 6 AC01 sequence types were scored in these 18 stocks which also presented null alleles indicating further, undefined diversity. Many more stocks are available from this area: it would be very interesting to compare data from a larger population set from this region to see if any more evidence is forthcoming regarding *Leishmania* population structure. Isoenzyme-based studies from other localized areas in Brazil have also reported a lack of heterogeneity among *L. V. braziliensis* stocks: no heterogeneity was observed among 15 *L. V. braziliensis* stocks from Ceará State (NE Brazil, de Queiroz *et al.*, 1994) or among 13 stocks from São Paulo State (SE Brazil, Yoshida *et al.* 1990). The observations of the same zymodeme, microdeme and AC01 sequence type

Chapter 4 - Brazil including Tres Bracos and Corte de Pedra

occurring in different stocks, seen here among stocks from Tres Braços / Corte de Pedra, supports the proposed hypothesis of clonal propagation (Tibayrenc *et al.*, 1991).

L. V. braziliensis heterogeneity has, however, been widely reported from many Latin American countries (Kreutzer & Christensen, 1980; Evans et al., 1984; Saravia et al., 1985; Revollo et al., 1992; Felinto de Brito et al., 1993; Kreutzer, 1996; Cupolillo et al., 1994; 1995; Chouicha et al., 1997; Delgado et al., 1997; Lucas et al., 1998; Saravia et al., 1998). L. V. braziliensis stocks from a single area (Amariji, Pernambuco State), were shown to very heterogeneous, with 4 zymodemes, 4 microdemes and 3 AC01 sequence types scored from just 4 stocks. This region has an ongoing CL epidemic and it would be extremely interesting to examine more stocks by both IEA and microsatellite analysis. Isoenzyme and microdeme heterogeneity among L. V. braziliensis stocks was more common among stocks originating from different geographical areas. Heterogeneity may arise from mutational or recombinational processes, the latter of which may be asexual or sexual. The application of microsatellite analysis to larger numbers of localized Leishmania populations containing heterogeneous stocks will likely provide further evidence regarding the mechanisms(s) generating this diversity and may identify alleles which are associated with geographical origin, clinical presentation etc.

CHAPTER 5 - NICARAGUA: SYMPATRIC TRANSMISSION OF L. V. BRAZILIENSIS AND L. V. PANAMENSIS

5.1 Introduction

Nicaragua, the largest country in Central America, is bordered between Honduras (north) and Costa Rica (south) and between the Pacific Ocean and Caribbean Sea on the east and west respectively. Half of Nicaragua is covered by forests, lakes and rivers. The low, fertile Pacific coast region, which includes most of the country's major cities, produces cotton, cattle, sugarcane, and basic food crops (rice, corn, and beans). Coffee is grown in the cooler central highlands bordering Honduras. The forested Caribbean (Atlantic) lowlands are sparsely populated.

CL and MCL are endemic in Nicaragua, notably in northern, central and eastern regions (Darce *et al.*, 1991). Between 1980 and 1987, more than 9500 cases of leishmaniasis were notified (Desjeux, 1992b). Visceral disease has not been reported from Nicaragua but is thought likely to occur in northern regions along the border with Honduras (WHO, 1990). L. V. braziliensis and L. V. panamensis occur sympatrically in the mountainous regions of northern Nicaragua, whereas L. V. panamensis is the major species present in the Atlantic rainforest regions of southern Nicaragua (Belli *et al.*, 1994). Putative L. V. braziliensis / L. V. panamensis hybrids have also been reported from Nicaragua (Missoni *et al.*, 1986; Darce *et al.*, 1991; Belli *et al.*, 1994), all from northern foci (Belli *et al.*, 1994).

5.2 Nicaraguan Leishmania stocks

Stocks representative of the observed isoenzyme and RFLP patterns from the Belli study (Belli *et al.*, 1994) were selected for further analysis: these are listed in *table 5.2*. *Figure 5.2* shows a map of Nicaragua indicating the geographical distribution of stocks characterized.

These stocks, including putative L. V. braziliensis and L. V. panamensis parental and L. V. braziliensis /L. V. panamensis hybrid stocks, have been previously characterized by IEA, serodeme and RFLP analysis (Missoni et al., 1986; Darce et al., 1991; Belli et al., 1994); several have recently been further analyzed using RAPD analysis (Noyes et al., 1996).

Table 5.2:

Leishmania stocks from Nicaragua.

ISOLATE	HOST	DEPARTMENT	LEISHMANIA	LESION
CODE NO.	SPECIES	/ AREA	SPECIES Wige	TYPE CI
	Fiuman	JIIIOTOga / BOCay	L. V. Drazmensis	
MHOM/NI/88/XD09	Human	Jinotega / Bocay	Lb (Lpa hybrid)	CL
MHOM/NI/90/HN196	Human	Jinotega / Bocay	L. V. braziliensis§	MCL
MHOM/NI/88/XD05	Human	Jinotega / Bocay	L. V. panamensis†	CL
MHOM/NI/88/XD42	Human	Jinotega / Pantasma	Lb /Lpa hybridt	CL
MHOM/NI/88/XD36	Human	Jinotega / Pantasma	Lb /Lpa hybrid†	CL
MHOM/NI/92/WSL03	Human	Zelaya Norte / Guayabo	Lb /Lpa hybrid§	CL
MHOM/NI/88/XD13	Human	Zelaya Norte / Wamblan	Lb /Lpa hybrid†	CL
MHOM/NI/89/XD19	Human	Zelaya Norte / Leymus	L. V. braziliensis§	MCL
MHOM/NI/89/XD20	Human	Zelaya Norte / Waslala	Lb /Lpa hybrid§	CL
MHOM/NI/92/WSL06	Human	Zelaya Norte / Wasiala	Lb /Lpa hybrid§	CL
MHOM/NI/84/NICA2	Human	Zelaya Norte / Waslala	Lb /Lpa hybrid±	CL
MHOM/NI/87/ZE09	Human	Zelaya Sur/ Nva Guinea	L. V. panamensis†	CL
MHOM/NI/89/XD17	Human	Ocotal / Murra	L. V. braziliensis†	CL
MHOM/NI/89/AZ01	Human	Ocotal / Murra	L. V. braziliensis§	MCL
MHOM/NI/88/XD39	Human	Matagalpa / Paiwas	L. V. panamensis§	MCL
MHOM/NI/89/XD41	Human	Matagalpa / Rio Blanco	Lb /Lpa hybrid§	CL
MHOM/NI/91/ZF01	Human	Matagalpa / El Coyolar	L. V. panamensis§	CL
MHOM/NI/88/XD45	Human	Chontales / Los Chiles	L. V. panamensis†	CL
MHOM/NI/89/XD03	Human	Rio Sn Juan / Azucenas	L. V. panamensis§	CL

Table 5.2: Leishmania stocks from Nicaragua. ⁶⁰- Lb /Lpa hybrid indicates a putative L. V. braziliensis / L. V. panamensis hybrid. All stocks were obtained from the cryobank at LSHTM. § - stocks examined by Belli et al. (1994); † - stocks examined by Darce et al. (1991); \pm - stocks examined by Missoni et al. (1986). [•] Lesion type: CL, cutaneous lesion; MCL, mucocutaneous lesion. Note comments on page173-4: 2 stocks previously identified as putative hybrids (MHOM/NI/92/WSL03 and MHOM/NI/92/ WSL06) were identified here as L. V. panamensis; 1 stock previously identified as L. V. panamensis (MHOM/NI/88/XD05) was identified here as a putative hybrid (possible mis-labeling in LSHTM cryobank ?).



Figure 5.2: Map of Nicaragua showing the geographical distribution of stocks.

Figure 5.2: Map of Nicaragua showing the geographical distribution of stocks examined (from Belli et al., 1994). (\bullet) - L. V. braziliensis; (\circ) - L. V. panamensis; (\bullet) - putative L. V. braziliensis / L. V. panamensis hybrid.

5.3 Isoenzyme analysis

L. V. panamensis stocks could be differentiated from L. V. braziliensis stocks using 8 enzyme loci: MPI, NHi1, NHi2, 6PGD, PGM, PEPD, ASAT and ALAT. Examples of isoenzyme banding patterns observed in the stocks from Nicaragua are shown in *figure 5.3.1.1* for the enzymes MPI (monomeric), 6PGD (dimeric) and NHi 1/2 (1 monomeric; 2 tetrameric). Banding patterns observed in the Nicaraguan stocks are shown diagramatically alongside reference strain patterns, in *figure 5.3.1.2*, for all enzymes studied.

In agreement with previous work involving these Nicaraguan stocks (Darce *et al.*, 1991; Belli *et al.*, 1994), three distinct groups were identified: among the 20 stocks studied here, 5 were identified as *L. V. braziliensis*, 7 as *L. V. panamensis* and 8 as putative *L. V. braziliensis / L. V. panamensis* hybrids. However, three discrepancies were noted. Two stocks (WSL03 and WSL06) reported as hybrids by Belli *et al.*

(1994) were characterized as *L. V. panamensis*: in their report, the IEA pattern shown photographically for one of these stocks (WSL03) at one of the loci showing hybrid phenotypes (6PGD) is not conclusive. Another stock (XD05) reported as *L. V. panamensis* by Belli *et al.* (1994) was characterized here as a hybrid. The XD05 stock used here (two separate stocks with the same name but different year of isolation are listed in Belli's report) was from Bocay / Jinotega and was also studied by Darce *et al.* (1991). These authors reported, although not giving results by individual stock names, that stocks examined from this area were either *L. V. braziliensis* or hybrids.









174

Figure 5.3.1.1: Isoenzyme banding patterns of Nicaraguan stocks (continued)

NHi 1/2



Lane: 1 2 3 4 5 6 7 8 9 10 11 12

6PGD



Lane 1 2 3 4 5 6 7 8 9 10 11 12

Figure 5.3.1.1: Isoenzyme variation in stocks from Nicaragua using NHi and 6PGD. Lane 1 - L. V.braziliensis reference strain LTB300; lane 2 - L. V. peruviana reference strain LC39; lane 3 - L. V.guyanensis reference strain M4147; lane 4 - L. V. panamensis reference strain LS94; lane 5 - L. V.lainsoni reference strain M6426; lane 6 - L. V. sp. n. reference strain IM2832; lane 7 - L. L. chagasi reference strain M975; lane 8 - L. L. mexicana reference strain IM2832; lane 9 - L. L. amazonensis reference strain M2269; lane 10 - XD28; lane 11 - XD09; lane 12 - ZF01.







176



177

Chapter 5 - Nicaragua





Figure 5.3.1.2:Diagrammatic representation of IEA banding patternsobserved in reference strains and Nicaraguan stocks (c'td).

5.3.1 Zymodemes

Only three zymodemes were scored from the Nicaraguan stocks using these eleven enzymes. A single zymodeme was scored for both L. V. braziliensis (LON200) and L. V. panamensis (LON202) stocks: these zymodemes are identical to the reference strains for L. V. braziliensis (LTB300) and L. V. panamensis (LS94) respectively. The zymodeme pattern of the L. V. braziliensis / L. V. panamensis hybrid (LON 216) appears to be an exact juxtaposition of the two parental type zymodemes, with one allele from each parental type evident at each variable enzyme locus. Table 5.3.1.1 summarizes the IEA allele and zymodeme designations for the Nicaraguan stocks.

	Table 5.3.1.1:	Allele and z	ymodeme d	esignations :	for Nic	araguan stoc
--	----------------	--------------	-----------	---------------	---------	--------------

	ENZYME												
STOCK CODE NO.	MPI	NHi 1	NHi 2	NHd	ES	PEPD	PGM	6PGD	GPI	G6PD	ASAT	ALAT	Z*
XD28, XD17, XD19 HN196, AZ01	6/6	2/2	7/7	4/4	4/4	2/2	3/3	5/5	4/4	2/2	5/5	4/4	LON200
XD42, XD05, XD09 XD13, XD20, XD36 XD41, NICA2	6/8	2/3	5/7	4/4	4/4	2/7	3/4	5/7	4/4	2/2	2/5	3/4	LON216
ZF01, ZE09, XD03 XD39, XD45, WSL03 WSL06	8/8	3/3	5/5	4/4	4/4	7/7	4/4	7/7	4/4	2/2	2/2	3/3	LON202

Table 5.3.1.1: allele and zymodeme designations for Nicaraguan stocks. Z = zymodeme; * LON numbers above 200 have been reassigned by the author as previous records have been lost (zymodemes are assigned based on the IEA conditions used, i.e. for *Leishmania Viannia* species). For full enzyme names, see chapter 2. See *figure 3.2.2* for allele scoring scheme. Data given assumes diploidy, see introduction.

Nicaraguan stock zymodemes are summarized by geographical origin and clinical presentation in *table 5.3.1.2.* Stocks within each group were shown to be identical using IEA, irrespective of geographical origin. Four stocks were isolated from MCL lesions: 3 of these stocks were identified as *L. V. braziliensis* and 1 as *L. V. panamensis.* In agreement with Darce *et al.* (1991), the hybrid isoenzyme genotype was not found to be associated with MCL (0/4 stocks examined).

Table 5.3.1.2 Summary of Nicaraguan stock zymodemes, by geographical origin and clinical presentation.

STOCK	Z*	GEOGRAPH	ICAL ORIGIN	ISOLATED	CLINICAL
CODE NO.		AREA	DEPARTMENT	FROM:	TYPE
XD28	LON200	Bocay	Jinotega	Human	CL
XD09	LON216	Bocay	Jinotega	Human	CL
HN196	LON200	Bocay	Jinotega	Human	MCL
XD05	LON216	Bocay	Jinotega	Human	CL
XD42	LON216	Pantasma	Jinotega	Human	CL
XD36	LON216	Pantasma	Jinotega	Human	CL
WSL03	LON202	Guayabo	Zelaya Norte	Human	CL
XD13	LON216	Wamblan	Zelaya Norte	Human	CL
XD19	LON200	Leymus	Zelaya Norte	Human	MCL
XD20	LON216	Waslala	Zelaya Norte	Human	CL
WSL06	LON202	Waslala	Zelaya Norte	Human	CL
NICA2	LON216	Wasiala	Zelaya Norte	Human	CL
ZE09	LON202	Nova Guinea	Zelaya Sur	Human	CL
XD17	LON200	Мигта	Ocotal	Human	CL
AZ01	LON200	Мигта	Ocotal	Human	MCL
XD39	LON202	Paiwas	Matagalpa	Human	MCL
XD41	LON216	Rio Blanco	Matagalpa	Human	CL
ZF01	LON202	El Coyolar	Matagalpa	Human	CL
XD45	LON202	Los Chiles	Chontales	Human	CL
XD03	LON202	Azucenas	Rio San Juan	Human	CL

Table 5.3.1.2: summary of Nicaraguan stock zymodemes, by geographical origin and clinical presentation. Z = zymodeme (assigned based on the IEA conditions used, i.e. for Leishmania Viannia species): LON200 - L. V. braziliensis; LON202 - L. V. panamensis; LON216 - L. V. braziliensis / L. V. panamensis hybrids. For full enzyme names, see chapter 2. See figure 3.2.2 for allele scoring scheme. Data given assumes diploidy, see introduction. Note that WSL03 and WSL06 have been characterized as L. V. panamensis: and XD05 as a hybrid, in contrast to Belli *et al.* (1994): see text.

5.4 RAPD analysis

Notwithstanding the fact that this technique was discontinued, for reasons discussed in chapter 3, several observations regarding stocks from Nicaragua deserve comment. Three stocks, one from each of the three zymodemes observed among the Nicaraguan sample, were examined using RAPD. *Figure 5.4.1* shows RAPD profiles generated from reference strains and these three Nicaraguan stocks (XD28, *L. V. braziliensis*; ZF01, *L. V. panamensis*; XD09, putative hybrid), using 3 primers (A3, L1 and H2).

Fourteen RAPD primers generated profiles which discriminated between the reference strains of L. V. braziliensis and L. V. panamensis. Stocks identified as L. V. braziliensis or L. V. panamensis from Nicaragua, although generating patterns which were similar to their counterpart reference strains, often differed by the presence and/or absence of minor bands (e.g. arrow 3, figure 5.4.1). This observation was not unexpected since the reference strains originated from geographically distant locations (Brazil and Panama respectively). Eight of the 14 primers found to separate L. V. braziliensis and L. V. panamensis also generated profiles for the putative hybrid stock which were heterozygous between those of the parental type profiles, i.e. contained bands "inherited" from both (e.g. arrows 1 and 2, figure 5.4.1). Two hybrid profiles (D1 and H2) included bands which were not present in either putative parent (e.g. arrow 5, figure 5.4.1). This band is, however, present in the profile of the L. V. braziliensis reference strain. The origin of such bands, as with all RAPD bands, is unknown but not unexpected since the putative parental types are unlikely to be the actual parents of the putative hybrid.

These observations illustrate the potential usefulness of this technique for inferring relationships between organisms in the absence of information from other sources, as discussed in chapters 1 and 3.

Notes to figure 5.4.1. (see next page) RAPD profiles generated from parental and hybrid type Nicaraguan stocks using primers A3, L1 and H2, visualized after electrophoresis through 1.2% agarose gel and ethidium bromide staining. Lane 1 - L. V. braziliensis reference strain, LTB300, lane 2 - L. V. panamensis reference strain, LS94. Lanes 3 - 5, Nicaraguan stocks: lane 3 - XD28 (L. V. braziliensis); lane 4 - XD09 (putative hybrid L. V. braziliensis / L. V. panamensis); lane 5 - ZF01 (L. V. panamensis). Lane 6 - negative control, lane 7 - 1kb molecular weight (mw) marker (Gibco BRL, UK). Stocks in lanes 4 and 5 are reversed in primer L1 picture. Arrows (numbered 1 - 5) illustrate the following: arrow 1 - band from L. V. braziliensis stock from Nicaragua but not in the reference strain, arrow 4 - band found in all subgenus Viannia stocks; arrow 5 - band found in putative hybrid but not present in Nicaraguan but not present in Nicaraguan parental strains (present in L. V. braziliensis reference strain).









5.5 PCR-RFLP analysis of rRNA gene ITS region

PCR-RFLP analysis of the rRNA gene ITS region was used to examine all stocks from the Nicaraguan sample. L. V. braziliensis and L. V. panamensis stocks could be distinguished by RFLP analysis using the enzymes BstUI, ScrFI, Fspl, TaqI and SphI. Hybrid stocks were shown, using each of these enzymes, to generate fragments of the same apparent size as the L. V. panamensis stocks. Figure 5.5.1 shows the RFLPs of Nicaraguan stocks from digestion using BstUI. Using several enzymes, RFLP profiles of hybrid stocks contained multiple heteroduplex bands which were not seen in L. V. braziliensis or L. V. panamensis stocks (arrow* figure 5.5.1).

Figure 5.5.1: RFLP analysis of rRNA ITS amplification products from Nicaraguan stocks using BstUL.



Figure 5.5.1: RFLP analysis of ribosomai RNA (rRNA) internal transcribed spacer (ITS) region amplification products from Nicaraguan stocks, visualized after electrophoresis through 6% polyacrylamide gel and silver staining. Lanes 1 & 17: 1kb molecular weight (mw) marker (Gibco BRL). L. V. braziliensis stocks: lanes 5, 7, 8, 16 (XD19, XD28, XD28, HN196). L. V. panamensis stocks: lanes 10, 12, 13, 14, 15 (XD39, WSL03, WSL06, ZE09, ZF01). Hybrid stocks: lanes 2, 3, 4, 6, 9, 11 (XD05, XD09, XD13, XD20, XD36, XD42). L. V. braziliensis and L. V. panamensis stocks can be differentiated by the smaller size of the upper fragment at approx. 200bp in L. V. braziliensis stocks. Hybrid stocks appear identical to the L. V. panamensis stocks except for the presence of 2 heteroduplex bands (arrow *) at approx. 340/360bp. Intra-specific variation can be seen for L. V. braziliensis stock XD28 (lanes 7 & 8, additional fragments at approx. 220bp) and for the hybrid XD42 (lane 11, doublet upper band at approx. 200bp)

Chapter 5 - Nicaragua

Diversity between stocks identified, using IEA, as the same type (species or hybrid) was rare, but is illustrated here for two stocks, XD28 (L. V. braziliensis) and XD42 (hybrid), using BstUI (lanes 7/8 and 11, figure 5.5.1). Greater diversity may have been undetected since the PAGE system used did not resolve fragments with small differences in size which made scoring difficult. PCR-RFLP analysis is, like microsatellite analysis, amenable to multiplexing. PCR products from different loci labeled with different fluorophores could be pooled for RFLP analysis. To ensure that all RFLP fragments were labeled, the PCR reaction mix would have to include labeled dNTPs (as opposed to the labeled primers used for microsatellite analysis). Fragments from different loci could be identified from the myriad of fragments generated by using a different label for each locus. Multiplexed RFLP products could thus be analyzed using the Genescan® and Genotyper® approach as described previously. PCR-RFLP analyses of Viannia parasites based on the gene loci and ITS regions of the gp63 and rRNA arrays have recently been reported (Cupolillo et al., 1995; Victoir et al., 1998). Other gene families such as the mini-exon array (Hassan et al., 1993; Fernandes et al., 1994) have also been characterized in Viannia species and could be included in this approach: combined results from several such targets will provide multilocus data for population and/or phylogenetic analyses.

5.6 <u>Microsatellite analysis</u>

5.6.1 PCR of microsatellite loci

As observed among the Brazilian stocks, several Nicaraguan stocks had null alleles at one or more of the microsatellite loci. Two stocks (XD17 and XD41) failed to amplify any of the 3 loci and one stock (XD45) failed to amplify AC01. These stocks originated from different geographical areas. PCR products generated from Nicaraguan stocks for all loci were within the expected size range.

5.6.2 Non-denaturing PAGE analysis

5.6.2.1 <u>AC01</u>

Figure 5.6.2.1 shows AC01 products from Nicaraguan stocks analyzed using nondenaturing PAGE and silver staining.

As for Brazilian stocks, 2 intensely-staining bands were observed for Nicaraguan stocks at this locus. Using PAGE, products from L. V. braziliensis and L. V. panamensis stocks could be separated by size (e.g. lanes 2 and 3, respectively, figure 5.6.2.1), with the exception of the L. V. braziliensis stock AZ01, the product of which was similar to L. V. panamensis stocks (data not shown, discussed in sections 5.6.3.1 and 5.7). Products of two different sizes were evident among the L. V. panamensis stocks (compare product sizes in lanes 3-5 with those in 6 & 8, figure 5.6.2.1). The putative hybrid stocks generated profiles that were an exact combination of those from one of the L. V. panamensis types (smaller allele) and L. V. braziliensis. In these hybrid profiles, the products correlating to L. V. panamensis were weak by comparison with those correlating to L. V. braziliensis. It is not clear why this should be so: it is possible that there is a mutation in the primer binding site of this allele in hybrid stocks such that the L. V. panamensis allele is amplified with less efficiency. Another possibility is that microsatellite alleles containing significantly different numbers of repeats may be amplified with different efficiencies (alleles with smaller numbers of repeats would be expected to be amplified more readily than alleles with large numbers of repeats). PCR amplification of a locus at which the alleles did vary significantly in repeat number would be biased towards the allele which was amplified more readily. This allele would dominate the reaction resulting in the generation of substantially more product.



Figure 5.6.2.1: AC01 products from Nicaraguan stocks analyzed using non-denaturing 6% PAGE and silver staining. Lane 1 - L. V. braziliensis reference strain, LTB300; lane 2 - HN196; lane 3 - WSL06; lane 4 - WSL03, lane 5 - ZF01; lane 6 - ZE09; lane 7 - XD42; lane 8 - XD39; lane 9 - XD36; lane 10 - XD28; lane 11 - XD20; lane 12 - XD19; lane 13 - XD13; lane 14 - XD09; lane 15 - XD05; lane 16 - 100 bp molecular weight marker (Gibco BRL, UK). PCR products from L. V. braziliensis stocks are in lanes 2, 10 and 12; L. V. panamensis stocks are in lanes 3 - 6 & 8, hybrids stocks are in lanes 7, 9, 11 & 13 - 15. The allelic band is the lowest intensely-staining band in each lane. Note the additional band (at set distance from the allelic band in each lane) and heteroduplex bands (indicated): see text for discussion.

Heteroduplex bands were also observed in Nicaraguan stocks: as with Brazilian stocks, Nicaraguan stocks designated as both homozygous and heterozygous for AC01 generated these molecules (designated as such using Genotyper® data, section 5.6.3). Heteroduplex bands are particularly evident in profiles from hybrid stocks, wherein 4 heteroduplex bands are present. Diploidy is a working hypothesis for *Leishmania*: whilst four heteroduplex bands could be generated by the reannealing of heterozygote allelic strands with products resulting from slippage or incorrect base insertion during PCR amplification, aneuploidy cannot be ruled out. Ploidy will be discussed further in chapter 7. Additional bands present in AC01 profiles have been discussed previously. AC01 PAGE patterns for these stocks correlate with, and are further discussed under, Genotyper® results (section 5.6.3.1).

Chapter 5 Nicaragua

5.6.2.2 AC16

Figure 5.6.2.2 shows the AC16 products from Nicaraguan stocks. All Nicaraguan stocks exhibited heteroduplex molecules at approx. 350bp. L. V. panamensis stocks could be grouped by the presence of three additional heteroduplex bands at approx. 370 - 400bp. Genotyper® results score the majority of these stocks as homozygous at this locus so the origin of these heteroduplex molecules is not clear. The L. V. panamensis stock XD45 did not generate these heteroduplex molecules but had an additional band, the origin of which was not determined from Genotyper® results (see section 5.6.3). Comparisons between XD45 and other L. V. panamensis stocks at the other two loci was not possible since XD45 had null alleles for both. L. V. braziliensis and hybrid stock profiles appeared similar: slight mobility differences were observed between the allelic bands of different stocks (e.g. between lanes 7 and 8) in these two groups. The overall profiles generated with Nicaraguan stocks at the AC16 locus correlate well with Genotyper® results.





Figure 5.6.2.2: AC16 products from Nicaraguan stocks analyzed using non-denaturing 6% PAGE and silver staining. Lane 1 - L. V. braziliensis reference strain, LTB300; lane 2 - HN196; lane 3 - WSLO6; lane 4 - WSL03; lane 5 - ZF01; lane 6 - ZE09; lane 7 - XD45; lane 8 - XD42; lane 9 - XD39; lane 10 - XD36; lane 11 - XD28; lane 12 - XD20; lane 13 - XD19, lane 14 - XD13; lane 15 - XD09; lane 16 - XD05; lane 17 - 100 bp molecular weight marker (Gibco BRL, UK). L. V. braziliensis stocks lanes 2, 11 and 13; L. V. panamensis stocks lanes 3 - 7 & 9, hybrids stocks lanes 8, 10, 12 & 14 - 16. Note the additional band (at set distance from the allelic band in each lane) and heteroduplex bands (indicated): see text for discussion.

5.6.2.3 <u>AC52</u>

Figure 5.6.2.3 shows AC52 products from Nicaraguan stocks. Allele size differences were clearly evident at this locus, but as with all PAGE analyses, these could not be quantified with any precision: stocks were grouped, as previously, by comparison of overall profiles. Two distinct allelic band sizes were observed in *L. V. braziliensis* stocks (lanes 5, 7 & 15): XD28 and HN196 (lanes 7 & 15) bands were larger than those of XD19 (lane 5). Products from XD19 appeared to be the same size as *L. V. panamensis* stocks (lanes 9, 11 - 14). *L. V. panamensis* products were generally less intense than those of *L. V. braziliensis*. Most *L. V. braziliensis* and hybrid stocks had prominent heteroduplex molecules which were not present in *L. V. panamensis* stocks (see lanes 12 - 14). In the profiles of several hybrid stocks, faint bands corresponding to allelic *L. V. panamensis* (or XD19) bands were visible (lanes 6, 8 & 10). These results generally concur with those of Genotyper® and are discussed further in section 5.6.3.



Figure 5.6.2.3 AC52 products from Nicaraguan stocks (PAGE)

Figure 5.6.2.3 AC52 products from Nicaraguan stocks analyzed using non-denaturing 6% PAGE and silver staining. Lane 1 - 100bp molecular weight marker (Gibco BRL, UK); lane 2 - XD05; lane 3 - XD09; lane 4 - XD13; lane 5 - XD19; lane 6 - XD20; lane 7 - XD28; lane 8 - XD36; lane 9 - XD39; lane 10 - XD42; lane 11 - ZE09; lane 12 - ZF01; lane 13 - WSL03; lane 14 - WSL06; lane 15 - HN196. L. V. braziliensis stocks lanes 5, 7 & 15; L. V paramensis stocks lanes 9, 11 - 14; hybrid stocks lanes 2 - 4, 6, 8 & 10. Note the additional bands (at set distances from the allelic band(s) in each lane) and heteroduplex bands (indicated): see text for discussion.

5.6.3 Genescan@ and Genotyper@ 2.0 analyses

Table 5.6.3 (page 192) summarizes the allele sizes scored, using Genotyper®, at each microsatellite locus in stocks from Nicaragua, listed by species type, geographical origin and by clinical presentation (all stocks were isolated from humans). Results are discussed for each microsatellite locus and by multilocus genotype.

5.6.3.1 <u>AC01</u>

In agreement with PAGE results, 2 different AC01 allele sizes were observed in Nicaraguan L. V. panamensis stocks: 241 (4 stocks, genotype 241/241) and 245 (2 stocks, genotype 245/245). The size of the AC01 allele observed in all Nicaraguan L. V. braziliensis stocks bar one was 227 (3 stocks, genotype 227/227): this is the same genotype scored amongst the majority of Brazilian L. V. braziliensis stocks. Also in agreement with PAGE results, one L. V. braziliensis stock, AZ01, generated a peak of 245 which is associated here with L. V. panamensis stocks. This stock will be discussed further in section 5.7.

Two peaks (227 and 241) were observed in putative hybrid stocks: these corresponded to alleles from *L. V. braziliensis* and *L. V. panamensis*, respectively, providing further evidence of the hybrid status of these stocks. In agreement with observations from PAGE analysis, the *L. V. panamensis* allele was much less intense than that of *L. V. braziliensis*: possible reasons for this difference have been discussed previously.

A single hybrid stock appeared homozygous, genotype 227/227: it is possible that the *L. V. panamensis* allele in this stock was so weakly amplified that it was not detected; it is also possible that this stock was indeed homozygous at this locus and did not have the 241 allele. *Figure 5.6.3.1* illustrates the Genotyper® peaks obtained for each of the genotypes 227/227 (*L. V. braziliensis* stock XD19), 241/241 (*L. V. panamensis* stock WSL03) and 227/241 (hybrid stock XD13).

There was no apparent association between geographical origin and AC01 genotype. Stocks isolated from MCL lesions (3 *L. V. braziliensis*, 1 *L. V. panamensis*) were genotype 227/227 (as in Brazil) or 245/245 (2 stocks including AZ01). The 241 and 245 alleles are not exclusive to *L. V. panamensis*: 241 was also observed in the *L. V. shawi* reference strain (genotype 241/241) and 245 was observed in the *L. V.* sp. n. stock IM2832 (genotype 227/245).

Chapter 5 - Nicaragua

AC01 Genotyper® results for these stocks can be explained by variation of the number of dinucleotide repeats as revealed by sequencing (see section 5.6.4).

2

Figure 5.6.3.1

14

Genotyper® output from the AC01 locus for selected Nicaraguan stocks, illustrating the peaks observed in putative parental and hybrid profiles.



Figure 5.6.3.1: Genotyper Φ output from the AC01 locus for selected Nicaraguan stocks, illustrating the peaks observed in putative parental and hybrid profiles. Numbers across the top of the figure represent peak size, in bases. The exact sizes of individual peaks were scored automatically by the software. The scale to the right of each trace indicates the peak intensity. See chapter 3 for discussion of allele scoring. Top - hybrid stock XD13 (genotype 227/241); middle - L. V. braziliensis stock XD19; bottom - L. V. panamensis stock WSL03 (genotype 242/241). Note the lower intensity of the 241 allele in the hybrid stock: see text for discussion.

			Microsatellite locus							
Stock				AC01 AC16			16	AC52		
code no.	Species	Area / Department	Lesion	ailele	allele	allele	allele	allele	allele	
•†	∞†§±		type*	1 a	2 ^b	1	2	1	2	
LTB300	Ь	Brazil	MCL	227	227	244	244	283	283	
XD28	bţ	Jinotega / Bocay	CL	227	227	250	252	313	313	
HN196	Ъ§	Jinotega / Bocay	MCL	227	227	250	250	307	307	
XD19	bş	Zelaya Norte / Leymus	MCL	227	227	254	254	297	299	
XD17	ьт	Ocotal / Murra	CL	x	x	x	x	x	x	
AZ01	bş	Ocotal / Murra	MCL_	245	245	250	252	291	291	
XD09	b/pa†	Jinotega / Bocay	CL	227	241	250	250	291	309	
XD05 ²	b/pa†	Jinotega / Bocay	CL	227	241	250	250	291	309	
XD42	b/pa†	Jinotega / Pantasma	CL	227	241	250	250	291	313	
XD36	b/pat	Jinotega / Pantasma	CL	227	241	250	250	291	307	
XD13	b/pa†	Zelaya Norte / Wamblan	CL	227	241	250	250	291	309	
XD20	b/pa§	Zelaya Norte / Waslala	CL	227	241	250	250	291	309	
NICA2	b/pa±	Zelaya Norte / Wasiala	CL	227	227	250	250	291	309	
XD41	b/pa§	Matagalpa / Rio Blanco	CL	x	x	x	x	x	x	
WSL031	pa§	Zelaya Norte / Guayabo	CL	241	241	244	244	291	291	
WSL06 ¹	pa§	Zelaya Norte / Wasiala	CL	241	241	244	244	299	299	
ZE09	pa†	Zelaya Sur/ Nva Guinea	CL	245	245	244	244	291	291	
XD39	pa§	Matagalpa / Paiwas	MCL	245	245	244	244	291	291	
ZF01	paş	Matagalpa / El Coyolar	CL	241	241	244	244	295	295	
XD45	pa†	Chontales / Los Chiles	CL	x	x	250	252	F	F	
XD03	paş	Rio Sn Juan / Azucenas	CL	227	241	250	250	291	309	
LS94	pa	Panama	CL	239	239	244	244	299	299	

Table 5.6.3Microsatellite allele sizes of stocks from Nicaragua by species type,
geographical origin and clinical presentation.

Table 5.6.3: Summary of microsatellite allele sizes of stocks from Nicaragua by species type, geographical area and by clinical type. Code names in **bold** type indicate reference strains which, although not from Nicaragua, are included for comparison. All stocks were of human origin. ^{co}Species: b - braziliensis; pa - paramensis; b/pa - putative braziliensis/panamensis hybrid; ^lWSL03 and WSL06 were identified using IEA as *L. V. panamensis*, not putative braziliensis/panamensis hybrids as reported by Belli *et al.* (1994). ²XD05 (MHOM/NI/88/XD05) was identified using IEA as putative braziliensis/panamensis hybrid, not *L. V. panamensis* as reported by Belli *et al.* (1994). [§] - stocks examined by Darce *et al.* (1991); [‡] - stocks examined by Missoni *et al.* (1986). ^{*} Lesion type: CL, cutaneous lesion; MCL, mucocutaneous lesion. Assignment and scoring of alleles are explained in chapter 3. X indicates that no PCR product was generated. F - failed to generate scorable Genotyper@ peak.

5.6.3.2 <u>AC16</u>

Three genotypes were scored amongst the 4 Nicaraguan L. V. braziliensis stocks: 250/250; 250/252 and 254/254. All hybrid stocks were genotype 250/250. The genotype 244/244 was the most common genotype observed in Nicaraguan L. V. panamensis stocks. Two other genotypes were scored in L. V. panamensis stocks, 250/250 and 250/252: these were also seen in L. V. braziliensis stocks. AC16 genotype 244/244, here associated only with L. V. panamensis stocks, was scored in many L. V. braziliensis stocks from Brazil. One L. V. panamensis stock, XD45, presented an additional band using PAGE (see figure 5.6.2.2): this stock was scored as 250/252 using Genotyper®. Other stocks also genotyped as 250/252 did not have this PAGE band, the origin of which is thus not clear. Four genotypes were observed among stocks isolated from MCL lesions: 250/250, 250/252, 254/254 and 244/244 (as in Brazilian L. V. braziliensis stocks, here a L. V. panamensis stock).

5.6.3.3 <u>AC52</u>

Many AC52 genotypes were scored among Nicaraguan stocks. L. V. braziliensis stocks presented 4 genotypes: three were homozygous with allele sizes of 291^* , 307 and 313; a fourth stock was heterozygous with alleles of 297 and 299 (this AC52 genotype was also scored in the L. V. shawi reference strain M15065. Four genotypes were scored from 5 L. V. panamensis stocks: four were homozygous with allele sizes of 291 (2 stocks), 295 and 299; the fifth stock was heterozygous with alleles of 291 and 309. In agreement with AC01 results, one L. V. braziliensis stock, AZ01*, had a genotype which was found in L. V. panamensis stocks.

Three genotypes, all heterozygous, were scored amongst the hybrid stocks: 291/307 (1 stock), 291/309 (5 stocks) and 291/313 (1 stock). These AC52 genotypes, as do AC01 genotypes, support the status of these stocks as hybrids between *L. V. braziliensis* and *L. V. panamensis*. As observed in PAGE analysis at the AC01 locus, peaks equivalent to *L. V. panamensis* alleles in these hybrid stocks were weaker than peaks equivalent to alleles of *L. V. braziliensis*. For this locus however, the hypothesis that alleles with longer repeat regions have lower amplification efficiencies (section 5.6.2.1) is not applicable because it is the *L. V. braziliensis* allele which contains the larger number of dinucleotide repeats. It is more likely here that there is a mutation in the primer binding site at this locus in alleles of *L. V. panamensis* stocks: this could also explain

why the products of L. V. panamensis stocks were generally less intense than those of L. V. braziliensis. (see figure 5.6.2.3).

Many Brazilian L. V. braziliensis stocks were homozygous for the AC52 allele 283: the alleles observed in several Nicaraguan L. V. braziliensis stocks are considerably larger. This is in agreement with Russell *et al.* (1999) who, from sequence data, scored AC52 repeat numbers in Nicaraguan and Brazilian L. V. braziliensis (2 and 7 examined) stocks as 21/23 and 9 (& rarely 15 and 16) respectively. These differences would, assuming size differences equate to changes in the number of repeats, result in allele sizes of 307 (283 + 24) and 311 (283 + 28), sizes in the range observed from Nicaraguan stocks (one exact match). These size differences illustrate an association between geographical origin and allele size. The correlation observed between Genotyper[®] (here) and sequencing data (Russell *et al.*, 1999) indicates that the Genotyper[®] technique is generally suitable for the analysis of microsatellite loci.

5.6.3.4 Multilocus analysis of microsatellite data

Microsatellite genotype data is compared with species type, as determined using IEA, in *table 5.6.3.4*. This presentation highlights 3 stocks (AZ01, XD03 and NICA2) for which microsatellite and IEA data appear contradictory.

AZ01 was identified as L. V. braziliensis using IEA. Microsatellite analysis using AC01 and AC52 however, identifies this stock as L. V. panamensis (AC16 results are ambiguous). XD03 was identified as L. V. panamensis using IEA. Microsatellite analysis shows that this stock is identical to the majority of hybrid stocks at all 3 loci (but, see section 5.6.4.2). Microsatellite results for NICA2 (hybrid stock) differed from IEA results at just one locus, AC01, where NICA2 failed to generate a L. V. panamensis peak: for reasons discussed previously regarding the amplification of L. V. panamensis alleles at this locus, this ambiguity is probably not significant.

Microdemes observed in stocks from Nicaragua are summarized in *table 5.6.3.5.* Fourteen microdemes were scored among the 18 Nicaraguan stocks assigned: each L. *V. braziliensis* stock (4) had a different microdeme, *L. V.* panamensis stocks (7) could be separated into 6 microdemes. Putative hybrid stocks (7) were separated into 4 microdemes; three hybrid stocks had identical multilocus genotypes, microdeme LMS34. In agreement with results for the Brazilian stocks, most microdemes contained a single stock. MCL stocks were LMS29, LMS30, LMS31 and LMS39.

Table 5.6.3.4:

Comparison of microsatellite (Genotyper®) and isoenzyme data, by genotypes observed at each microsatellite locus

		Stocks:	species status assigned us	ing IEA data
Locus	Observed genotype	L. V. braziliensis	L. V. braziliensis / L. V. panamensis hybrid	L. V. panamensis
	227 / 227	LTB300, XD28, HN196, XD19	NICA2	
AC01	227 / 241		XD09, XD42, XD36, XD13, XD20, XD05	XD03
	241 / 241			WSL03, WSL06, ZF01
	245 / 245	AC01		ZE09, XD39
	244 / 244	LTB300		LS94, WSL03, WSL06, ZE09, XD39, ZF01
AC16	250 / 250	HN196	XD09, XD42, XD36, XD13, XD20, NICA2, XD05	XD03
	250 / 252	XD28, AZ01		XD45
	254 / 254	XD19		
	283 / 283	LTB300		
	291 / 291	AZ01		LS94, WSL03, ZE09, XD39
	291 / 307		XD36	
AC52	291 / 309		XD09, XD13, XD20, NICA2, XD05	XD03
	295 / 295			ZF01
	299 / 299			WSL06
	297 / 299	XD19		
	307 / 307	HN196		
	313/313	XD28		

Table 5.6.3.4 comparison of microsatellite (Genotyper®) and isoenzyme data, by genotypes observed at each microsatellite locus, Code names in **bold** indicate reference strains, not of Nicaraguan origin. Stocks which are <u>underlined</u> have conflicting microsatellite and IEA results, see text.

Notes to *table 5.6.3.5* (next page): summary of microdemes observed in Nicaraguan stocks. † microdeme is the term coined here to describe *Leishmania* sharing the same microsatellite multilocus genotype (i.e. combined genotype data from the 3 microsatellite loci: AC01, AC16 and AC52). "LMS" numbers have been used to identify different microdemes (i.e. LMS1, 2, 3 etc.). Stocks which failed to generate PCR products at any of the three loci were not assigned a microdeme number because the genetic basis of failure to amplify could not be determined. §*Viannia* species codes: b - *braziliensis*, p - *peruviana*; g - *guyanensis*; s - *shawi*; pa - *panamensis*, 1 - *lainsoni*; * *L. V.* sp. n. (reference strain IM2832, Grimaldi et al., 1991); h - putative L. V. braziliensis / L. V. panamensis hybrid. Stocks which are in *italics* are from MCL lesions. F - failed to generate scorable Genotyper@ peak.
Microdeme	N	licros	atellite	locus	genotyp	De	Geogra	phical origin	Stock(s)	Viannia
+	A	201	A	C16	A	C52	State/Dept	ate/Dept Region/Area		species
LMSI	227	227	244	244	283	283	BRAZIL		LTB300	b
LMS2	231	231	ND	ND	ND	ND	PERU		LC39	pe
LMS3	225	241	242	242	ND	ND	BRAZIL		M4147	g
LMS4	241	241	246	248	297	299	BRAZIL		M15065	S
LMS5	239	239	244	244	291	291	PANAMA		LS94	pa
LMS6	231	231	240	270	281	283	BRAZIL		M6426	1
LMS7	227	245	X	X	X	X	BRAZIL		IM2832	*
LMS28	227	227	250	252	313	313	Jinotega	Bocay	XD28	
LMS29	227	227	250	250	307	307	Jinotega	Bocay	HN196	b
LMS30	227	227	254	254	297	299	Zelaya Norte	Leymus	XD19	
LMS31	245	245	250	252	291	291	Ocotal	Мигта	AZ01	
LMS32	227	241	250	250	291	307	Jinotega	Pantasma	XD36	
LMS33	227	227	250	250	291	309	Zelaya Norte	Wasiala	NICA2	
							Jinotega	Восау	XD05, XD09	h
LMS34	227	241	250	250	291	309	Zelaya Norte	Wamblan	XD13	
				İ				Waslala	XD20	
LMS35	227	241	250	250	291	313	Jinotega	Pantasma	XD42	
LMS36	241	241	244	244	291	291	Zelaya Norte	Guayabo	WSL03	
LMS37	241	241	244	244	295	295	Matagalpa	El Coyolar	ZF01	
LMS38	241	241	244	244	299	299	Zelaya Norte	Waslala	WSL06	
1 1/620	246	245	244	244	201	201	Zelaya Sur	Nova Guinea	ZE09	pa
LM339	245	245	244	244	271	271	Matagalpa	Paiwas	XD39	
LMS40	227	241	250	250	291	309	Rio San Juan	Azucenas	XD03	
LMS41	X	X	250	252	F	F	Chontales	Los Chiles	XD45	
not assigned±	X	X	X	x	X	x	Ocotal	Murra	XD17	b
							Matagalpa	Rio Blanco	XD41	h h

Table 5.6.3.5: Summary of microdemest observed in Nicaraguan stocks.

Chapter 5 - Nicarama

5.6.4 AC01 Sequencing

Of the 17 Nicaraguan stocks which generated a PCR product for AC01, all produced readable AC01 sequence data. *Figure 5.6.4.1* shows the alignment of sequences from the AC01 microsatellite locus for Nicaraguan stocks and reference strains.

5.6.4.1 AC01 sequence variation among Nicaraguan stocks

L. V. braziliensis and L. V. panamensis stocks could be separated based on the size of the repeat (8 : 14-17 repeats respectively), by two bases changes at positions 25 and 28 ([TTGT : CTGC]), by two bases changes at positions 135/6 ([TT : CG]) and by a single base change at position 144 ([T : A]), respectively. These differences appear to be species or complex specific. The [C] at position 25 was observed in all L. V. panamensis stocks examined but not in the L. V. guyanensis or L. V. shawi reference strains. L. V. panamensis, like L. V. guyanensis, did not have the 4 base insert immediately prior to the repeat seen in L. V. shawi stocks. One of the Brazilian stocks identified as L. V. shawi (M11367) did not have this insert either and, comparing sequence data (see figure 4.5.4.1), appears to be more similar to L. V. panamensis stocks (see section 5.7).

Among L. V. panamensis stocks, 6 sequence types were scored. Differences in the number of repeats were evident in these stocks: 3 stocks (ZF01, WSL06 and WSL03) had n = 15 and 3 (ZE09, XD03 and XD39) had n = 17. Stocks sharing the same number of repeats appeared also to have shared base types at positions 115 and 123. Four AC01 sequence types were scored among Nicaraguan L. V. braziliensis stocks: all bar one derived from very minor differences (single base changes or bases scored as heterozygous) in the downstream flanking region. One L. V. braziliensis stock (AZ01) was identical to L. V. panamensis at all the discriminatory positions noted above and in the number of repeats (n = 17): this stock will be discussed further in section 5.7. As discussed in chapters 3 and 4, the sequence upstream of the repeat itself, excluding the differences noted above, was conserved. All L. V. braziliensis stocks.

Species	Strain/stock*	10	20	30	40	50	60	70	80
ş	code no.	1	Ţ	I	ł	1	ł	I.	1
b	LTB300 type	ATGTGCCTCT	CCCACCCTTA	GTGCTTGTCT	TCTTCCTGCT	TTGCCTCTCT	GTCTGT	GTGTGTGTGT	GTG
b	XD19								
b	HN196								
h	NICA2 type								
h	XD13								
h	XD36								
h	XD09								
h	XD42								
h	XD20								
pa	LS94			CC					TGTGTGT
ра	ZF01			CC					TGTGTGT
pa	WSL06			CC					TGTGTGT
pa	WSL03			cc					TGTGTGT
pa	ZE09			CC					TGTGTGT
b	AZ01			CC					TGTGTGT
pa	XD03			CC					TGTGTGT
pa	XD39			CC				• • • • • • • • • • •	TGTGTGT
g	M4147			C		C			TGTGTGT
8	M15065						TGTC		TGTGTCT
pe	LC39								TGTG
1	M6426			AC	C	T.GC	C		TGTG

Figure 5.6.4.1 Sequence alignment of the AC01 locus from Nicaraguan stocks and Viannia reference strains.

Notes to figure 5.6.4.1: sequence alignment of the AC01 locus from Nicaraguan stocks and Viamia reference strains. ⁵Species abbreviation: b - braziliensis, papanamensis, h - putative braziliensis / panamensis hybrid, g - guyanensis, s - shawi, pe - peruviana, I - lainsoni. Species designation determined using IEA. Stock type: LTB300 type (LTB300 and XD28); NICA2 type (NICA2 and XD05). Reference strains are indicated in **bold** type. Bases in blue type form the microsatellite repeat region; bases in red type illustrate the TCCGC end motif; bases in green type are invariant downstream in all stocks. Numbers above sequence data indicate the base position used for alignment purposes. Only 165 bases of the total AC01 product were aligned and used for phylogenetic analysis (see chapter 3). See Appendix IV for base coding system.

Species	Strain/stock*	90	100) 110	120	13	140) 150	160 165
ş	code no	I	1		1	1	1	1	
b	LTB300 type		-TCCGCTTCA	GTGGGCCGAT	CCGTTTCACT	TTTTGCCGGT	GACGTTTGTG	TGCTGACGTG	TCTGGTGGCC TCWCA
b	XD19		÷						A
b	HN196		Y		Y			W	
h	NICA2 type						A		A
h	XD13						A		
h	XD36						A		
h	XD09		Y				A		
h	XD42		YW		к		М		.S
h	XD20		W		S.	W	M		.SMT
ра	LS94	GTGTG	YW	M	Y		CG	AY	
ра	ZF 01	GTGTGTG	Y	M	Y		CG	A	W.A
pa	WSL06	GTGTGTG	Y	M	Y	KW	CG		
ра	WSL03	GTGTGTG	CW	M	Y	KSW	CG		GK W.A
pa	ZE09	GTGTGTGTGT	GCW	M	C		CG	AY	
b	AZ01	GTGTGTGTGTGT	GCW	Y	CW		CG	AY	.SGK A.A
ра	XD03	GTGTGTGTGT	GYW	M	CW	W	SCG	AY	G A.A
ра	XD39	GTGTGTGTGT	GYW	.KSAY	c	GSSW	SCG	AY	.SGK.S A.A
g	M4147	GTGYGYS	SCA	.KSAY	SCW	GY.SSA	SCG	SAY	.SGK W.A
8	M15065	GCG				K	CK	₩	A
pe	LC39						K		K W.A
I	M6426			T.				W	

Figure 5.6.4.1 Sequence alignment of the AC01 locus from Nicaraguan stocks and Viannia reference strains (continued).

Notes to figure 5.6.4.1: sequence alignment of the AC01 locus from Nicaraguan stocks and Viannia reference strains. ⁵Species abbreviation: b - braziliensis, pa - panamensis, h - putative braziliensis / panamensis hybrid, g - guyanensis, s - shawt, pe - peruviana, 1 - lainsoni. Species designation determined using IEA. Stock type: LTB300 type (LTB300 and XD28); NICA2 type (NICA2 and XD05). Reference strains are indicated in **bold** type. Bases in blue type form the microsatellite repeat region; bases in red type illustrate the TCCGC end motif; bases in green type are invariant downstream in all stocks. Numbers above sequence data indicate the base position used for alignment purposes. Only 165 bases of the total AC01 product were aligned and used for phylogenetic analysis (see chapter 3). See Appendix IV for base coding system.

Chapter 5 - Nicaragua

Genotyper[®] data for the putative L. V. braziliensis / L. V. panamensis hybrid stocks at this locus indicated that they were heterozygous. Sequencing analysis showed however, that they were identical to L. V. braziliensis stocks in the upstream flanking region and in the number of dinucleotide repeats scored (n = 8). The downstream flanking region was generally more variable in hybrid stocks than in L. V. braziliensis stocks, with 6 sequence types scored from 7 stocks. As discussed previously, it is likely that sequencing analysis only detected the L. V. braziliensis allele in these stocks because of the relatively poor amplification of the L. V. panamensis allele.

Relationships among the Nicaraguan stocks based on AC01 sequence comparisons will be discussed in section 5.7.

5.6.4.2 Comparison of AC01 data from sequencing and Genotyper®

Sequence data correlated exactly with Genotyper® data for all homozygous L. V. braziliensis and L. V. panamensis stocks, and for the stock AZ01 which appears to have isoenzyme loci corresponding to L. V. braziliensis and microsatellite loci corresponding to L. V. panamensis (see section 5.7). Data from putative hybrid stocks (genotype 227/241) did not concur for reasons discussed previously.

The Genotyper® result for the stock XD03 at AC01 was identical to most other hybrid stocks i.e. having 2 alleles, genotype 227 / 241. The AC01 sequence of this stock, however, indicated that this stock was equivalent to *L. V. panamensis* stocks with the AC01 allele 245 (i.e. n = 17). XD03 was identified as *L. V. panamensis* using IEA. Sequencing and Genotyper® analyses of all Nicaraguan stocks were repeated and generated the same results. The observed discrepancy between sequencing and Genotyper® results is difficult to explain. All other stocks of AC01 genotype 227/241 had n = 8 (the *L. V. panamensis* AC01 allele 241 is assumed to have been "outamplified" during cycle sequencing, see previous sections). Here, only sequence from the *L. V. panamensis* allele is scored and this is not of the expected size (245 instead of 241). The most plausible explanation is that DNA samples were repeatedly mixed up, either for Genotyper® or sequencing analysis: the phylogenetic position of this stock using microsatellite analysis (section 5.7) has been disregarded here.

5.7 Phylogenetic analysis of Nicaraguan stocks

Phylogenetic analysis of isoenzyme data (multilocus data) included data from all *Leishmania* reference strains. Phylogenetic analysis of sequence data from the microsatellite locus AC01 included data from all reference strains of the *Viannia* subgenus, with the exception of *L. V.* sp. n. reference strain IM2832 which failed to produce readable sequence data. The AC01 microsatellite locus is specific to the *Viannia* subgenus hence it was not possible to include reference strains from the subgenus *Leishmania* in this phylogenetic analysis.

5.7.1 Phylogenetic analysis of isoenzyme data

Isoenzyme phylogenies were constructed, as described in chapter 2, using a maximumlikelihood approach, quartet puzzling (Strimmer & von Hassler, 1996). Figure 5.7.1.1 shows the consensus cladogram constructed after quartet puzzling analysis of isoenzyme data from *Leishmania* reference strains and from Nicaraguan stocks. The latter spanned 2 species (*L. V. braziliensis* and *L. V. panamensis*) of the 2 main *Viannia* subgenus complexes (*L. V. braziliensis* and *L. V. guyanensis* complexes, respectively), and included putative *L. V. braziliensis / L. V. panamensis* (i.e. speciescomplex) hybrids.

The Leishmania subgenus reference strains (PP75, L. L. chagasi, BEL21, L. L. mexicana and M2269, L. L. amazonensis) plus the two outlying Viannia reference strains M6426 (L. V. lainsoni) and IM2832 (L. V. sp. n., Grimaldi et al., 1991) were clearly distinct (78% quartet support) from species in the 2 main species complexes of the Viannia subgenus.

The 2 main Viannia species complexes (L. V. braziliensis and L. V. guyanensis complexes) were clearly separated from the other reference strains (63% and 88% quartet support, respectively). Three clades are evident in figure 5.7.1.1. The first clade (88% quartet support) contains reference strains and Nicaraguan stocks belonging to the L. V. guyanensis complex (3 zymodemes: LON203 - L. V. shawi; LON 123 - L. V. guyanensis and LON202 - L. V. panamensis). Within the L. V. guyanensis and LON202 - L. V. panamensis). Within the L. V. guyanensis and L. V. shawi reference strains (53% quartet support), forming a discrete cluster. The second clade (100% quartet support) contains all Nicaraguan



Figure 5.7.1.1: Cladogram showing the results of quartet puzzling for Nicaraguan stocks and reference strains based on zymodemes Values on branches are "quartet supports", figures which are similar to bootstraps (Strimmer & von Hassler, 1996). These values are the percentage support of each branch, representing the number of quartets supporting each given branching order. Branch lengths are arbitrary. Zymodemes: LON202 - L. V. paramensis, LON200 - L. V. braziliensis, LON216 - putative L. V. braziliensis / L. V. paramensis hybrids.

stocks identified as putative L. V. braziliensis / L. V. panamensis hybrids (single zymodeme, LON216). The third clade (97% quartet support) contains reference strains and stocks belonging to the L. V. braziliensis complex (2 zymodemes: LON200 - L. V. braziliensis and LON201 - L. V. peruviana).

The Nicaraguan stocks within each cluster showed a remarkable lack of isoenzyme variation with a single zymodeme observed in each. The putative hybrid stocks clustered slightly closer to *L. V. braziliensis* stocks than to *L. V. panamensis* stocks.

As with the Brazilian stocks, isoenzyme data from the Nicaraguan stocks was also analyzed using a similarity approach. Dendrograms were constructed by UPGMA using the SYN-TAX-pc package (Podani, 1993). Figure 5.7.1.2 shows the dendrogram constructed for these stocks from a similarity coefficient matrix. Stocks are classified according to similarity (range 0 - 1).

The clustering observed using this method concurs with that from quarter puzzling. The Viannia and Leishmania subgenera are clearly defined with all reference strains clustering in the correct subgenus (c.f. figure 4.6.1.2). As with the cladogram generated by quartet puzzling, the L. V. braziliensis and L. V. guyanensis species complexes are also clearly separated. Similarly, the putative L. V. braziliensis / L. V. panamensis hybrids share greater similarity with members of the L. V. braziliensis complex.

5.7.2 Phylogenetic analysis of sequence data from AC01

Molecular phylogenies for sequence data from the AC01 microsatellite locus were constructed as previously described. Figure 5.7.2.1 shows an unrooted phylogram of the neighbour-joining consensus tree for the Nicaraguan stocks and reference strains based on sequence data from the microsatellite locus AC01. The reliability of tree topologies was assessed by the bootstrap method (n = 1000). Only branches with greater than 50% bootstrap support are indicated. Note the trifurcating layout and the different scale in this figure compared to figure 4.6.2.1.

The IEA and AC01 sequence phylogenies generated from data for Nicaraguan stocks are generally concordant, with the notable exceptions of the Nicaraguan hybrid stocks and AZ01, which are discussed in greater detail below.





Figure 5.7.1.2: UPGMA dendrogram constructed from the similarity coefficient matrix for Nicaraguan stocks and reference strains based on isoenzyme data. Similarities between zymodemes (y-axis, range 0-1) were calculated, using SYN-TAX-pc (Podani, 1993), by averaging the Jaccard coefficients calculated from data from twelve enzyme loci. Stocks / strains marked with * are reference strains. Zymodemes (LON numbers) are indicated on branches.





Figure 5.7.2.1: unrooted phylogram of the neighbour-joining consensus tree for the Nicaraguan stocks and reference strains based on sequence data from the microsatellite locus AC01. Numbers indicate the percentage of bootstrap support for each branch. The scale bar represents 0.001 (0.1%) nucleotide changes per sequence position. Numbers in parentheses indicate the number of dinucleotide repeats (n) scored (for L. V. guyanensis and L. V. shawi stocks the repeats are not perfect). Clusters I and II stocks / strains are L. V. braziliensis, cluster III stocks / strains are L. V. peruviana, cluster IV stocks / strains are, with the exception of M4147 (L. V. guyanensis reference strain) L. V. panamensis, V is L. V. shawi and VI is L. V. lainsoni (both reference strains).

In addition to the *L. V. lainsoni* reference strain M6426 (VI), used here as an outgroup, 2 major subdivisions are evident, corresponding to the *L. V. braziliensis* complex (clusters I - III) and the *L. V. guyanensis* complex (clusters IV and V).

Among the L. V. braziliensis complex, three groups can be seen. I contains Nicaraguan stocks identified by IEA as putative L. V. braziliensis / L. V. panamensis hybrids. II includes Nicaraguan stocks identified as L. V. braziliensis and the braziliensis reference strain LTB300. III contains 2 L. V. peruviana reference strains: LC39 (n = 10) and LC1152 (n = 13). In the AC01 phylogenetic analysis of Brazilian stocks, these strains were oddly placed in different species complex clusters; it is possible in the previous analysis that a mistake was made during data entry for the phylogenetic analysis which resulted in the misplacing of LC1152.

Cluster IV contains Nicaraguan L. V. panamensis stocks and the L. V. panamensis and L. V. guyanensis reference strains (LS94 and M4147, respectively). Nicaraguan L. V. panamensis stocks could, by AC01 sequence analysis, be split into 2 groups based on the number of dinucleotide repeats and variation in the downstream flanking region (see figure 5.6.4.1). L. V. panamensis stocks with n = 17 (ZE09, XD03 and XD39), along with the L. V. guyanensis reference strain M4147 and the Nicaraguan stock AZ01, formed a discrete group (62% bootstrap support) within cluster IV.

The stock AZ01 was identified as L. V. braziliensis using IEA. This stock was, however, found to have 17 dinucleotide repeats using sequence analysis and an allele size of 245 using Genotyper, both indicative of L. V. panamensis status. The most parsimonious explanation for this stock is the occurrence of a recombination event between braziliensis and panamensis stocks resulting in the exchange of IEA and AC01 markers. A similar event appears to have occurred in the stock NICA2, which was identified as a putative hybrid using IEA, but scored as braziliensis by both sequence and Genotyper, analyses (n = 8 and an allele size of 227). This stock therefore appears to be heterozygous at enzyme loci but homozygous for AC01, possibly suggesting that partial recombination, involving the enzyme loci only, has occurred.

All strains / stocks in groups I and II were shown, using sequence analysis, to have 8 dinucleotide repeats. As discussed in sections 5.6.3.1 and 5.6.4.1, results for the

putative hybrid stocks from sequence and Genotyper[®] analyses were not in agreement. Genotyper[®] analysis suggested the presence of 2 alleles (227 and 241) in these hybrid stocks, corresponding to alleles from both *L. V. braziliensis* and *L. V. panamensis*. The phylogenetic position of these stocks should be treated with caution until the discrepancies between sequence and Genotyper[®] analyses have been resolved. This also applies to the phylogenetic position of the *L. V. panamensis* stock, XD03 (heterozygous 227 / 241 using Genotyper[®] but 17 dinucleotide repeats by sequence analysis), as previously discussed.

5.8 Discussion

Very little diversity was observed among the Nicaraguan stocks using IEA: just 3 zymodemes were scored, corresponding to *L. V. braziliensis*, *L. V. panamensis* and putative *braziliensis* / *panamensis* hybrids. In agreement with results from Brazilian stocks, microsatellite analysis revealed greater diversity than IEA. However, the two methods used for microsatellite analysis (sequencing and Genotyper®) were not always in agreement, especially for the putative hybrid stocks. This made phylogenetic analyses difficult to interpret. These discordant results need to be investigated further and resolved.

Nevertheless, using both IEA and AC01 microsatellite analysis (Genotyper® data), the patterns observed in the putative L. V. braziliensis / L. V. panamensis hybrid stocks concur with a juxtaposition of the parental patterns. Although these stocks were not cloned, the IEA results cannot be explained by a mixture of genotypes. These observations, assuming that Leishmania are diploid and that these markers are inherited in a Mendelian fashion, suggest that these hybrids may be the equivalent of an F1 generation resulting from a total recombination event between the parental species types. The precise mechanism involved in generating these hybrids remains, in the absence of experimental genetic exchange, unclear. Banuls *et al.* (1999) have recently examined the Nicaraguan hybrids using IEA and RAPD. These authors also raised the possibility that these hybrids may, in fact, represent an ancestral type from which the parental types descended, but suggested that this explanation was less parsimonious than genetic recombination.

CHAPTER 6 - HUÁNUCO, PERU: SYMPATRIC TRANSMISSION OF L. V. BRAZILIENSIS AND L. V. PERUVIANA

6.0 Leishmaniasis in Peru

Cutaneous leishmaniasis (CL) and mucocutaneous leishmaniasis (MCL) are endemic in Peru. Leishmania Viannia braziliensis and L. V. peruviana are the most frequently reported species causing CL (Desjeux, 1992b; Lucas et al., 1994, 1998) however, CL due to infection with L. V. guyanensis, L. V. lainsoni and L. Leishmania amazonensis have also been reported (Lucas et al., 1994, 1998; Franke et al., 1990). Mucosal leishmaniasis (ML), with mucosal involvement by contiguity, has been described for all species causing CL in Peru (Lucas et al., 1998). ML is, however, distinct from MCL which involves the metastatic spread of parasites to mucosal sites: MCL is principally associated with L. V. braziliensis infection (Lucas et al., 1998). DCL has also been reported in Peru (Lucas et al., 1998) but VL is absent (Desjeux, 1992a).

L. V. braziliensis and L. V. peruviana have traditionally been associated with different clinical forms of disease. L. V. braziliensis is regarded as the most notorious species since infection may, in addition to CL, result in MCL ("espundia") (Romero et al., 1987). CL resulting from L. V. peruviana infection, also known as "uta" or Andean CL (ACL), generally presents as relatively benign, self-limiting, cutaneous lesions. CL caused by L. V. braziliensis infection presents a more variable clinical picture ranging from uta-like lesions to multiple aggressive lesions which require therapeutic intervention (Llanos-Cuentas, 1993). However, different grades of lesion severity have been described for both species (Llanos-Cuentas, personal comm.) and clinical examination alone cannot pinpoint accurately the infecting species causing CL in a region where both species occur sympatrically.

Leishmania species have often been incriminated as the cause of CL on the basis of known geographical range. L. V. peruviana is found primarily between 800 - 3000m a.s.l. in the western Andes and inter-Andean valleys where it is associated with rural agricultural practices (Guerra, 1988). L. V. braziliensis is found predominantly at lower altitudes in Arnazonian areas and is associated with activities in forested areas (Guerra, 1988).

There are regions, however, as in the present study area in the unforested Andean Department of Huánuco (Peru), where the Andean and forest regions meet and in which *L. V. peruviana* and *L. V. braziliensis* exist sympatrically.

L. V. braziliensis and L. V. peruviana are remarkably similar and the subtlety of discriminatory power available until relatively recently has impeded the study of genetic diversity amongst them. The two species can be differentiated biochemically using isoenzyme analysis (IEA) (Arana *et al.*, 1990), random amplified polymorphic DNA (RAPD) (Dujardin *et al.*, 1995b, Banuls *et al.*, 2000), PCR-RFLP of the gp63 gene intergenic region (Victoir *et al.*, 1998), molecular karyotyping (Dujardin *et al.*, 1995b) and microsatellite analysis (Russell *et al.*, 1999).

6.1 <u>Study site</u>

Prior to 1983, sporadic cases of cutaneous leishmaniasis were reported from valleys in the Department of Huánuco, Peru (Llanos-Cuentas, 1993). A concurrent CL and MCL epidemic has occurred in the region since the mid 1980's. Leishmania stocks were isolated from humans and dogs as part of a study entitled "The epidemiology of Leishmania braziliensis in the unforested eastern Andean highlands of Huánuco Department, Peru" (Davies, C.R. & Llanos-Cuentas, E.A. ISC grant: contract no. C11-CT93-0036). Figure 6.1 shows a map of the study site, which is situated at an altitude of 2000-3000m. a.s.l., an altitude commonly associated with uta. The study site encompassed rural and peri-urban villages around Huánuco City and from the neighbouring province of Ambo, an area circa 40km in diameter, in three valleys: Higueras, Quera and Chinobamba. Two of the valleys, Quera and Higueras, are relatively dry, with mostly xerophytic vegetation, although some irrigation is in situ for cultivation purposes. The third valley, Chinobamba, is more moist: the land here is largely employed in crop cultivation (coffee, potatoes and maize). Most adults of all three valleys are employed in agricultural practices; the extent of visits to forested areas is difficult to ascertain. A full demographic and clinical census of the populations inhabiting villages and peri-urban sites around Huánuco City was carried out as part of this study. Households in these areas routinely keep dogs, donkeys, pigs and chickens. Leishmania transmission appears to be domestic (C. Davies, personal comm.). The principal sand fly vector in the area is suspected to be Lutzomyia (Helcocyrtomyia) tejadai, an anthropophilic species found both inside and outside houses (C. Davies, personal comm.).



Figure 6.1: Map of Huanuco study sites, Huanuco Department, Peru.

Figure 6.1: Map of Huanuco study site. Shaded areas are situated 2000-3000 above sea-level. Map of Peru (inset) shows location of Huanuco . Department. (Courtesy of Clive Davies).

6.2 Leishmania reference strains and Huanuco stocks

6.2.1 Leishmania reference strains

The reference strains used for comparison include those listed in *table 3.1* (Chapter 3), plus five additional *L. V. peruviana* stocks, two isolated from humans (MHOM/PE/94/LC1152; MHOM/PE/84/LC26) and three isolated from canine lesions (MCAN/PE/92/LRP709; MCAN/PE/92/LRP737; MCAN/PE/92/LRP906). These additional reference strains all originate from the Department of Ancash, Peru.

6.2.2 Huanuco stocks

Fifty-nine *Leishmania* stocks were isolated: 45 from humans and 14 from dogs. These stocks have not been characterized previously, although seven other stocks from the region have been analyzed by IEA, RAPD and molecular karyotyping (Dujardin *et al.*, 1995b). The Huanuco stocks are listed in *table 6.2.2*.

STOCK CODE NO.	HOST	VALLEY / DISTRICT / VILLAGE	LESION
	SPECIES		TYPE
MCAN/PE/95/HR78	Dog	Quera / El Valle / Conchumayo	CL
MCAN/PE/95/HR80	Dog	Quera / El Valle / Conchumayo	CL
MCAN/PE/95/HR108	Dog	Quera / El Valle / Conchumayo	CL
MCAN/PE/95/HR110	Dog	Quera / El Valle / Conchumayo	CL
MHOM/PE/94/LC2421	Human	Quera / El Valle / Conchumayo	MCL
MHOM/PE/94/LC2422	Human	Quera / El Valle / Conchumayo	CL
MHOM/PE/94/LC2412	Human	Quera / El Valle / Pomacucho	CL
MHOM/PE/94/LC2452	Human	Quera / El Valle / Huayrajirca	CL
MHOM/PE/94/LC2679	Human	Quera / El Valle / Huaycho	CL
MHOM/PE/95/LC2829	Human	Quera / El Valle / Mirache	MCL
MHOM/PE/95/LC2898	Human	Quera / El Valle / Pachabamba	CL
MHOM/PE/95/LC2900	Human	Quera / El Valle / Naosa	MCL
MHOM/PE/95/LC2875	Human	Chinobamba / Churubamba / Churubamba	CL
MCAN/PE/95/HR529	Dog	Chinobamba / Churubamba / Quechaloma	CL
MHOM/PE/94/LC2434	Human	Chinobamba / Churubamba / Quechaloma	CL
MHOM/PE/94/LC2450	Human	Chinobamba / Churubamba / Quechaloma	CL
MHOM/PE/94/LC2683	Human	Chinobamba / Churubamba / Quechaloma	CL
MCAN/PE/95/HR701	Dog	Chinobamba / Churubamba / Vilcabamba	CL
MHOM/PE/95/LC2901	Human	Chinobamba / Churubamba / Vilcabamba	MCL
MHOM/PE/94/LC2435	Human	Chinobamba / Churubamba / Vilcabamba	CL
MCAN/PE/95/HR399	Dog	Chinobamba / Churubamba / Chinobamba	CL
MCAN/PE/95/HR410	Dog	Chinobamba / Churubamba / Chinobamba	CL
MCAN/PE/95/HR413	Dog	Chinobamba / Churubamba / Chinobamba	CL
MCAN/PE/95/HR419	Dog	Chinobamba / Churubamba / Chinobamba	CL
MCAN/PE/95/HR424	Dog	Chinobamba / Churubamba / Chinobamba	CL
MCAN/PE/95/HR434	Dog	Chinobamba / Churubamba / Chinobamba	CL
MCAN/PE/95/HR797	Dog	Chinobamba / not determined	CL
MCAN/PE/95/HR799	Dog	Chinobamba / not determined	CL
MHOM/PE/94/LC2484	Human	Chinobamba / Churubamba / Chinobamba	MCL
MHOM/PE/94/LC2485	Human	Chinobamba / Churubamba / Chinobamba	MCL
MHOM/PE/94/LC2487	Human	Chinobamba / Churubamba / Chinobamba	CL
MHOM/PE/94/LC2489	Human	Chinobamba / Churubamba / Chinobamba	CL

Table 6.2.2 Leishmania stocks from Huánuco, Peru.

STOCK CODE NO.	HOST	VALLEY / DISTRICT / VILLAGE	LESION
	SPECIES		ТУРЕ
MHOM/PE/94/LC2511	Human	Chinobamba / Churubamba / Chinobamba	CL
MHOM/PE/94/LC2523	Human	Chinobamba / Churubamba / Chinobamba	MCL
MHOM/PE/95/LC3150	Human	Chinobamba / not determined	CL
MHOM/PE/94/LC2491	Human	Chinobamba / Churubamba / Huayucro	CL
MHOM/PE/94/LC2520	Human	Chinobamba / Churubamba / Cascay	CL
MHOM/PE/94/LC2522	Human	Chinobamba / Churubamba / Cascay	CL
MHOM/PE/94/LC2551	Human	Higueras / Kichki / Limapampa	CL
MHOM/PE/94/LC2553	Human	Higueras / Kichki / Limapampa	MCL
MHOM/PE/94/LC2558	Human	Higueras / Kichki / Limapampa	CL
MHOM/PE/94/LC2570	Human	Higueras / Kichki / Huancapallac	CL
MHOM/PE/94/LC2572	Human	Higueras / Yarumayo / Chullay	CL
MHOM/PE/94/LC2576	Human	Higueras / Yarumayo / Chullay	CL
MHOM/PE/94/LC2603	Human	Higueras / Yarumayo / Cozo	MCL
		CTTY / BOROUGH / DISTRICT	
MHOM/PE/95/LC2904	Human	Huánuco City /Heroes de Jactay / Las Moras	CL
MHOM/PE/95/LC2905	Human	Huánuco City /Heroes de Jactay / Las Moras	CL
MHOM/PE/95/LC2841	Human	Huánuco City / Amarilis / San Luis	MCL
MHOM/PE/95/LC2842	Human	Huánuco City / Amarilis / San Luis	MCL
MHOM/PE/95/LC2851	Human	Huánuco City / Amarilis / San Luis	CL
MHOM/PE/95/LC2873	Human	Huánuco City / Amarilis / San Luis	CL
MHOM/PE/95/LC2902	Human	Huánuco City / Amarilis / San Luis	CL
MHOM/PE/95/LC2790	Human	Huánuco City / Amarilis / Gonzalez Prada	CL
MHOM/PE/95/LC2843	Human	Huánuco City/Amaralis / Hermilio Valdizan	CL
MHOM/PE/95/LC2876	Human	Huánuco City / Marabamba	CL
MHOM/PE/95/LC2866	Human	Huánuco City / Sariapampa	CL
		PROVINCE / DISTRICT	
MHOM/PE/95/LC2825	Human	Ambo Province / Las Pampas	CL
MHOM/PE/95/LC2844	Human	Ambo Province / Quicacan	CL
MHOM/PE/95/LC2877	Human	Ambo Province / Huacar District	CL

Table 6.2.2 Leishmania stocks from Huánuco, Peru (continued).

6.3 <u>Isoenzyme analysis</u>

Isoenzyme polymorphism was observed in the 59 Huanuco stocks at all enzyme loci bar NHi2. Four stocks (HR419, HR529, HR701 and LC2679) accounted for virtually all polymorphism at 6 loci: PGM, 6PGD, GPI, G6PD, ASAT and ALAT. Two stocks (LC2435 and LC2877) repeatedly lacked detectable enzyme activity at PGM, 6PGD and ALAT. The remaining 53 Huanuco stocks were polymorphic for MPI, NHi1, NHd, ES and PEPD. Banding patterns observed in the reference strains and Huanuco stocks are shown diagramatically, along with allele designations, for all enzymes in *figure 6.3.1*.

6.3.1 Polymorphic enzymes

6.3.1.1 MPI

MPI is the only enzyme that is described as consistently discriminating between L. V. braziliensis and L. V. peruviana on TSGE (Arana et al., 1989). The Huanuco stocks were therefore classified initially according to their MPI profile: 6 MPI phenotypes were scored among Huánuco stocks (see figures 6.3.1 and 6.3.1.1).

Twenty-five stocks were identified as L. V. braziliensis (MPI type 1, inferred genotype 6/6) and 4 as L. V. peruviana (MPI type 2, inferred genotype 7/7). A double-banded pattern (MPI type 3, inferred genotype 6/7) suggestive of a L. V. braziliensis / L. V. peruviana hybrid was observed in 26 stocks. This double-banded phenotype is the expected banding pattern for a heterozygote using a monomeric enzyme such as MPI. Mixed cultures were excluded by both equal band intensities and the occurrence of the same phenotype in biological clones isolated from stocks from the same area (Dujardin et al. 1995b). This MPI phenotype was also observed in 2 stocks (including the widely used reference strain M2903) from Brazil: in these stocks however, polymorphism was not observed at any other locus. Three stocks presented another double-banded MPI pattern (MPI type 4, inferred genotype 6/9): the lower band in this pattern corresponded to that of L. V. braziliensis but the upper band did not correspond to any of the reference strains used. This MPI pattern was not observed among the 7 Huanuco samples analyzed by Dujardin et al. (1995b), however, this phenotype has been previously reported in stocks from Belize (Evans et al., 1984). Three stocks generated banding patterns (MPI type 5, inferred genotype 5/6) identical to the L. V. sp. n. reference strain IM2832 and one stock was L. V. lainsoni (MPI type 6, inferred genotype 4/4).

Figure 6.3.1.1:

MPI phenotypes observed in the Huanuco stocks.



Figure 6.3.1.1: MP1 phenotypes observed in the Huanuco stocks. Lanes 1-3: reference strains: lane 1: L. V. braziliensis (LTB300); lane 2: L. V. peruviana (LC39); lane 3: L. V. lainsoni (M6246). Lanes 4-12: Huanuco stocks: lanes 4 & 5: L. V. peruviana (HR78 & LC2484); lanes 6, 8 & 9: L. V. braziliensis (LC2422, LC2487 & LC2412); lane 7: putative L. V. braziliensis / L. V. peruviana (HR399); lanes 10 & 11: L. V. braziliensis variant (MP1 type 4, LC2603 & LC2842); lane 12: L. V. lainsoni (LC2679).

6.3.1.2 <u>NHd</u>

The enzyme NHd was shown to differentiate between the reference strains of L. V. braziliensis and L. V. peruviana however it did not differentiate these species, as identified by MPI type, in the Huánuco stocks. Nonetheless a triple-banded hybrid L. V. braziliensis / L. V. peruviana phenotype was observed in all 26 stocks with MPI type 3 (putative L. V. braziliensis / L. V. peruviana hybrids). A triplet is the expected banding pattern for a heterozygote and a dimeric enzyme such as NHd. The presence of the more intense central band in this triplet adds further evidence to refute the possibility of mixed cultures: the latter would generate a doublet with just 2 bands corresponding to the outer bands of the triplet. Dujardin *et al.* (1995b), also reported finding a triplet pattern with NHd (their NH2) in several stocks from the region. The L. (V) peruviana NHd allele was not observed among the Huánuco stocks: this allele was either not sampled from the small number of L. V. peruviana stocks identified, or originated from elsewhere. Figure 6.4.1.2 shows the NHd phenotypes observed among Huánuco stocks.

6.3.1.3 NHi

The enzyme system NHi consists of two loci: NHi1 (faster migration) and NHi2 (slower migration) and does not differentiate between the reference strains of L. V. braziliensis and L. V. peruviana.

Nhi2 was monomorphic for all of the Huánuco stocks bar LC2679 (L. V. lainsoni). Excluding the L. V. lainsoni stock LC2679, Huánuco stocks could be divided into 2 groups based on their Nhi1 pattern: 23 stocks generated a single band identical to that observed in L. V. braziliensis, L. V. peruviana or L. V. sp. n. (IM2832) reference strains; the remaining 35 stocks (6 L. V. braziliensis stocks and all stocks with MPI types 3 and 4) generated a poorly resolved double-banded pattern which appeared to consist of one band from L. V. braziliensis or L. V. peruviana ("allele 2") plus another band ("allele 3") similar to that produced by reference stocks from the L. V. guyanensis complex (see figure 6.3.1). Figure 6.3.1.3 shows the NHi phenotypes observed among Huánuco stocks.

6.3.1.4 <u>ES and PEPD</u>

Excluding the L. V. lainsoni stock LC2679, Huanuco stocks could be divided into 3 groups based on their ES and PEPD profiles: group 1 - stocks identical to L. V. braziliensis/L. V. peruviana, group 2 - stocks identical to L. V. sp. n. reference strain IM2832 and group 3 - stocks sharing bands from 1 and 2.

At the PEPD locus, group 1 stocks (36) had the inferred PEPD genotype 2/2, group 2 stocks (3) had the inferred PEPD genotype 4/4 and group 3 stocks (19 stocks, i.e. all MPI type 1 stocks bar 3) had the inferred PEPD genotype 2/4.

At the ES locus, group 1 stocks (39) had the inferred PEPD genotype 4/4, group 2 stocks (3) had the inferred PEPD genotype 5/5 and group 3 stocks (16 stocks, i.e. all MPI type 1 stocks bar 3) had the inferred PEPD genotype 4/5.



Figure 6.3.1.2: NHd phenotypes observed among Huanuco stocks. Lanes 1-3: reference strains: lane 1: L. V. braziliensis (LTB300); lane 2: L. V. peruviana (LC39); lane 3: L. V. lainsoni (M6246). Lanes 4-12: Huanuco stocks: lanes 4 & 5: L. V. peruviana (HR78 & LC2484); lanes 6, 8 & 9: L. V. braziliensis (LC2422, LC2487 & LC2412); lane 7: putative L. V. braziliensis - L. V. peruviana hybrid (HR399); lanes 10 & 11: L. V. braziliensis variant (MPI type 4, LC2603 & LC2842); lane 12: L. V. lainsoni (LC2679). Note different mobilities of reference strain and Huanuco stocks of L. V. peruviana.





Figure 6.3.1.3: NHi phenotypes observed among Huanuco stocks. Lanes 1-3: reference strains: lane 1: L. V. braziliensis (LTB300); lane 2: L. V. peruviana (LC39); lane 3: L. V. lainsoni (M6246). Lanes 4-12: Huánuco stocks: lanes 4 & 5: L. V. peruviana (HR78 & LC2484); lanes 6, 8 & 9: L. V. braziliensis (LC2422, LC2487 & LC2412); lane 7: putative L. V. braziliensis / L. V. peruviana hybrid (HR399); lanes 10 & 11: L. V. braziliensis variant (MPI type 4, LC2603 & LC2842); lane 12: L. V. lainsoni (LC2679).

6.3.2 Zymodemes

The 59 Huanuco stocks showed a remarkable degree of isoenzyme diversity considering that they originated from such a small area. Three species *L. V. peruviana, L. V. braziliensis* and *L. V. lainsoni* and a new, as yet unnamed parasite, *L. V.* sp. n. (type strain IM2832) were found to occur sympatrically in the region. *L. V. guyanensis* and *L. L. amazonensis*, previously reported from Peru (Lucas *et al.*, 1994; 1998) were not identified among the Huanuco stocks.

Twelve zymodemes were identified using eleven enzyme systems (12 loci). Alleles and zymodeme designations for the Huánuco stocks are summarized in *table 6.3.2.1*: allele designations are the same as those given in *figure 6.3.1*. Stocks characterized as L. V. lainsoni and L. V. sp. n. (type strain IM2832) accounted for two of the zymodemes scored. LON227 included 3 dog stocks which were characterized as L. V. sp. n.: these stocks differed from the L. V. sp. n. reference strain IM2832 (LON5) at 3 loci. LON228 contained a single human stock, LC2679, which differed from the L. V. lainsoni reference strain M6426 (LON204) at two loci (NHi1 and GPI).

Ten zymodemes encompassed the remaining 55 stocks which could be split into 3 groups: L. V. braziliensis (LON222 - LON226), L. V. peruviana (LON217) and putative L. V. braziliensis / L. V. peruviana hybrids (LON218 - 221). L. V. braziliensis stocks differed from the reference strain LTB300 (LON200) at 1 or 2 loci; L. V. peruviana stocks differed from the reference strain LC39 (LON201) at a single locus. In contrast to the putative L. V. braziliensis / L. V. praziliensis / L. V. panamensis hybrids from Nicaragua, the genotypes of these hybrids were not a simple combination of "parental" genotypes. All 4 L. V. peruviana stocks were zymodeme LON217. Most L. V. braziliensis stocks were LON225 (15/25): 4 other zymodemes were scored, 3 containing 3 stocks each and one containing a single stock. Most hybrid stocks were LON218 (23/26): 3 other zymodemes were scored, each containing a single stock.



219



220





Notes to <u>table 6.3.2.1</u> (next page): allele and zymodeme designations for Huanuco stocks. Code names given in **bold** are reference strains. Z = zymodeme; * LON numbers above 200 have been reassigned by the author as previous records have been lost (zymodemes are assigned based on the IEA conditions used, i.e. for *Leishmania Viannia* species). For full enzyme names, see chapter 2. See *figure 6.3.1* for allele scoring scheme. Data given assumes diploidy, see introduction.

Table 6.3.2.1:

Allele and zymodeme designations for Huánuco stocks.

						EN	ZYME						1
STOCK CODE NOs.	MPI	NHi 1	NHi 2	NHd	ES	PEPD	PGM	6PGD	GPI	G6PD	ASAT	ALAT	Z*
HR78, LC2434, LC2484, LC2851	7/7	2/2	7/7	4/4	4/4	2/2	3/3	5/5	4/4	2/2	5/5	4/4	LON217
HR80, HR108, HR110, HR399, HR413, HR424, HR434, HR797, HR799, LC2422, LC2485, LC2511, LC2553, LC2570, LC2576, LC2790, LC2825, LC2841, LC2844, LC2901, LC2902, LC2904, LC3150	6/7	2/3	7/7	4/5	4/4	2/2	3/3	5 / 5	4/4	2/2	5/5	4/4	LON218
HR410	6/7	2/3	7/7	4/5	4/5	2/2	3/3	5/5	4/4	2/2	5/5	4/4	LON219
LC2435	6/7	2/3	7/7	4/5	4/4	2/2	-	5/5	4/4	2/2	5/5	-	LON220
LC2877	6/7	2/3	7/7	4/5	4/4	2/2	-	-	4/4	2/2	5/5	4/4	LON221
LC2551, LC2603, LC2842	6/9	2/3	7/7	4/4	4/4	2/2	3/3	5/5	4/4	2/2	5/5	4 / 4	LON222
LC2450, LC2452, LC2523,	6/6	2/3	7/7	4/4	4/4	2/4	3/3	5/5	4/4	2/2	5/5	4/4	LON223
LC2520, LC2876, LC2900,	6/6	2/3	7/7	4/4	4/4	2/2	3/3	5/5	4/4	2/2	5/5	4/4	LON224
LC2412, LC2421, LC2487, LC2489, LC2491, LC2522, LC2558, LC2572, LC2683, LC2829, LC2843, LC2866, LC2875, LC2898, LC2905,	6/6	2/2	7/7	4/4	4/5	2/4	3/3	5/5	4/4	2/2	5/5	4/4	LON225
LC2873	6/6	2/2	7/7	4/4	4/4	2/4	3/3	5/5	4/4	2/2	5/5	4/4	LON226
HR419, HR529, HR701	5/6	2/2	7/7	4/4	5/5	4/4	2/2	5/5	2/2	3/3	5/5	4/4	LON227
LC2679	4/4	1/1	7/7	3/3	2/2	5/5	2/2	8/8	5/5	5/5	1/1	5/5	LON228

223

("hapter 6 - Hubmaco, Peru

6.3.2.1 Zymodemes: associations with host species, clinical presentation and geographical origin

Huánuco stock zymodemes are summarized by host species, clinical presentation and by geographical origin in *table 6.3.2.2.* Of these 59 stocks, 45 (76.3%) were of human origin and 14 (23.7%) were isolated from dogs.

Among the 45 stocks isolated from humans, 3 (6.7%) were identified as L. V. peruviana, 25 (55.6%) as L. V. braziliensis, 16 (35.6%) as putative L. V. braziliensis / L. V. peruviana hybrids and a single stock (2.2%) as L. V. lainsoni, a species only recently reported from Peru (Lucas et al., 1994). Eleven of the 45 (24.4%) human stocks originated from patients with MCL. Six MCL stocks were identified as L. V. braziliensis (6/25 stocks from 4/5 L. V. braziliensis zymodemes), underlining the association of this species with MCL. A single MCL stock was identified as L. V. peruviana. Lucas et al. (1998) attributed a single case of ML in Peru to L. V. peruviana but made no comment on this observation in the text. No distinction was made between ML and MCL in their study however, so it is possible that the lesion from which this stock originated was ML, as defined previously, not MCL. Nevertheless, the isolation here of L. V. peruviana from an MCL lesion is an important finding linking yet another Viannia species with this clinical presentation. The putative L. V. braziliensis/L. V. peruviana hybrid was also associated with MCL (4/26 stocks, 1/4 hybrid zymodemes). The implication of these findings is that all patients from Huanuco infected with L. V. braziliensis, L. V. peruviana and putative hybrids thereof, must be considered at risk of developing MCL.

Fourteen stocks were isolated from dogs (from cutaneous lesions on ears or nose): these dogs came from 2 of the regions in the study (Quera and Chinobamba). Four zymodemes were scored from these dogs: LON217 (*L. V. peruviana*, 1 stock); LON 218 and LON219 (putative hybrid, 9 and 1 stocks respectively) and LON227 (*L. V.* sp. n., 3 stocks). The 3 dog stocks accounting for LON227 (HR419, HR529 and HR701) were almost identical to the *L. V.* sp. n. reference strain IM2832, which has been proposed to be a new species in the subgenus *Viannia* (Grimaldi *et al.*, 1991): as far as I am aware, this is the first time that this *Viannia* phenotype has been identified in Peru or in dogs (IM2832 is a sand fly isolate from Amazonas State, Brazil). Interestingly, this phenotype was not sampled among the human stocks. *L. V. braziliensis* was not found among the dog stocks examined.

			HUMAN STOCKS										DOG STOCKS								
				C	La					M	CLp			Human						Dog	All
			F	legior	<u>ş</u>		T ¹	Region§ T ² stocks				Region§					stocks	stocks			
Species†	<u>Z</u> *	A	с	н	Q	нс	All	A	с	н	Q	нс	Ali	Total	A	с	н	Q	нс	Total	Total
ре	217	-	1	-	-	1	2	-	1	-		-	1	3	-	-	-	1	-	1	4
	218	2	3	2	1	3	11	-	2	1	-	1	4	15	-	6		3	-	9	24
b/pe	219	-	-	-	-	-	0	-	-	-	-	-	0	0	-	1	-	-	-	1	1
	220	-	1	-	-	-	1	-	-	-	-	-	0	1	-	-	-	-	-	0	1
	221	1	-	-	-	-	1	-	_	-	-	-	0	1	-	-	-	-	-	0	1
	222	-	-	1	-	-	1	-	-	1	-	I	2	3	-	-	-	-	-	0	3
	223	-	1	-	1	-	2	-	1	-	-	-	1	3	-	-	-	-	-	0	3
b	224	-	1	-	-	1	2	-	-	•	1	_	1	3	-	-	-	-	-	0	3
	225	-	5	2	2	3	12	-	-	-	2	-	2	14	-	-	-	-	-	0	14
	226	-	-	-	-	1	1	-	-	-	-	-	0	1	-	-	-	-	-	0	1
*	227	-	-	-	-	-	0	-	-	-	-	-	0	0	-	3	-	-	-	3	3
1	228	-	-	-	1	-	1	-	-	-	-	-	0	1	-	-	-	-	-	0	1
							34						11	45						14	59

Table 6.3.2.2: Su

Summary of Huanuco stock zymodemes by host species, clinical presentation and by geographical region.

Table 6.3.2.2: summary of Huanuco stock zymodemes by host species, clinical presentation and by geographical region. Z = zymodeme; * LON numbers above 200 have been reassigned by the author as previous records have been lost (zymodemes are assigned based on the IEA conditions used, i.e. for *Leishmania Viannia* species). † Species codes: pe - *peruviana*; b - *braziliensis*; b/pe - putative *braziliensis / peruviana* hybrid; * *L. V.* sp. n. (IM2832, Grimaldi *et al.*, 1991); I - *lainsoni*. T¹ - total number of human CL stocks examined. T² - total number of MCL stocks examined. §Region codes: A - Ambo; C - Chinobamba; H - Higueras, Q - Quera, HC - Huanuco City. *CL - cutaneous leishmaniasis; ^bMCL - mucocutaneous leishmaniasis. Chapter 6 - Huánuco Peru

Chapter 6 - Huamuco. Peru

The finding here of L. V. peruviana and L. V. braziliensis / L. V. peruviana hybrid phenotypes in dogs and humans from the same area indicates that both are exposed to the same infective sand fly populations. Dogs are reservoirs of visceral disease and the isolation of apparently identical parasites from dogs and humans has led to the belief that they are also reservoir hosts of CL/MCL. The role of dogs in the epidemiology of CL has recently been extensively reviewed by Reithinger & Davies (1999).

It is difficult to pinpoint any association between geographical origin and zymodeme for these stocks because the area from which they originated was so small and it is not therefore possible to say how much movement occurred between the different regions and villages therein. Nevertheless, it is interesting to note that the 3 dog stocks of zymodeme LON227 all came from the same region (Chinobamba): this zymodeme (species) was not found in human stocks from this region or in dog or human stocks from other regions. It is tempting to speculate that these dogs may have had contact with a sylvatic cycle. Zymodeme LON218 (putative hybrid) was sampled among stocks from all regions and from both human and dog stocks. The zymodeme LON225 (*L. V. braziliensis*), although sampled from 4/5 regions, was only scored from human stocks: in fact, *L. V. braziliensis* was not scored from dogs at all. Too few stocks from each zymodeme were identified to allow any further comment.

6.4 Microsatellite analysis

6.4.1 PCR of microsatellite loci

6.4.1.1 Null alleles

As observed among the Brazilian and Nicaraguan stocks, several Huanuco stocks had null alleles at one or more of the microsatellite loci. Two *L. V. peruviana* stocks (LC2484 and LC2851) had null alleles at all three microsatellite loci. Two hybrid dog stocks (HR399 and HR410) failed to amplify AC52.

6.4.1.2 Agarose gel electrophoresis

PCR products generated from Huanuco stocks for all loci were within the expected size range. Size variation was evident, especially using AC16 and AC52 (data not shown), however the resolution was insufficient for characterization.

Chapter 6 - Huánuco, Peru

6.4.2 Non-denaturing PAGE analysis

6.4.2.1 <u>AC01</u>

Figure 6.4.2.1 shows AC01 products from Huánuco stocks analyzed using nondenaturing PAGE and silver staining.

Figure 6.4.2.1 AC01 products from Huánuco stocks (PAGE analysis)



Figure 6.4.2. F. AC01 products from Huånuco stocks analyzed using non-denaturing 6% PAGE and silver staining. Lane 1 - 100 bp molecular weight marker (Gibco BRL, UK); lane 2 - LC2873; lane 3 - LC2842; lane 4 - LC2841; lane 5 - LC2905; lane 6 - LC2603; lane 7 - LC2570; lane 8 - LC2558; lane 9 - LC2551; lane 10 - LC2523; lane 11 - LC2487; lane 12 - LC2485; lane 13 - LC2484. PCR products from L. V. braziliensis stocks (identified using IEA) are in lanes 2, 3, 5 6, 8 -11; an L. V. peruviana stock is in lane 13 hybrids, putative hybrid stocks are in lanes 4, 7 & 12. Note the prominent heteroduplex bands in hybrid and some L. V. braziliensis lanes.

Many pattern types could be scored among the Huánuco stocks at AC01 using PAGE: the majority of stocks appeared to be heterozygous with 2 allelic bands. All stocks presented the locus-specific additional band, described previously: heteroduplexes were particularly evident for apparently heterozygous stocks. For example, in *figure 6.4.2.1*, stocks in lanes 3, 6 and 9 have 2 clearly-resolved allelic bands and 2 heteroduplex bands which are widely spaced. Stocks in lanes 4, 7 and 12 appear to have 2 allelic bands (poorly resolved): they also have 2 heteroduplex bands but these are lower and more closely spaced. Stocks in lanes 5, 8 and 11 also appear to have 2

allelic bands but their heterozygous bands are more closely spaced than in the 2 previous patterns. Stocks in lanes 2, 10 and 13 have very weak heteroduplex bands; the band in lane 2 is clearly larger than the bands in lanes 10 and 13.

Stocks identified as L. V. braziliensis / L. V. peruviana hybrids using IEA presented several different PAGE patterns, all with heteroduplex bands. AC01 PAGE pattern type correlated almost exactly with zymodeme. For example, the PAGE pattern generated by all stocks of zymodeme LON222 were identical (lanes 3, 6 and 9); this pattern was not found in other stocks. All stocks of zymodeme LON218 generated the pattern seen, for example, in lane 4.

6.4.2.2 <u>AC16</u>

Figure 6.4.2.2 shows the AC16 products from Huånuco stocks. As with AC01, AC16 products from different stocks could be grouped according to pattern type. The interpretation of these PAGE patterns was complicated owing to the presence of small differences in the size of the allelic band, heterozygous allelic bands, additional locus-specific bands and heteroduplex molecules. One PAGE pattern, however, could be correlated with a zymodeme: all stocks of zymodeme LON223 generated the pattern visible in lane 1 of figure 6.4.2.2.

6.4.2.3 <u>AC52</u>

Figure 6.4.2.3 shows AC52 products from Huånuco stocks. Stocks could be grouped for AC52, as for other loci, by comparison of pattern type. The interpretation of AC52 PAGE patterns was complicated owing to the multiplicity of apparently allelic bands (see lanes 3, 8, 9 & 11), additional bands and heteroduplex bands. Clear differences in the sizes of the allelic bands were evident (see lanes 2, 6, 10 & 13). AC52 PAGE patterns could usually be correlated with species type and/or zymodemes: for example, 3 bands (possibly 2 allelic, 1 additional) were generated by hybrid stocks (lanes 8, 9 & 11); all L. V. peruviana stocks and one hybrid stock (LC2422) generated large products; all stocks from zymodeme LON223 generated the pattern observed in lane 6.

Chapter 6 - Huamico, Peru

Figure 6.4.2.2 AC16 products from Huanuco stocks (PAGE analysis)



Figure 6.4.2.2: AC16 products from Huánuco stocks analyzed using non-denaturing 6% PAGE and silver staining. Lane 1 - LC2450; lane 2 - LC2683; lane 3 - LC2901; lane 4 - LC2435; lane 5 - LC2484; lane 6 - LC2485; lane 7 - LC2487; lane 8 - LC2523; lane 9 - LC2551; lane 10 - LC2558; lane 11 - LC2570; lane 12 - LC2603; lane 13 - additional L. V. peruviana reference strain, LC1152; lane 14 - Ikbp molecular weight marker (Gibco BRL, UK). L. V. prauliensis stocks (identified using IEA) lanes 1, 2, 7-10 & 12, L. V. peruviana stock lane 5; hybrid stocks lanes 3, 4, 6 & 11.



Figure 6.4.2.3 AC52 products from Huánuco stocks (PAGE analysis)

mw: bp

Figure 6.4.2.3: AC52 products from Huånuco stocks analyzed using non-denaturing 6% PAGE and silver staining. Lane 1 - 1kbp molecular weight marker (Gibco BRL, UK); lane 2 - LC2422; lane 3 - LC2679; lane 4 - LC2898; lane 5 - LC2875; lane 6 - LC2450; lane 7 - LC2683; lane 8 - LC2901; lane 9 - LC2435; lane 10 - LC2484; lane 11 - LC2485; lane 12 - LC2487; lane 13 - LC2523; lane 14 - LC2844. L. V. braziliensis stocks - lanes 4, 5 - 7, 12 & 13; L. V peruviana stock - lane 10; hybrid stocks - lanes 2, 8, 9, 11 & 14, L. V. lainsoni stock - lane 3.

6.4.3 Genescan® and Genotyper® analyses

6.4.3.1 <u>AC01</u>

Eight distinct AC01 alleles were scored amongst the Huanuco stocks giving rise to 9 genotype combinations: table 6.4.3.1a summarizes the AC01 genotypes observed in Huanuco stocks using Genotyper®. Figure 6.4.3.2 illustrates the AC01 alleles observed in Huanuco stocks using Genotyper®. AC01 Genotyper® results for these stocks can generally be explained by variation of the number of dinucleotide repeats (see AC01 sequencing, section 6.4.4).

Five Peruvian L. V. peruviana reference strains were included for comparison in this study: alleles from these strains, and from the L. V. peruviana reference strain LC39, were more variable in size, but generally larger, than those associated with L. V. braziliensis (allele sizes: 231, 235, 235 and 237). Only 2 of the 4 L. V. peruviana stocks from Huånuco generated an AC01 product. These stocks generated single peaks of sizes 235 (HR78) and 241 (LC2484) and were assumed to be homozygous, genotypes 235/235 and 241/241 respectively. The genotype 235/235 was also found in 1 of the L. V. peruviana reference strains; the allele 241 and inferred genotype 241/241 is associated with Nicaraguan L. V. panamensis stocks (see chapter 5).

From previously examined populations (chapters 4 and 5), the AC01 allele 227 was associated with L. V. braziliensis stocks and was usually scored as a single peak with the inferred homozygous genotype 227 / 227. In contrast to L. V. braziliensis stocks from Brazil and Nicaragua, however, Huanuco stocks identified as L. V. braziliensis using IEA did not present the homozygous 227/227 genotype. Indeed, all L. V. braziliensis stocks bar one presented a heterozygous pattern (i.e. 2 peaks were scored). Most L. V. braziliensis stocks clearly presented the heterozygous genotype 227/231. This genotype was also scored in 3 L. V. braziliensis stocks from Brazil. The variation among Huanuco L. V. braziliensis stocks also included 3 other heterozygous patterns: 227/233 (1 stock), 227/237 (3 stocks) and 229/233 (3 stocks). A single L. V. braziliensis stock, LC2873, presented a single peak, inferred homozygous genotype 237/237. Two L. V. braziliensis AC01 genotypes correlated with zymodemes: all 3 stocks with genotype 227/237 were LON222 and all 3 stocks with the genotype 229/233 were LON223.

Table 6.4.3.1a:

Comparison of AC01 genotypes observed in Huanuco stocks using Genotyper®, by species

Observed	Stock	s: species status	IEA data†		
genotype	b	h	ре	*	1
225 / 225		_	-	HR701	-
227 / 227	LTB300	-	-	-	
227 / 231	LC2412, LC2421 LC2487, LC2489 LC2491, LC2522 LC2558, LC2572 LC2683, LC2829 LC2866, LC2875 LC2898, LC2905 LC2520, LC2900 LC2843	HR108, HR110 HR399, HR410 HR413, HR424 HR434, HR797 HR799, LC2435 <i>LC2485</i> , LC2570 LC2576, LC2790 LC2825, <i>LC2841</i> LC2844, LC2877 <i>LC2901</i> , LC2902 LC2904, LC3150	-	-	-
227 / 233	LC2876			-	
227 / 237	LC2551, LC2603 LC2842 (all LON222)	_			-
227 / 245				IM2832	-
229 / 231	-	HR80	-		LC2679
229 / 233	LC2450, LC2452, <i>LC2523</i> (all LON223)	-	-	-	
231 / 231	-	-	LC26, LC39, LRP906	-	M6426
231 / 235	-		LRP709		
235 / 235	-		LRP737, HR78	HR419	
237 / 237	LC2873	LC2422	LC1152		
241 / 241	_		LC2484	-	-
F	_	-	-	HR529	
X		-	LC2434, LC2851	-	-
ND		LC2511, LC2553	_		

Table 6.4.3.1a: Comparison of AC01 genotypes observed in Huánuco stocks using Genotyper[®], by species. Species abbreviations: b - L. V. braziliensis; h - putative L. V. braziliensis / L. V. peruviana hybrid; pe - L. V. peruviana; * - L. V. sp. n.; 1 - L. V. lainsoni. Status determined from isoenzyme analysis. X - did not generate a PCR product (null allele(s)). F - failed to generate scorable Genotyper[®] peak. ND - not done. Code names in **bold** indicate reference strains, not from Huánuco. Stocks which are in *italics* are from MCL lesions.
Figure 6.4.3.2 Genotyper® output from the AC01 locus for selected Huánuco stocks, illustrating the peaks observed in putative parental and hybrid profiles.



Figure 6.4.3.2: Genotyper \oplus output from the AC01 locus for selected Huánuco stocks, illustrating the peaks observed in putative parental (single peaks, traces 1 - 3 & 6 - 8) and hybrid (double peaks, traces 4 & 5) profiles. Numbers across the top of the figure represent peak size, in bases. The exact sizes of individual peaks were scored automatically by the software. The scale to the right of each trace indicates the peak intensity. See chapter 3 for discussion of allele scoring. Traces 1 - 3 & 6 - 8: peaks with alleles sizes corresponding to 8, 10, 13. 7, 12 and 15 dinucleotide repeats, respectively.

All bar 2 of the putative hybrid stocks generated peaks corresponding to the genotype 227/231 which might be expected from a heterozygous locus having one 227 allele (i.e. from *L. V. braziliensis*) and one 231 allele (i.e. from *L. V. peruviana*). The 231 allele was not sampled among *L. V. peruviana* stocks from Huanuco, but was observed in 4 of the *L. V. peruviana* reference strains. One putative hybrid stock (LC2422) generated a single peak of size 237 (inferred genotype 237/237): this genotype was also observed in one *L. V. braziliensis* stock (LC2873) and one of the additional *L. V. peruviana* reference strains, LC1152.

Only 2/3 IM2832-type stocks generated scorable AC01 data: both generated single peaks of sizes 225 and 235 which were assumed to indicate homozygous genotypes 225/225 and 235/235 respectively. Both of these alleles were distinct from those scored in the IM2832 reference strain (227 / 245).

Most of the Huanuco stocks identified using IEA as L. V. braziliensis and putative L. V. braziliensis / L. V. peruviana hybrids were heterozygous at the AC01 locus. The most common AC01 genotype was 227/231, which accounted for 39/54 (72.2%) of Huanuco stocks. The allele 231, in addition to its presence in the genotype 227/231, was found in the homozygous state in 3 L. V. peruviana stocks (all additional L. V. peruviana reference strains) and in the L. V. lainsoni reference strain, M6426. This allele was also scored from the inferred genotype 229/231 in 2 stocks (1 putative hybrid and the Huánuco L. V. lainsoni stock, LC2679). These observations, in agreement with Russell et al. (1999), indicate that the alleles scored at this locus alone are not always correct in discriminating between Viannia species, as identified by IEA. The fact that most Huanuco L. V. braziliensis stocks were heterozygous suggests that the AC01 locus is not linked to enzyme loci: there is, in fact, no reason why different sets of markers should be simultaneously homozygous or heterozygous.

There appears to be considerably greater diversity in AC01 allele size amongst Huánuco stocks than was observed among Brazilian and Nicaraguan stocks. Huánuco stocks were isolated from a small geographical area in which, it is proposed that, the natural transmission cycle had recently been interrupted by the introduction of a new species. This interruption resulted in the current epidemic and provided the ideal setting for the occurrence of genetic recombination events. *Table 6.4.3.2* compares the allele sizes scored in *L. V. peruviana* and *L. V. braziliensis* stocks / strains from this study with data from Peruvian strains reported by Russell *et al.* (1999).

	L. V. peruviana	L. V. braziliensis
	null	227
	231	229
This thesis	235	231
	237	233
	241	237
No. stocks examined	10	25
	227 (8*)	221 (5)
	229 (9)	231 (10)
Russell et al. $(1999)^{\dagger}$	231 (10)	
	233 (11*)	
	237 (13)	
	239 (14)	
No. stocks examined	27	3

Table 6.4.3.1b: AC01 alleles scored in Peruvian stocks and strains

[†] sizes calculated, assuming no insertions or deletions in flanking regions, from the number of dinucleotide repeats reported (given in brackets - scored from sequence data). See *table 3.8.5.1* (page 136, chapter 3) for calculation. alleles scored once each from a heterozygous stock (inferred equivalent genotype 227/233).

There was no apparent correlation between AC01 allele/genotype and clinical presentation. Ten out of eleven stocks isolated from MCL lesions generated AC01 products and were assigned to 4 genotypes: 227/231 (3/17 L. V. braziliensis and 3/22 hybrid stocks); 227/237 (1/3 stocks - L. V. braziliensis); 229/233 (2/3 stocks - L. V. braziliensis) and 241/241 (1/1 stock - L. V. peruviana). The inferred genotype 241/241 was previously found associated with L. V. panamensis stocks from Nicaragua, where it was not associated with MCL. There was no apparent association between geographical origin and AC01 genotype.

6.4.3.2 <u>AC16</u>

Ten distinct AC16 alleles were scored amongst the Huánuco stocks giving rise to 13 genotype combinations: *table 6.4.3.2* summarizes the AC16 genotypes observed in Huánuco stocks using Genotyper[®].

Of the six *L. V. peruviana* reference strains used, 3 (including LC39) failed to generate scorable Genotyper D data, 2 generated a single peak of 266 (inferred homozygous genotype 266 / 266) and 1 generated 2 peaks of 244 and 274 (inferred heterozygous genotype 244 / 274). Two Huanuco *L. V. peruviana* stocks failed to amplify this locus; the two remaining stocks presented different profiles.

Table 6.4.3.2:

Comparison of AC16 genotypes observed in Huánuco stocks using Genotyper®, by species

Observed	Stocks: species status assigned using IEA data										
genotype	b	h	ре	*	1						
240 / 240	-	HR399, HR410, HR799	-	-	-						
240 / 244		LC2570, LC2435	-		-						
240 / 270		-	-	-	M6426						
244 / 244	LTB300, LC2520, LC2843 LC2873	HR108, HR110, HR413, HR424, HR434, <i>LC2485</i> , LC2576, LC2790 <i>LC2841</i> , LC2844 <i>LC2901</i> , LC2904 LC3150, LC2877	-	-							
244 / 246	LC2551, LC2603 LC2842 (all LON222)	-	-		-						
244 / 248	_	HR797	-	-							
244 / 250	LC2900	LC2825	-	_	-						
244 / 266		-	HR78								
244 / 274	LC2876		LC26		-						
248 / 248	LC2412, LC2421 LC2487, LC2489 LC2491, LC2522 LC2558, LC2572 LC2683, LC2866 LC2875, LC2898 LC2905		-	-	-						
248 / 262	LC2450, LC2452 LC2523 (all LON223)		-	-	-						
250 / 250	_	_	LC2484	-	_						
250 / 252		LC2422, HR80	_	HR529, HR701	LC2679						
250 / 254				HR419	-						
266 / 266			LRP737, LC1152	_							
F	LC2829	LC2902	LC39, LRP906 LRP709	-							
X		-	LC2434, LC2851	IM2832							
ND	-	LC2511, LC2553	-	-							

Table 6.4.3.2: Comparison of AC16 genotypes observed in Huånuco stocks using Genotyper Φ , by species. Species abbreviations: b - L. V. braziliensis; h - putative L. V. braziliensis / L. V. peruviana hybrid; pe - L. V. peruviana; \bullet - L. V. sp. n.; l - L. V. lainsoni. Status determined from isoenzyme analysis. X - did not generate a PCR product (null allele(s)). F - failed to generate scorable Genotyper Φ peak. ND - not done. Code names in bold indicate reference strains, not from Huánuco. Stocks which are in *italics* are from MCL lesions.

HR78 generated 2 peaks (inferred heterozygous genotype 244 / 266, the 266 allele was also identified in 2 *L. V. peruviana* reference strains), whilst LC2484 generated a single peak of 250 (inferred homozygous genotype 250 / 250).

Three Huanuco L. V. braziliensis stocks generated a single peak, inferred genotype 244 / 244, identical to the L. V. braziliensis reference strain LTB300. The majority of Huanuco L. V. braziliensis stocks (13/25), however, presented a single peak of 248 (inferred homozygous, genotype 248 / 248). Five stocks generated double peaks: three stocks (all stocks from zymodeme LON222) generated peaks of 244 and 246 (inferred genotype 244 / 246), one stock generated peaks of 244 and 250 (inferred genotype 244 / 250) and another stock generated peaks of 244 and 274 (inferred genotype 244 / 274, the 274 allele was also scored from one of the L. V. peruviana reference strains).

In the absence of sequence data for AC16 it is difficult to make comparisons between these results and those of Russell *et al.* (1999), however, in their study, allele sizes of *L. V. peruviana* (inferred from the number of dinucleotide repeats scored, n, range 7-12) were, with a single exception (n = 7), considerably smaller than those identified from *L. V. braziliensis* stocks (n, range 15-56).

The majority of putative L. V. braziliensis / L. V. peruviana hybrid stocks (14/26) generated a single peak, size 244, for the AC16 locus (inferred genotype, homozygous 244 / 244). Three stocks generated single products of 240 (inferred genotype 240 / 240) and 6 stocks generated double peaks with inferred genotypes of 240 / 244 (2 stocks), 244 / 248 (1 stock), 244 / 250 (1 stock) and 250 / 252 (2 stocks).

All 3 IM2832-type Huanuco stocks generated 2 AC16 products: two were heterozygous 250/252 and the third 250/254.

Six AC16 genotypes were observed among stocks isolated from MCL lesions: 244 / 244 (also associated with MCL in Brazilian *L. V. braziliensis* stocks and a single Nicaraguan *L. V. panamensis* stock), 244 / 246, 244 / 250, 248 / 248, 248 / 262 and 250 / 250 (also associated with MCL from a single Brazilian *L. V. braziliensis* stock).

6.4.3_3 <u>AC52</u>

Many Huanuco stocks failed to generate readable Genotyper[®] data from the AC52 locus. Of those stocks which could be scored (34/59), 5 distinct AC52 alleles were scored in 4 genotype combinations: *table 6.4.3.3* summarizes the AC52 genotypes observed among Huanuco stocks using Genotyper[®].

Table 6.4.3.3: Comparison of AC52 genotypes observed in Huanuco stocks using Genotyper®, by species

Observed	Sto	cks: species stat	us assigned usin	g IEA data	
genotype	b	h	ре		1
265 / 295	LC2520, LC2843 LC2876, LC2551 <i>LC2603, LC2421</i> LC2487, LC2489 LC2491, LC2522 LC2558, LC2572 LC2683, LC2866 LC2875, LC2898 LC2905, <i>LC2829</i>	LC2570, LC2435 HR108, HR110, HR413, HR424, HR434, LC2485 LC2576, LC2790 LC2841, LC2844 LC2877, LC2904 LC3150, LC2825	-	_	-
281 / 283	-		-	-	M6426
283 / 283	LTB300	_		-	
287 / 289	-		-	HR701	-
295 / 295	-		LC26		LC2679
291 / 291	_	_		HR419	
F	LC2900, LC2842 LC2412, LC2450 LC2452, LC2523 LC2873	HR799, <i>LC2901</i> HR797, HR80, LC2422, LC2902	HR78, <i>LC2484</i> LRP906, LRP709, LRP737, LC1152, LC39	HR.529	
X	-	HR399, HR410	LC2434, LC2851	IM2832	-
ND	-	LC2511, LC2553			_

Table 6.4.3.3: Comparison of AC52 genotypes observed in Huånuco stocks using Genotyper, by species. Species abbreviations: b - L. V. braziliensis; h - putative L. V. braziliensis / L. V. peruviana hybrid; pe - L. V. peruviana; * - L. V. sp. n.; l - L. V. lainsoni. Status determined from isoenzyme analysis. X - did not generate a PCR product (null allele(s)). F - failed to generate scorable Genotyper peak. ND - not done. Code names in bold indicate reference strains, not from Huåmuco. Stocks which are in *italics* are from MCL lesions.

Chapter 6 - Huànuco, Peru

Only 1 L. V. peruviana strain (LC26) generated a scorable Genotyper® peak, size 295, inferred genotype 295 / 295. All other reference strains, and the 2 Huanuco stocks which produced a PCR amplification product for AC52, failed to produce peaks.

All L. V. braziliensis and putative L. V. braziliensis / L. V. peruviana stocks from Huánuco which were scorable generated 2 peaks of sizes 265 and 295, with an inferred heterozygous genotype of 265 / 295. The 265 allele has not been described from other populations or reference strains. The 295 allele was scored from an inferred homozygous genotype 295 / 295 in 2 Huánuco stocks: the L. V. lainsoni stock LC2679 and from one of the L. V. peruviana reference strains, see above. This genotype was also scored from a single Nicaraguan L. V. panamensis stock. The 2 other inferred AC52 genotypes were accounted for by 2 IM2832-type stocks, inferred genotypes 287 / 289 and 291 / 291: the latter has also been scored from the Nicaraguan population, mostly from L. V. panamensis and putative L. V. braziliensis / L. V. panamensis hybrid stocks.

It is not clear why so many of the Huánuco stocks failed to generate readable AC52 Genotyper® data: the AC52 products were analyzed at the same time in the same lanes of the same gel as AC01 and AC16 products which were easily scored. The run was repeated with the same results; AC52 peaks were very short and diffuse, and could not be reliably scored.

Comparing the AC52 non-denaturing PAGE results for these stocks (figure 6.4.2.3) with stocks from other populations (figure 4.5.2.4, Brazil and figure 5.6.2.3, Nicaragua), it is evident that, overall, the AC52 patterns from Huánuco stocks are more complex, with many additional bands staining as intensely as the presumed allelic bands. These many bands may account for the multiple diffuse peaks of low intensity seen in the Huánuco stocks using Genotyper. A possible explanation for these results could be slippage during PCR amplification resulting in various product sizes, however, slippage presumably occurs randomly and it would therefore be unlikely that consistent pattern types would have been scored from PAGE analysis.

Russell et al. (1999), using sequence analysis, also found the AC52 locus to be complex, notably in L. V. peruviana isolates, with a long dinucleotide repeat sequence

interspersed with (TA) inserts. They found fewer (TA) inserts and smaller numbers of repeats in isolates of other species.

Genotyper® results at other microsatellite loci for the Huanuco stocks indicated that this population was extremely heterogeneous. AC52 Genotyper® scoring was impossible for some stocks in all groups (i.e. identified as species, hybrids and the IM type). To address this problem, several approaches could be taken: (i) re-examining the PCR conditions used for this locus; (ii) using an AC52 probe (derived from a single AC52 PCR product) under high- stringency conditions on Southern blots of genomic DNA-RFLPs to see if this locus has been duplicated in these stocks.

6.4.3.4 Multilocus analysis of microsatellite data

Missing data complicated assigning microdemes in the Huanuco stocks: stocks with data at all 3 loci were assigned to microdemes first, followed by stocks with data at both AC01 and AC16. Microdeme numbers were assigned (solely on Genotyper® data) with increasing AC01, then AC16, allele size, where possible. Stocks with data from AC52 plus one other locus were assigned to given microdemes based on similar PAGE patterns at the locus for which Genotyper® data was missing. Stocks with data from only 1 locus were not assigned to a microdeme (scored "?") but grouped with other similar stocks based on the data available. Microdemes observed in stocks from Huanuco are shown in *table 6.4.3.4*.

Twenty microdemes, all new, were scored from this data: 3 from the additional L. V. peruviana reference strains and 17 from the Huánuco stocks. Of those scored from Huánuco stocks, 10 contained a single stock and 4 contained more than 2 stocks. Of the 7 microdemes which contained more than one stock, 3 (LMS46, LMS48 and LMS52) contained stocks identified using IEA as different species / types. Microdemes LMS46 (identified by IEA as putative L. V. braziliensis / L. V. peruviana hybrid stocks plus 3 stocks L. V. braziliensis) and LMS51 (all L. V. braziliensis) contained the majority of stocks (17 and 13 respectively). Two microdemes correlated exactly with 2 L. V. braziliensis zymodemes: all 3 stocks identified as LON222 were assigned to microdeme LMS50 and all 3 stocks identified as LON223 were microdeme LMS53. Stocks isolated from MCL lesions were assigned to 6 microdemes.

Table 6.4.3.4 Summary of microdemes[†] observed in stocks from Huanuco and additional L. V. peruviana reference strains.

Microdeme	Microsatellite locus genotype						Stock(s)	Viannia
+	AC	:01	AC	:16	AC	:52		species§
LMS1	227	227	244	244	283	283	LTB300	b
LMS2	231	231	F	F	F	F	LC39	pe
LMS6	231	231	240	270	281	283	M6426	I
LMS7	227	245	X	X	X	X	IM2832	*
LMS42	225	225	250	252	287	289	HR 701	*
	F	F					HR529	*
LMS43	227	231	240	240	x	X	HR399, HR410	b/pe
LMS44	227	231	240	240	F	F	HR799	b/pe
LMS45	227	231	240	244	265	295	LC2570, LC2435	b/pe
LMS46	227	231	244	244	265	295	HR108, HR110, HR413, HR424, HR434, <i>LC2485</i> , LC2576, LC2790, <i>LC2841</i> , LC2844, LC2877, LC2904, LC3150	b/pe
			r	E	-		1C2820	
1			r	Г	E	E		b/me
I MS47	227	221	244	248	r F	F	LC2901	h/pe
I MS49	227	231	244	240	265	205	1002825	b/pe
LIVISTO	221	231	244	230	205 E	<u>275</u> F	102000	b
I MS49	227	232	244	274	265	295	1 C2876	<u> </u>
LMS49	227	233	244	246	265	295	1 C2551 1 C2603	<u> </u>
LIVESSU	221	231	244	240	F	F	1C2842	
LMS51	227	231	248	248	265	295 F	LC2421, LC2487, LC2489, LC2491, LC2522, LC2558, LC2572, LC2683, LC2866, LC2875, LC2898, LC2905	b
LMS52	229	231	250	252	295	295	1.C2679	
	/				F	F	HR80	b/pe
LMS53	229	233	248	262	F	F	LC2450, LC2452, LC2523	b
LMS54	231	231	244	274	295	295	LC26	pe
(? LMS2)	231	231	F	F	F	F	LRP906	pe
?	231	235	F	F	F	F	LRP709	pe
LMS55	235	235	244	266	F	F	HR78	pe
LMS56	235	235	250	254	291	291	HR419	*
LMS57	235	235	266	266	F	F	LRP737	pe
LMS58	237	237	244	244	F	F	LC2873	b
LMS59	237	237	250	252	F	F	LC2422	b/pe
LMS60	237	237	266	266	F	F	LC1152	pe
LMS61	241	241	250	250	F	F	LC2484	pe
?	227	231	F	F	F	F	LC2902	b/pe
±NA	X	X	X	X	X	X	LC2434, LC2851	pe
ND							LC2511, LC2553	b/pe

Table 6.4.3.4 (previous page): summary of microdemes[†] observed in stocks from Huanuco and additional *L. V. peruviana* reference strains. [†] microdeme is the term coined here to describe *Leishmania* sharing the same microsatellite multilocus genotype (i.e. combined genotype data from the 3 microsatellite loci: AC01, AC16 and AC52). "LMS" numbers have been used to identify different microdemes (i.e. LMS1, 2, 3 etc.). "?" - not assigned owing to missing data. Code names in **bold** type indicate reference strains which, although not from Huanuco, are included for comparison. [±]NA - stocks which failed to generate PCR products (X) at any of the three loci were not assigned a microdeme number because the genetic basis of failure to amplify could not be determined. F - failed to generate scorable Genotyper pak. ND - not done. §*Viannia* species codes: b - *braziliensis*; pe - *peruviana*, 1 - *lainsont*; * *L. V.* sp. n. (reference strain IM2832, Grimaldi *et al.*, 1991); b/pe - putative *L. V. braziliensis*; / *L. V. peruviana*, 1 - *lainsont*; * *L. V. braziliensis*; pe - *peruviana*; 1 - *lainsont*; * *L. V. sp.* n. (reference strain IM2832, Grimaldi *et al.*, 1991); b/pe - putative *L. V. braziliensis*; / *L. V. peruviana*; 1 - *lainsont*; * *L.*

6.4.4 AC01 sequencing

Readable AC01 sequence data was obtained from all 55 Huanuco stocks which generated a PCR product for the AC01 locus. *Figure 6.4.4.1* shows the alignment of sequences from the AC01 microsatellite locus for Huanuco stocks and *Viannia* reference strains.

6.4.4.1 Interpretation of sequence data from Huanuco stocks

The large numbers of Huanuco stocks which were shown, using Genotyper®, to have 2 distinct AC01 alleles complicated sequence analysis. Direct cycle sequencing of these stocks generated products from both alleles which overlapped when electrophoresed. For most stocks, differences in allele size could be interpreted in terms of different numbers of dinucleotide repeats (see below). Since the sequence upstream of the repeat was conserved, downstream sequences could be aligned and compared. Despite the overlapping downstream sequence in stocks heterozygous for this locus, sequence data was usually unambiguous to manual interpretation. All samples were coded for sequencing and subsequent analysis to prevent bias from manual scoring. Scoring was carried out using the standard coding system format for nucleotides and nucleotide combinations (*Appendix IV*).

Species	Strain/stock*	10	20	30	40	50	60	70	80
§	code no	+	1		ł	1	1	1	1
b	LTB300	ATGTGCCTCT	CCCACCCTTA	GTGCTTGTCT	TCTTCCTGCT	TTGCCTCTCT	GTCTGT	GTGTGTGTGT	GTG
pe	LC2484								
•	HR419 group								
1	M6426			AC	c	T.GC	c		TGTG
1	LC2679				c	T.GC	c		
5	M15065			c					TGTGCGT
pa	LS94			cc					TGTGTGT
8	M4147					c			TGYGYGT
pe	LC39 group [†]								TGTG
pe	LC1152								TGTGTGT
pe	LRP737 [†]								TGTGTGT
pe	HR78								TGTGTGT
h/b	LC2422 group								TGTGTGT
Ь	LC2452 group								TG
b	consensus 1								TSYG
h/b	consensus 2						R	s	TSYG
b	LC2551 group						R	s	TSYGYKT

Figure 6.4.4.1 Sequence alignment of the AC01 locus from Huanuco stocks and Viannia reference strains: bases 1 - 80.

Notes to figure 6.4.4.1 sequence alignment of the AC01 locus from Huanuco stocks and Viannia reference strains. Species abbreviation b - braziliensis; 1 - lainsont, s - shawi, pa - panamensis; g - guyamensis; pe - peruviana, h - putative braziliensis / peruviana hybrid. Species / type designation determined using IEA. \dagger indicates additional L V. peruviana reference strain Reference strains are indicated in bold type. Stock groups: HR419 group (HR419, HR529 and HR701); LC39 group (LC39, \dagger LC26, \dagger LRP709 and \dagger LRP906); LC2422 group (LC2422 and LC2873); LC2452 group (LC2452, LC2450 and LC2523); LC2551 group (LC2551, LC2603 and LC2842) Consensus groups 1 and 2 are explained in the text. Numbers above sequence data indicate the base position used for alignment purposes. Bases in coloured type are explained in the text. Only these 165 bases of the total AC01 product were aligned and used for phylogenetic analysis (see chapter 3). See Appendix IV for base coding system.

Species	Strain/stock*	90	100) 11	0 12	0 13	0 140	150	160	165
§	code no.	1	Ĩ.	1	1		1	}	1	
b	LTB300		-TCCGCTTCA	GTGGGCCGAT	CCGTTTCACT	TTTTGCCGGT	GACGTTTGTG	TGCTGACGTG	TCTGGTGGCC	TCWCA
pe	LC2484						AK	W		
•	HR419 group									
1	M6426			T.		c		W		A
1	LC2679			T		c		N		W.T
8	M15065	GCG					CK	W		A
pa	LS94	GTGTG	YW	M	Y		CG	AY	KK	A
g	M4147	GTGYGYS	SCA	.KSAY	SCW	GY.SSA	SCG	SAY	.SGK H	W.A
pe	LC39 group [†]				••••		K	W	K i	W.A
pe	LC1152 [†]	GTG	YW	M	Y	KSW	K	W	K K	W.A
pe	LRP737 [†]	GTG	YW	M	Y	KSW	G	W	K 1	W.A
pe	HR78	GTG	CW	MR	C	GM	R	A	GKM. 7	A.A
h/b	LC2422 group	GTG	Y	M	Y	KW	K	W	K I	W
ь	LC2452 group		SYS.KYYW	SW.K.SVSMK	MYSYKW.W	YYSSSK	.WSAYG	YKS.SAYK	.S.SKK.KSV I	нм.
b	consensus 1		-YYYSMK.SR	.YV.RY.SR.	YYSWC.YWY.	GYYG.YSRSD	K.YGY.	KA.DKRYV	KYG.SY.VHH 1	A.A
h/b	consensus 2		-YYYSMKTSR	.YS.RYMSR.	YYSWC.YWY.	GYYG.YSASD	K.YGYK	KA.DKRYV	KYG.SY.SH. /	A.A
b	LC2551 group	SWG	SSCKMW	SY.KKY.AM.	YYK.GY.RS.	KWCGKYY.KD	KR.KKAY	.SYD.WS.YS	DVRKM W	M

Figure 6.4.4.1 Sequence alignment of the AC01 locus from Huanuco stocks and Viannia reference strains (c'td): bases 81 - 165.

Notes to figure 6.4.4.1 sequence alignment of the AC01 locus from Huanuco stocks and *Viannia* reference strains. [§]Species abbreviation: b - *braziliensis*; 1 - *lainsont*; s - *shavi*; pa - *panamensis*; g - *guyanensis*; pe - *peruviana*, h - putative *braziliensis* / *peruviana* hybrid. Species / type designation determined using IEA. \dagger indicates additional L. V. peruviana reference strain. Reference strains are indicated in **bold** type. * Stock groups: HR419 group (HR419, HR529 and HR701); LC39 group (LC39, \dagger LC26, \dagger LRP709 and \dagger LRP906); LC2422 group (LC2422 and LC2873); LC2452 group (LC2452, LC2450 and LC2523); LC2551 group (LC2551, LC2603 and LC2842). Consensus groups 1 and 2 are explained in the text. Numbers above sequence data indicate the base position used for alignment purposes. Bases in coloured type are explained in the text. Only these 165 bases of the total AC01 product were aligned and used for phylogenetic analysis (see chapter 3). See Appendix IV for base coding system.

6.4.4.2 AC01 sequence variation among Huanuco stocks

Based on overall AC01 sequence, Huanuco stocks could be divided into 9 groups, 6 of which contained more than 1 stock. All stocks within each of 7 groups had identical sequences. Stocks in the remaining 2 groups (consensus groups 1 and 2) showed a high degree of similarity with other members in their group but varied in downstream sequence. Consensus group 2 stocks could be differentiated from consensus group 1 stocks based on sequence variation at positions 51 (in the upstream region, see i, below) and 53 (at the beginning of the microsatellite repeat, see ii, below). Owing to the high level of conservation in these regions, this variation was assumed to be more significant than downstream variation in separating the stocks within consensus groups 1 and 2. Consensus group 1 contained all L. V. braziliensis stocks (identified using IEA) bar 3 with inferred AC01 genotype 227 / 231 (from Genotyper® data). Consensus group 2 contained all putative L. V. braziliensis / peruviana hybrid stocks with the genotype 227 / 231 plus the 3 L. V. braziliensis stocks. Sequence variation will be considered in 3 sections: i) sequence upstream of the dinucleotide (microsatellite) repeat (bases 1-57 of the alignment shown in figure 6.4.4.1); ii) the microsatellite repeat unit (bases 58-91) and iii) sequence downstream of the dinucleotide repeat (bases 92-165).

i) sequence upstream of the dinucleotide (microsatellite) repeat (bases 1-57)

In agreement with AC01 sequencing results from stocks in other populations (chapters 4 and 5), there was little sequence variation upstream of the dinucleotide repeat region (bases 1-57) in Huånuco stocks: 4 different upstream sequences could be identified. One stock (LC2679), identified using IEA as *L. V. lainsoni*, had an identical sequence in this region to that of the *L. V. lainsoni* reference strain M6426. All 3 IM2832-type stocks (*HR419, HR529 and HR701) had a "C" at position 28, a feature identified in all non-*L. V. braziliensis* complex isolates. Many stocks (21/59: LC2484, HR78, LC2422 group [3], LC2452 group [3] and consensus [14]) had upstream sequence identical to that of the *L. V. braziliensis* reference strain LTB300. The remaining stocks (30/59: LC2551 group [3] and consensus 2 [27]) had a double peak at position 51 with "A" and "G" scored (R). Strains used as additional *L. V. peruviana* references were, like the *L. V. braziliensis* reference strain LC39, identical in upstream sequence to that of the *L. V. braziliensis* reference strain LTB300.

ii) the microsatellite repeat unit (bases 58-91)

Among the Huanuco stocks, the number of "TG" repeats scored ranged from 5 - 13 (in blue type). The L. V. lainsoni stock LC2679 had 5 dinucleotide repeats, half the number observed in the L. V. lainsoni reference strain M6426. All 3 IM2832-type stocks (*HR419, HR529 and HR701) had 7 repeats. Both of these repeat numbers (5 and 7) are lower than any described by Russell et al. (1999) for this locus. Of the additional L. V. peruviana reference strains examined, 3 (LC26, LRP709 and LRP906) had 10 repeats, the same number as the L. V. peruviana reference strain LC39; 2 others however, (LC1152 and LRP737) had 13. Both 10 and 13 repeats were commonly scored in L. V. peruviana isolates by Russell et al. (1999). Among the Huanuco stocks identified as L. V. peruviana, only 1 stock (HR78) had 13 repeats; none had 10. A single L. V. peruviana stock (LC2484) had 8 repeats: this number of repeats is associated with I., V. braziliensis (chapters 4 and 5 here, and Russell et al., 1999). Similarly, a single stock identified as L. V. braziliensis (LC2873) and a single hybrid stock (LC2422) had 13 repeats. These stocks will be discussed further in section 6.4.4.3, below. Of the remaining stocks, all had 2 overlapping repeat sequences which could generally be interpreted in terms of 8 and 9 repeats, 8 and 10 repeats or 8 and 13 repeats (see figure 6.4.4.2, below).

iii) sequence downstream of the dinucleotide repeat (bases 92-165)

For stocks homozygous for AC01, the dinucleotide repeats were followed by a highly conserved motif of "TCCGC (red type)." This motif was used to denote the start of the downstream flanking region. All stocks in the HR419 group (IM2832 type) were identical to the *L. V. braziliensis* reference strain, LTB300, in the downstream region. A single stock, LC2484, assigned as *L. V. peruviana* using IEA, differed from the *L. V. braziliensis* reference strain, LTB300, at only 3 positions. The *L. V. lainsoni* stock, LC2679, was similar, but not identical, to the *L. V. lainsoni* reference strain M6426. *L. V. peruviana* stocks in the LC39 group (n = 10) were all identical and differed from the *L. V. braziliensis* reference strain, LTB300, in 5 downstream bases. Stocks in the LC2422 group, LC1152 and LRP709 (all n = 13, all identified as *L. V. peruviana*, bar LC2873 in the LC2422 group) were remarkably similar in the downstream region, with the same heterozygous peaks usually scored at the same nucleotide positions. Another *I. V. peruviana* stocks, with n = 13, HR78, shared less overall similarity with other n = 13 stocks, however the same nucleotide positions were often variable.

Figure 6.4.4.2: Dinucleotide repeat region illustrating the overlapping sequence reads in stocks with 2 distinct alleles.

i) Stocks with 2 alleles containing 8 and 10 dinucleotide repeats

<u>§</u> §	<u>p. (n)</u>	allele		sequence			
b	(8)	LTB300 group	1	GTCTGT	GTGTGTGTGT	GTGTCCGCTT	CAGTGGGCCG
pe	(10)	LC39 group	2	GTCTGT	GTGTGTGTGT	GTGTGTGTCC	GC TTCAGTGG
h	(8/10)	EXPECTED	3	GTCTGT	GTGTGTGTGT	GTGTSYGYYY	SMKTSRGYSG
		consensus group 1				• • • • • • • • • • •	v.
		consensus group 2		R	s		

ii) LC2551 group stocks (alleles containing 8 and 13 dinucleotide repeats)

9 <u>Sp (n)</u>	allele		sequence			
b (8)	LTB300 group	1	GTCTGT	GTGTGTGTGT	GTGTCCGC TT	CAGTGGGCCG
pe (13)	LC1152 group	2	GTCTGT	GTGTGTGTGT	GTGTGTGTGT	GTGTCCGCYW
h (8/13)	EXPECTED	3	GTCTGT	GTGTGTGTGT	GTGTSYGYKT	SWGTSSGCYD
	LC2551 group		R			CK

iii) LC2452 group stocks (alleles containing 9 and 10 dinucleotide repeats)

§ <u>Sp. (n</u>)	allele		sequence			
? (9)	?	1	GTCTGT	GTGTGTGTGT	GTGTGTCCGC	TTCAGTGGGC
pe (10)	LC39 group	2	GTCTGT	GTGTGTGTGT	GTGTGTGTCC	GCTTCAGTGG
? (9/10)	EXPECTED	3	GTCTGT	GTGTGTGTGT	GTGTGTSYSC	KYYWSWGKGS
	LC2452 group			•••••	• • • • • • • • • • •	•••••

Notes to figure 6.4.4.2 dinucleotide repeat region, illustrating the overlapping sequence reads in stocks with 2 distinct alleles. The **EXPECTED** sequence of heterozygous stocks (i.e. having 1 copy each of the alleles in rows 1 & 2) is given in row 3 of each example 1) – iii) (see Appendix IV for base coding system). The sequences observed in heterozygous stocks are compared to this expected sequence. Dots indicate that an identical base was scored to that expected. §Sp. species abbreviation: b - braziliensis, pe - peruviana, h - putative braziliensis / peruviana hybrid, ? - unknown. Species / type designation determined using IEA. Consensus groups 1 and 2 are explained in the text. (n): number of dinucleotide repeats, bases in red indicate the TCCGC motif used to identify the end of the dinucleotide repeat region, bases in green indicate bases which are invariant despite the overlapping sequence reads.

In 4 Huánuco groups (consensus group 1, consensus group 2, LC2551 group and LC2452 group), significant variation was scored in the downstream flanking region owing to the superimposition of sequence generated from 2 distinct alleles. The number of dinucleotide repeats (blue type) contained in each allele was determined by identifying the TCCGC end motif in each (red type, see *figure 6.4.4.2*). It was possible to align the mixed sequences based on sites which were invariant (in green type), despite the overlap.

Figures 6.4.4.3 and 6.4.4.4 illustrate the expected sequence that would be presented by stocks with alleles encoding AC01 products with 8 / 10 and 8 / 13 dinucleotide repeats, respectively. This expected sequence was then compared with observed sequence(s). For stocks in consensus groups 1 and 2, the overlapping sequences could be explained by the presence of 2 alleles, one with 8 dinucleotide repeats and the other with 10 (see *figure 6.4.4.3*). For stocks in the LC2551 group, the overlapping sequences could be explained by the presence of 2 alleles, one with 8 dinucleotide repeats and the other with 10 (see *figure 6.4.4.3*). For stocks in the LC2551 group, the overlapping sequences could be explained by the presence of 2 alleles, one with 8 dinucleotide repeats and the other with 13 (see *figure 6.4.4.4*). For these 3 groups, the 2 alleles had been previously scored from Peruvian stocks and reference strains. Differences between expected and observed sequences of heterozygous stocks may be explained by the fact that the 2 sequences used as "parental" 8 and 10 dinucleotide repeat-containing alleles were from strains not originating from Huanuco.

In one group of stocks (LC2452 group), significant downstream variation was observed which, like the 3 previous groups, suggested overlapping sequence from alleles containing different numbers of repeats. For the LC2452 group stocks, however, 9 dinucleotide repeats were scored unequivocally: this number of repeats had not been scored among any of the stocks examined previously, although Russell *et al.* (1999) recorded this allele in *Viannia* stocks (scored once each from *L. V. braziliensis*, *L. V. peruviana* and *L. V. panamensis* isolates from Bolivia, Peru and Colombia, respectively). By identifying the TCCGC end motif, the other allele in LC2452 group stocks was found to contain 10 repeats (see figure 6.4.4.2).

Species	Strain / stock	10	20		40	50	60		80
b (0)	TR200 group	ATCTCCCTCT	i دردم		TOTTCOTCOT	TTCCCTCTCTCT	CTCTCT	CREREDERE	CECECCCCE
U (0)	L 1 B300 group	AIGIGEETET	CONCOLIA	GIGCIIGICI	ICIICCIGCI	Induction	GIC 1GI	010101010101	GIGICCGCII
pe (10)	LC39 group	ATGTGCCTCT	CCCACCCTTA	GTGCTTGTCT	TCTTCCTGCT	TTGCCTCTCT	GTCTGT	GTGTGTGTGT	GTGTGTGTCC
h (8/10)	EXPECTED	ATGTGCCTCT	CCCACCCTTA	GTGCTTGTCT	TCTTCCTGCT	TTGCCTCTCT	GTCTGT	GTGTGTGTGT	GTGTSYGYYY
	consensus group 1								
	consensus group 2	• • • • • • • • • • •	•••••				R	s	
		90	100) 11() 120) 13(0 140) 150	2
		ł	1	1	1		I	1	
b (8)	LTB300 group	CAGTGGGCCG	ATCCGTTTCA	CTTTTTGCCG	GTGACGTTTG	TGTGCTGACG	TGTCTGGTGG	CCTCWCA	
pe (10)	LC39 group	GCTTCAGTGG	GCCGATCCGT	TTCACTTTTT	GCCGGTGACG	TKTGTGTGCW	GACGTGTCKG	GTGGCCWCAC	ł
h (8/10)	EXPECTED	SMKTSRGYSG	.RYCSRTYYSW	YTYWYTKYYK	GYSRSKKWYG	TKTGYKKRCD	KRYSTGKYKG	SYKSHCWCACA	ł
	consensus group 1	v.		CGG	D.T	.GT.A	VG.	TV.HA	
	consensus group 2		M	CGG	A.D.T	.GA	VG.		

Expected sequences of stocks heterozygous at the AC01 locus for alleles encoding 8 and 10 dinucleotide repeats.

Notes to figure 6.4.4.3 Expected sequences of stocks heterozygous at the AC01 locus for alleles encoding 8 and 10 dinucleotide repeats. Species abbreviation b - braziliensis, pe - peruviana, h - putative braziliensis peruviana hybrid. Species / type designation determined using IEA. Consensus groups 1 and 2 are explained in the text. Numbers above sequence data indicate the base position used for alignment purposes. Bases in blue type indicate the dinucleotide repeats, bases in red type indicate the TCCGC motif used to identify the end of the dinucleotide repeat region; bases in green type indicate bases which are invariant despite the overlapping sequence reads. See Appendix IV for base coding system.

248

Figure 6.4.4.3

Chapter 6 Huanuco, Peru

Figure 6.4.4.4	Expected sequences of stocks heterozy	gous at the AC01 locus for alleles encoding 8 and 13 dinucleotide repeats
----------------	---------------------------------------	---------------------------------------------------------------------------

Species 1	Strain / stock	10	20	30	40	50	60	70	80
		1	1	(1	1	(1	1
b (8)	LTB300 group	ATGTGCCTCT	CCCACCCTTA	GTGCTTGTCT	TCTTCCTGCT	TTGCCTCTCT	GTCTGT	GTGTGTGTGT	GTGTCCGCTT
pe (13)	LC1152 group	ATGTGCCTCT	CCCACCCTTA	GTGCTTGTCT	TCTTCCTGCT	TTGCCTCTCT	GTCTGT	GTGTGTGTGT	GTGTGTGTGT
h (8/13)	EXPECTED	ATGTGCCTCT	CCCACCCTTA	GTGCTTGTCT	TCTTCCTGCT	TTGCCTCTCT	GTCTGT	GTGTGTGTGT	GTGTSYGYKT
	LC2551 group						R.,	s	

		90	100	110	12	1 13	7 140	1 15	5 100
		t.	1	I.	1	I.	1	1	
b (8)	LTB300 group	CAGTGGGCCG	ATCCGTTTCA	CTTTTTGCCG	GTGACGTTTG	TGTGCTGACG	TGTCTGGTGG	CCTCWCA	
pe (13)	LC1152 [†]	GTGTCCGC YW	CAGTGGGCMG	ATCCGTYTCA	CTTTKTGCSG	GWGACGTKTG	TGTGCWGACG	TGTCKGGTGG	CCWCACA
h (8/13)	EXPECTED	SWGTSSGCYD	MWSYGKKYMR	MTYYKTBYCR	STKWBKKYBG	KDKRCKKDYG	TGTSYDGWSG	YSTCDSRTGG	CCWCACA
	LC2551 group	ск	CA	G	CGY.	A		V.K	.M.MW

Notes to figure 6.4.4.4. Expected sequences of stocks heterozygous at the AC01 locus for alleles encoding 8 and 13 dinucleotide repeats. Species abbreviation b - braziliensis, pe - peruviana, h - putative braziliensis / peruviana hybrid. Species / type designation determined using IEA. Numbers above sequence data indicate the base position used for alignment purposes. Bases in blue type indicate the dinucleotide repeats, bases in red type indicate the TCCGC motif used to identify the end of the dinucleotide repeat region; bases in green type indicate bases which are invariant despite the overlapping sequence reads. See Appendix IV for base coding system.

Chapter 6 - Huámuco, Peru

6.4.4.3 Comparison of AC01 data from sequencing and Genotyper®

Table 6.4.4.3 compares the number of dinucleotide repeats scored from sequence data with Genotyper results. For the majority of Huanuco stocks and additional L. V. peruviana reference strains, the number of dinucleotide repeats scored from sequence analysis correlated exactly with allele size scored from Genotyper analysis (assuming an allele size of 227 equates to 8 dinucleotide repeats, see chapter 3). This correlation was all the more impressive since the majority of stocks were scored as heterozygous using Genotyper D. For example, all L. V. braziliensis stocks scored using Genotyper as 227 / 231 were found to have overlapping sequence reads which could be interpreted as 2 alleles differing by 2 repeats (i.e. 8 and 10), giving allele sizes which varied by 4 base pairs.

For several stocks, small discrepancies were noted. As discussed previously (chapters 3 - 5), these could have arisen from insertions or deletions ('indels'') in the extreme 5' sequence (not used in the alignments), thus giving rise to Genotyper® sizes larger or smaller, respectively, than expected by extrapolation from the number of dinucleotide repeats. This explanation could possibly account for the observed disparity between methods for the *L. V. peruviana* stocks LRP737 and HR78 where 13 repeats were scored from sequence data (expected allele size 237) but where Genotyper® results indicated an allele size of 235 (? 2 base deletion). Similarly, in stocks scored using Genotyper® as heterozygous for ACO1, a small indel in one of the 2 alleles could account for the discrepancy between the 2 methods. For example, in stocks of the LC2542 group, Genotyper® analysis scored the stocks as 229 / 233 (suggesting alleles containing 9 and 11 repeats) whereas sequence data indicated that 9 and 10 repeats were in fact, present - equivalent to a 2 base insertion in the 233 allele.

In a few stocks, however, there were significant differences between the 2 methods in which large indels would be required to explain Genotyper[®] data or where Genotyper[®] results indicated a stock heterozygous for AC01 but where sequence data suggested a single allele. For example, a 14 base pair insertion would be required to account for the disparity between these methods for the *L. V. peruviana* stock LC2484 (Genotyper[®] peak of 241, expected number of repeats = 15: 8 repeats were scored by sequence analysis). For the Huánuco *L. V. lainsoni* stock LC2679, 5 repeats were repeatedly scored by sequence analysis suggesting a single allele (Genotyper[®] data of 229 / 231 suggesting 2 distinct alleles, inferring 9 and 10 dinucleotide repeats). The

Chapter 6 – Huámuco, Peru

significant discrepancies are difficult to explain. These may be the result of labeling errors: all stocks concerned should be re-examined by both techniques to confirm this.

Table 6.4.4.3:	Comparison of the number of dinucleotide repeats scored
	from sequence data with Genotyper® results.

Strain, stock or group from	No. stocks	Species designation	No. of dinucleotide	AC01 Genotyper®
sequencing data [†]		using IEA§	repeats scored	data±
M6426 [†]	-	1	10	231 / 231
LC2679	1	1	5	229 / 231
HR419 group	3	IM2832 type	7	225 / 225 (1) 235 / 235 (1) failed (1)
LTB300 [†]	-	b	8	227 / 227
consensus group 1	14	b	8 / 10	227 / 231
LC2452 group	3	b	9 / 10	229 / 233
LC2551 group	3	b	8 / 13	227 / 237
LC2873	1	b	13	237 / 237
consensus group 2	27	b / pe	8 / 10	227 / 231 227 / 233 (1) 229 / 231 (1)
LC2422	1	b/pe	13	237 / 237
LC2484	1	pe	8	241/241
LC39 group [†]	4	pe	10	231 / 231 (3) 231 / 235 (1)
LC1152 [†]	1	pe	13	237 / 237
LRP737 [†]	1	pe	13	235 / 235
HR78	1	pe	13	235 / 235

[†] indicates a reference strain. Consensus groups 1 and 2 are explained in the text. Species abbreviations: b - braziliensis; pe - peruviana, b / pe - putative braziliensis / peruviana hybrid; 1 - lainsoni; IM2832-type - potential new Viannia species (Grimaldi et al., 1991). [±] numbers in parentheses indicate the number of stocks / strains with the given Genotyper@ data.

6.4.4.4 Comparison of microsatellite and isoenzyme data

Two Huanuco stocks had conflicting results by IEA and AC01 microsatellite locus analysis which need to be readdressed to eliminate the possibility of an error. The stock LC2873, identified as *L. V. braziliensis* using IEA, was found to be highly similar to *L. V. peruviana* stocks and strains both in the number of dinucleotide repeats

scored (13), DNA sequence and in Genotyper® allele size (237). This number of repeats was not found to be associated with *L. V. braziliensis* by Russell *et al.* (1999), although they examined relatively few isolates.

Another stock, LC2484, was identified using IEA as L. V. peruviana but found to have just 8 dinucleotide repeats from sequence analysis, a repeat number associated with L. V. braziliensis (here, Russell et al., 1999). However, this stock generated a Genotyper® peak of 241 suggestive of 15 repeats; L. V. peruviana stocks generally have greater numbers of repeats at the AC01 locus compared to L. V. braziliensis stocks (here, Russell et al., 1999), however 15 repeats have only been described in L. V. panamensis stocks (again, here, Russell et al., 1999).

6.5 Phylogenetic analysis of Huánuco stocks

Phylogenetic analysis of isoenzyme data (multilocus data) included data from all *Leishmania* reference strains. Phylogenetic analysis of sequence data from the microsatellite locus AC01 included data from all reference strains of the *Viannia* subgenus, with the exception of *L. V.* sp. n. reference strain IM2832 which failed to produce readable sequence data. The AC01 microsatellite locus is specific to the *Viannia* subgenus hence it was not possible to include reference strains from the subgenus *Leishmania* in this phylogenetic analysis.

6.5.1 Phylogenetic analysis of isoenzyme data

Isoenzyme phylogenies were constructed, as described in chapter 2, using a maximumlikelihood approach, quartet puzzling (Strimmer & von Hassler, 1996). Figure 6.5.1.1 shows the consensus cladogram constructed after quartet puzzling analysis of isoenzyme data from *Leishmania* reference strains and from Huanuco stocks. Stocks from Huanuco spanned 3 Viannia species (L. V. braziliensis, L. V. peruviana and L. V. lainsoni) plus a potentially new Viannia species (L. V. sp. n.: Grimaldi et al., 1991), and included putative L. V. braziliensis / L. V. peruviana hybrids.

Using this approach, the Huánuco L. V. lainsoni stock LC2679 clustered with the reference strain for this species (M6426), although the former had a distinct zymodeme (LON228 and LON206, respectively). The status of the potential new Viannia species (type and reference strain IM2832, LON205) as distinct from other members of this subgenus is not supported by this analysis. IM2832 differed from the

Chapter 6 - Huamuco, Peru

L. V. braziliensis reference strain LTB300 at 9/12 loci: the 3 Huanuco stocks (HR419, HR529 and HR701, LON227), which were found to be identical to IM2832 at 6 of these 9 loci, did, however, cluster separately from other stocks (73% support). It seems likely that the data for IM2832 was incorrectly entered for this analysis (compare the position of IM2832 in this analysis with its position in *figure 6.5.1.2*).

Reference strains of the L. V. guyanensis complex (L. V. guyanensis, M4147; L. V. shawi, M15065 and L. V. panamensis, LS94) formed a clade (68% support): the close relationship reported between L. V. shawi and L. V. guyanensis (Shaw et al., 1991) is supported here, by 92% of quartets.

All strains and stocks identified as *L.V. peruviana* were identified in a discrete clade (77% quartet support), with the *L. V. peruviana* stocks from Huanuco (LON217) separate from the reference strains (LON201) of this species (83% support).

The L. V. braziliensis (LON225) and putative hybrid (LON218) zymodemes were supported by less than 50% of quartets: stocks belonging to each of these zymodemes thus have separate (collapsed) branches and do not appear as a discrete cluster. Several clusters can, however, be seen for other Huanuco L. V. braziliensis and putative hybrid stock zymodemes. Stocks from L. V. braziliensis zymodeme LON223 forms a discrete cluster (76% quartet support). Two L. V. braziliensis zymodemes, LON222 and LON224, form a clade (76% support) with stocks from each zymodeme clustering as a group (79% and 74% support, respectively). Two putative hybrid stocks (LC2435 - LON220 and LC2877 - LON221) also formed a clade (99% support). The separate standing of these stocks is likely due to the absence of bands at 2 enzyme loci (see section 6.3).

Figure 6.5.1.1 (next page): Cladogram showing the results of quartet puzzling for Huánuco stocks and reference strains based on isoenzyme data. Values on branches are "quartet supports", figures which are similar to bootstraps (Strimmer & von Haeseler, 1996). These values are the percentage support of each branch, representing the number of quartets (n = 1000) supporting each given branching order. Branches with less than 50% support have been collapsed. Branch lengths are arbitrary. Zymodemes and species designations are indicated. "h" indicates stocks which were putative braziliensis | peruviana hybrids. "" indicates LON219 (HR410) and LON226 (LC2873).



As with stocks from previous chapters, isoenzyme data from the Huanuco stocks was also analyzed using a similarity approach. Dendrograms were constructed by UPGMA using the SYN-TAX-pc package (Podani, 1993). Figure 6.5.1.2 shows the dendrogram constructed for these stocks from a similarity coefficient matrix. Stocks are classified according to similarity (range 0 - 1).

Greater resolution is obtained using this analysis with each Viannia species-complex being identified. Similarly to its position in figure 6.5.1.1, the L. V. lainsoni stock LC2679 clusters with the reference strain M6426: using this analysis, however, these stocks show more similarity to, and cluster with, reference strains from the Leishmania subgenus. Reference strains of the L. V. guyanensis complex form a discrete cluster, again with L. V. shawi and L. V. guyanensis sharing more similarity than with L. V. panamensis.

In contrast to its position in *figure 6.5.1.1*, the reference strain and potential new *Viannia* species IM2832 (*L. V.* sp. n., Grimaldi *et al.*, 1991) clearly forms a distinct group, together with the Huanuco dog stocks HR419, HR529 and HR701. This group clusters between the *L. V. braziliensis* and *L. V. guyanensis* species complexes.

The majority of Huanuco stocks were identified as belonging to the L. V. braziliensis complex, i.e. were L. V. braziliensis, L. V. peruviana or putative hybrids thereof. Using the UPGMA approach, based on IEA data, all stocks from each zymodeme were shown to cluster together. Four major clades are evident: clades 1 and 4 contain L. V. braziliensis stocks (comprising stocks from 4 and 2 zymodemes, respectively), clade 3 contains L. V. peruviana stocks (from 2 zymodemes) and clade 3 contains all putative braziliensis / peruviana hybrid stocks (from 4 zymodemes). The L. V. braziliensis clade 4 (containing stocks from zymodemes LON225 and LON226) falls outside the clades containing the reference strains of both L. V. braziliensis (LON200) and L. V. peruviana (LON201).

Figure 6.5.1.2 (next page): UPGMA dendrogram constructed from the similarity coefficient matrix for Huanuco stocks and reference strains based on isoenzyme data. Similarities between zymodemes (y-axis, range 0-1) were calculated, using SYN-TAX-pc (Podani, 1993), by averaging the Jaccard coefficients calculated from data from twelve enzyme loci. Stocks / strains marked with * are reference strains. Zymodemes (LON numbers) are indicated on branches. "h" indicates putative *braziliensis / peruviana* hvbrid stocks.

Chapter 6 - Huamuco, Peru

Banuls *et al.* (1999), based on data from IEA and RAPD analysis, argue that the species status of *L. V. peruviana* is valid. In their analyses, all *L. V. peruviana* stocks and strains clustered into a discrete typing unit (DTU) which showed "a clear structuring between the 2 species". In this analysis, based on IEA data, *L. V. peruviana* strains and stocks, whilst forming a discrete clade (clade 3), did so nestled amongst the *L. V. braziliensis* clades (1 and 4). This observation does not, therefore, support their suggestion that *L. V. peruviana* is an indisputable species in its own right.

6.5.2 Phylogenetic analysis of sequence data from AC01

Molecular phylogenies for sequence data from the AC01 microsatellite locus were constructed as previously described. *Figure 6.5.2* shows an unrooted phylogram of the neighbour-joining consensus tree for the Huanuco stocks and reference strains based on sequence data from the microsatellite locus AC01. The reliability of tree topologies was assessed by the bootstrap method (n = 1000).

The interpretation of this phylogram in meaningful terms is difficult bearing in mind discussions from previous chapters regarding the applicability of current software for analyzing sequence data from microsatellite loci. Nevertheless, several clusters are evident which will be discussed briefly.

The phylogram was drawn using the L. V. lainsoni reference strain M6426 as an outgroup. L. V. lainsoni was used as the most distinct species in the Viannia subgenus (the PCR primers used to amplify this microsatellite locus do not amplify subgenus Leishmania stocks or parasites from other genera (Russell et al., 1999). The Huanuco L. V. lainsoni stock, LC2679, grouped closely with the reference strain in cluster VII.

Figure 6.5.2 (next page): unrooted phylogram of the neighbour-joining consensus tree for the Huanuco stocks and reference strains based on sequence data from the microsatellite locus AC01. Numbers indicate the percentage of bootstrap support for each branch. The scale bar represents 0.001 (0.1%) nucleotide changes per sequence position. Code names in bold type indicate reference strains. Numbers in parentheses indicate the number of dinucleotide repeats (n) scored (for the reference strains of L V. guyamensis, M4147, and L. V. shawi, M15065, the repeats are not perfect). The species designations for, and explanation of, clusters are given in the text.





The three dog stocks, HR419, HR529 and HR701, shown using IEA to be closely related to the L. V. sp. n. reference strain IM2832, cluster together (cluster VI) and, like L. V. lainsoni stocks, are distinct from strains and stocks of the two main Viannia subgenus.

Using this analysis, two of the reference strains from the L. V. guyanensis complex (panamensis - LS94 and shawi - M15065) are distinct from members of the L. V. braziliensis complex, however, the guyanensis reference strain (M4147) nestles between L. V. peruviana stocks and the majority of Huanuco stocks. From looking at the AC01 sequence alignment (figure 6.4.4.1), this is likely due to ambiguity in the downstream sequence of all of these stocks.

Cluster I contains 2 L. V. braziliensis reference strains (LTB300 and M2903) which, like the vast majority of L. V. braziliensis stocks, had alleles with n = 8. As discussed previously, the only Huanuco stock with n = 8 was LC2484, which was identified using IEA as L. V. peruviana and not L. V. braziliensis. The vast majority of Huanuco stocks grouped together in cluster III. This cluster contained all stocks which were heterozygous for alleles containing 8 and 10 dinucleotide repeats. As discussed previously, the stocks in this cluster were identified using IEA as L. V. braziliensis (18) and putative L. V. braziliensis / L. V. peruviana hybrids (23). Cluster II contains all 3 LON222 stocks which were heterozygous for AC01 with alleles containing 8 and 13 dinucleotide repeats. The remaining Huanuco stocks and all L. V. peruviana reference strains formed a distinct group outside of clusters II and III. All Huanuco stocks with an AC01 allele where n = 13 were grouped together in cluster IV and were most closely related to the 2 L. V. peruviana reference strains (LC1152 and LRP737) which also had n = 13. As discussed previously, only one of these 3 stocks had been identified as L. V. peruviana using IEA. The final group of Huanuco stocks formed cluster V which comprised all stocks belonging to LON223 (L. V. braziliensis) which each had 2 AC01 alleles containing 9 and 10 dinucleotide repeats.

Thus, the relationships between Huanuco stocks from the phylogenetic analysis of AC01 microsatellite sequence data shows that these stocks comprise a complex population. Several groupings (i.e. clusters II, V and VI) however, correlate with those identified using IEA data and hence are likely "true" groups of stocks. Some stocks did not, however, cluster with the same stocks as had been observed from the

Chapter 6 - Huamuco, Peru

phylogenetic analysis of IEA data. These observations may be explained by several points. Firstly, as reported by Russell *et al.* (1999), these microsatellite markers do not always correlate with *Leishmania* species status as determined using IEA. Secondly, there is no reason why microsatellite alleles should be homozygous in an individual homozygous for enzyme loci, or indeed vice versa. Lastly, as discussed previously, the phylogeny constructed from the analysis of microsatellite data is likely to be inherently inaccurate owing to the lack of an appropriate and applicable model for such loci.

Other molecular markers have been used to construct phylogenies of Viannia stocks in which L. V. peruviana formed a discrete DTU [karyotyping, Dujardin et al., 1993b, 1995b; gp63 gene analysis, Victoir et al., 1998; RAPD analysis, Dujardin et al., 1995b and Banuls et al., 1999; 2000]. However, the Huanuco stocks have only recently become available and hence could not have been included in previous studies. It would be interesting to see the phylogenetic positions of these stocks, particularly those of zymodernes LON225 and LON226, using data from the molecular markers and techniques employed by the aforementioned authors.

Chapter 6 - Huamuco, Peru

6.6 Population genetic analyses

It has been proposed that *Leishmania* have a clonal population structure, with sexual reproduction rare or absent in natural populations (Tibayrenc *et al.*, 1990). To test this hypothesis, data from the Huanuco stocks were examined for evidence of segregation and recombination, basic consequences of sexual reproduction, using standard tests. The validity of using these approaches for the analysis of data from these organisms is discussed, in context with the results, in section 6.6.3.

6.6.1 Test for Hardy-Weinberg equilibrium

Observations of deviations from Hardy-Weinberg (HW) equilibrium, fixed heterozygosity and "missing" genotypes indicate the absence of possible segregation and therefore suggest a clonal population structure. Here, Hardy-Weinberg (HW) exact tests were used to provide evidence for segregation at 10 loci: HW analysis was not possible at 4 enzyme loci as they were monomorphic for all stocks, HW analysis was not carried out on data from the microsatellite locus AC52 because data were missing from many stocks. Exact P-values describing departure from HW equilibrium (panmixia) were calculated as described in chapter 2, by comparing observed and expected genotype frequencies.

Observed and expected allele and genotype frequencies for the Huánuco stocks based on IEA and microsatellite data, respectively, are given in *tables V-1* and *V-2 (Appendix* V). The Huánuco stock identified as *L. V. lainsoni*, LC2679, was excluded from all analyses as it was distinct from the other stocks at almost every enzyme locus. The three stocks identified as IM2832-type (HR419, HR529 and HR701), proposed to be a new *Viannia* species (Grimaldi *et al.*, 1991), were included in the analysis because the alleles scored for these stocks were identical to those of the *L. V. braziliensis/L. V. peruviana* stocks at 2 polymorphic loci (NHi1 and NHd) and 4 monomorphic loci (NHi2, 6PGD, ASAT and ALAT). The polymorphism observed at 3 loci (PGM, GPI and G6PD) was due entirely to these stocks, which were homozygous for a different allele to that of the *L. V. braziliensis / L. V. peruviana* stocks. At 2 loci, ES and PEPD, the IM2832-type stocks were homozygous for a different allele to that observed in *L. V. braziliensis / L. V. peruviana* stocks homozygous for these loci. However, many *L. V. braziliensis / L. V. peruviana* stocks were heterozygous for these loci, with one allele apparently from the IM2832-type stocks. *Table 6.6.1* summarizes the results of

Chapter 6 - Huámuco, Peru

HW exact tests, by locus, for IEA and microsatellite data from these stocks. For each locus, an exact P-value is given to describe the probability of departure from HW equilibrium having occurred by chance. In addition, the observed and expected numbers of heterozygotes are shown along with a calculation of the fixation index (Fis, Weir & Cockerham, 1984).

Table 6.6.1. Summary of Hardy-Weinberg exact test results[†], by locus, for IEA and microsatellite data from Huánuco stocks[±]

Locus	N	A	Deviation	exact	Heterozygotes		Fis*	
			from HW [†]	P-value	Observed	Expected	(W & C)	
MPI	58	5	No	0.5212	32	28,287	-0.133	
NHi1	_58	2	Yes	0.0010	35	24.652	-0.425	
NHd	58	2	No	0.0508	26	20.348	-0.281	
ES	58	2	No	0.4035	16	17.983	+0.111	
PEPD	58	2	No	0.7115	19	19.783	+0.040	
PGM	58	2	Yes	0.0000	0	5.740	+1.000	
GPI	58	2	Yes	0.0000	0	5.740	+1.000	
G6PD	58	2	Yes	0.0000	0	5.740	+1.000	
AC01 ^M	53	8	Yes	0.0000	47	36.581	-0.288	
AC16 ^M	52	10	Yes	0.0000	18	37.854	+0.527	
NHi2	58	1						
6PGD	58	1	only one allele, analysis not possible					
ASAT	58	1						
ALAT	58	1						

± analysis excludes L. V. lainsoni stock LC2679.

N Number of stocks analyzed for locus

A Number of alleles scored at locus from all stocks

- <u>HW exact tests</u> results were computed using Genepop (Raymond & Roussett, 1995), see Chapter 2 for details. Deviation from HW equilibrium assumes a 5% confidence interval (i.e. P < 0.05). P-values: a complete enumeration was determined for loci with up to 4 alleles and an estimation of exact P-value was determined by the Markov chain method for loci with more than 4 alleles.
- **A**

Fis:

Within-population fixation index:

<u>He - Ho</u> He

where He is the expected number of heterozygotes and Ho is the observed number of heterozygotes. Range of Fis = -1 > 0 < +1. +1 indicates that all stocks were homozygous at that locus. A negative value indicates an excess of heterozygotes. Fis computed, using Genepop, as in Weir & Cockerham (W & C, 1984).

Clonal reproduction, by definition, generates individuals identical to the original clone (assuming no mutation). Observations of: 1. deviation from Hardy-Weinberg (HW) equilibrium, 2. fixed heterozygosity and 3. the absence of possible segregation types are all evidence of clonal expansion. Results from these tests are discussed by the above points:

1. Deviation from Hardy-Weinberg (HW) equilibrium:

Analysis of the Huanuco stocks (10 loci, were amenable to statistical testing, see *table* 6.6.1) has shown that 3 enzyme loci (MPI, ES and PEPD) showed no deviation from Hardy-Weinberg (HW) equilibrium (P > 0.05) suggesting segregation of alleles at these loci. 1 enzyme locus (NHd) showed marginal deviation at the 5% level. The remaining 6 loci, however, (4 enzyme and both microsatellite loci) showed considerable deviation (P < 0.001) from HW equilibrium indicating that alleles at these loci were not segregating, an observation suggestive of clonal propagation.

2. Fixed heterozygosity:

Heterozygosity appears not to be totally fixed. Although several multilocus genotypes were sampled repeatedly (suggestive of clonal expansion), many other combinations were also identified, albeit less frequently. In addition, not all heterozygotes were identical. An unusual finding in this population is the observation of excess heterozygotes (denoted by a negative Fis value). Excess heterozygote form. This finding will be discussed further in section 6.6.3.

3. The absence or lack of possible segregation genotypes.

The absence or lack of possible segregation genotypes (see table V-1, Appendix V) was evident but particularly at loci where more than 2 alleles were scored (and therefore particularly at microsatellite loci). As a result of large numbers of alleles being scored, the expected numbers of many genotypes at several loci were very small. Given that some alleles (enzyme and microsatellite) were rare and that a relatively small number of stocks were analyzed, it is not surprising that some genotypes were not sampled. Thus, the lack of all possible segregation genotypes here is not necessarily the result of a lack of segregation, but possibly due to the small sample size.

Chapter 6 - Huanuco. Peru

6.6.2 Tests of linkage disequilibrium

Tests of linkage disequilibrium (LD, non-random association of alleles) between alleles at different loci can be used to provide evidence for recombination. As with segregation, evidence of a lack of recombination suggests that the population is not mating randomly and hence may have a clonal population structure. Here, the occurrence of recombination between loci was examined using three statistical tests for detecting LD: the Index of Association (I_A; Maynard-Smith *et al.*, 1993); the D' index (Lewontin, 1964) and the r^2 (square of the correlation coefficient) index (Hill and Robertson, 1968). The tests differ in that the I_A index examines the data for evidence of LD in the population over all loci, whereas the D' and r^2 indices examine pairwise combinations of loci and hence may be able to pinpoint which loci are involved in generating any LD.

6.6.2.1 Index of Association (I_A)

The Index of Association (I_A) has been proposed as a method for characterizing the population structures of microorganisms (Maynard-Smith *et al.*, 1993). The index is a relatively simple measurement of the degree of association between loci in a given population: I_A is expected to be zero in a large pannictic population. Values of I_A which differ significantly from zero may represent departure due to linkage disequilibrium (LD).

In addition to the I_A value obtained for the population as a whole, Maynard-Smith *et al.* (1993) state that this test can also be used to elucidate underlying population structures. LD may, they argue, be due in part to the presence of sub-groups (which may consist of, for example, discrete taxonomic units or geographically-separated populations) in the population. Linkage disequilibrium observed in the population as a whole may be resolved by calculating the I_A values of these sub-groups. It may be shown from the I_A values of the latter, that one or more subgroups is/are responsible for the LD in an otherwise clonal population. In addition, these authors suggest that the bias due to the presence of over-represented clones can be removed by comparing the I_A value obtained for all individuals in a population with the I_A values obtained from analyzing only the electrophoretic / microsatellite types (ETs or MSTs) observed.

So, according to Maynard-Smith *et al.* (1993), the structure of a given population may be categorized by comparing values of I_A calculated for the whole population, subgroups and electrophoretic types thereof. Different types of population structure are described including "true" clonality, "epidemic" clonality, and panmixia. Organisms exhibiting true clonality generate clones which are stable in space and time, organisms exhibiting epidemic clonality undergo occasional bouts of clonal propagation although the species is basically sexual. Using I_A values, parasitic protozoa have been categorized (see Maynard-Smith *et al.*, 1993) as clonal (*Trypanosoma cruzi*), epidemic (*T. brucei*) and epidemic/panmictic (*Plasmodium falciparum*).

The equations required to calculate I_A values (taken from Maynard-Smith *et al.*, 1993) are given in *Appendix V*, with allele frequencies of whole Huanuco population and of subgroups therein by locus (*table V-3*). I_A values were calculated for all individuals and for 4 sub-groups: sub-groups were based on taxonomic groups identified using IEA (subgroup 1 [SG1] - stocks identified as *L. V. braziliensis*; SG2 - stocks identified as *L. V. peruviana*; SG3 - stocks identified as putative *L. V. braziliensis / L. V. peruviana* hybrids; SG4 - stocks identified as IM2832-type). I_A results are given in detail in *table V-4 (Appendix V)* and summarized in *table 6.6.2.1*, along with I_A values from previous studies involving trypanosomatids.

Table 6.6.2.1: IA values for Huánuco Leishmania stocks and for other trypanosomatids.

Parasite	No. of isolates	IA	Reference
Leishmania Viannia	58	0.97 (all loci) 0.59 (enzyme loci) 2.77 (microsat. loci)	present work
T. cruzi	524	2.63 +/- 0.06	Zhang et al. (1988)
T. brucei	321 73 67 44 142	0.45 +/- 0.007 0.53 +/- 0.003 1.14 +/- 0.002 0.80 +/- 0.009 0.30 +/- 0.00	Mihok et al. (1989) Stevens & Tibayrenc (1996) Stevens & Tibayrenc (1996) Stevens & Tibayrenc (1996) Stevens & Tibayrenc (1996)

Chapter 6 - Huámuco, Peru

The I_A values for the Huanuco Leishmania stocks obtained for enzyme loci and for all loci combined are relatively small and comparable to values obtained for *Trypanosoma brucei* (Mihok *et al.*, 1989; Stevens & Tibayrenc, 1996). *T. brucei* is considered to exhibit epidemic clonality, i.e. undergoing occasional bouts of clonal propagation although the species is basically sexual. The I_A value for microsatellite loci (2) is more comparable to that of *T. cruzi* (Zhang *et al.*, 1988) which is considered to exhibit true clonality, i.e. generating clones which are stable in space and time. The I_A value calculated from microsatellite data for the Huanuco Leishmania stocks is, however, higher owing to the absence of possible genotypes. It is not possible to determine whether these genotypes are actually missing or have not been sampled since the sample size was too small to score all the possible, rare genotypes from these hypervariable loci.

It is noteworthy that different levels of taxonomic speciation are employed by the various researchers of these genera. Strains of *T. cruzi*, for example, are known to be as different genetically as some species of *Leishmania* (Miles *et al.*, 1983). Researchers studying *Leishmania* have tended to liberally impose species status based on distinct clinical presentation and geographical location, despite the fact that the parasites involved are genetically very similar. This discordant level of taxonomic speciation complicates comparison of I_A values between different genera.

In an attempt to pinpoint any groups of stocks responsible for the reduced I_A values, the population was divided into 4 taxonomic subgroups (see *table V-4*, *Appendix V*). The validity of this approach for these stocks is questionable since the sample sizes of subgroups 2 and 4 were very small and contained only 1 electrophoretic type, making further analysis impossible. The overall I_A values for those subgroups which could be analyzed increased from the value calculated for all stocks together, possibly indicating that stocks in each subgroup had a clonal structure but that some exchange had occurred between groups. One of the subgroups (SG3), for example, appeared to be products of a clonal expansion following a recombination event between 2 supposed species (*L. V. braziliensis x L. V. peruviana*).

According to Maynard-Smith *et al.* (1993), it is possible to distinguish "true" and "epidemic" clonality by comparing the I_A values from the whole population with those

Chapter 6 Huamuco. Peru

of the electrophoretic and microsatellite types, thereby eliminating the bias from overrepresented clones. If the I_A value is reduced in the latter it is taken that the organism exhibits epidemic clonality.

Accordingly, I_A values of ETs and MSTs were calculated using enzyme and microsatellite data for the whole population and for each subgroup (see *table-V-4*, *Appendix V*). I_A was found to increase in all but one instance. According to Maynard-Smith *et al.* (1993), in the cases where I_A was found to increase, this would be interpreted as indicative of clonal structure masked in the whole population analysis by a superabundance of clones. In a single case the I_A value decreased when comparing the I_A value of all stocks with that of just the ETs (IEA data, SG1); this suggests that the stocks within this SG are not necessarily clonal but, perhaps the products of genetic exchange event(s).

These results reinforce the note of caution sounded by Stevens and Tibayrenc (1996), regarding the danger of inadvertently combining inappropriate data from distantly related (but apparently similar) groups of stocks. If this analysis is correct (and which is by no means clear to the author), the results indicate that the Huanuco "population" in fact contained several different populations of stocks which might more properly be considered separately. Furthermore, the structure of each subgroup that was amenable to analysis was found, bar one, to be clonal (again, assuming that this analysis is correct), even after the removal of apparently over-represented clones.

6.6.2.2 D' and r^2 indices

The D' index (Lewontin, 1964) and the r^2 (square of the correlation coefficient) index (Hill and Robertson, 1968) provide a more conventional examination for evidence of LD because they compare observed and expected data from pairwise combinations of loci. As a result, both tests can "pinpoint" where linkage, if any, occurs. For a comprehensive explanation and description of D' and r^2 indices, see Hedrick (1985).

The D' index is a measure of the disequilibrium (D) between the most common allele at each of two loci compared to its theoretical maximum (Dmax). D' has a potential range between +/- 1.0. Zero indicates no linkage; non-zero values signify incomplete linkage and 1 denotes complete linkage (i.e. the loci are always inherited together). The + or - value reflects which alleles at the loci were compared. Since more than 2 alleles may be scored at a locus, a negative value may arise if the allele frequency of the most common allele is less than the sum of the remaining allele frequencies.

The regression correlation coefficient, r^2 , provides a more conservative analysis (i.e. approaches 0 or 1 slowly) than D' because r^2 analysis is not skewed by a missing gametic type. The r^2 index takes into account the level of significance (P value) and has a range of 0 to 1.0 (0 indicating no linkage and 1 indicating complete linkage).

Calculations of D' and r^2 values require the identification of all gametic types (as opposed to allelic types for segregation tests at a single locus). It is assumed for these analyses that *Leishmania* are diploid at all loci examined and that all alleles, at each locus, are co-dominant (see introduction). The gametes which make up each of the observed phenotypes can be identified except for the two types present in the double heterozygote which, because the alleles are co-dominant, are indistinguishable (see notes in *Appendix V*). Hence D' and r² calculations are based on maximum-likelihood estimates of gametic frequencies. The equations required to calculate D' and r² values are given in *Appendix V*.

D' and r^2 values were calculated for pairwise combinations of loci. Only those polymorphic loci where the frequency of the most common allele was less than 0.90 were, used for analyses of LD because these loci were highly informative statistically. A test of significance (Q) was calculated for each D score from which probability (P) values were assigned. Results were scored as not significant at a P value of > 0.05 (i.e. more than 5% probability that the result could have occurred by chance). D' and r^2 results are summarized in *table 6.6.2.2* overleaf.

The majority of pairwise combinations (see *table 6.6.2.2*) generated high D' values indicating linkage and suggesting that the inheritance of those locus combinations resulted from clonal reproduction. Seven pairwise combinations, however, showed a deviation (D' < 0.7) from complete linkage (D' approximately = 1), of which only 3 (MPI x NHi1, AC16 x NHi1 and AC16 x NHd) gave values which were statistically-significant. These D' values were, however, still large compared to values reported for LD in *Plasmodium falciparum* (Conway *et al.*, 1999).
Table 6.6.2.2

Summary of linkage disequilibrium results[†], by pairwise locus combination, for IEA and microsatellite data from Huánuco stocks[±]

2-locus combination [†]	N	D,	P value	r ²
MPI ⁶ x NHi1 ²	58	0.687	Sig. P < 0.005	0.383
MPI ⁶ x NHd ⁴	58	1.000	Sig. P < 0.005	0.548
MPI ⁶ x ES ⁴	58	- 0.579	Not Sig.	0.041
MPI ⁶ x PEPD ²	58	- 0.644	Not Sig.	0.060
NHi1 ² x NHd ⁴	58	1.000	Sig. P < 0.005	0.664
NHi1 ² x ES ⁴	58	- 0. 976	Sig. P < 0.025	0.097
NHi1 ² x PEPD ²	58	- 0.981	Sig. P < 0.01	0.115
NHd ⁴ x ES ⁴	58	- 1.000	Sig. P < 0.05	0.068
NHd ⁴ x PEPD ²	58	- 1.000	Sig. P < 0.05	0.080
ES ⁴ x PEPD ²	58	0.946	Sig. P < 0.005	0.763
AC01227 x MPI6	53	1.000	Sig. P < 0.005	0.313
AC01227 x NHi12	53	1.000	Sig. P < 0.005	0.313
AC01227 x NHd4	53	1.000	Sig. P < 0.005	0.241
AC01 ²²⁷ x ES ⁴	53	1.000	Sig. P < 0.005	0.185
AC01227 x PEPD2	53	1.000	Sig. P < 0.005	0.194
AC16244 x MPI6	52	- 0.333	Not Sig.	0.070
AC16244 x NHi12	52	- 0.489	Sig. P < 0.01	0.146
AC16244 x NHd4	52	- 0.576	Sig. P < 0.01	0.129
AC16244 x ES4	52	0.882	Sig. P < 0.01	0.145
AC16 ²⁴⁴ x PEPD ²	52	0.805	Sig. P < 0.01	0.143
AC01227 x AC16244	51	0.191	Not Sig.	0.032

+

number in superscript after each enzyme locus indicates the most common allele (A_1) used for all calculations (see *Appendix V*).

Chapter 6 Huámuco. Peru

Complete linkage is represented by an r^2 index of 1; the r^2 values for the 3 pairs of loci which generated a statistically-significant, non-1 value for D' were all less than 0.400. These low values suggest that incomplete linkage was present between these pairs of loci.

The calculated r^2 values were generally low compared to D' values. The D' and r^2 indices are not directly comparable, however, because the latter is calculated using the value of D (not D', which is effectively the maximum D possible from the Dmax). The low r^2 values obtained in this analysis illustrate the lack of statistical power due to the relatively small sample size. The small sample size is of particular relevance to the pairwise combinations involving microsatellite loci since many alleles were present at these loci.

6.6.3 Summary of population genetic analyses

The results herein provide evidence for a degree of genetic exchange having occurred amongst Huánuco stocks. This evidence stems from 3 observations:

- segregation: a lack of deviation from HW equilibrium was observed at 3 enzyme loci (MPI, ES and PEPD);
- 2. <u>linkage disequilibrium</u>: relatively low I_A values were obtained for enzyme and combined loci;
- <u>linkage disequilibrium</u>: D' values of significantly less than 1 were obtained for
 3 pairwise locus combinations (MPI x NHi1; AC16 x NHi1 and AC16 x
 NHd, shown in **bold** type in *table 6.6.2.2.*)

The finding of evidence for genetic exchange is, perhaps, the result of the short time period and the small geographical area over which the samples were isolated (evidence of exchange between samples collected from wide geographical areas is less likely).

Population genetic analyses were carried out here primarily to provide statistical support for the proposed occurrence of genetic exchange among these stocks, however, these tests also give an insight into the prevalent mode of reproduction occurring in the population. The combined findings suggest the population structure is predominantly clonal, despite at least one recombinational event. Observations of stocks with different zymodemes sharing microsatellite markers, and vice versa, emphasizes the fact that these genetic markers were independent. The expansive clonal propagation of one of the putative hybrid types suggests an increased fitness relative to the parental strains present.

The results presented indicate that genetic exchange as a method for generating genetic diversity cannot be rejected. There are, however, several doubts as to the applicability of classical population genetic analyses to *Leishmania* and to this population specifically. *Leishmania* reproduce in a predominantly clonal manner whereas classical analyses assume a starting point of random mating. Segregation tests are suitable for examining data from organisms of known, fixed, ploidy. Whilst *Leishmania* are thought to be diploid at most housekeeping (i.e. enzyme) loci, there have been reports using karyotype analysis which suggest a multiclonal structure / aneuploidy (Pages *et al.*, 1989; Bastien *et al.*, 1990) and a recent report has described multiclonal strains, identified by the analysis of microsatellite loci, in the related trypanosomatid, *Trypanosoma cruzi* (Oliveira *et al.*, 1998). Population genetic tests have traditionally been carried out on data from a single species; the *Leishmania* stocks analyzed here comprised at least 2 notional (very closely-related) species and putative hybrids thereof. It would not have been possible to analyze each group individually in a manner which was statistically meaningful.

6.7 Discussion of the Huánuco population

By comparison with the Brazilian and Nicaraguan populations, a tremendous level of diversity was identified among the Huánuco stocks. The results and analyses described indicate that this may be due, in part at least, to genetic exchange. [As discussed previously, there is accumulating evidence for the occurrence of genetic exchange in *Leishmania*. Possible reasons why genetic exchange does not occur (or is not identified) more frequently are discussed in greater detail in chapter 7.]

The Huanuco stocks showed a remarkable degree of diversity: 12 zymodemes were identified from 12 enzyme loci and 17 microdemes identified from 3 microsatellite loci. In addition, 9 AC01 sequence types were scored. Several of the inferred genotypes were sampled only once whereas others were repeatedly sampled from both geographically-separate locations and different host species.

Using IEA, 3 species and a new, as yet unnamed, parasite, L. V. sp. n. (type strain IM2832, Grimaldi *et al.*, 1991) were found to occur sympatrically in the region. This is a remarkable finding in itself considering the small geographical area (circa 40km^2) concerned. Stocks were identified as: L. V. peruviana (4 stocks, 6.8%), L. V. braziliensis (25 stocks, 42.4%), L. V. lainsoni (1 stock, 1.7%) and L. V. sp. n. (3 stocks, 5.1%). In addition, 26 stocks (44.1%) were identified as putative L. V. braziliensis / L. V. peruviana hybrids.

A single human stock was identified as L. V. lainsoni: this is the second report of this species occurring in Peru (Lucas *et al.*, 1994). At the present time it cannot be excluded that this patient may have acquired L. V. lainsoni infection elsewhere.

Three dog stocks were almost identical to the type strain, IM2832, of a potential new species of the *Viannia* subgenus (*L. V.* sp. n., Grimaldi *et al.*, 1991). As far as I am aware, this is the first report of this *Viannia* phenotype occurring in Peru or in dogs. These stocks will be discussed in greater detail later.

L. V. peruviana was recorded in the Department of Huánuco prior to the current epidemic, however CL was seldom encountered and no cases of MCL were recorded (Llanos-Cuentas 1993). The recent introduction of L. V. braziliensis (Dujardin et al., 1995b) has, without doubt, resulted in the current epidemic since it is this species, and hybrids consisting of this species and L. V. peruviana, which have been isolated from more than 85% of recent cases (this thesis; Dujardin et al., 1995b). L. V. braziliensis may have been introduced by human immigration from another region or by human or canine intrusion into an unidentified sylvatic transmission cycle. It is recognized that L. V. braziliensis can be transmitted by a wider range of sand fly species than other New World Leishmania species (Killick-Kendrick, 1999). It seems likely therefore, that L. V. braziliensis rapidly adapted to the local vector species after introduction. Presumably, one or more local vector species and/or reservoir hosts (including humans) had a prior L. V. peruviana infection such that dual infection facilitated the occurrence of recombination. It is not known where (i.e. in host or vector) any such exchange would occur (see chapter 7 for discussion), however mixed Leishmania infections have been reported from both sand flies and humans (Barrios et al., 1994; and Silveira et al., 1984; Mehbratu et al., 1991; Ibrahim et al., 1994; Hernandes-Montes et al., 1998, respectively).

Seventeen microdemes were scored from 54 Huánuco stocks: 3 microdemes were shared between different species / hybrid groups and 4 contained more than 1 stock. Microdemes did not always correlate with species status as identified using IEA. Stocks isolated from MCL lesions were assigned to 6 microdemes. Nine AC01 sequence types were scored from 55 Huánuco stocks: 1 sequence type was shared by several stocks identified as *L. V. braziliensis* and *L. V. braziliensis / L. V. peruviana* hybrids and 6 sequence types contained more than one stock. The numbers of zymodemes, microdemes and AC01 sequence types scored among the Huánuco stocks are summarized, by species type, in *table 6.7.1*.

Table 6.7.1 Summary of zymodemes, microdemes and AC01 sequence types scored from Huánuco stocks, by species group.

Species	No. stocks	No. zymodemes (59 stocks ⁴)	No. microdemes (54 stocks ^b)	No. AC01 sequence types (55 stocks ^c)
L. V. braziliensis	25	5	7 ^d	4 ^r
L. V. peruviana	4	1	2	2
L. b / L. pe hybrids	26	4	8 ^{d,s}	2 ^r
L. V. lainsoni	1	1	1°	1
L. V. sp. n.	3	1	2	1
Totals	59	12	17 ^{d,e}	9 ^r

Notes to *table 6.7.1*: ^a 59 stocks were isolated and examined; ^b 54 stocks could be assigned to a microdeme (see section 6.4.3.4); ^a AC01 sequence type was determined for all 55 stocks which amplified the AC01 locus; ^d 2 microdemes were shared by several stocks identified as L. *V. braziliensis or L. braziliensis / L. peruviana* (L. b / L. pe) hybrids; ^e 1 stock identified as a *L. braziliensis / L. peruviana* hybrid shared a microdeme with the Huánuco L. *V. lainsoni* stock; ^f several stocks identified as *L. V. braziliensis / L. peruviana* hybrids had an identical AC01 sequence type.

Of the 59 Huánuco stocks, 45 (76.3%) were of human origin and 14 (23.7%) were isolated from dogs. Eleven of the human stocks (24.4%) were isolated from mucocutaneous lesions. Six MCL stocks were identified as *L. V. braziliensis*, underlining the association of this species with MCL. A single stock isolated from an MCL lesion was identified as *L. V. peruviana*: this is an important finding linking yet another *Viannia* species with this clinical presentation. *L. V. braziliensis / L. V. peruviana* hybrids were also associated with MCL. The implication of these findings is that all patients from Huánuco infected with *L. V. braziliensis*, *L. V. peruviana* and

Chapter 6 Huámuco, Peru

putative hybrids thereof, must be considered at risk of developing MCL. *Table 6.7.2* summarizes the host and clinical origin of the Huanuco stocks, by species group.

Species		Humans	Dogs	No.	
	CL	MCL	Total	(CL)	stocks
L. V. braziliensis	19	6	25	0	25
L. V. peruviana	2	1	3	1	4
L. b / L. pe hybrids	12	4	16	10	26
L. V. lainsoni	1	0	1	0	1
L. V. sp. n.	-	-	0	3	3
Totais	34	11	45	14	59

Table 6.7.2 Summary of the host and clinical origin of the Huánuco stocks, by species group.

Fourteen stocks were isolated from dogs: one was identified as L. V. peruviana, 10 as putative L. V. braziliensis / L. V. peruviana hybrids and 3 as L. V. sp. n. ('IM2832-type').

The finding here of identical L. V. peruviana and L. V. braziliensis / L. V. peruviana hybrid phenotypes in dogs and humans from the same area would seem to support the widely held belief that dogs can act as reservoir hosts of CL and MCL (a subject reviewed extensively by Reithinger & Davies, 1999). Of course, this finding may simply reflect the fact that dogs and humans are exposed to the same infective sand fly populations. However, several species types were found only in humans or dogs; for example, L. V. braziliensis was not isolated from dogs and the 'IM2832-type' L. V. sp. n. was not isolated from humans. The latter finding may suggest that dogs and humans are not, in fact, exposed to the same sand fly populations or that dogs and humans differ in their susceptibility to infection with the various parasite types present. The role of dogs as reservoirs of human infection cannot be proven conclusively without further ecoepidemiological investigation of the dynamics of the transmission cycle. Entomological studies are needed to investigate biting rates, host preferences, etc., further parasitological studies are needed to compare the genetic diversity of stocks from an extended vector and host range (to include dogs, humans, other domestic and sylvatic reservoirs and especially local sand fly fauna), and xenodiagnosis studies are needed to investigate whether local sand flies can become infected from identified animal reservoirs.

The origin of the stocks identified as 'IM2832-type' is unclear. IM2832 itself is a sand fly isolate from Amazonas State, Brazil (Grimaldi *et al.*, 1991): to my knowledge, this strain has not been reported since its original discovery, from Brazil or from any other location. One can speculate that these 3 dogs, all from the same village, contracted this parasite strain after interrupting a previously undisturbed sylvatic cycle, perhaps on a foray into the forest. This phenotype was not sampled among the human stocks however so, if this speculation were true and assuming that dogs may be reservoirs of human infection, it would appear that *Lu. tejadai*, the sand fly species responsible for the peri-domestic transmission of the other species and hybrids in the study site, may not be capable of transmitting the IM2832-type strain. The 'IM2832-type' *L. V.* sp. n. dog stocks and the *L. V. lainsoni* stock LC2679 were shown, using phylogenetic analysis of IEA data, to be genetically distinct from Huanuco stocks of the two main *Viannia* subgenus complexes.

The geographical separation and distinct clinical presentations of CL and MCL in Peru have supported the concept that these diseases are caused by discrete *Leishmania* species (Shaw, 1994). *L. V. braziliensis* and *L. V. peruviana* stocks can now be differentiated using a wide range of phenotypic and genotypic approaches, (Arana *et al.*, 1990; Dujardin *et al.*, 1993b; 1995b; Victoir *et al.*, 1998; Banuls *et al.*, 1999; Russell *et al.*, 1999), nevertheless the two species are remarkably similar. In this study, phylogenetic analysis based on IEA data showed that *L. V. peruviana* strains and stocks formed a discrete clade nestled amongst the *L. V. braziliensis* clades; this observation contradicts that of Banuls *et al.*, (1999; 2000) who found that stocks of *L. V. peruviana* (not including those from Huanuco) formed a DTU.

The Huånuco population contained 26 putative L. V. braziliensis / L. V. peruviana hybrid stocks. In contrast to the L. V. braziliensis / L. V. panamensis hybrids from Nicaragua, the genotypes of these hybrids were not all simple combinations of the 'parental' genotypes present. At least 11 genotypes (A - K) were scored from these 26 stocks: these genotypes are summarized in *table 6.7.3*.

Table 6.7.3 Summary of genotypes scored from putative L. V. braziliensis /

Genotype	Zymodeme	Microdeme	Stocks	No. stocks
A	LON218	LMS43	HR399	1
B	LON218	LMS44	HR799	1
С	LON218	LMS45	LC2570	1
D	LON218	LMS46	HR108, HR110, HR413, HR424, HR434, LC2485, LC2576, LC2790, LC2841, LC2844, LC2901, LC2904, LC3150	13
E	LON218	LMS47	HR797	1
F	LON218	LMS48	LC2825	1
G	LON218	LMS52	HR80	1
Н	LON218	LMS59	LC2422	1
I	LON219	LMS43	HR410	1
J	LON220	LMS45	LC2435	1
K	LON221	LMS46	LC2877	1
	LON218	ND	LC2511, LC2553	2
	LON218	NS	LC2902	1

L. V. peruviana hybrid stocks.

Stocks in blue type were isolated from MCL lesions ND indicates that microsatellite analysis was not done, NS indicates that a microdeme could not be assigned because Genotyper® analysis failed at 2/3 loci.

The majority of stocks (14/26) had genotype D. These stocks may have arisen from clonal expansion after a genetic exchange event between local strain(s) of *L. V. braziliensis* and *L. V. peruviana* and appear to have increased fitness compared to the parental strains. Increased fitness is evident from the excess of these heterozygous stocks compared to any genotypes scored from the 'parental' species. It is not known whether this increased fitness is due to greater virulence and/or pathogenicity or because it is transmitted with greater facility by the vector species in the area. It is noteworthy that, among hybrid stocks, MCL was associated only with genotype D. The other hybrid genotypes scored each contained a single stock. These stocks may have arisen by mutation from genotype D, or from separate genetic exchange events involving different combinations of loci.

CHAPTER 7 –

EXPERIMENTAL INVESTIGATION OF GENETIC EXCHANGE

7.1 Introduction

Genetic transformation involves introducing exogenous DNA into a host organism. The first transfection system for a trypanosomatid resulted in transient gene expression (Bellofatto & Cross, 1989). Stable transformation is dependent on genes which, when expressed, confer a constant selectable phenotype to the host organism. Stable transformation of a trypanosomatid was first described for *Leishmania* (Laban *et al.*, 1990; Kapler *et al.*, 1990).

A number of vectors that permit the stable expression of exogenous genes in *Leishmania* have now been described (Laban *et al.*, 1990; Cruz & Beverley, 1990; LeBowitz *et al.*, 1990, 1991; Coburn *et al.*, 1991; Panton., 1991; Kelly *et al.*, 1992; Freedman & Beverley, 1993; Brooks *et al.*, 2000). Some of these vectors are also functional in other trypanosomatid genera e.g. *pTEX* functions in both *Leishmania* and *Trypanosoma cruzi* (Kelly *et al.*, 1992). Clayton (1999) has recently reviewed genetic manipulation of the Kinetoplastida.

Numerous selectable markers [parentheses give selective drug / method of selection] have been used in *Leishmania* including neomycin phosphotransferase II [G418], hygromycin B phosphotransferase [hygromycin B], *Sh ble* [phleomycin, zeocin], puromycin acetyltransferase [puromycin], N-acetyl glucosamine-1-phosphate transferase [tunicamycin], green fluorescent protein (GFP) of *Aequorea victoria* [fluorescence], thymidine kinase from the Herpes simplex virus, HSV [ganciclovir, negative selection] and blasticidin-S-deaminase [blasticidin S] (Kapler *et al.*, 1990; Laban *et al.*, 1990; Cruz *et al.*, 1991; Lee & van der Ploeg, 1991; Leowitz *et al.*, 1992, Liu & Chang, 1992; Freedman & Beverley, 1993; Ha *et al.*, 1996; Brooks *et al.*, 2000, respectively).

Transformation can be mediated by integration or by episomal (shuttle) vectors. For integration, specific targeting fragments are required in the vector to promote homologous recombination. Cruz & Beverley (1990) and Lee & van der Ploeg (1991)

showed that gene targeting is mediated exclusively by homologous recombination in kinetoplastid organisms, and that linearized constructs integrate with much higher efficiencies. Tobin *et al.* (1991) showed that *Leishmania* contain the enzymatic machinery necessary for homologous recombination.

Several research groups have attempted to demonstrate genetic exchange in trypanosomatids. Success has been achieved only in *Trypanosoma brucei* (Jenni *et al.*, 1986) and, despite continued research, the precise mechanism of exchange remains unclear (reviewed by Gibson & Stevens, 1999). The existence of "natural" hybrids has been reported in other trypanosomatid genera such as those described in the Nicaraguan and Huanuco *Leishmania* populations (Darce *et al.*, 1991; Belli *et al.*, 1994; Dujardin *et al* 1995b; see also chapters 5 and 6), indicating that exchange may occur in these organisms.

The aim of this experiment was to attempt to demonstrate genetic exchange in *Leishmania*. Previously, a single, unsuccessful attempt to demonstrate exchange using Old World species of the subgenus *Leishmania* had been reported (Panton *et al.*, 1991). These authors obtained a mixed infection in sand flies but did not find any hybrid progeny among 2500 scored. Other attempts have been made using Old World strains (David Evans, personal comm.), however, to the best of my knowledge, there are no reports of these experiments, presumably due to lack of success.

For reasons, some of which will be discussed later, the experiment described herein proved too ambitious given the time available, and was thus discontinued. This chapter has been included because some of the findings may be of use to future work and the discussion raises important points regarding the generation of diversity in *Leishmania*.

7.2 Strategy

In theory, the experiment appears relatively simple (see *figure 7.2*): stocks from two species (parents 1 and 2, respectively) would be transformed to different drug resistance phenotypes (markers A and B, respectively) and crossed. Any progeny that exhibited both drug-resistance phenotypes (i.e. having both markers) were likely to be the product of a genetic exchange event, and could be selected on media containing both selective drugs.



(figure adapted from Gibson & Stevens, 1999)



7.3 Selection of *Leishmania* strains for transformation

The Nicaraguan Leishmania population comprised stocks belonging to two Viannia species (L. V. braziliensis and L. V. panamensis) and putative hybrid stocks. As discussed in chapter 5, these hybrids exhibit markers from both Nicaraguan parental strains of these two species at all discriminatory loci studied. Since all these loci are unlikely to be located on a single chromosome, it is possible that these hybrids may be the product of an exchange event involving haploid chromosomes, one set from each parental strain. Hybrid stocks were independently isolated from humans living in areas separated by large distances and geographical barriers and, as such, it is possible that this genetic exchange event may have occurred more than once. Nicaraguan stocks of the two 'parental' species were selected for transformation because it was felt more likely that they would be capable of generating hybrid offspring from crossing experiments than would randomly isolated strains of the two species. In hindsight, it is likely that stocks from the Huanuco population would have been more suitable for this study in that they were isolated from an epidemic population in a small area: these stocks were, not available at the start of this experiment.

Nicaraguan "parental" L. V. braziliensis and L. V. panamensis stocks (XD28 and ZF01 respectively) were selected for transformation to different drug resistance phenotypes (G418 and hygromycin or phleomycin). Another L. V. braziliensis stock, M2903, was also selected as an "integration control" because it was the origin of the targeting fragments which were going to be incorporated into the vector.

7.4 <u>Construction of transformation vectors</u>

As described previously, trypanosomatids have unusual mechanisms for gene transcription. To ensure that transformation vectors function in the target cell, vectors in current use carry processing elements (spliced leader and polyadenylation signals) involved in mRNA maturation. To simplify the analysis of progeny, it was intended here to construct vectors incorporating specific targeting fragments to promote integration of the drug resistance gene into the *Leishmania* genome by homologous recombination.

7.4.1 Construction of the vector conferring neomycin resistance

The vector pTEX-neo-133P (see figure 7.4.1) was constructed from two existing plasmids: pT3T719U-133P (see figure 2.12.2, Chapter 2, Amorim et al., 1996) and pTEX (figure 7.4.1; Kelly et al., 1992). pT3T719U-133P contains 133P, a 2.5kb sequence spanning parts of the coding and untranslated regions of the 70kDa heat shock protein gene (hsp70) from L. V. braziliensis stock M2903. pTEX contains the neomycin resistance gene (neo^r) and multiple cloning site (MCS) flanked by T. cruzi processing elements derived from glyceraldehyde phosphate dehydrogenase (gapdh). pTEX is maintained episomally and has been shown to confer resistance to neomycin in T. cruzi and Leishmania spp. The vector also contains the ampicillin resistance gene (amp^r) for selection purposes.

Briefly, the 2.5kb 133P fragment was isolated from the pT3T719U-133P vector by agarose gel electrophoresis after digestion with *Eco*RI. After purification, 133P was ligated to *Eco*RI-linearized *pTEX*. *E. coli* cells were transformed with the construct. Plasmids containing the 133P insert (*pTEX-neo-133P*, figure 7.4.1) were selected by size after gel electrophoresis. At each stage of construction, the products were checked for correct size and orientation using restriction enzyme analysis and gel electrophoresis.



Promastigote cultures of each stock (M2903, XD28 and ZF01) were electroporated (see Chapter 2 for details), both with and without the addition of vector DNA (DNA+ and mock-transfected controls, respectively).

In initial experiments, cultures of the *L. V. braziliensis* stocks M2903 and XD28 (both DNA+ and mock-transfected controls) appeared to have survived the electroporation procedure, but died after transfer to fresh liquid medium even in the absence of the neomycin analogue, G418. This may have occurred due to the over-dilution of electroporated cells or because the promastigotes were damaged by the electroporation procedure. A second experiment was successful: both DNA+ XD28 and M2903 stocks were transformed and grew well at concentrations of G418 up to 100ug/mL. Mock-transfected control cultures died after transfer to fresh liquid medium containing even low concentrations of G418. The presence of the neomycin gene in these stocks was confirmed by Southern analysis, however, the size of the band indicated that the construct was maintained episomally and had not integrated.

Cultures of the *L. V. panamensis* stock ZF01 (both DNA+ and mock-transfected controls) survived the electroporation procedure and grew well in liquid medium in the absence of G418. ZF01 DNA+ parasites grew well in the presence of 25 μ g/mL and 50 μ g/mL G418, suggesting that they too had been transformed to neomycin resistance. These parasites, however, failed to grow and subsequently died, in medium containing higher G418 concentrations (75 and 100 μ g/mL).

Southern analysis of DNA extracted from this "transformed" stock using ³²P-labeled neomycin gene probes (isolated from two different sources, pGEM-Neo and pTEX) showed that the Neo^r gene was absent. The probes were shown to be labeled and specific for the neomycin gene by their use on of a digest of a cosmid containing the neomycin gene (Martin Taylor, personal communication). It was concluded that the observed growth in medium containing G418 was not a result of transformation, but was indicative of natural resistance. Mock-transfected ZF01 control cultures died in the presence of even low concentrations of G418, indicating that the isolate as a whole was not resistant, however it is possible that a subpopulation of ZF01 grew out under G418 selection. A mutation may have occurred which conferred resistance to G418; any organism carrying such a mutation would obviously have a selective advantage and would thus expand rapidly. "Natural" resistance to G418 and related drugs has

been noted previously; such resistance may be conferred by as little as a single base change in rRNA genes (John Kelly, personal communication).

7.4.2 Construction of the vector conferring phleomycin resistance

The second vector was initially designed to contain the gene encoding resistance to hygromycin (hygromycin B phosphotransferase, hyg') however preliminary studies showed that ZF01, and 2 other L. V. panamensis stocks, exhibited natural resistance to this drug in addition to neomycin. It was subsequently decided to base the second vector on pTEX-ble (Nozaki & Cross, 1994), a plasmid vector containing the gene encoding resistance to phleomycin (ble'), because all L. V. panamensis stocks tested (ZF01, ZE09, LS94) proved susceptible to this drug.

Initially, the same approach to construction was taken as for pTEX-neo-133P, i.e. with 133P being inserted into the pTEX-ble vector. This, however, proved impossible owing to the presence of unmapped restriction sites in pTEX-ble. A second approach was attempted whereby a 0.5kb fragment containing the phleomycin resistance (ble^r) marker from pTEX-ble was inserted into pT3T719U-133P.

In practice, construction of the vectors proved to be extremely difficult and only one of the vectors, *pTEX-neo-133P*, was successfully constructed. Construction of the *ble^r* vector was not completed.

It became evident, after encountering experimental difficulties, that this experiment was not going to be completed within an acceptable time. After the investment of many hours, this work was discontinued to allow time to concentrate on other areas of the project. Only one of the required vectors was successfully constructed. This vector was used to transform two *L. V. braziliensis* stocks, but failed to integrate and was maintained episomally. The *L. V. panamensis* stock exhibited natural resistance to G418 thus preventing the selection of transformants. The transformed stocks are stored, for future use, in the PMBBU cryobank at LSHTM. 7.5

Discussion of genetic transformation and genetic exchange

7.5.1 Vectors and integration by homologous recombination

For this study, integrative vectors were required to promote stable drug resistance phenotypes, thus facilitating the analysis of progeny from crossing experiments. To this end, constructs carrying hsp70 gene fragments from *L. V. braziliensis* were designed to promote homologous recombination into the *hsp70* gene array. Successful transformation was achieved using the linearized construct however the vector did not appear to integrate and was maintained episomally, even in the stock from which the *hsp70* targeting fragments were derived.

Failure of integration is, in hindsight and after the report of Papadopoulou & Dumas (1997), not surprising given the level of sequence homology required to promote homologous recombination. Papadopoulou & Dumas (1997) found that the frequency of homologous recombination between vector and chromosomal target DNA sequence was influenced by many factors. The most critical parameters for efficient homologous recombination in *Leishmania* were shown to be (1) the degree of homology between the vector and the genomic target (isogenic DNA or very high degree of homology critical) and (2) the length of homologous sequences (optimal length 1 - 2 kb on each side). The targeted locus was also shown to be important because (i) recombination frequency increased with increasing copies of the target locus and (ii) variable levels of targeting efficiency were observed at different chromosomal locations. In addition, variable targeting frequencies were exhibited by different strains of the same *Leishmania* species.

The hsp70 targeting fragments incorporated into the targeting vector pTEX-neo-133P were approximately 1.25 kb in length on either side of the resistance marker which, according to Papadopoulou & Dumas (1997) is the optimal length required to facilitate homologous recombination. The hsp70 targeting fragments, however, included part of the more variable hsp70 array untranslated region (see figure 2.13.2, Chapter 2). According to Papadopoulou & Dumas (1997), sequence divergence drastically reduces the efficiency of gene targeting in *Leishmania*. It is seems likely that the sequence homology between the vector and *L. V. panamensis* and *L. V. braziliensis* strain XD28 were sufficiently divergent to prevent homologous recombination.

It is more difficult to explain why the vector did not integrate into the hsp70 array of the L. V. braziliensis strain M2903 because this strain was the source of the hsp70targeting fragments. Again, however, failure to integrate might have been the result of sequence divergence: M2903 is a widely-used reference strain and its continued maintenance in culture in different laboratories may have resulted in accumulated sequence differences, presumably in the variable region, between our M2903 strain and that used by Amorim *et al.* (1996) who constructed pT3T719U-133P. It is also possible that the linearized construct did integrate in some cells. The presence of other cells transformed with multiple episomal copies would, however, in the presence of G418, likely have 'out-selected' cells containing a single *neo* copy.

Cross-species homologous recombination has recently been reported In *Leishmania* using a vector targeting a highly conserved gene (Krobitsch & Clos, 2000).

7.5.2 Crossing experiments and potential site of genetic exchange

Panton *et al.* (1991) performed experimental crosses in *Phlebotomus papatasi* using two cloned strains of *L. major.* The clones contained numerous restriction fragment length polymorphisms and were selected for resistance to methotrexate or tunicamycin by gene amplification (i.e. not by transfection). Promastigotes from the gut contents of flies shown to have mixed infections were plated onto selective media. Despite examining 2500 progeny for the presence of both drug resistance markers, no hybrid parasites were scored. The authors concluded that the frequency of genetic exchange in their cross was less than 4×10^{-4} .

One possible reason why these authors failed to generate hybrid offspring could have been that genetic exchange between *Leishmania* may not occur in the sand fly vector but in the host. Genetic exchange in the related kinetoplastid *Trypanosoma brucei* has been shown to occur in the tsetse vector (Jenni *et al.*, 1985). However, *T. brucei* subsp. parasites circulate in the blood of the host and do not have an intracellular stage: genetic exchange seems unlikely in this environment because the chances of encountering another parasite are low. By comparison, *Leishmania* species reside inside professional phagocytic cells such as macrophages, Langerhan's and Küppfer cells. Co-inoculation of more than one *Leishmania* species or strain could result in the co-localization of parasites inside the host phagocyte thereby allowing close proximity for genetic exchange events. Indeed, Kreutzer *et al.* (1994) reported data of nuclear

DNA content for amastigotes and promastigotes (measured using quantitative (Feulgen) microspectrophotometry), which suggested that sexual reproduction may take place inside the intracellular amastigote form.

7.5.3 Models for genetic exchange; ploidy

Based on observations of apparently triploid offspring from experimental crossing experiments using *T. brucei* (Gibson *et al.*, 1992; Gibson & Whittington, 1993; Gibson & Bailey, 1994), Gibson (1995) proposed 2 models for the mechanism of genetic exchange, both involving fusion and meiosis. The first model involves the fusion of diploids followed by meiosis to produce diploid progeny. In the second model, the processes are reversed with meiosis followed by fusion of haploids. In *T. brucei*, evidence for meiosis comes from the Mendelian inheritance patterns observed for housekeeping genes and the high frequency of triploid hybrids; evidence for cell fusion in *T. brucei* rests on observations of the inheritance patterns of kinetoplast DNA (for a review, see Gibson & Stevens, 1999). To date, no haploid stage has been substantiated for *T. brucei*, or for any other trypanosomatid.

Similar evidence for genetic exchange has been documented in Leishmania. Lanotte & Rioux (1990) reported the apparent cell fusion of cultured L. infantum and L. tropica promastigotes. Indirect evidence of diploidy, genetic recombination and Mendelian inheritance in natural Leishmania populations comes from data generated by IEA, RAPD, RFLP and molecular karyotype analyses (Evans et al., 1987; Darce et al., 1991; Kelly et al., 1991; Bonfante-Garrido et al., 1992; Belli et al., 1994; Dujardin et al., 1995b; Banuls et al., 1997). Diploidy is taken as the working hypothesis for Leishmania, however several groups have also reported evidence of aneuploidy, mostly from molecular karyotpye analysis (e.g. Bastien et al. 1990). Evidence of aneuploidy has also been reported in T. cruzi (Gibson & Miles, 1986, see Gibson & Stevens, 1999). Leishmania have been shown to have highly plastic genomes in which chromosomal rearrangements are common (reviewed in chapter 1). Bastien et al. (1990) suggested a mosaic structure for some strains of L. infantum after isolating clones with different karyotypes from the same strain. This finding is supported by clonal variation within an L. V. braziliensis isolate using isoenzyme analysis (Cuba-Cuba et al., 1991).

7.5.4 Detection of heterozygotes

So, why are natural hybrids or products of genetic exchange not detected more often?

The detection of genetic exchange requires that molecular markers can be differentiated. If all markers examined are identical in all stocks (as in Três Braços, Brazil) it would not be possible to tell whether stocks had undergone genetic exchange (although clonal expansion seems more likely). In addition, genetic exchange may not necessarily result in the exchange of all markers studied (as in the Nicaraguan stocks); partial recombination involving sets of markers also occurs (as described here in the Huánuco stocks). Highly discriminatory techniques have only relatively recently been introduced. With the application of more markers, genetic exchange events may become more evident.

Genetic exchange requires the presence of mixed infections. Mixed Leishmania infections have been reported both in sand flies and in humans (Barrios et al., 1994; and Silveira et al., 1984; Mehbratu et al., 1991; Ibrahim et al., 1994; Hernandes-Montes et al., 1998, respectively). Such reports are, however, relatively rare.

Many *Leishmania* species have restricted ranges due to suitable vector and host availability. Hence, even if a new strain or species is introduced into an area, it is possible that the vector species present may be refractory to infection or may not capable of transmitting the new parasite.

Genetic exchange may result in loss or gain of fitness. Offspring resulting from a genetic exchange event may be less fit than the "parental" strains and be lost as the result of selection processes.

Past studies have investigated strains which have been amassed from wide geographical areas over a large time scale. Often, the population size has been too small and has been biased towards human isolates. The ideal population in which to find evidence of genetic exchange would likely comprise many isolates from sand fly vectors, human, domestic and sylvatic hosts from a small geographical area in an epidemic situation.

CHAPTER 8 - CONCLUDING REMARKS

The Viannia subgenus is found only in the New World, where the vast range of habitats and diverse fauna therein are likely to have played a major role in the diversification of parasites. Species of the Leishmania subgenus Viannia were originally thought to exist in discrete transmission cycles and to be responsible for distinct clinical presentations. However, four Viannia species have now been associated with MCL, a clinical presentation previously attributed to infection with L. V. braziliensis. Recent reports describing putative species have fueled the debate regarding the taxonomic status and population structure of these parasites.

The aim of this project was to use molecular methods to reassess inter- and intraspecific relationships in the *Leishmania* subgenus *Viannia*. By exploring the extent of genetic diversity occurring in natural *Viannia* populations and the occurrence of genetic recombination across currently recognized species, it was hoped that this project would provide an insight into the changing epidemiology of cutaneous and mucocutaneous leishmaniasis (CL and MCL) in Latin America.

A panel of reference strains was selected, including representatives of the subgenus *Leishmania*, for initial comparisons of established and novel methods. The most discriminatory techniques were subsequently used to examine *Viannia* stocks from three endemic areas (Três Braços, Bahia State, Brazil; Nicaragua; and Huánuco, Peru) in detail. In addition, stocks isolated from widely separated geographical areas in Brazil were examined.

Isoenzyme analysis (IEA) was used as a gold standard to confirm the identity of isolates and, alongside other markers, to study diversity, gene flow and phylogenetic relationships. Overall, 27 zymodemes were scored from 125 *Viannia* strains and stocks (6 species and 1 potential new species) using 12 enzyme loci. Nineteen stocks belonging to eight zymodemes were associated with MCL, including one *L. V. peruviana* stock, a finding not previously reported.

Random amplification of polymorphic DNA (RAPD) data were of value in detecting and confirming the presence of L. V. braziliensis / L. V. panamensis hybrids among Nicaraguan stocks, although it was found to be difficult to reproduce these data on occasions. PCR-RFLP analysis of the rRNA ITS region was found to be highly informative and allowed distinction of reference species even within the Viannia species complexes.

During the course of this project, collaborators in Cambridge isolated microsatellites specific for the subgenus *Viannia*. Three PCR-based strategies were employed to examine all *Viannia* strains and stocks at three microsatellite loci. Non-denaturing PAGE analysis was found to be a sensitive method of typing stocks, although data interpretation was complicated by the presence of multiple bands and artefactual bands. The presence of heteroduplexes was a useful indicator of multiple alleles, not only in the putative inter-species hybrids.

Using a Genescan@ / Genotyper@ approach, it was possible to identify allele(s) present in *Viannia* stocks at each microsatellite locus. Moreover, putative hybrid stocks could be detected by the presence of double peaks at one or more of the loci examined. The term "microdeme" was adopted for a stock, or group of stocks, defined by a combination of microsatellite characters (Genotyper@ data). Tremendous diversity was revealed, with at least 61 microdemes scored over all stocks.

DNA sequencing was performed on PCR products of the AC01 locus, generating data on the number of dinucleotide repeats and variation in the flanking regions. At least 33 AC01 sequence types were scored compared to 9 AC01 allele sizes identified using Genotyper[®]. There was general agreement between dinucleotide repeat number and allele size, as determined by sequencing and Genotyper[®], respectively; although in several instances bias was noted in the resolution of alleles which could not always be related to the length of the microsatellite product.

Microsatellite data were exploited for both population genetics (Genotyper® data) and phylogenetics (AC01 sequence data).

Little genetic diversity was found among *L. V. braziliensis* stocks from Três Braços and Corte de Pedra, Brazil, although more than one microdeme was identified. In contrast, each of the four *L. V. braziliensis* stocks from Amariji, Pernambuco State, had a distinct genotype. Isoenzyme and microdeme heterogeneity among Brazilian *L. V. braziliensis* stocks was more common among stocks originating from different geographical areas.

Among the Nicaraguan stocks, 3 zymodemes were scored which corresponded to *L. V. braziliensis*, *L. V. panamensis* and putative *braziliensis* / *panamensis* hybrids. The presence of inter-species hybrids was supported by RAPD and microsatellite data.

In Huánuco, Peru, L. V. braziliensis and L. V. peruviana and L. V. braziliensis / L. V. peruviana hybrid forms were found sympatrically in the same locality. Furthermore, all three forms were found in humans and two of them in dogs. The presence of indistinguishable stocks occurring sympatrically in humans and dogs strongly implicates the dog as a potential reservoir of infection. Two other parasites were also described in the Huánuco population: a single human isolate was identified as L. V. lainsoni and three dog isolates were identified as a potential new species, previously described in a sand fly isolate from Manaus, Brazil. Mucocutaneous disease, usually associated with L. V. braziliensis, was also associated with L. V. peruviana and L. V. braziliensis / L. V. peruviana hybrids in Huánuco.

Isoenzyme and microsatellite data were used to provide an estimation of segregation (Hardy-Weinberg analysis) and linkage disequilibrium (I_A and D' / r^2 indices) in the Huánuco *Viannia* population. Analysis of combined data suggested that at least one, and probably several, recombination events had occurred among these stocks. However, the population did not conform to panmixia, implying that propagation appeared to be predominantly clonal. Interpretation of linkage disequilibrium indices was complicated by a lack of statistical power, a result of the large numbers of possible microsatellite alleles relative to the small number of stocks analyzed. The expansive clonal propagation of one of the putative hybrid types suggests an increased fitness of this parasite relative to the parental strains present.

Phylogenetic analysis of these stocks generally supported intuitive concepts of species groups and relationships, despite limitations in the range and number of samples, and despite equivocal models, e.g., for phylogenetic analysis of microsatellite data. In agreement with previous reports, the subgenus *Viannia* was, from IEA data, monophyletic.

It is apparent from pre-existing literature that the boundaries between species of the two major Viannia species complexes have doubtful validity in view of the description of hybrid forms between L. V. braziliensis and L. V. panamensis; L. V. braziliensis and L. V. peruviana, and L. V. braziliensis and L. V. guyanensis. Work reported in this thesis confirms these doubts. In particular, the status of L. V. peruviana as a species distinct from L. V. braziliensis was, in contrast to previous reports, not supported. L. V. lainsoni stocks, and those of a putative new species, were shown to be distinct from stocks of the braziliensis and guyanensis species complexes. A recently revised classification has suggested that the two most recently described Viannia species, L. equatorensis and L. colombiensis should, in fact, be considered as members of a new Trypanosomatidae section, Paraleishmania.

Genetic recombination in the subgenus Viannia has fundamental implications in terms of outbreaks of new pathogenic forms, e.g., as shown here with the L. V. braziliensis / L. V. peruviana hybrids which were associated with MCL, and for the spread of drug resistance. L. V. braziliensis strains in particular have shown themselves to be exceptionally adaptable to changing environments. Experimental studies of genetic recombination using transformed laboratory strains were discontinued owing to limited resources, however these studies are clearly of interest.

The insight gained into the epidemiology of these *Viannia* populations suggests a disease agent that can undergo genetic exchange and can exist in several distinct transmission cycles. Interruption of natural sylvatic cycles can lead to founder effects, clonal propagation, and subsequent widespread dissemination of biological clones.

In future work there is further scope for the application of PCR-RFLP analysis to molecular taxonomy, population genetics and phylogenetics of *Leishmania*. Additional targets suitable for PCR-RFLP analysis are likely to emerge from the *Leishmania* genome sequencing programme. The application of Genescan® / Genotyper® would facilitate high throughput scoring of PCR-RFLP data, although sequence comparisons of such targets would provide the ultimate approach to molecular phylogenetics.

The application of microsatellites to the population genetics and phylogenetics of *Leishmania* is in its infancy. Further microsatellite loci, which are amenable to both Genotypers and DNA sequence analysis, are required. The chromosomal location of these markers should be investigated by Southern blotting, using labeled oligonucleotides and high stringency conditions. As yet, there are no microsatellite markers available for the subgenus *Leishmania*. Microsatellites are likely to be particularly valuable for studies of gene flow between closely related *Leishmania* populations, rather than for the definition of species specific profiles.

In retrospect, we would like to have had more isolates from the Huánuco region in order to detect predicted additional genotypes, thus strengthening the analyses of segregation and linkage. To exclude the possibility of polyploid or aneuploid genotypes, it would be useful to quantitate the DNA content per cell of Huánuco stocks representative of the putative parental and hybrid forms. More isolates are available from Três Braços in Brazil, but in view of their relative lack of diversity it seems less important to extend the study of that locality. In contrast, the extraordinary diversity scored in isolates from Amariji merits further investigation.

If the epidemiology of the subgenus *Viannia* were comparable to that of the African trypanosome, *T. brucei*, in Africa, an intensive study of genetic diversity in an undisturbed sylvatic transmission cycle would complement this project. Such a study would be very difficult logistically because it is likely that large numbers of mammalian reservoir hosts and vector species would need to be examined, although it might be possible to exploit existing isolate collections.

By analogy with the research on the geographical distribution of *T. cruzi* in Latin America, where approximately 500 isolates have now been examined by a variety of methods, it would be of interest to map *Viannia* genotypes in all major endemic areas. A history of emergence and spread of biological clones might then become apparent which could be linked to vector or reservoir host distribution. Among reservoir hosts, the dog may have a significant role in the dissemination of *Viannia* far beyond regions with natural, sylvatic reservoir hosts, a role which clearly needs clarifying. This would help to predict where vector distribution could encourage new suburban outbreaks, as have recently occurred in Brazil. Furthermore, if the importance of the dog as a reservoir host is confirmed, this should provide impetus for the development of new vaccines, drugs and diagnostic methods for canine leishmaniasis.

Molecular comparisons, e.g., through PCR-RFLP or DNA and amino acid sequencebased phylogenetics will also allow an insight into the link between mammal host species and *Leishmania* species, or between vector species and *Leishmania* species, and the possible phenomenon of co-evolution. This may reveal whether *Leishmania* was originally a parasite of insects which ventured into mammals, as is frequently proposed, or a parasite of mammals that hijacked arthropod vectors for transmission.

We remain convinced that genetic transformation should allow experimental studies of the mechanisms of hybridization and recombination in *Leishmania*.

There is no doubt that leishmaniasis is a major public health problem in Latin America and around the world, and that the *Viannia* subgenus is spreading in Latin America and encroaching on suburban areas. The epidemiology is thus highly dynamic. This project has expanded the molecular approaches to studying the epidemiology and taxonomy of the subgenus *Viannia* and has produced a new perception of the variety and complexity of *Viannia* populations.

REFERENCES

Aebischer, T. (1994) Recurrent cutaneous leishmaniasis: a role for persistent parasites? Parasitol. Today 10:25-8.

Aguilar, C.M., Rangel, E.F., Garcia, L., Fernandez, E., Momen, H., Grimaldi, G. Jnr., & de Vargas, Z. (1989) Zoonotic cutaneous leishmaniasis due to *Leishmania (Viannia) braziliensis* associated with domestic animals in Venezuela and Brazil. *Mem.Inst.Oswaldo Cruz* 84:19-28.

Al-Gindan, Y., Omer, A.H.S., Al-Humeidan, Y., Peters, W. & Evans, D.A. (1983) A case of mucocutaneous leishmaniasis in Saudi Arabia caused by *Leishmania major* and its response to treatment. *Clin.Exp. Dermatol.* 8:185-8.

Almeida, M.C., Cuba-Cuba, C.A., Moraes, M.A.P. & Miles, M.A. (1996) Dissemination of Leishmania (Viannia) braziliensis. J.Comp.Path. 115:311-6.

Amos, W., Sawcer, S.J., Feakes, R.W. & Rubinsztein, D.C. (1996) Microsatellites show mutational bias and heterozygote instability. *Nat. Genet.* 13:390-1.

Altés, J., Riera, M., Udina, M., Galmés, A., Balanzat, J., Balesteros, A., Buades, J., Salvá, F. & Villalonga C. (1991) Visceral leishmaniasis: another HIV-associated opportunistic infection? Report of eight cases and review of the literature. *AIDS* 5:201-7.

Alvar, J., Balesteros, J.A., Soler, R., Benito, A., van Eys, G.J.M., Schoone, G.J. & Cabrer, B. (1990) Mucocutaneous leishmaniasis due to *Leishmania (Leishmania) infantum:* biochemical characterization. *Am.J.Trop.Med.Hyg.* 43:614-8.

Alvar, J., Blazquez, J. & Najera, R. (1989) Association of visceral leishmaniasis and HIV infection. J.Inf.Dis. 160:560-1.

Amorim, A.G., Carrington, M., Miles, M.A., Barker, D.C. & de Almeida, M.L. (1996) Identification of the C-terminal region of 70kDa heat shock protein from *Leishmania (Viannia) braziliensis* as a target the humoral immune response. *Cell Stress Chaperones* 1:177-87.

Amos, W. & Rubinstzein, D.C. (1996) Microsatellites are subject to directional evolution. Nat. Genet. 12:13-4.

Anonymous (1985) International Code of Zoological Nomenclature, 3rd Ed., p.338, International Trust for Zoological Nomenclature, University of California Press, Berkeley and Los Angeles.

Apostol, B.L., Black, W.C., Reiter, P. & Miller, B.R. (1996) Population genetics with RAPD-PCR markers: the breeding structure of *Aedes aegypti* in Puerto Rico. *Heredity* 76:325-334.

Ara, M., Maillo, C., Peon, G., Clavel, A., Cuesta, J., Grasa, M.P. & Carapeto, F.J. (1998) Visceral leishmaniasis with cutaneous lesions in a patient infected with human immunodeficiency virus. Br.J.Dermatol. 139:114-7.

Arana, M., Evans, D.A., Zolessi, A., Llanos-Cuentas, A. & Arevalo, J. (1990) The biochemical characterization of *Leishmania* (Viannia) braziliensis and L. (V) peruviana by isoenzyme electrophoresis. Trans. R.Soc. Trop. Med. Hyg. 84:526-9.

Arias, J. (1996) Programa de controle de Leishmaniose Tegumentar Americana (LTA): situação epidemiológica. Ministreio da Saúde, Fundação Nacional de Saúde, Coordenação Nacional de Dermatologia Sanitária, Brasilia, Brasil.

Armijos, R.X., Weigel, M.M., Aviles, H., Makdonado, R. & Racines, J. (1998) Field trial of a vaccine against New World cutaneous leishmaniasis in an at-risk child population: safety, immunogenicity and efficacy during the first 12 months of follow-up. *J.Inf.Dis.* 177:1352-7.

Ashley, M.V. & Dow, B.D. (1994) The use of microsatellite analysis in population biology: background, methods and potential applications. *EXS* 69:185-201.

Atta, A.M., Oliveira, D., Correa, J., Atta, M.L., Almeida, R.P. & Carvalho, E.M. (1998) Antileishmanial IgE antibodies: a marker of active disease in visceral leishmaniasis. *Am.J.Trop.Med.Hyg.* 59: 426-30.

Badaro, R., Benson, D., Eulalio, M.C., Freire, M., Cunha, S., Netto, E.M., Pedral-Sampaio, D., Madureira, C., Burns, J.M., Houghton, R.L., David, J.R. & Reed, S.G. (1996) rK39: a cloned antigen of *Leishmania chagasi* that predicts active visceral leishmaniasis. *J.Inf.Dis.* 173:758-61.

Banuls, A.L., Dujardin, J.C., Guerrini, F., De Doncker, S., Jacquet, D., Arevalo, J., Noel, S., Le Ray, D. & Tibayrenc, M. (2000) Is *Leishmania (Viannia) peruviana* a distinct species? A MLEE/RAPD evolutionary genetics answer. *J. Eukaryot Microbiol.* 47:197-207.

Banuls, A.L., Guerrini, F., Le Pont, F., Barrera, C., Espinel, I., Guderian, R., Echeverria, R. & Tibayrenc, M (1997) Evidence for hybridization by multilocus enzyme electrophoresis and random amplified polymorphic DNA between *Leishmania braziliensis* and *Leishmania panamensis/guyanensis* in Ecuador. *J.Eukaryot.Microbiol.* 44:408-11.

Banuls, A.L., Hide, M. & Tibayrenc, M. (1999) Molecular epidemiology and evolutionary genetics of *Leishmania* parasites. *Int.J.Parasitol.* 29:1137-47.

Barker, D.C. & Butcher, J. (1983) The use of DNA probes in the identification of leishmaniasis: discrimination between isolates of the *Leishmania mexicana* and *L. braziliensis* complexes. *Trans.R. Soc. Trop. Med. Hyg.* 77:285-97.

Barral-Netto, M., Barral, A., Brodskyn, C., Carvalho, E.M. & Reed, S.G. (1995) Cytotoxicity in human mucosal and cutaneous leishmaniasis. *Parasite Immunol*. 17:21-8.

Barral-Netto, M., Barral, A., Brownell, C.E., Skeiky, Y.A.W., Ellingsworth L.R., Twardzik, D.R. & Reed, S.G. (1992) Transforming growth factor- β in leishmanial infection: a parasite escape mechanism. *Science* 257:545-8.

Barral, A., Pedral-Sampaio, D., Grimaldi, G., Momen, H., McMahon-Pratt, D., Ribeiro de Jesus, A., Almeida, R., Badaró, R., Barral-Netto, M., Carvalho, E.M. & Johnson, W. (1991) Leishmaniasis in Bahia, Brazil: evidence that *Leishmania amazonensis* produces a wide spectrum of clinical disease. *Am.J.Trop.Med.Hyg.* 44:536-46.

Barreto, A.C., Cuba, C.C., Marsden, P.D., Vexenat, J.A. & De Belder, M. (1981) Caracteristicas epidemiologicas da leishmaniose tegumentar americana em uma regiao endemica do Estado da Bahia, Brasil. I. Leishmaniose humana. Boll.Oficina Sanitaria Pan AM 90:415-24.

Barrett, M.P., MacLeod, A., Tovar, J., Sweetman, J.P., Tait, A., Le Page, R.W.F. & Melville, S.E. (1997) A single locus minisatellite sequence which distinguishes between *Trypanosoma brucei* isolates. *Mol.Biochem.Parasitol.* **86**:95-9.

Barrett, T.V. & Senra, M.S. (1989) Leishmaniasis in Manaus, Brazil. Parasitol. Today 5:255-7.

Barrios, M., Rodriguez, N., Feliciangeli, D.M., Ulrich, M., Telles, S., Pinardi, M.E. & Convit, J. (1994) Coexistence of two species of *Leishmania* in the digestive tract of the vector *Lutzomyia ovallesi*. *Am.J. Trop. Med. Hyg.* **51**:669-75.

Bastien, P., Blaineau, C. & Pages, M. (1992) Sex, lies and karyotype. Parasitol. Today 8:174-7.

Bastien, P., Blaineau, C., Taminh, M., Rioux, J-A., Roizes, G. & Pages, M. (1990) Interclonal variations in molecular karyotype in *Leishmania infantum* imply a mosaic strain structure. *Mol. Biochem. Parasitol.*, 40:53-61.

Belli, A.A., Miles, M.A. & Kelly, J.M. (1994) A putative Leishmania panamensis / Leishmania braziliensis hybrid is a causative agent of human cutaneous leishmaniasis in Nicaragua. Parasitology 109:435-42.

Bellofatto, V. & Cross. G.A.M. (1989) Expression of a bacterial gene in a trypanosomatid protozoan. Science 244:1167-9

Berenguer, J., Moreno, S., Cercenado, E., Juan, C.L., Quirós, B., de la Fuente, A.G. & Bouza, E. (1989) Visceral leishmaniasis in patients infected with Human Immunodeficiency Virus (HIV). Ann.Intern.Med. 111:129-32.

Berhe, N., Hailu, A. & Gemetchu, T. (1995) Human immunodeficiency virus and recurrence of cutaneous leishmaniasis long after healed localized cutaneous leishmaniasis due to *Leishmania* aethiopica. Trans. R. Soc. Trop. Med. Hyg. **89**:400-1.

Berman, J.D. (1997) Human leishmaniasis:clinical, diagnostic and chemo-therapeutic developments in the last 10 years. *Clin.Inf.Dis.* 24:684-703.

Beverley, S.M., Ismach, R.B. & McMahon-Pratt, D. (1987) Evolution of the genus *Leishmania* as revealed by comparisons of nuclear DNA restriction fragment patterns. *Proc.Natl.Acad.Sci. USA* 84: 484-8.

Beverley, S.M., Ellenberger, T.E., Iovannisci, D.M., Kapler, G.M., Petrillo-Peixoto, M. & Sina, B.J. (1988) Gene amplification in *Leishmania* pp. 431-8. In: Englund, P.T. & Sher, A. (Eds) *The biology* of parasitism. Alan R Liss Inc., NY.

Birky, C.W. (1983) Relaxed cellular controls and organelle heredity. Science 222:468-75.

Birnboim, H.C. & Doly, J. (1979) A rapid alkaline extraction procedure for screening recombinant plasmid DNA. Nucl. Acids Res 7:1513-23.

Bishop, R.P. & Akinsehinwa, F. (1989) Characterization of *Leishmania donovani* stocks by genomic DNA heterogeneity and molecular karyotype. *Trans.R.Soc.Trop. Med.Hyg.* **83**:629-34.

Bishop, R.P. & Miles, M.A. (1987) Chromosome size polymorphisms of Leishmania donovani. Mol.Biochem.Parasitol. 24:263-72.

Blackwell, J.M., Hale, C., Roberts, M.B., Ulczak, O.M., Liew, F.Y. & Howard, J.G. (1985) An H-IIlinked gene has a parallel effect on *Leishmania major* and *Leishmania donovani* infections in mice. *Immunogenetics* 21:385-95.

Blaineau, C., Bastien, P. & Pages, M. (1992) Multiple forms of chromosomes I, II and V in a restricted population of *Leishmania infantum* contrasting with monomorphism in individual strains suggest haploidy or automixy. *Mol.Biochem.Parasitol.* **50**:197-204.

Blaineau, C., Bastien, P., Rioux, J.A., Roizes, G. & Pages, M. (1991) Long-range restriction maps of size-variable homologous chromosomes in *Leishmania infantum*. Mol.Biochem.Parasitol. 46:292-302.

Bogdan, C., Gessner, A., Solbach, W. & Rollinghof, M. (1996) Invasion, control and persistence of *Leishmania* parasites. *Curr.Opinions in Immunol.* 8:517-25.

Bogliolo, A.R., Lauria-Pires, L. & Gibson, W.C. (1996) Polymorphisms in *Trypanosoma cruzi*:: evidence of genetic recombination. Acta Tropica 61:31-40.

Bonfante-Garrido, R., Melendez, E., Barroeta, S., Mejia de Alejos, M.A., Momen, H., Cupolillo, E., McMahon-Pratt, D. & Grimaldi, G. Jnr. (1992) Cutaneous leishmaniasis in western Venezuela caused by infection with *Leishmania venezuelensis* and *L. braziliensis* variants. *Trans.R.Soc.Trop.Med.Hyg.* 86:141-8.

Bonfante-Garrido, R., Urdaneta, I., Urdaneta, R. & Alvarado, J. (1991) Natural infection of cats with Leishmania in Barquisimeto, Venezuela. Trans. R. Soc. Trop. Med. Hyg. 85:53.

Borst, P. (1986) Discontinuous transcription and antigenic variation in trypanosomes. Ann. Rev. Biochem. 55: 701-32.

Bradley, D.J. (1977) Regulation of *Leishmania* populations within the host. II. Genetic control of acute susceptibility of mice to *Leishmania donovani* infection. *Clin.Exptl.Immunol.* **30**:130-40.

Brandão-Filho, S.P., Campbell-Lendrum, D., Brito, M.E.F, Shaw, J.K. & Davies, C.R. (1999) Epidemiological surveys confirm an increasing burden of cutaneous leishmaniasis in north-east Brazil. *Trans.R.Soc.Trop.Med.Hyg.* 93:488-94.

Britto, C., Ravel, C., Bastien, P., Blaineau, C., Pages, M., Dedet, J-P & Wincker, P. (1998) Conserved linkage groups associated with large-scale chromosomal rearrangements between Old World and New World Leishmania genomes. Gene 222:107-17.

Brooks, D.R., McCulloch, R., Coombs, G.H. & Mottram, J.C. (2000) Stable transformation of trypanosomatids through targeted chromosomal integration of the selectable marker gene encoding blasticidin S deaminase. *FEMS Microbiol.Lett.* **186**:287-91

de Bruijn, M.H.L. & Barker, D.C. (1992) Diagnosis of New World leishmaniasis: specific detection of species of the *Leishmania braziliensis* complex by amplification of kinetoplast DNA. Acta Tropica 52: 145-58.

Bryceson, A.D.M. (1969) Diffuse cutaneous leishmaniasis in Ethiopia. I. The clinical and histological features of the disease. Trans. R. Soc. Trop. Med. Hyg. 63:708-737.

Bryceson, A.D.M. (1970a) Diffuse cutaneous leishmaniasis in Ethiopia. III Immunological studies. Trans. R. Soc. Trop. Med. Hyg. 64:380-7.

Bryceson, A.D.M. (1970b) Diffuse cutaneous leishmaniasis in Ethiopia. II. Treatment. Trans.R.Soc. Trop.Med.Hyg. 64:369-79.

Button, L.L., Russell, D.G., Klein, H.L., Medina-Acosta, E., Karess, R.E. & McMaster, W.R. (1989) Genes encoding the major surface glycoprotein in *Leishmania* are tandemly linked at a single chromosomal locus and are constitutively transcribed. *Mol.Biochem.Parasitol.* 32:271-84.

Cabrera, M., Shaw, M-A., Sharples, C., Williams, H., Castes, M., Convit, J. & Blackwell, J.M. (1995) Polymorphism in tumor necrosis factor genes associated with mucocutaneous leishmaniasis. *J.Exp.Med.* 182:1259-64.

Callen, D.F., Thompson, A.D., Shen, Y., Phillips, H.A., Richards, R.I., Mulley, J.C. & Sutherland, G.R. (1993) Incidence and origin of "null" alleles in the (AC)n microsatellite markers. *Am.J.Hum.Genet.* 52:922-7.

Carranza, S., Giribet, G., Ribera, C., Baguna, J & Riutort, M. (1996) Evidence that two types of rDNA coexist in the genome of *Dugesia (Schmidtea) mediterranea* (Platyhelminthes, Turbellaria, Tricladida). *Mol.Biol.Evol.* 13:824-32.

Carrasco, H.J., Frame, I.A., Valente, S.A. & Miles, M.A. (1996) Genetic exchange as a possible source of genomic diversity in sylvatic populations of *Trypanosoma cruzi*. Am.J.Trop.Med.Hyg. 54:418-24.

Carvalho, E.M., Johnson, W.D., Barreto, E., Marsden, P.D., Costa, J.L.M., Reed, S.G. & Rocha, H. (1985) Cell mediated immunity in American cutaneous and mucosal leishmaniasis. *J.Immunol.* 135:4144-8.

Childs, G.E., Lightner, L.K., McKinney, L., Groves MG, Price EE & Hendricks LD (1984) Inbred mice as model hosts for cutaneous leishmaniasis. I. Resistance and susceptibility to infection with *Leishmania* braziliensis, L. mexicana, and L. aethiopica. Ann. Trop. Med. Parasitol. 78:25-34.

Chouicha, N., Lanotte, G., Pratlong, F., Cuba-Cuba, C.A., Velez, I.D. & Dedet, J.P. (1997) Phylogenetic taxonomy of *Leishmania (Viannia) braziltensis* based on isoenzymic study of 137 isolates. *Parasitology* 115:343-8. Chunge, C.N., Owate, J., Pamba, H.O. & Donno, L. (1990) Treatment of visceral leishmaniasis in Kenya by aminosidine alone or combined with sodium stibo-gluconate. *Trans.R.Soc.Trop.Med.Hyg.* 84:221-5.

Clark, C.G. (1992) Riboprinting: a molecular approach to the identification and taxonomy of protozoa. *pp*. D4.1-4.4. In: Lee, J.J. & Soldo, A.T. (Eds.) *Protocols in Protozoology*. Allen Press (USA).

Clark, C.G. & Diamond, L.S. (1991) The Laredo strain and other Entamoeba histolytica-like amoebae are Entamoeba moshkovskii. Mol.Biochem.Parasitol. 46:11-18.

Clark, C.G. & Pung, O.J. (1994) Host specificity of ribosomal DNA variation in sylvatic Trypanosoma cruzi from North America. Mol.Biochem.Parasitol. 66:175-9.

Clauvel, J.P., Couderc, L.J., Belmin, J., Daniel, M.T., Rabian, C. & Seligmann, M. (1986) Visceral leishmaniasis complicating acquired immunodeficiency syndrome (AIDS). *Trans.R.Soc.Trop.Med.Hyg.* **80**:1010-11.

Clayton, C.E. (1999) Genetic manipulation of Kinetoplastida. Parasitol. Today 15:372-8.

Coburn, C.M., Otteman, K.M., McNeely, T., Turco, S.J. & Beverley, S.M. (1991) Stable DNA transfection of a wide range of trypanosomatids. *Mol.Biochem. Parasitol.* 46:169-80.

Convit, J., Ulrich, M., Fernandez, C.T. Tapia FJ, Caceres-Dittmar G, Castes M. & Rondon AJ (1993) The clinical and immunological spectrum of American cutaneous leishmaniasis. Trans. R. Soc. Trop. Med. Hyg. 87:444-8

Convit, J., Pinardi, M.E. & Rondon, A.J. (1972) Diffuse cutaneous leishmaniasis: A disease due to an immunological defect of the host. Trans. R. Soc. Trop. Med. Hyg. 66:603-10.

Conway, D.J., Roper, C., Oduola, A.M.J., Arnot, D.E., Kremsner, P.G., Grobusch, M.P., Curtis, C.F. & Greenwood, B.M. (1999) High recombination rate in natural populations of *Plasmodium falciparum*. *Proc.Natl.Acad Sci. USA* 96:4506-11.

Costa, J.M.L., Marsden, P.D., Llanos-Cuentas, E.A., Netto, E.M., Carvalho, E.M., Barral, A., Rosa, A.C., Cuba-Cuba, C.A., Magalhæs, A.V. & Barreto, A.C. (1986) Disseminated cutaneous leishmaniasis in a field clinic in Bahia, Brazil: A report of eight cases. *J. Trop.Med.Hyg.* **89**:319-23.

Coukell, A.J. & Brogden, R.N. (1998) Lipsosomal amphotericin B. Therapeutic use in the management of fungal infections and visceral leishmaniasis. Drugs 55:585-612.

Coura, J.R., Galvao-Castro, B. & Grimaldi, G. Jnr. (1987) Disseminated american cutaneous leishmaniasis in a patient with AIDS. Mem. Inst. Oswaldo Cruz \$2:581-2.

Croft, S.L., Urbina, J.A. & Brun, R. (1997) Chapter 14: Chemotherapy of human leishmaniasis and trypanosomiasis *pp.* 245-57 In: Hide G. *et al.* (Eds) *Trypanosomiasis and Leishmaniasis*. CAB International.

Cruz, A. & Beverley, S.M. (1990) Gene replacement in parasitic protozoa. Nature 348:171-3.

Cruz, A.K., Coburn, C.M. & Beverley, S.M. (1991) Double targeted gene replacement for creating null mutants *Proc.Natl.Acad.Sci. USA* **38**:7170-4.

Cruz, A.K., Titus, R. & Beverley, S.M. (1993) Plasticity in chromosome number and testing of essential genes in *Leishmania* by targeting. *Proc.Natl.Acad.Sci. USA* 90:1599-603.

Cuba-Cuba, C.A., Evans, D.A., Rosa, A de C. & Marsden, P.D. (1991) Clonal variation within a mucosal isolate derived from a patient with Leishmania (Viannia) braziliensis infection. Rev.Inst.Med. Trop. Sao Paulo 33:343-50.

Cuba-Cuba, C.A., Miles, M.A., Vexenat, A., Barker, D.C., McMahon-Pratt, D., Butcher, J., Barreto, A.C. & Marsden, P.D. (1985) A focus of mucocutaneous leishmaniasis in Tres Bracos, Bahia, Brazil: characterization and identification of *Leishmania* stocks isolated from man and dogs. *Trans.R.Soc. Trop.Med.Hyg*, **79**:500-7.

Cuba-Cuba, C.A., Netto, E.M., Marsden, P.D., Rosa, A.C., Llanos-Cuentas, E.A. & Costa, J.L.M. (1986) Cultivation of *Leishmania braziliensis braziliensis* from skin ulcers in man under field conditions. *Trans.R. Soc. Trop.Med. Hyg.* 80:456-7.

Cupolillo, E., Grimaldi, G. Jnr. & Momen, H. (1994) A general classification of New World Leishmania using numerical taxonomy. Am.J. Trop. Med. Hyg. 50:296-311.

Cupolillo, E., Grimaldi, G. Jnr., Momen, H. & Beverley, S.M. (1995) Intergenic region typing (IRT): A rapid molecular approach to the characterization and evolution of *Leishmania*. *Mol.Biochem.Parasitol.* 73:145-55.

Cupolillo, E., Medina-Acosta, E., Noyes, H., Momen, H. & Grimaldi, G. Jnr. (2000) A revised classification for *Leishmania* and *Endotrypanum*. *Parasitol. Today* 16:142-4.

Darce, M., Moran, J., Palacios, X., Belli, A., Gomez-Urcuyo, F., Zamora, D., Valle, S., Gantier, J.C., Momen, H. & Grimaldi, G. Jnr. (1991) Etiology of human cutaneous leishmaniasis in Nicaragua. *Trans.R.Soc. Trop. Med. Hyg.* **85**:58-9.

David, V.A. & Menotti-Raymond, M. (1998) Automated DNA detection with fluorescence-based technologies. Chapter 10, pp. 337-69. In: Hoelzel, A.R. (Ed.) Molecular genetic analysis of populations: a practical approach. Second Edition. I.R.L. Press (Oxford).

Davies, C.R., Llanos-Cuentas, E.A., Pyke, S.D.M. & Dye, C. (1995) Cutaneous leishmaniasis in the Peruvian Andes: an epidemiological study of infection and immunity. *Epidemiol.Infect.* 114:297-318.

Degrave, W., Fernandes, O., Campbell, D., Bozza, M. & Lopes, U. (1994a) Use of molecular probes and PCR for detection and typing of *Leishmania* - a mini-review. *Mem.Inst. Oswaldo Cruz* 89:463-9.

Degrave, W., Fernandes, O., Thiemann, O., Wincker, P., Britto, C., Cardoso, A., Pereira, J.B., Bozza, M., Lopes, U. & Morel, C. (1994b) Detection of *Trypanosoma cruzi* and *Leishmania* using the polymerase chain reaction. *Mem.Inst.Oswaldo Cruz* 89:367-8.

Delgado, O., Cupolillo, E., Bonfante-Garrido, R., Silva, S., Belfort, E., Grimaldi, G. Jnr., Momen, H. (1997) Cutaneous leishmaniasis in Venezuela caused by infection with a new hybrid between Leishmania (Viannia) braziliensis and L. (V.) guyanensis. Mem.Inst.Oswaldo Cruz 92:581-2.

Desjeux, P. (1992a) Information on the epidemiology and control of the leishmaniases by country or territory. World Health Organization document WHO/ Leish/91.30.

Desjeux, P. (1992b) Human leishmaniases: epidemiology and public health aspects. World Health Stat. Q. 45:267-75.

Di Rienzo, A., Peterson, A.C., Garza, J.C., Valdes, A.M., Slatkin, M. & Freimer, N.B. (1994) Mutational processes of simple-sequence repeat loci in human populations. Proc.Natl.Acad. Sci. USA 91:3166-70.

Dogra, J. & Saxena, V.N. (1996) Itraconazole and leishmaniasis: a randomised double-blind trial in cutaneous disease. Int.J.Parasitol. 26:1413-5.

Drocourt, D., Calmels, T., Reynes, J.P., Baron, M. & Tiraby, G. (1990) Cassettes of *Streptoalloteichus* hindustanus ble gene for transformation of lower and higher eukaryotes to phleomycin resistance. Nucl. Acid Res. 18:4009.

Dube, A., Sharma, P., Srivastava, J.K., Miara, A., Naik, S. & Katiyar, J.C. (1998) Vaccination of langur monkeys (*Presbytis entellus*) against *Leishmania donovani* with autoclaved *L. major* plus BCG. *Parasitology* 116:219-21.

Dujardin, J.C., Banuls, A-L., Dujardin, J.P., Arevalo, J., Tibayrenc, M. & Le Ray, D. (1998) Comparison of chromosome and isoenzyme polymorphism in geographical populations of *Leishmania* (Viannia) peruviana. Parasitology 117:547-54.

Dujardin, J.C., Banuls, A-L., Llanos-Cuentas, A., Alvarez, E., de Donecker, S., Jacquet, D., Le Ray, D., Arevalo, J. & Tibayrenc, M. (1995b) Putative *Leishmania* hybrids in the Eastern Andean valley of Huanuco, Peru. Acta Tropica 59:293-307.

Dujardin, J.C., de Donecker, S., Victoir, K., Le Ray, D., Hamers, R. & Arevalo, J. (1994) Size polymorphism of chromosomes bearing gp63 genes in *Leishmania* of the braziliensis complex: indication of a rearrangement of the gp63 genes in *L. braziliensis* and *L. peruviana*. Ann. Trop. Med. Parasitol. 88:445-8.

Dujardin, J.C., Dujardin, J.P., Tibayrenc, M., Timperman, G., de Donecker, S., Jacquet, D., Arevalo, J., Llanos-Cuentas, A., Guerra, H., Bermudez, H., Hamers, R. & Le Ray, D. (1995a) Karyotype plasticity in Neotropical Leishmania: an index for measuring genomic distance among L.(V.) peruviana and L. (V.) braziliensis populations. Parasitology 110:21-30.

Dujardin, J.C., Gajendran, N., Arevalo, J., Llanos-Cuentas, A., Guerra, H., Gomez, J., Arroyo, J., De Donecker, S., Jacquet, D., Hamers, R. & Le Ray, D. (1993a) Karyotype polymorphism and conserved characters in the *Leishmania (Viannia) braziliensis* complex explored with chromosome-derived probes. *Ann.Soc.Belge de Med.Trop.* 73:101-18.

Dujardin, J.C., Llanos-Cuentas, A., Caceres, A., Arana, M., Dujardin, J.P. Guerrini, F., Gomez, J., Arroyo, J., De Donecker, S., Jacquet, D., Hamers, R., Guerra, H., Le Ray, D. & Arevalo, J. (1993b) Molecular karyotype variation in *Leishmania (Viannia) peruviana*: indication of geographical populations in Peru distributed along a north-south cline. *Ann. Trop. Med. Pstol.* **87**:335-47.

Ebert, F. (1987) Isoenzyme studies on *Leishmania* stocks from Peru by ultrathin-layer isoelectricfocusing. *Trop.Med.Parasitol.* 38:37-40.

Echevarría, J., Campos, P., Chang, J., Cuéllar, L., Gotuzzo, E., Paz, L. & Llanos-Cuentas, A. (1993) Mucocutaneous leishmaniasis and AIDS: case report. *Trans.R.Soc. Trop. Med. Hyg.* 87:186.

Ellis, J.T. & Crampton, J.M. (1991) Differences between Leishmania (Leishmania) chagasi, L. (L.) infantum and L. (L.) donovani as shown by DNA fingerprinting. Mem.Inst.Oswaldo Cruz 86:479-81.

Eperon, S. & McMahon-Pratt, D. (1989) Extracellular amastigote-like forms of *Leishmania* panamensis and *L. braziliensis*. II Stage- and species-specific monoclonal antibodies. *J. Protozool.* **36**:510-8.

Eresh, S., Mendoza-Leon, A. & Barker, D.C. (1993) A small chromosome of *Leishmania (Viannia)* braziliensis contains multicopy sequences which are complex specific. Acta Tropica 55:33-46.

Escobar, M.A., Saravia, N.G. & Weigle, K. (1996) Concurrent mucosal leishmaniasis and pulmonary tuberculosis: two case reports and review. *J.Clin.Infect. Dis.* 23:836-7.

Espinoza, J., Skinner, A.C., Davies, C.R., Llanos-Cuentas, A., Arevalo, J., Dye, C., McMaster, W.R., Ajioka, J.W. & Blackwell, J.M. (1995) Extensive polymorphism at the Gp63 locus in field isolates of Leishmania peruviana. Mol.Biochem.Parasitol. 72:203-13.

Estoup, A., Garnery, L., Solignac, M. & Cornuet, J.M. (1995a) Microsatellite variation in honey bee (*Apis mellifera L.*) populations: hierarchical genetic structure and test of the infinite allele and stepwise mutation models. *Genetics* 140:679-95.

Estoup, A., Tailliez, C., Cornuet, J.M. & Solignac, M. (1995b) Size homoplasy and mutational processes of interrupted microsatellites in two bee species, *Apis mellifera* and *Bombus terrestris* (Apidae). Mol. Biol. Evol. 12:1074-84.

Evans, D.A., Godfrey, D., Lanham, S., Lanotte, G., Modabber, F. & Schnur, L. (1989) Handbook on isolation, characterization and cryopreservation of *Leishmania*. UNDP/World Bank/WHO special programme for research and training in tropical diseases (TDR), Geneva, Switzerland.

Evans, D.A., Kennedy, W.P.K., Elbihari, S., Chapman, C.J., Smith, V. & Peters, W. (1987) Hybrid formation within the genus *Leishmania*. *Parasitologia* 29:165-73.

Evans, D.A., Lanham, S.M., Baldwin, C.I. & Peters, W. (1984) The isolation and isoenzyme characterization of *Leishmania braziliensis* subsp. from patients with cutaneous leishmaniasis acquired in Belize. *Trans.R.Soc. Trop.Med.Hyg.* **78**:35-42.

Falqueto, A., Sessa, P.A., Varejao, J.B., Barros, G.C., Momen, H. & Grimaldi, Jnr. G. (1991) Leishmaniasis due to *Leishmania braziliensis* in Espirito Santo State, Brazil. Further evidence on the role of dogs as a reservoir of infection for humans. *Mem.Inst.Oswaldo Cruz* **36**:499-500.

Felinto de Brito, M.E., Brandao, S.P., Salles, N.R., Cupolillo, E., Grimaldi, G. Inr. & Momen, H. (1993) Human cutaneous leishmaniasis due to a new enzymatic variant of *Leishmania (Viannia)* braziliensis occuring in Pernambuco, Brazil. Mem.Inst. Oswaldo Cruz **88**:633-4.

Fernandes, O., Murthy, V.K., Kurath, U., Degrave, W.M. & Campbell, D.A. (1994) Mini-exon gene variation in human pathogenic *Leishmania* species. *Mol.Biochem.Parasitol.* **66**:261-71.

Fernandes, A.P., Nelson, K. & Beverley, S.M. (1993) Evolution of nuclear ribosomal RNAs in kinetoplastid protozoa: perspectives on the age and origins of parasitism. *Proc.Natl.Acad.Sci. USA* 90:11608-12.

Fiorentino, D.F., Zlotnik, A., Mossmann, T.R., Howard, M., Moore, K.W. & O'Garra, A. (1991) IL-10 acts on the antigen-presenting cell to inhibit cytokine production. *J.Immunol.* 146:3444-51.

Fischer, S.G. & Lerman, L.S. (1983) DNA fragments differing by single base pair substitutions are separated in denaturing gradient gels: correspondence with melting theory. *Proc.Natl.Acad.Sci. USA* **80**:1579.

da Fonseca, A.L., Vallochi, A.L., Furtado, G. de C., Abrahamsohn, I. de A. & Lima, G.M. (1997) Immune response and protection in mice inoculated with *Leishmania amazonensis* clones expressing different degrees of virulence. *Parasitol.Res.* 83:690-7.

França, F., Lago, E.L., Tada, M.S., Costa J.M.L., Val, K., Oliviera, J., Costa, M.A, Osaki, M., Cheever, L., Netto, E.M., Barreto, E.C., Johnson, E.D. & Marsden, P.D. (1991) An outbreak of *Leishmania* (Viannia) braziliensis infection. Mem.Inst. Oswaldo Cruz **36**:169-74.

Franke, E.D., Lucas, C.M., Tovar, A.A., Kruger, J.H., De Rivera, M.V. & Wignall, F.S. (1990) Diffuse cutaneous leishmaniasis acquired in Peru. Am.J. Trop. Med. Hyg. 43:260-2.

Freedman, D.J. & Beverley, S.M. (1993) Two more independent selectable markers for stable transfection of Leishmania. Mol.Biochem.Parasitol. 62: 33-44.

Fu, G., Perona-Wright, G. & Barker, D.C. (1998) Leishmania braziliensis: characterization of a complex-specific subtelomeric repeat sequence and its use in the detection of parasites. Exptl. Parasitol. 90: 236-43.

Garza, J.C. & Freimer, N.B. (1996) Homoplasy for size at microsstellite loci in humans and chimpanzees. Genome Res. 6:211-7.

Garza, J.C., Slatkin, M., Freimer, N.B. (1995) Microsatellite allele frequencies in humans and chimpanzees, with implications for constraints on allele size. *Mol.Biol.Evol.* 12:594-603.

Gasser, R., Nansen, P. & Guldberg, P. (1996) Fingerprinting sequence variation in ribosomal DNA of parasites by DGGE. *Molecular and Cellular Probes* 10:99-105.

Ghalib, H.W., Eltoum, E.A., Kroon, C.C.M. & El Hassan, A.M. (1992) Identification of Leishmania from mucosal leishmaniasis by recombinant DNA probes. Trans. R. Soc. Trop. Med. Hyg. 86:158-60.

Ghalib, H.W., Piuvezam, M.R., Skeiky, Y.A.W., Siddig, M., Hashim, F.A., El Hassan, A.M., Russo, D.M. & Reed, S.G. (1993) Interleukin 10 production correlates with pathology in human *Leishmania* donovani infections. *J.Clin.Invest.* 92:324-9.

Giannini, S.H., Curry, S.S., Tesh, R.B. & Van der Ploeg, L.H.T. (1990) Size-conserved chromosomes and stability of molecular karyotypes in cloned stocks of *Leishmania major* Mol.Biochem.Parasitol. **39**:9-22.

Gibson, W.C. (1989) Analysis of a genetic cross between Trypanosoma brucei rhodesiense and T. b. brucei. Parasitology 99:391-402.

Gibson, W.C. (1990) Trypanosome diversity in Lambwe Valley, Kenya - sex or selection? *Parasitol. Today* 6:342-3.

Gibson, W.C. (1995) The significance of genetic exchange in trypanosomes. Parasitol. Today 11:465-8.

Gibson, W.C. & Bailey, M. (1994) Genetic exchange in *Trypanosoma brucei*: evidence for meiosis from analysis of a cross between drug-resistant transformants. *Mol.Biochem.Parasitol.* **64**: 241-52.

Gibson, W.C., Borst, P. & Fase-Fowler, F. (1985) Further analysis of intraspecific variation in *Trypanosoma brucei* using restriction site polymorphisms in the maxicircle of kinetoplast DNA. *Mol.Biochem.Parasitol.* 15:21-36.

Gibson, W.C. & Garside, L. (1990) Kinetoplast DNA minicircles are inherited from both parents in genetic hybrids of *Trypanosoma brucei*. Mol.Biochem.Parasitol. 42:45-54.

Gibson, W.C., Garside, L. & Bailey, M. (1992) Trisomy and chromosome size changes in hybrid trypanosomes from a genetic cross between African trypanosomes in vitro. Mol. Biochem.Parasitol. 52:189-200.

Gibson, W.C., Marshall, T.F. & Godfrey, D.G. (1980) Numerical anlysis of enzyme polymorphisms: a new approach to the epidemiology and taxonomy of trypanosomes of the subgenus *Trypanozoon*. *Adv.Parasitol.* 18:175-246.

Gibson, W.C. & Miles, M.A. (1986) The karyotype and ploidy of *Trypanosoma cruzi*. *EMBO.J.* 5: 1299-305

Gibson, W.C. & Stevens, J. (1999) Genetic exchange in the Trypanosomatidae. Adv. Parasitol. 43:1-46.

Gibson, W.C. & Whittington, H. (1993) Genetic exchange in trypanosoma bruce: selction of hybrid trypanosomes by introduction of genes conferring drug resistance. *Mol.Biochem.Parasitol.* **60**:19-26.

Godfrey, D.G. & Kilgour, V. (1976) Enzyme electrophoresis in characterizing the causative organism of Gambian trypanosomiasis. *Trans.R.Soc.Trop.Med.Hyg.* **70**:219-24.

Golino, A., Duncan, J., Zeluff, B., Golino, A., Duncan, JM, Zeluff, B., DePriest, J. McAllister HA, Radovancevic B., & Frazier OH (1992) Leishmaniasis in a heart transplant patient. J. Heart Lung Trans. 11:820.

Gomes, R.F., Macedo, A.M., Pena, S.D.J. & Melo, M.N. (1995) *Leishmania (Viannia) braziliensis:* genetic relationships between strains isolated from different areas of Brazil as revealed by DNA fingerprinting and RAPD. *Exp. Parasitol.* **30**:681-7.

Gramiccia, M., Smith, D.F., Angelici, M.C., Ready, P.D. & Gradoni, L. (1992) A kinetoplast DNA probe diagnostic for *Leishmania infantum.* Parasitology 105:29-34.

Grimaldi, G. Jnr. (1995) Meetings on vaccine studies towards the control of leishmaniasis. Mem.Inst. Oswaldo Cruz. 90:553-6.

Grimaldi, G. Jnr., David, J.R. & McMahon-Pratt, D. (1987) Identification and distribution of New World Leishmania species characterized by serodeme analysis using monoclonal antibodies. *Am.J.Trop.Med.Hyg.* 36:270-87.

Grimaldi, G. Jnr., Kreutzer, R.D., Hashiguchi, Y., Gomez L, E.A., Mimori, T. & Tesh, R.B. (1992) Description of *Leishmania equatorensis* sp.n. (Kinetoplastida: *Trypanosomatidae*), a new parasite infecting arboreal mammals in Ecuador. *Mem. Inst. Oswaldo Cruz* 87:221-8. Grimaldi, G. Jnr. & McMahon-Pratt, D. (1996) Monoclonal antibodies for the identification of New World Leishmania species. Mem.Inst.Oswaldo Cruz 91:37-42.

Grimaldi, G. Jnr., Momen, H., Naiff, R.D., McMahon-Pratt, D. & Barrett, T.V. (1991) Characterization and classification of leishmanial parasites from humans, wild mammals, and sand flies in the Amazon region of Brazil. Am.J.Trop.Med.Hyg. 44:645-61.

Grimaldi, G. & Tesh, R.B. (1993) Leishmaniases of the New World: current concepts and implications for future research. *Clin.Micro.Rev.* 6:230-50.

Grimaldi, M.C. & Cronau-Roy, B. (1997) Microsatellite allelic homplasy due to variable flanking sequences. J.Mol. Evol. 44:336-40.

Guerra, H. (1988) Distribution of Leishmania in Peru. pp. 135-45. In: Walton, B.C., Wijeyaratne, P.M. & Modabber, F. (Eds.) Research on control strategies for the leishmaniases IDRC-MR 184e, Ottawa.

Guevara, P., Alonso, G., da Silveira, J.F., de Mello, M., Scorza, J.V., Anez, N. & Ramirez, J.L. (1992) Identification of new world *Leishmania* using ribosomal gene spacer probes. *Mol.Biochem.Parasitol.* 56:15-26.

Guevara, P., Ramirez, J.L., Rojas, E., Scorza, J.V., Gonzalez, N. & Anez, N. (1993) Leishmania braziliensis in blood 30 years after cure. Lancet 341:1341.

Guevara, P., Rojas, E., Gonzalez, N., Scorza, J.V., Anez, N., Valera, M. & Ramirez, J.L. (1994) Presence of *Leishmania braziliensis* in blood samples from cured patients or at different stages of immunotherapy. *Clin.Diag.Lab.Immunol.* 1:385-9.

Guizani, I., Van Eys, G.J.J.M., Ismail, R.B. & Dellagi, K. (1994) Use of recombinant DNA probes for species identification of Old World *Leishmania* isolates. *Am.J.Trop.Med.Hyg.* 50:632-40.

Gutierrez, Y., Salinas, G.H., Palma, G. Gutierrez Y, Salinas GH, Palma G, Valderrama LB, Santrich CV, Saravia NG (1991) Correlation between histopathology, immune response, clinical presentation, and evolution in *Leishmania braziliensis* infection. *Am.J.Trop.Med.Hyg.* **45**:281-9.

Ha, D.S., Schwarz, J.K., Turco, S.J. & Beverley, S.M. (1996) Use of the green fluorescent protein as a marker in transfected *Leishmania*. *Mol.Biochem.Parasitol*. 77:57-64.

Hall, L.R. & Titus, R.G. (1995) Sandfly vector saliva selectively modulates macrophage functions that inhibit killing of *Leishmania major* and nitric oxide production. *J.Immunol.* 155:3501-6.

Hamada, H., Petrino, M.G. & Kakumaga, T. (1982) A novel repeated element with Z-DNA forming potential is widely found in evolutionarily diverse eukaryotic genomes. *Proc.Natl.Acad. Sci. USA* 79:6465-9

Handman, E. (1997) Leishmania vaccines: old and new Parasitol. Today 13:236-8.

Handman, E., Symons, F.M., Baldwin, T.M., Curtis, J.M. & Scheerlinck, J.P. (1995) Protective vaccination with promastigote surface antigen 2 from *Leishmania major* is mediated by a TH1 type of immune response. *Infect.Immun.* 63:4261-7.

Hanekamp, T. & Langer, P.J. (1991) Molecular karyotype and chromosomal location of genes ecnoding two major surface glycoproteins, gp63 & gp46/M2, Hsp70 & B-tubulin in cloned strains of several *Leishmania* species. *Mol.Biochem.Parasitol.* **48**:27-38.

Hanman, C.A., Zhao, F., Shaw, J.J. & Lainson, R. (1991) Monoclonal antibodies for the identification of New World Leishmania. Trans. R. Soc. Trop. Med. Hyg. 85:220.

Harith, A.E., Kolk, A.H.J., Kager, P.A., Leeuwemburg, J., Muigai, R., Kiugu, S. & Laarman, J.J. (1986) A simple and economical direct agglutination test for serodiagnosis and sero-epidemiological studies of visceral leishmaniasis. *Trans.R.Soc. Trop.Med.Hyg.* **80**:583-7.

Harris, H. & Hopkinson, D.A. (1976) Chapter 4, The enzymes. In: Handbook of enzyme electrophoresis in human genetics. Amsterdam: North Holland.

Harris, E., Kropp, G., Belli, A., Rodriguez, B. & Agabian, N. (1998) Single-step multiplex PCR assay for characterization of New World *Leishmania* complexes. *J.Clin.Micro.* 36:1989-95.

Harry, M., Poyet, G., Romana, C.A. & Solignac, M. (1998) Isolation and characterization of microsatellite markers in the bloodsucking bug *Rhodnius pallescens* (*Heteroptera*, *Reduviidae*). *Mol.Ecol.* 7:1784-6.

Hassan, M.D.Q., Ghosh, A., Ghosh, S.S., Gupta, M., Basu, D., Mallik, K.K. & Adhya, S. (1993) Enzymatic amplification of mini-exon-derived RNA gene spacers of *Leishmania donovani* : primers and probes for DNA diagnosis. *Parasitology* 107:509-17.

Hawkins, J.R. (1997) Finding Mutations: the basics. IRL Press, Oxford.

Hedrick, P. W. (1985) Genetics of Populations. Jones & Bartlett Publishers, Inc. Boston, USA.

Hernández-Montes, O., Monroy-Ostria, A., McCann, S. & Barker, D.C. (1998) Identification of Mexican Leishmania species by analysis of PCR amplified DNA. Acta Trop. 71:139-53.

Hernández, D., Rodriguez, N., Martínez, C., García, L. & Convit, J. (1993) Leishmania braziliensis causing visceral leishmaniasis in a patient with human immunodeficiency virus infection, identified with the aid of the polymerase chain reaction. Trans. R. Soc. Trop. Med. Hyg. 87:627-8.

Herwaldt, B.L., Arana, B.A., & Navin, T.R. (1992) The natural history of cutaneous leishmaniasis in Guatemala. J.Infect.Dis. 165:518-27.

Hill, J.O., Awwad, M. & North, R.J. (1989) Elimination of CD4 suppressor T cells from susceptible BALB/c mice releases CD8 T lymphocytes to mediate protective immunity against *Leishmania*. *J.Exp.Med.* 169:1819-27.

Hill, W.G. & Robertson, A. (1968) Linkage disequilibrium in finite populations. *Theor.Appl.Genet.* 38: 226-31.

Ho, J.L., He, S.H., Rios, M.J.C. & Wick, E.A. (1992) Interleukin-4 inhibits human macrophage activation by tumor necrosis factor, granulocyte-monocyte colony stimulating factor, and interleukin-3 for antileishmanial activity and oxidative burst capacity. J.Inf.Dis. 165:344-51.

Hommel, M., Peter, W., Ranque, J., Quilici, M. & Lanotte, G. (1978) The micro-ELISA technique in the serodiagnosis of visceral leishmaniasis. Ann. Trop. Med. Parasitol. 72:213-8.

Howard, M.K., Kelly, J.M., Lane, R.P. & Miles, M.A. (1991) A sensitive repetitive DNA probe that is specific to the *Leishmania donovani* complex and its use as an epidemiological and diagnostic reagent. *Mol.Biochem.Parasitol.* 44:63-72.

Ibrahim, M.E., Smyth, A.J., Ali, M.H., Barker, D.C. & Kharazmi, A. (1994) The polymerase chain reaction can reveal the occurrence of naturally mixed infections with *Leishmania* parasites. *Acta Tropica* 57:327-32.

Inga, R., De Doncker, S., Gomez, J., Lopez, M., Garcia, R., Le Ray, D., Arevalo, J. & Dujardin, J-C. (1998) Relation between variation in copy number of ribosomal RNA encoding genes and size of harbouring chromosomes in *Leishmania* of the subgenus *Viannia*. *Mol.Biochem.Parasitol*. 92:219-28.

Iovannisci, D.M. & Beverley, S.M. (1989) Structural alterations of chromosome 2 in *Leishmania major* as evidence for diploidy, including spontaneous amplification of the mini-exon array. *Mol. Biochem.Parasitol.*, 34:177-88.

Jaccard, P. (1908) Nouvelles recherches sur la distribution florale. Bull.Soc.Vaudoise Sci.Nat. 44:223-70.
Jaffe, C.L., Bennett, E., Grimaldi, G. Jnr. & McMahon-Pratt, D. (1984) Production and characterization of monoclonal antibodies against *Leishmania donovani* for immunodiagnosis. *J.Immunol.* 133:440-7.

Jardim, A., Funk, V., Caprioli, R.M. & Olafson, R.W. (1995) Isolation and characterization of the *Leishmania donovani* kinetoplastid membrane protein-11, a major immunoreactive membrane glycoprotein. *Biochem.J.* 305:307-13.

Jeffreys, A.J., Monckton, D.G., Tamaki, K., Neil, D.L., Armour, J.A.L., MacLeod, A., Collick, A., Allen, M. & Jobling, M. (1993) Minisatellite variant repeat mapping: application to DNA typing and mutation analysis. pp. 125-39. In: Pena, S.D.J. et al. (Eds) DNA Fingerprinting: State of the Science. Birkhauser Verlag (Switzerland).

Jeffreys, A.J., Wilson, V. & Thein, S.L. (1985) Hypervariable "minisatellite" regions in human DNA. Nature 314:67-73

Jenni, L., Marti, S., Schweizer, J., Betschart, B., Le Page, R.W.F., Wells, J.M., Tait, A., Paindavoine, P., Pays, E. & Steinert, M. (1986) Hybrid formation between African trypanosomes during cyclical transmission. *Nature* 322:173-5.

Jensen, A.T., Gaafar, A., Ismail, A., Christensen, C.B., Kemp, M., Hassan, A.M., Kharazami, A. & Theander, T.G. (1996) Serodiagnosis of cutaneous leishmaniasis: assessment of an enzyme-linked immunosorbent assay using a peptide sequence from gene B protein. *Am.J.Trop.Med.Hyg.* 55:490-5.

Jeronimo, S.M.B. & Pearson, R.D. (1989) Leishmaniasis: an update. Curr. Opinions in Inf. Dis. 2:631-8.

Jha, T.K., Sundar, S., Thakur, C.P., Bachmann, P., Karbwang, J., Fischer, C., Voss, A. & Berman, J. (1999) Miltefosine, an oral agent, for the treatment of Indian visceral leishmaniasis. *N.Engl.J.Med.* 341:1795-800.

Jones, T.C., Johnson, W.D., Barretto, A.C., Lago E, Badaro R, Cerf B, Reed SG, Netto EM, Tada MS, Franca TF, Wiese, K., Golightly, L., Fikrig, E., Costa, J.M.L., Cuba, C.C. & Marsden, P.D. (1987) Epidemiology of American cutaneous leishmaniasis due to *Leishmania braziliensis braziliensis*. J.Inf.Dis. 156:73-83.

Kain, K.C., Craig, A.A. & Ohrt, C. (1996) Single stranded conformational polymorphism analysis differentiates *Plasmodium falciparum* treatment failures from reinfections. *Mol.Biochem.Parasitol.* 79:167-75.

Kapler, G.M., Coburn, C.M. & Beverley, S.M. (1990) Transfection of the human parasite *Leishmania* delineates a 30kb region sufficient for extrachromosomal replication and expression. *Mol.Cell.Biol.* 10:1084-94.

Kazan, K., Manners, J.M. & Cameron, D.F. (1993) Inheritance of random amplified polymorphic DNA markers in an interspecific cross in the genus *Stylosanthes*. Genome 36:50-6.

Kelly, J.M. (1993) Isolation of RNA and DNA from *Leishmania*. pp123-31 In: Hyde, J.E. (Ed.) *Methods in molecular biology, Vol.21: Protocols in molecular parasitology*, Humana Press, Totowa, NY.

Kelly, J.M. (1995a) Trypanosomatid shuttle vectors: new tools for the functional dissection of parasite genomes. *Parasitol.Today* 11:447-51

Kelly, J.M. (1997) Genetic transformation of parasitic protozoa. Adv. Parasitol. 39:227-270.

Kelly, J.M., Law, J.M., Chapman, C.J., van Eys, G.J.J.M. & Evans, D.A. (1991) Evidence of genetic recombination in *Leishmania*. Mol.Biochem.Parasitol. 46:253-64.

Kelly, J.M., Taylor, M.C., Rudenko, G. & Blundell, P.A. (1995b) Transfection of the African and American trypanosomes. pp349-60. In: Nickoloff, J.A. (Ed.) Methods in Molecular Biology Vol 47: Protocols for electroporation and electroprofusion of microorganisms. Humana Press, Totowa, NY. Kelly, J.M., Ward, H. M., Miles, M.A. & Kendall, G. (1992) A shuttle vector which facilitates the expression of transfected genes in *Trypanosoma cruzi* and *Leishmania*. Nucl.Acid Res. 20:3963-69.

Killick-Kendrick, R. (1990) Phlebotomine vectors of the leishmaniases: a review. Med.Vet.Entomol. 4:1-24.

Killick-Kendrick, R. (1999) The biology and control of Phlebotomine sand flies. *Clinics in Dermatol.* 17:279-89.

Killick-Kendrick, R., Killick-Kendrick, M., Focheux, C., Dereure, J., Puech, M.P. & Cadiergues, M.C. (1997) Protection of dogs from bites of phlebotomine sandflies by deltamethrin collars for control of canine leishmaniasis. *Med. Vet Entomol.* 11:105-11.

Kimura, M. & Ohta, T. (1978) Stepwise mutation model and distribution of allelic frequencies in a finite population. *Proc Natl Acad Sci U S A* 75:2868-72.

Kreutzer, R.D. (1996) Genetic similarity among Central and South American populations of Leishmania (Viannia) braziliensis. Am.J.Trop.Med.Hyg. 55:106-10.

Kreutzer, R.D. & Christensen, H.A. (1980) Characterization of *Leishmania* spp. by isoenzyme electrophoresis. *Am.J. Trop. Med. Hyg.* 29:199-208.

Kreutzer, R.D., Corredor, A., Grimaldi, G. Jnr., Grogl, M., Rowton, E.D., Young, D.G., Morales, A., McMahon-Pratt, D., Guzman, H. & Tesh, R.B. (1991) Characterization of *Leishmania colombiensis* sp.n. (Kinetoplastida: Trypanosomatidae), a new parasite infecting humans, animals, and Phlebotomine sand flies in Colombia and Panama. *Am.J.Trop.Med.Hyg.* 44:662-75.

Kreutzer, R.D., Souraty, N. & Semko, M.E. (1987) Biochemical identities and differences among *Leishmania* species and subspecies. *Am.J.Trop.Med.Hyg.* 36:22-32.

Kreutzer, R.D., Yemma, J.J., Grogl, M., Tesh, R.B. & Martin, T.I. (1994) Evidence of sexual reproduction in the protozoan parasite *Leishmania* (Kinetoplastida: *Trypanosomatidae*). Am.J.Trop. Med.Hyg. 51:301-7.

Krobitsch, S. & Clos, J. Cross-species homologous recombination in *Leishmania donovani* reveals the sites of integration. *Mol. Biochem. Parasitol.* 107:123-8.

Kweider, M., Lemesre, J.L., Darcy F, Kusnierz JP, Capron A. & Santoro F. (1987) Infectivity of Leishmania braziliensis promastigotes is dependent on the increasing expression of a 65000-dalton surface antigen. J.Immunol. 138:299-305.

Laban, A., Tobin, J.F., de Lafaille, M.A.C. & Wirth, D.F. (1990) Stable expression of the bacterial neo^F gene in Leishmania enrietti. Nature 343:572-4.

Lainson, R. & Shaw, J.J. (1972) Leishmaniasis in the New World: taxonomic problems. Brit.Med. Bull. 28:44-8.

Lainson, R. & Shaw, J.J. (1987) Evolution, classification and geographical distribution. Chapter 1, pp 1-120. In: Peters, W. & Killick-Kendrick, R. (Eds.) The Leishmaniases in Biology and Medicine Vol. 1. Biology and Epidemiology. Academic Press (London).

Lainson, R., Shaw, J.J., Silveira, F.T., de Souza, A.A.A., Braga, R.R. & Ishikawa, E.A.Y. (1994) The dermal leishmaniases of Brazil, with special reference to the eco-epidemiology of the disease in Amazonia. *Mem.Inst.Oswaldo Cruz* 89:435-43.

Lainson, R., Shaw, J.J., Ready, P.D., Miles, M.A. & Povoa, M. (1981) Leishmaniasis in Brazil: XVI. Isolation and identification of *Leishmania* species from sand flies, wild mammals and man in north Para State, with particular reference to *L. braziliensis guyanensis* causative agent of "pian-bois". *Trans.R.* Soc. Trop. Med. Hyg. 75:530-6.

Landfear, S.M., McMahon-Pratt, D. & Wirth, D.M. (1983) Tandem arrangement of tubulin genes in the protozoan parasite *Leishmania enrietti*. Mol.Cell.Biol. 3:1070-6.

Lanham, S.M., Grendon, J.M., Miles, M.A., Povoa, M.M. & de Souza, A.A. (1981) A comparison of electrophoretic methods for isoenzyme characterization of trypanosomatids. I: Standard stocks of *Trypanosoma cruzi* zymodemes from northeast Brazil. *Trans.R.Soc.Trop.Med.Hyg.* **75**, 742-50.

Lanotte, G. & Rioux, J.A. (1990) Fusion cellulaire chez les Leishmania (Kinetoplastida, Trypanosomatidae). C.R.Acad.Sci.Paris, t.310, serie III:285-88.

Larsen, C.L. (1994) Genotypic diversity in Leishmania braziliensis from Trés Braços, Brazil. MSc project, LSHTM, London.

Le Blancq, S.M., Cibulskis, R.E. & Peters, W. (1986) Leishmania in the Old World: 5. Numerical analysis of isoenzyme data. Trans. R. Soc. Trop. Med. Hyg. 30:517-24.

LeBowitz, J.W., Coburn, C.M. & Beverley, S.M. (1991) Simultaneous transient expression assays of the trypanosomatid parasite *Leishmania* using beta-galatosidase and beta-glucoronidase as reporter enzymes. *Gene* 103:119-23.

LeBowitz, J.W., Coburn, C.M., McMahon-Pratt, D. & Beverley, S.M. (1990) Development of a stable Leishmania expression vector and application to the study of parasite surface antigens. Proc.Natl.Acad.Sci. USA 87:9736-40.

LeBowitz, J.W., Cruz, A. & Beverley, S.M. (1992) Thymidine kinase as a negative selectable marker in *Leishmania major. Mol.Biochem.Parasitol.* 51:321-5.

Lee, M.G., Atkinson, B.L., Giannini, S.H. & van der Ploeg, L.H. (1988) Structure and expression of the hsp 70 gene family of Letshmania major. Nucl. Acids Res. 16:9567-85

Lee, M.G.S. & Van der Ploeg, L.H.T. (1991) Homologous recombination and stable transfection in the parasitic protozoa *Trypanosoma brucei*. Science 250:1583-7.

Lehmann, T., Besansky, N.J., Hawley, W.A., Fahey, T.G., Kamau, L. & Collins, F.H. (1997) Microgeographic structure of *Anopheles gambiae* in western Kenya based on mtDNA and microsatellite loci. *Mol.Ecol.* 6:243-53.

Lehmann, T., Hawley, W.A. & Collins, F.H. (1996a) An evaluation of evolutionary constraints on microsatellite loci using null alleles. *Genetics* 144:1155-63.

Lehmann, T., Hawley, W.A., Kamau, L., Fontenille, D., Simard, F. & Collins, F.H. (1996b) Genetic differentiation of *Anopheles gambiae* populations from East and west Africa: comparison of microsatellite and allozyme loci. *Heredity* 77:192-200.

Leiby, D.A., Kanesa-thasan, N., Scott, P. & Nacy, C.A. (1994) Leishmaniasis. pp 87-118. In: Kierszenbaum, F. (Ed.) Parasitic Infections and the Immune System. Academic Press (London).

Leon, W., Fouts, D. & Manning, J. (1978) Sequence arrangement of the 16S and 26S rRNA genes in the pathogenic haemoflagellate Leishmania donovani. Nucl. Acid Res. 5:491-504.

Lessa, E.P. (1992) Rapid surveying of DNA sequence variation in natural populations. *Mol.Biol.Evol.* 9:323-30.

Levinson, G. & Gutman, G.A. (1987) Slipped-strand mispairing: a major mechanism for DNA sequence evolution. *Mol.Biol.Evol.* 4:203-21.

Lewontin, R.C. (1964) The interaction of selection and linkage. I: General considerations: heterotic models. *Genetics* 49:49-67.

Liew, F.Y., Hale, C. & Howard, J.G. (1982) Immunologic regulation of experimental cutaneous leishmaniasis. V. Characterization of effector and specific suppressor T cells. J.Immunol. 128: 1917-22.

Liew, F.Y. & O'Donnell, C.A. (1993) Immunology of leishmaniasis. Adv Parasitol. 32:161-259.

Lighthall, G.K. & Giannini, S.H. (1992) The chromosomes of Leishmania. Parasitol. Today 8: 192-9.

Llanos-Cuentas, A.E. (1993) Risk factors associated with the transmission of Andean cutaneous leishmaniasis. PhD thesis, University of London.

Llanos-Cuentas, A.E., Roncal, N., Villaseca, P., Paz, L., Ogusuku, E., Pérez, J.E., Cáceres, A. & Davies, C.R. (1999) Natural infections of *Leishmania peruviana* in animals in the Peruvian Andes. *Trans.R. Soc. Trop. Med. Hyg.* **93**: 15-20.

Llanos-Cuentas, A., Echevarria, J., Chang, J., Campos, P., Cieza, D., Lentnek, D., Gonzalez, D., Robbins, M. & Williams, R. (1991) Randomized trial of amphotericin B complex (ABLC) versus Fungizone in patients with mucocutaneous leishmaniasis. *Mem.Inst.Oswaldo Cruz* 86: Suppl.1 238.

Locksley, R.M., Heinzel, F.P., Fankhauser, J.E., Nelson, C.S. & Sadick, M.D. (1988) Cutaneous host defence in leishmaniasis: interaction of isolated dermal macrophages and epidermal Langerhans cells with the insect-stage promastigote. *Infect.Immun.* 56:336-42.

Lopes, U.G., Momen, H., Grimaldi, G. Jnr., Marzochi, M.C., Pacheco, R.S. & Morel, C.M. (1984) Schizodeme and zymodeme characterization of *Leishmania* in the investigation of foci of visceral and cutaneous leishmaniasis. *J.Parasitol.* 70:89-98.

Lopes, U.G. & Wirth, D.F. (1986) Identification of visceral Leishmania species with cloned sequences of kinetoplast DNA. Mol.Biochem.Parasitol. 20:77-84.

Lopez, M., Inga, R., Cangalaya, M., Echevarria, J., Llanos-Cuentas, A., Orrego, C. & Arevalo, J. (1993) Diagnosis of *Leishmania* using the polymerase chain reaction: a simplified procedure for field work. *Am.J.Trop.Med.Hyg.* 49:348-56

Lucas, C.M., Franke, E.D., Cachay, M.I., Tejada, A., Carrizales, D. & Kreutzer, R.D. (1994) Leishmania (Viannia) lainsoni; first isolation in Peru. Am.J. Trop. Med. Hyg. 51:533-37.

Lucas, C.M., Franke, E.D., Cachay, M.I., Tejada, A., Cruz, M.E., Kreutzer, R.D., Barker, D.C., McCann, S.H.E. and Watts, D.M. (1998) Geographical distribution and clinical description of leishmaniasis cases in Peru. *Am.J.Trop.Med.Hyg.* 59:312-7.

Lui, X & Chang, K-P. (1992) The 63-kilobase circular amplicon of tunicamycin-resistant *Leishmania* amazonensis contains a functional N-acetylglucosamine-1-phosphate transferase gene that can be used as a dominant slectable marker in transfection. *Mol.Cell.Biol.* 12:4112-22.

Lynch, M & Milligan, B.G. (1994) Analysis of population genetic structure with RAPD markers. Mol.Ecol. 3:91-9.

Macedo, A.M., Melo, M.N., Gomes, R.F.& Pena, S.D.J. (1992) DNA fingerprints: a tool for identification and determination of the relationships between species and strains of *Leishmania*. *Mol.Biochem.Parasitol.* 53:63-70.

Machado, E.S., Braga, M da P., Da-Cruz, A.M., Countinho, S.G., Viera, A.R.M., Rutowitsch, M.S., Cuzzi-Maya, T., Grimaldi, G. Jnr. & Menezes, J.A. (1992) Disseminated American mucocutaneous leishmaniasis caused by *Leishmania braziliensis braziliensis* in a patient with AIDS: a case report. *Mem.Inst. Oswaldo Cruz* 87:487-92.

Magill, A.J., Grogl, M., Gasser, R.A. Jr., Sun, W. & Oster, C.N. (1993) Visceral infection caused by Leishmania tropica in veterans of Operation Desert Storm. N.Engl.J.Med 328:1383-7.

Marsden, P.D. (1986) Mucosal leishmaniasis ("espundia" escornel 1911). Trans.R.Soc. Trop.Med. Hyg. 80:859-76.

Marsden, P.D. (1994a) Mucosal leishmaniasis due to Leishmania (Viannia) braziliensis in Tres Braços, Bahia-Brazil. Rev.Soc.Bras.Med.Trop. 27:93-101.

Marsden, P.D. (1994b) Personal experience with diagnostic and therapeutic aspects of human Leishmania (Viannia) braziliensis in Três Braços. Mem.Inst. Oswaldo Cruz 59:485-7. Martinez, J.E., Arias, A.L., Escobar, M. & Saravia, N.G. (1992) Haemoculture of *Leishmania* (Viannia) braziliensis from two cases of mucosal leishmaniasis: re-examination of haematogenous dissemination. *Trans.R.Soc. Trop. Med. Hyg.* 86:392-4.

Martinez, J.E., Travi, B.L., Valencia, A.Z. & Saravia, N.G. (1991) Metastatic capability of Leishmania (Viannia) panamensis and Leishmania (Viannia) guyanensis in golden hamsters. J. Parasitol. 77:762-8.

Marzochi, M.A. & Marzochi, K.B. (1994) Tegumentary and visceral leishmaniases in Brazil: emerging anthropozoonosis and possibilities for their control. *Cad.Saude Publ. Rio de Janeiro* 10 suppl. 2:359-75.

Marzochi, K.B., Marzochi, M.A., Silva, A.F., Grativol, N., Duarte, R., Confort, E.M. & Modabber, F. (1998) Phase I study of an inactivated vaccine against American tegumentary leishmaniasis in normal volunteers in Brazil. *Mem.Inst.Oswaldo Cruz* 93:205-12.

Maslov, D.A., Lukes, J., Jirku, M. & Simpson, L. (1996) Phylogeny of trypanosomes as inferred from the small and large subunit rRNAs: implications for the evolution of parasitism in the trypanosomatid protozoa. *Mol.Biochem.Parasitol.* 75:197-205.

May, B. (1992) Starch gel electrophoresis of allozymes. pp.1-27. In: Hoelzel, A.R. (Ed.) Molecular genetic analysis of populations. A Practical Approach. IRL Press, Oxford.

Maynard-Smith, J., Smith, N.H., O'Rourke, M. & Spratt, B.G. (1993) How clonal are bacteria? Proc. Natl. Acad. Sci. USA 90:4384-8.

Mayrink, W., Costa, C.A., Magalhaes, P.A., Melo, M.N., Dias, M., Oliveira-Lima, A., Michalik, M.S.M. & Williams, P. (1985) A field trial of a vaccine against American Dermal Leishmaniasis: experience in the State of Espirito Santo, Brazil. Ann. Trop. Med. Parasitol. 79:259-69.

Mayrink, W., Genaro, O., Franca-Silva, J.C., Costa, R.T., Tafuri, W.L., Peixoto, V.P.C., da Silva, A.R., Reis, A.B., Williams, P. & Costa, C.A. (1996) Phase I and II open clinical trials of a vaccine against Leishmania chagasi infection in dogs. *Mem. Inst.Oswaldo Cruz* 91:695-7.

McMahon-Pratt, D. & David, J.R. (1981) Monoclonal antibodies that distinguish between New World species of *Leishmania*, *Nature* 291:581.

McMahon-Pratt, D., Bennett, E. & David, J.R. (1982) Monoclonal antibodies that distinguish subspecies of Leishmania braziliensis. J.Immunol. 129:926-7.

McMahon-Pratt, D., Bennett, E., Jaffe, C.L. & Grimaldi, G. Jnr. (1985) Subspecies and species-specific antigens of *Leishmania mexicana* characterized by monoclonal antibodies. *J.Immunol.* 134:1935-40.

McMahon-Pratt, D., Rodriguez, D., Rodriguez, J.R., Zhang, Y., Manson, K., Bergman, C., Rivas, L., Rodriguez, J.F., Lohman, K.L. & Ruddle, N.H. (1993) Recombinant vaccinia viruses expressing gp46/M-2 protect against *Leishmania* infection. *Infect.Immun.* 61:3351-9.

McMahon-Pratt, D., Traub-Cesko, Y., Lohman, K.L., Rogers, D.D. & Beverley, S.M. (1992) Loss of the gp46/M-2 surface membrane glycoprotein gene family in the *Leishmania braziliensis* complex. *Mol.Biochem.Parasitol.* **50**:151-60.

Medina-Acosta, E., Karess, R.E. & Russell, D.G. (1993) Structurally distinct genes for the surface protease of *Leishmania mexicana* are developmentally regulated. *Mol. Biochem. Parasitol.* 57:31-46

Mebrahtu, YB, Lawyer PG, Hendricks LD, Muigai R, Oster CN, Perkins PV, Koech DK, Pamba H, Roberts CR. (1991) Concurrent infection with *Leishmania donovani* and *Leishmania major* in a Kenyan patient: clinical description and parasite characterization. Am J Trop Med Hyg 45:290-6.

Melby, P.C., Andrade-Narvaez, F.J., Darnell, B.J., Valencia-Pacheco, G., Tryon, V.V. & Palomo-Cetina, A. (1994) Increased expression of proinflammatory cytokines in chronic lesions of human cutaneous leishmaniasis. *Infect.Immun.* 62:837-42.

Mendoza-Leon, A., Havercroft, J.C. & Barker, D.C. (1995) The restriction fragment length polymorphism analysis of the β-tubulin gene region in New World Leishmania Parasitology 111:1-9.

Michaelides, K., Schwaab, R., Lalloz, M.R.A., Schmidt, W. & Tuddenham, E.G.D. (1995) Mutational analysis: new mutations. pp.255-88. In: McPherson et al., (Eds.) PCR2: A practical approach. IRL Press, Oxford.

Mihok, S., Otieno, L.H. & Darji, N. (1990) Population genetics of *Trypanosoma brucei* and the epidemiology of sleeping sickness in the Lambwe Valley, Kenya. *Parasitology* 100:219-33.

Miles, M.A. (1983) The epidemiology of South American trypanosomiasis - biochemical and immunological approaches and their relevance to control. *Trans.R.Soc.Trop.Med.Hyg.* **77**:5-23.

Miles, M.A., Lanham, S.M., de Souza, A.A. & Povoa, M. (1980a) Further enzymic characters of *Trypanosoma cruzi* and their evaluation for strain identification. *Trans.R.Soc.Trop.Med.Hyg.* 74:221-37.

Miles, M.A., Lainson, R., Shaw, J.J., Povoa, M. & de Souza, A.A. (1981) Leishmaniasis in Brazil: XV. Biochemical distinction of *Leishmania mexicana amazonensis*, *L. braziliensis braziliensis* and *L. braziliensis guyanensis* - aetiological agents of cutaneous leishmaniasis in the Amazon Basin of Brazil. *Trans. R.Soc. Trop. Med. Hyg.*, 75:524-9.

Miles, M.A., Povoa, M., de Souza, A.A., Lainson, R. & Shaw, J.J. (1980b) Some methods for the enzymic characterization of Latin-American Leishmania with particular reference to Leishmania mexicana amazonensis and subspecies of Leishmania hertigi. Trans. R. Soc. Trop. Med. Hyg. 74:243-52.

Missoni, E., Morrelli, R., Balladares, L., Berrios, J., de Solan, M.D., Baldwin, C.I. & Evans, D.A. (1986) Isolation and characterization of leishmanias for Nicaragua. *Trans.R.Soc.Trop.Med.Hyg.* 80:999-1000.

Modabber, F. (1990) Development of vaccines against leishmaniasis. Scand.J. Inf. Dis. 76:72-8.

Modabber, F. (1995) Vaccines against leishmaniasis. Ann. Trop. Med. Parasitol. 89: suppl. 1:83-8.

Moli, H., Flohé, S. & Röllinghof, M. (1995) Dendritic cells in *Leishmania major* immune mice harbor persistent parasites and mediate an antigen-specific T cell immune response. *Eur.J. Immunol.* 25:693-9.

Moll, H., Fuchs, H., Blank, C. & Rollinghof, M. (1993) Langerhans cells transport *Leishmania major* from the infected skin to the draining lymph node for presentation to antigen-specific T cells. *Eur.J.Immunol.* 23:1595-1601.

Mullis, K.B. & Faloona, F.A. (1987) Specific synthesis of DNA in vitro via a polymerase-catalyzed chain reaction. *Methods in Enzymology* 155:335-50.

Myers, R.M., Maniatis, T. & Lerman, L.S. (1987) Detection and localization of single base chain changes by denaturing gradient gel electrophoresis. *Methods in Enzymology* 155:501-27.

Myers, R.M., Sheffield, V.C. & Cox, D.R. (1990) Detection of single base changes in DNA: ribonuclease cleavage and denaturing gradient gel electrophoresis. *pp.* 95-139. In: Davies, K.E. (Ed.) Genome analysis: a practical approach. IRL Press, Oxford.

Nadler, S.A., Lindquist, R.L. & Near, T.J. (1995) Genetic structure of midwestern Ascaris suum populations: a comparison of isoenzyme and RAPD markers. J. Parasitol. 81:385-94.

Neal, R.A., Reeves, A. & Peters, W. (1990) Leishmania infecting man and animals in Saudi Arabia 7. Partial protection of mice against Leishmania major by prior infection with L. arabica. Trans.R. Soc. Trop. Med. Hyg. 84:233-8.

Noyes, H.A., Belli, A.A. & Maingon, R. (1996) Appraisal of various random amplified polymorphic DNA-polymerase chain reaction primers for *Leishmania* identification. *Am.J.Trop.Med.Hyg.* 55:98-105.

Nozaki, J. & Cross, G.A.M. (1994) Functional complementation of glycoprotein 72 in a Trypanosoma cruzi glycoprotein 72 null mutant. Mol. Biochem. Parasitol. 67:91-102.

Ohta, T. & Kimura, M. (1973) A model of mutation appropriate to estimate the number of electrophoretically detectable alleles in a finite population. *Genet.Res.* 22:201-4.

Oliveira, R.P., Broude, N.E., Macedo, A.M., Cantor, C.R., Smith, C.L. & Pena, S.D. (1998) Probing the genetic population structure of *Trypanosoma cruzi* with polymorphic microsatellites. *Proc.Natl.* Acad. Sci. USA 95:3776-80.

Oliveira, R.P., Macedo, A.M., Chiari, E. & Pena, S.D.J. (1997) An alternative approach to evaluating the intraspecific genetic variability of parasites. *Parasitol. Today* 13:196-200.

Oliveira-Neto, M.P., Pirmez, C., Rangel, E., Schubach, A. & Grimaldi, G. Jnr. (1988) An outbreak of american cutaneous leishmaniasis (*Leishmania braziliensis braziliensis*) in a periurban area of Rio de Janeiro City, Brazil: clinical and epidemiological studies. *Mem.Inst.Oswaldo Cruz* 83:427-35.

Oliveira-Neto, M.P., Schubach, A., Mattos, M., da-Costa, S.C. & Pirmez, C. (1997b) Intralesional therapy of American cutaneous leishmaniasis with pentavalent antimony in Rio de Janeiro, Brazil - an area of *Leishmania (V) braziliensis* transmission. *Int.J. Dermatol* 36:463-8.

Oliveira-Neto, M.P., Schubach, A., Mattos, M., Goncalves-Costa, S.C. & Pirmez, C. (1997a) A lowdose antimony treatment in 159 patients with American cutaneous leishmaniasis: extensive follow-up studies (up to 10 years). *Am.J.Trop.Med.Hyg.* 57:651-5.

Olsen, G.J. Lane, D.J., Giovannoni, S.J. & Pace, N.R. (1986) Microbial ecology and evolution: a ribosomal RNA approach. Ann.Rev.Microbiol. 40:337-65.

Opperdoes, F.R. (1987) Compartmentation of carbohydrate metabolism in trypanosomes. Am. Rev. Microbiol. 41:127-51.

Orita, M., Iwahana, H., Kanazawa, H., Hayashi, K. & Sekiya, T. (1989) Detection of polymorphisms of human DNA by gel electrophoresis as single-strand conformation polymorphisms. *Proc.Natl.Acad.Sci.* USA 86:2766-70.

Orti, G., Pearse, D.E. & Avise, J.C. (1997) Phylogenetic assessment of length variation at a microsatellite locus. *Proc.Natl.Acad.Sci. USA* 94:10745-9.

Osorio, Y., Gonzalez, S.J., Gama, V.L. & Travi, B.L. (1998) Reinfection in American cutaneous leishmaniasis: evaluation of clinical outcomes in the hamster model. *Mem.Inst.Oswaldo Cruz.* 93:353-6.

Ozensoy, S., Ozbel, Y., Turgay, N., Alkan, M.Z., Gul, K., Gilman-Sachs, A., Chang, K.P., Reed, S.G. & Ozcel, M.A. (1998) Serodiagnosis and epidemiology of visceral leishmaniasis in Turkey. *Am.J. Trop. Med. Hyg.* 59:363-9.

Pages, M., Bastien, P., Veas, F., Rossi, V., Bellis, M., Wincker, P., Rioux, J.A. & Roizes, G. (1989) Chromosome size and number polymorphisms in *Leishmania infantum* suggest amplification/deletion and possible genetic exchange. *Mol.Biochem.Parasitol.* 36:161-8.

Paindavoine, P., Zampetti-Bosseler, F., Pays, E., Schweizer, J., Guyaux, M., Jenni, L. & Steinert, M. (1986) Trypanosome hybrids generated in tsetse flies by nuclear fusion. *EMBO.J.* 5:3631-6.

Panton, L.J., Tesh, R.B., Nadeau, K.C. & Beverley, S.M. (1991) A test for genetic exchange in mixed infections of *Leishmania major* in the sandfly *Phlebotomus papatasi*. J.Protozool. 38:224-8.

Papadopoulou, B. & Dumas, C. (1997) Parameters controlling the rate of gene targeting frequency in the protozoan parasite Leishmania. Nucl. Acid Res. 25:4278-86.

Papadopoulou, G., Karagouni, E. & Dotsika, E. (1998) ISCOMs vaccine against experimental leishmaniasis. Vaccine 16:885-92.

Pappas, M.G., Hajkowski, R. & Hockmeyer, W.T (1983) Dot enzyme-linked immunosorbent assay (Dot-ELISA): a micro technique for the rapid diagnosis of visceral leishmaniasis. *J.Immunol. Methods* 64:205-14

Passos, V.M.A., Falcao, A.L., Marzochi, M.C.A., Gontijo, C.M.F., Dias, E.S., Barbosa-Santos, E.G.O., Guerra, H.L. & Katz, N. (1993) Epidemiological aspects of cutaneous leishmaniasis in a periurban area of the metropolitan region of Belo Horizonte, Minas Gerais, Brazil. *Mem.Inst.Oswalso Cruz.* 91:19-20.

Passos, V.M.A., Lasmar, E.B., Gontijo, C.M.F., Fernandes, O. & Degrave, W. (1996) Natural infection of a domestic cat (*Felis domesticus*) with *Leishmania (Viannia)* in the Metroploitan region of Belo Horizonte, State of Minas Gerais, Brazil. *Mem.Inst.Oswaldo Cruz.* 91:19-20.

Pearson, R.D. & Wilson, J. (1989) pp 31-81 In Walzer, P.D. & Genta, S.M. (Eds.): Parasite infections in the immunocompromised host. Marcel Dekker.

Pernberton, J.M., Slate, J., Bancroft, D.R. & Barrett, J.A. (1995) Nonamplifying alleles at microsatellite loci: a caution for for parentage and population studies. Mol.Ecol. 4:249-52.

Petzl-Erler, M.L., Belich, M.P. & Queiroz-Tellez, F. (1991) Association of mucosal leishmaniasis with HLA. Human Immunol. 32:254-60.

Piarroux, R., Trouve, V., Pratlong, F., Martini, A., Lambert, M. & Rioux, J.A. (1994) The use of isoelectric focusing on polyacrylamide gel for the enzymatic analysis of "Old World" *Leishmania* species. *Trans.R.Soc. Trop.Med. Hyg.* 88:475-8.

Pimienta, P.F.P., Saraiva, E.M.B., Rowton, E., Modi, G.B., Garraway, L.A., Beverley, S.M., Turco, S.J. & Sacks, D.L. (1994) Evidence that the vectorial competence of phlebotomine sand flies for different species of *Leishmania* is controlled by structural polymorphisms in the surface lipophosphoglycan. *Proc.Natl.Acad.Sci. USA* 91:9155-9.

Pirmez, C., Yamamura, M., Uyemura, K., Paes-Olivera, M., Conceicao-Silva, F. & Modlin, R.L. (1993) Cytokine patterns in the pathogenesis of human leishmaniasis. *J. Clin. Invest.* 91:1390-5.

Podani, J. (1993) SYN-TAX-pc. Computer programme for multivariate data analysis in ecology and systematics. Sciencia publishing, Budapest, Hungary.

Pogue, G.P., Koul, S., Less, N.S., Dwyer, D.M. & Nakhasi, H.L. (1995) Identification of intra- and interspecific *Leishmania* genetic polymorphisms by arbitrarily primed polymerase chain reactions and use of polymorphic DNA to identify differentially regulated genes. *Parasitol.Res.* **31**:282-90.

Pollard, V.W., Rohrer, S.P., Michelotti, E.F., Hancock, K. & Hajduk, S.L. (1990) Organization of minicircle genes for guide RNAs in *Trypanosoma brucei*. Cell 63:783-90.

Postigo, C., Llamas, R., Zarco, C. Rubio, R., Pulido, F., Costa, J.R. & Iglesias, L. (1997) Cutaneous lesions in patients with visceral leishmaniasis and HIV infection. *J.Infect.* 35:265-8.

Preston, P.M., carter, R.L., Leuchars, E., Davies, A.J.S. & Dumonde, D.C. (1972) Experimental cutaneous leishmaniasis. III. Effects of thymectomy on the course of infection of CBA mice with *Leishmaniai tropica*. *Clin. Exptl. Immunol.* 10:337-57.

Proudfoot, L., O'Donnell, C.A. & Liew, F.Y. (1995) Glycoinositolphospholipids of *Leishmania major* inhibit nitric oxide synthesis and reduce leishmanicidal activity in murine macrophages. *Eur.J.Immunol.* 25:745-50.

Qiao, Z., Miles, M.A. & Wilson, S.M. (1995) Detection of parasites of the Leishmania donovanicomplex by a polymerase chain reaction-solution hybridization enzyme-linked immunoassay (PCR-SHELA). Parasitology 110:269-75.

de Queiroz, R.G., Vasconcelos, I.A., Vasconcelos, A.W., Pessoa, F.A.C., De Sousa, R.N. & David, J.R. (1994) Cutaneous leishmaniasis in Ceara State in northeastern Brazil: incrimination of Lutzomyia whitmani (Diptera: Psychodidae) as a vector of Leishmania braziliensis in Baturite municipality. Am.J.Trop.Med.Hyg. 50:693-8.

Rab, M.A., Hassan, M., Buz, D., Mahmood, M.T. & Evans, D.A. (1992) The isolation and cultivation of *Leishmania infantum* from apparently normal skin of visceral leishmaniasis patients in northern Pakistan. *Trans.R.Soc. Trop.Med.Hyg.* **36**:620-1.

Ramirez, J.L. & Guevara, P. (1987) The ribosomal gene spacer as a tool for the taxonomy of *Leishmania*. Mol. Biochem. Parasitol. 22:177-83.

Ramirez, J.L. & Guevara, P. (1997) Persistent infections by Leishmania (Viannia) braziliensis. Mem.Inst.Oswaldo Cruz. 92:333-8.

Rangel, E.F., Lainson, R., Souza, A.A., Ready, P. & Azevedo, A.C.R. (1996) Variation between geographical populations of *Lutzomyia (Nyssomyia) whitmani* (Antunes & Coutinho, 1939) sensu lato (Diptera: Psychodidae: Phlebotominae) in Brazil. Mem.Inst.Oswaldo Cruz 91:43-50.

Raymond, M. & Rousset, F. (1995) Genepop (version 1.2): population genetics software for exact tests and ecumenicism. J.Hered. 86:248-9.

Rees, P.H. & Kager, W.C. (1987) Visceral leishmaniasis and post-kala-azar dermal lesihmaniasis. pp 583-611. In: Peters, W. & Killick-Kendrick, R. (Eds.) The Leishmaniases in Biology and Medicine Vol. 1. Biology and Epidemiology. Academic Press (London).

Reiner, N.E., Lo, R., Llanos-Cuentas, A., Guerra, H., Button, L.L. & McMaster, W.R. (1989) Genetic heterogeneity in Peruvian Leishmania isolates. Am.J.Trop.Med.Hyg. 41:416-21.

Reithinger, R. & Davies, C.R. (1999) Is the domestic dog (*Canis familiaris*) a reservoir host of American cutaneous leishmaniasis? A critical review of the current evidence. *Am.J.Trop.Med.Hyg.* 61:530-41.

Revollo, S., Dimier-David, L., David, C., Lyevre, P., Camacho, C. & Dedet, J.P. (1992) Isoenzyme charcterization of *Leishmania braziliensis braziliensis* isolates obtained from Bolivian and Peruvian patients. *Trans. R.Soc. Trop. Med. Hyg.* 86:388-91.

Ridley, D.S., Marsden, P.D., Cuba, C.C. & Barreto, A.C. (1980) A histological classification of mucocutaneous leishmaniasis in Brazil and its clinical evaluation. *Trans.R.Soc. Trop. Med. Hyg.* 74:508-14.

Rioux, J.A., Lanotte, G., Serres, E., Pratlong, F., Bastien, P. & Perieres, J. (1990) Taxonomy of *Leishmania*. Use of isoenzymes. Suggestions for a new classification. *Ann.Parasitol.Hum.Comp.* 65:111-25.

Robello, C., Gamarro, F., Castanys, S. & Alvarez-Valin, F. (2000) Evolutionary relationships in *Trypanosoma cruzi*: molecular phylogenetics supports the existence of a new major lineage of strains. *Gene* 246:331-8.

Roberts, S.C., Swihart, K.G., Agey, M.W., Ramamoorthy, R., Wilson, M.E. & Donelson, J.E. (1993) Sequence diversity and organization of the msp gene family encoding gp63 of *Leishmania chagasi Mol.Biochem.Parasitol.* 62:157-72.

Rocha, R.A., Sampaio, R.N., Guerra, M., Magalhaes A, Cuba CC, Barreto AC & Marsden PD (1980) Apparent Glucantime failure in five patients with mucocutaneous leishmaniasis. *J. Trop. Med. Hyg.* 83:131-9.

Rodriguez, N., Guzman, B., Rodas, A., Takiff, H., Bloom, B.R. & Convit, J. (1994) Diagnosis of cutaneous leishmaniasis and species discrimination of parasites by PCR and hybridization. *J.Clin.Micro.* 32:2246-52.

Rojas, J.C., Munoz, A., Herrero, M.V., Dobles, A. & Zeledon, R. (1994) Risk factors for domiciliary and peridomiciliary transmission of cutaneous leishmaniasis in Costa Rica. *Am.J.Trop.Med.Hyg.* 51:S104.

Romero, G.G., Arana, M., Lopez, M., Montoya, I, Bohl, R., Campos, M., Arevalo, J. & Llanos, A. (1987) Characterization of *Leishmania* species from Peru. *Trans.R.Soc. Trop.Med.Hyg.* **81**:14-24.

Rongnoparut, P., Yaicharoen, S., Sirichotpakorn, N., Rattanarithikul, R., Lanzaro, G.C. & Linthicum, K.J. (1996) Microsatellite polymorphism in Anopheles maculatus, a malaria vector in Thailand. Am.J. Trop. Med. Hyg. 55:589-94. Rosa, A.C., Cuba-Cuba, C., Vexenat, A., Barreto, A.C. & Marsden, P.D. (1983) Predominance of Leishmania braziliensis braziliensis in the regions of Três Braços and Corte de Pedra, Bahia. Trans.R.Soc.Trop.Med.Hyg. 82:409-10.

Rossi, V., Wincker, P., Ravel, C., Blaineau, C., Pages, M. & Bastien, P. (1994) Structural organization of microsatellite families in the *Leishmania* genome and polymorphisms at two (CA)n loci. *Mol.Biochem.Parasitol.* 65:271-82.

Russell, R., Iribar, M.P., Labson, B., Brewster, S., Blackwell, J.M., Dye, C. & Ajioka, J.W. (1999) Intra and inter-specific microsatellite variation in the *Leishmania* subgenus *Viannia*. *Mol.Biochem. Parasitol*. 103:71-7.

Russo, D.M., Barral-Netto, M., Barral, A. & Reed, S.G. (1993) Human T cell responses in *Leishmania* infections. p. 119 In: Sun, T. (Ed.) Progress in Clinical Parasitology. Vol III. Springer Verlag, NY.

Sacks, D.L. & Da Silva, R.P. (1987) The generation of infective stage *Leishmania major* promastigotes is associated with cell-surface expression and release of a developmentally regulated glycolipid. *J.Immunol* 139:3099-106.

Sacks, D.L., Pimenta, P.F.P., McConville, M.J., Schneider, P. & Turco, S.J. (1995) Stage-specific binding of *Leishmania donovani* to the sand fly vector midgut is regulated by conformational changes in the abundant surface lipophosphoglycan. *J.Exp.Med.* **181**:685-97.

Safjanova, V.M. (1982) Classification of the genus Leishmania Ross. pp. 95-101 [Russian]. In: The Leishmanias. Protozoology part 7. Academy of Sciences, Leningrad, USSR.

Saiki, R.K., Gelfand, D.H., Stoffel, S., Scharf, S.J., Higuchi, R., Horn, G.T., Mullis, K.B. & Erlich, H.A. (1988) Primer-directed enzymatic amplification of DNA with a thermostable DNA polymerase. *Science* 239:487-91.

Saito, N. & Nei, M. (1987) The neighbor-joining method: a new method for reconstructing phylogenetic trees. *Mol.Biol.Evol.* 4:406-25.

Spithill, T.W. & Samaras, N. (1985) The molecular karyotype of *Leishmania major* and mapping of alpha and beta tubulin gene families to multiple unlinked chromosomal loci. *Nucl.Acids Res.* 13:4155-69.

Samaras, N. & Spithill, W. (1987) Molecular karyotype of five species of *Leishmania* and analysis of gene locations and chromosomal rearrangements. *Mol.Biochem.Parasitol.* 25:279-91.

Sambrook, J., Fritsch, E.F. & Maniatis, T. (1989) Molecular Cloning: a Laboratory Manual. 2nd ed., Cold Spring Harbor Laboratory Press, New York.

Sampaio, R.N., Marsden, P.D., Llanos-Cuentas, E.A. Cuba-Cuba, C. & Grimaldi, G. Jnr. (1985) Leishmania mexicana amazonensis isolated from a patient with fatal mucosal leishmaniasis. Rev.Soc.Bras.Med.Trop. 18:273-4.

Santrich, C., Segura, I., Arias, A.L. & Saravia, N.G. (1990) Mucosal disease caused by Leishmania braziliensis guyanensis. Am.J. Trop. Med. Hyg. 42:51-5.

Sanz, M del Mar, Rubio R, Casillas A, Guijarro C, Costa JR, Martinez R. & de Dios Garcia J (1991) Visceral leishmaniasis in HIV infected patients. AIDS 5:1272-4.

Saravia, N.G., Holguin, A.F., McMahon-Pratt, D. & D'Allesandro, A. (1985) Mucocutaneous leishmaniasis in Colombia: *Leishmania braziliensis* subspecies diversity. *Am.J.Trop.Med.Hyg.* 34:714-20.

Saravia, N.G., Segura, I., Holguin, A.F., Santrich, C., Valderrama, L. & Ocampo, C. (1998) Epidemiologic, genetic, and clinical associations among phenotypically distinct populations of Leishmania (Viannia) in Colombia. Am.J. Trop. Med. Hyg. 59:86-94. Saravia, N.G., Valderrama L, Labrada M, Holguin AF, Navas C, Palma G. & Weigle KA (1989) The relationship of *Leishmania braziliensis* subspecies and immune response to disease expression in New World leishmaniasis. *J.Inf.Dis.* **159**:725-35.

Saravia, N.G., Weigle, K., Segura, I., Holmes, S., Pacheco, R., Labrada, L.A. & Goncalves, A. (1990) Recurrent lesions in human *Leishmania braziliensis* infection: Reactivation or reinfection? *Lancet* 336:398-402.

Schlötterer, C. (1998) Microsatellites. Chapter 7, pp. 237-261. In: Hoelzel, A.R. (Ed.) Molecular genetic analysis of populations: a practical approach. Second Edition. I.R.L. Press (Oxford).

Schlotterer, C., Amos, B. & Tautz, D. (1991) Conservation of polymorphic simple sequence loci in cetacean species. *Nature* 354:63-5.

Schlotterer, C., Ritter, R., Harr, B. & Brem, G. (1998) High mutation rate of a long microsatellite allele in *Drosophila melanogaster* provides evidence for allele-specific mutation rates. *Mol.Biol.Evol.* 15:1269-74.

Schlötterer, C. & Tautz, D. (1992) Slippage synthesis of simple sequence DNA. Nucl. Acid Res. 20:211-5.

Scholler, J.K., Reed, S.G. & Stuart, K. (1986) Molecular karyotypes of species and subspecies of *Leishmania*. Mol.Biochem.Parasitol. 20:279-93.

Schönian, G., Schweynoch, C., Zlateva, K., Oskam, L., Kroon, N., Gräser, Y. & Presber, W. (1996) Identification and determination of the relationships of species and strains within the genus *Leishmania* using single primers in the polymerase chain reaction. *Mol.Biochem.Parasitol.* 77:19-29.

Schubach, A., Haddad, F., Oliveira-Neto, M.P., Degrave, W., Pirmez, C., Grimaldi, G. Jnr & Fernandes, O. (1998) Detection of *Leishmania* DNA by polymerase chain reaction in scars of treated human patients. J.Inf.Dis. 178:911-4.

Schwartz, D.C. & Cantor, C.R. (1984) Separation of yeast chromosome-sized DNAs by pulsed field gradient gel electrophoresis. *Cell* 37:67-75.

Schweizer, J., Pospichal, H., Hide, G., Buchanan, N., Tait, A. & Jenni, L. (1994) Analysis of a new genetic cross between two East African *Trypanosoma brucei* clones. *Parasitology* 109:83-93.

Schweizer, J., Pospichal, H. & Jenni, L. (1991) Hybrid formation between African trypanosomes in vitro. Acta Tropica 49:237-40.

Scott, P. (1989) The role of Th1 and Th2 cells in experimental cutaneous leishmaniasis. *Exp. Parasitol.* 63:369-72.

Scott, P. (1991) IFN-gamma modulates the early development of Th1 and Th2 reponses in a murine model of cutaneous leishmaniasis. *J.Immunol.* 147:3149-55.

Scott, P., Natovitz, P. & Sher, A. (1986) B lymphocytes are required for the generation of T cells that mediate healing of cutaneous leishmariasis. *J.Immunol.* 137:1017-21.

Seaman, J., Pryce, D., Sondorp, H.E., Moody, A., Bryceson, A.D.M. & Davidson, R.N. (1993) Epidemic visceral leishmaniasis in Sudan: a randomized trial of aminosidine plus sodium stibogluconate versus sodium stibogluconate alone. *J.Inf.Dis.* **168**:715-20.

Shapiro, H. & Englund, P.T. (1995) The structure and replication of kinetoplast DNA. Ann. Rev. Microbiol. 49:117-43.

Shaw, J.J. (1994) Taxonomy of the genus Leishmania: present and future trends and their implications. Mem.Inst.Oswaldo Cruz 89:471-8. Shaw, J.J., Ishikawa, E.A., Lainson, R., Braga, R.R. & Silveira, F.T. (1991) Cutaneous leishmaniasis of man due to *Leishmania (Viannia) shawi* (Lainson, de Souza, Povoa, Ishikawa & Silveira) in Para State, Brazil. Ann. Parasitol. Hum. Comp. 66:243-6.

Silva, J.O., Lago, E., Noronha, E.F., Okida, H.M., Scandiuzzi, M.C.P., Haldi, S.I., Johnson, W.D. & Marsden, P.D. (1992) Ten years of occurrence of human Leishmania Viannia braziliensis (Lvb) infections in farms around Tres Braços, Bahia-Brazil. Rev. Soc. Brasil. Med. Trop. 25: 81-2.

Silveira, FT, Lainson R, Shaw JJ, Ribeiro R da S. (1984) Cutaneous leishmaniasis in Amazonia. Report of the 1st human case of mixed infection, determined by two different Leishmania species: Leishmania brasiliensis and Leishmania mexicana amazonensis. Rev.Inst.Med.Trop. Sao Paulo. 26:272-5

Silveira, F.T., Shaw, J.J., Braga, R.R. & Ishikawa, E. (1987) Dermal leishmaniasis in the Amazon region of Brazil: *Leishmania (Viannia) lainsoni* sp.n., a new parasite from the state of Para. *Mem.Inst.Oswaldo Cruz* 82:289-92.

Singer, S.M., Yee, J. & Nash, T.E. (1998) Episomal and integrated maintenance of foreign DNA in *Giardia lamblia*. Mol.Biochem.Parasitol. 92:59-69.

Smith, L.E., Rodrigues, M. & Russell, D.G. (1991) The interaction between CD8⁺ cytotoxicity cells and *Leishmania*-infected macrophages. *J.Exp.Med.* **174**:499-505.

Sogin, M.L., Elwood, H.J. & Gunderson, J.H. (1986) Evolutionary diversity of eukaryotic smallsubunit rRNA genes. *Proc. Natl. Acad. Sci. USA* 83:1383-7.

Sousa, A. de Q., Parise, M.E., Pompeu, M.M.L., Coehlo-Filho, J.M., Vasconcelos, I.A.B., Lima, J.W.O., Olivera, E.G., Vasconcelos, A.W., David, J.R. & Maguire, J.H. (1995) Bubonic leishmaniasis: a common manifestation of *Leishmania (Viannia) braziliensis* infection in Ceara, Brazil. *Am.J.Trop. Med.Hyg.* 53:380-5.

Spithill, T.W. & Samaras, N. (1985) The molecular karyotype of *Leishmania major* and mapping of alpha and beta tubulin gene families to multiple unlinked chromosomal loci. *Nucl.Acid Res.* 13:4155-69.

Steinkraus, H.B., Greer, J.M., Stephenson, D.C. & Langer, P. J. (1993) Sequence heterogeneity and polymorphic gene arrangements of the *Leishmania guyanensis* gp63 genes. *Mol.Biochem.Parasitol.* 62:173-86.

Stemberg, J., Tait, A., Haley, S., Wells, J.M., Le Page, R.W.F., Schweizer, J. & Jenni, L. (1988) Gene exchange in African trypanosomes: characterization of a new hybrid genotype. *Mol.Biochem.Parasitol.* 27:191-200.

Stemberg, J., Turner, C.M.R., Wells, J.M., Ranford-Cartwright, L.C., Le Page, R.W.F. & Tait, A. (1989) Gene exchange in African trypanosomes: frequency and allelic segregation. *Mol.Biochem. Parasitol*, 34:269-80.

Stevens, J. & Tibayrenc, M. (1996) Trypanosoma brucei s.l.: evolution, linkage and the clonality debate. Parasitology 112:481-8.

Stothard, J.R., Frame, I.A., Carrasco, H.J. & Miles, M.A. (1998a) On the molecular taxonomy of *Trypanosoma cruzi* using riboprinting. *Parasitology* 117:243-7.

Stothard, J.R., Frame, I.A., Carrasco, H.J. & Miles, M.A. (1998b) Temperature gradient gel electrophoresis (TGGE) analysis of riboprints from *Trypanosoma cruzi*. Parasitology 117:249-53.

Stothard, J.R., Frame, I.A. & Miles, M.A. (1997) Use of polymerase chain reaction-based single strand conformational polymorphism and denaturing gradient gel electrophoresis methods for detection of sequence variation of ribosomal DNA of *Trypanosoma cruzi*. *Im.J.Parasitol.* 27:339-43.

Stothard, J.R. & Rollinson, D. (1996) An evaluation of random amplified polymorphic DNA (RAPD) for the identification and phylogeny of freshwater snails of the genus *Bulinus* (Gastropoda: *Planorbidae*). *J.Moll. Stud.* 62: 65-75.

Stothard, J.R., Hughes, S. & Rollinson, D. (1996) Variation within the internal transcribed spacer (ITS) of ribosomal DNA genes of intermediate snail hosts within the genus *Bulinus* (Gastropoda: Planorbidae). Acta Trop. 61:19-29.

Strimmer, K. & von Haeseler, A. (1996) Quartet puzzling: A quartet maximum likelihood method for reconstructing tree topologies. Mol.Biol.Evol. 13:964-9.

Su, X.Z., Carucci, D.J. & Wellems, T.E. (1998) *Plasmodium falciparum*: parasite typing by using a multicopy microsatellite marker, PfRRM. *Exp.Parasitol.* 89:262-5.

Suman, G., Srivastava, J.K., Ray, S., Chandra, R., Srivastava, V.K. & Katiyar, J.C. (1993) Evaluation of enzyme-linked immunosorbent assay in the diagnosis of kala-azar in Malda district (West Bengal). *Indian J.Med. Res.* 97:242-6.

Sun, C-H., Chou, C-F. & Tai, J-H. (1998) Stable DNA transfection of the primitive protozoan pathogen Giardia lamblia. Mol. Biochem. Parasitol. 92:123-32.

Swofford, D.L. (1997) Phylogenetic analysis using parsimony PAUP⁺ 4.0. Sinauer, Sutherland, Massachusetts.

Tait, A. (1980) Evidence for diploidy and mating in trypanosomes. Nature 287:536-8.

Tautz, D. (1989) Hypervariability of simple sequences as a general source for polymorphic DNA markers. Nucl. Acid Res. 17:6463-72.

Tautz, D. & Renz, M. (1984) Simple sequences are ubiquitous repetitive components of eukaryotic genomes. Nucl.Acid Res. 12:4127-38.

Teklemariam, S., Hiwot, A.G., Frommel, D., Miko, T.L., Ganlov, G. & Bryceson, A. (1994) Aminosidine and its combination with sodium stibogluconate in the treatment of diffuse cutaneous leishmaniasis caused by *Leishmania aethiopica*. *Trans.R.Soc.Trop.Med.Hyg.* **38**:334-9.

Tibayrenc, M. (1995) Population genetics of parasitic protozoa and other microorganisms. Adv. Parasitol, 36:47-115.

Tibayrenc, M. & Ayala, F.J. (1991) Towards a population genetics of microorganisms: the clonal theory of parasitic protozoa. *Parasitol.Today* 7:228-32.

Tibayrenc, M., Kjellberg, F. & Ayala, F.J. (1990) A clonal theory for parasitic protozoa: the population structures of *Entamoeba*, *Giardia*, *Leishmania*, *Naegleria*, *Plasmodium*, *Trichomonas* and *Trypanosoma* and their medical and taxonomical consequences. *Proc.Natl.Acad.Sci. USA* 87:2414-8.

Tibayrenc, M., Neubauer, K., Barnabe, C., Guerrini, F., Skarecky, D. & Ayala, F.J. (1993) Genetic characterization of parasitic protozoa: parity between random primer DNA typing and multilocus enzyme electrophoresis. *Proc. Natl. Acad. Sci. USA* 90:1335-9.

Tibayrenc, M., Ward, P., Moya, A. & Ayala, F.J. (1986) Natural populations of *Trypanosoma cruzi*, the agent of Chagas' disease, have complex multiclonal structures. *Proc.Natl.Acad.Sci.USA* 87:2414-8.

Titus, R.G., Gueiros-Filho, F.J., de Freitas, L.A. & Beverley, S.M. (1995) Development of a safe live Leishmania vaccine line by gene replacement. Proc.Natl. Acad.Sci. USA 92:10267-71.

Titus, R.G. & Ribeiro, J.M.C. (1988) Salivary gland lysates from the sandfly Lutzomyia longipalpis enhance Leishmania infectivity. Science 239:1306-8.

Tobin, J.F., Laban, A. & Wirth, D.F. (1991) Homologous recombination in Leishmania enriettii. Proc.Natl.Acad.Sci. USA 38:864-8.

Travi, B.L., Osorio, Y. & Saravia, N.G. (1996) The inflammatory response promotes cutaneous metastasis in hamsters infected with *Leishmania (Viannia) panamensis. J.Parasitol.* 82:454-7.

Truc, P. & Tibayrenc, M. (1993) Population genetics of *Trypanosomo brucei* in Central Africa: taxonomic and epidemiological significance. *Parasttology* **106**:137-49.

Turner, C.M.R., Hide, G., Buchanan, N. & Tait, A. (1995) *Trypanosoma brucet*: inheritance of kinetoplast maxicircles in a genetic cross and their segregation during vegetative growth. *Exp. Parasitol.* **80**:234-41.

Turner, C.M.R., Sternberg, J., Buchanan, N., Smith, E., Hide, G. & Tait, A. (1990) Evidence that the mechanism of gene exchange in *Trypanosoma brucei* involves meiosis and syngamy. *Parasitology* 101:377-86.

Uliana, S.R.B., Affonso, M.H.T., Camargo, E.P. & Floeter-Winter, L.M. (1991) Leishmania : Genus identification based on a specific sequence of the 18S ribosomal RNA sequence. Exp. Parasitol. 72:157-63.

Uliana, S.R.B., Nelson, K., Beverley, S.M., Camargo, E.P. & Floeter-Winter, L.M. (1994) Discrimination amongst *Leishmania* by polymerase chain reaction and hybridization with small subunit ribosomal DNA derived oligonucleotides. *J.Euk Microbiol.* 41:324-30.

Valdes, A.M., Slatkin, M. & Freimer, N.B. (1993) Allele frequencies at microsatellite loci: the stepwise mutation model revisited. *Genetics* 133:737-49.

Van der Ploeg, L.H.T. (1986) Discontinuous transcription and splicing in trypanosomes. Cell 47:479-80.

Van Belkum, A., Ramesar, J., Trommelen, G. & Uitterlinden, A.G. (1992) Mini and microsatellites in the genome of rodent malaria parasites. *Gene* 118:81-6.

Van Eys, G.J.J.M., Schoone, G.J., Kroon, N.C.M. & Ebeting, S.B. (1992) Sequence analysis of small subunit ribosomal RNA genes and its use for detection and identification of *Leishmania* parasites. *Mol.Biochem.Parasitol.*, 51:133-42.

Vexenat, J.A., Barreto, A.C., Cuba, C.A. & Marsden P.D. (1986) Caracteristicas epidemiologicas da leishmaniose tegumentar americana em uma regiao endemica do Estado da Bahia, Brasil. III. Fauna flebotominica. *Mem.Inst. Oswaldo Cruz* **31**:293-301.

Vexenat, J.A. (1998) PhD thesis entitled: "The epidemiology of visceral leishmaniasis in Teresina, Piaui State, Brazil, with special emphasis on the diagnosis and transmissibility of canine infection". University of London

Viard, F., Franck, P., Dubois, M.P., Estoup, A. & Jarne, P. (1998) Variation of microsatellite size homoplasy across electromorphs, loci, and populations in three invertebrate species. *J.Mol.Evol.* 47:42-51.

Vickerman, K. (1994) The evolutionary expansion of the trypanosomatid flagellates. Int.J.Parasitol. 24:1317-31.

Victoir, K., Dujardin, J.C., de Donecker, S., Barker, D.C., Arevalo, J., Hamers, R. & leRay, D. (1995) Plasticity of gp63 gene organization in *Leishmania (Viannia) braziliensis* and *Leishmania (Viannia)* peruviana. Parasitology 111:265-73.

Victoir, K., Banuls A.L., Arevalo J, Llanos-Cuentas A, Harners R, Noel S, De Doncker S, Le Ray D, Tibayrenc M, Dujardin JC. (1998) The gp63 gene locus, a target for genetic characterization of *Leishmania* belonging to subgenus *Viannia*. *Parasitology* 117:1-13.

Vidigal, T.H., Neto, E.D., Carvalho, O.D.S. & Simpson, A.J.G. (1994) Biomphalaria glabrata: extensive genetic variation in Brazilian isolates revealed by random amplified polymorphic DNA analysis. *Exp. Parasitol.* 79:187-94.

Vieira, J.B., Lacerda, M.M. & Marsden, P.D. (1990) National reporting of leishmaniasis: the Brazilian experience. *Parasitol. Today* 6:339-40.

Villalba, E. & Ramirez, J.L. (1982) Ribosomal DNA of Leishmania brasiliensis number of ribosomal copies and gene isolation. J. Protozool. 29:438-41.

Wahren, B. (1996) Gene vaccines. Immunotechnology 2:77-83

Walton, B.C. & Valverde, L. (1979) Racial differences in espundia. Ann. Trop. Med. Parasitol. 73:23-9.

Wang, R., Kafatos, F.C. & Zheng, L. (1999) Microsatellite markers and genotyping procedures for Anopheles gambiae. Parasitol. Today 15:33-7.

Webb, J.R., Button, L. L. & McMaster, W.R. (1991) Heterogeneity of the genes encoding the major surface glycoprotein of *Leishmania donovani*. Mol.Biochem. Parasitol. 48:173-84.

Weigle, K., Santrich, C., Martinez, F., Valderrama, L. & Saravia, N.G. (1993) Epidemiology of cutaneous leishmaniasis in Colombia: a longitudinal study of the natural history, prevalence, and incidence of infection and clinical manifestations. J.Inf.Dis. 168:699-708.

Weigle, K. & Saravia, N.G. (1996) Natural history, clinical evolution, and the host-parasite interaction in New World cutaneous leishmaniasis. *Clinics in Dermatol.* 14:433-50.

Weigle, K.A., Saravia, N.G., Davalos, M., Moreno, L.H. & D'Allesandro, A. (1986) Leishmania braziliensis from the Pacific Coast region of Colombia: Foci of transmission, clinical spectrum and isoenzyme phenotypes. Am.J.Trop.Med.Hyg. 35:722-31.

Weir, B.S. & Cockerham, C.C. (1984) Estimating F-statistics for the analysis of population structure. Evolution 38:1358-70.

Wells, J.M., Prospero, T.D., Jenni, L. & Le Page, R.W.F. (1987) DNA contents and molecular karyotypes of hybrid Trypanosoma brucei. Mol. Biochem. Parasitol. 24:103-16.

Welsh, J. & McClelland, M. (1990) Fingerprinting genomes using PCR fingerprinting with arbitrary primers. Nucl.Acid Res. 18: 7213-8.

Williams, J.G.K., Hanafey, M.K., Rafalski, J.A. and Tingey, S.V. (1993) Genetic analysis using random amplified polymorphic DNA markers. *Methods Enzymol.* 218:704-40.

Williams, J.G.K., Kubelik, A.R., Livak, K.J., Rafolski, J.A. & Tingey, S.A. (1990) DNA polymorphisms amplified by arbitrary primers are useful as genetic markers. Nucl. Acid Res. 18:6531-5.

Wilson, S.M., McNerney, R., Moreno, M.B., Frame, I. & Miles, M.A. (1992) Adaptation of a radioactive *L. donovani* complex DNA probe to a chemiluminescent detection system gives enhanced sensitivity for diagnostic and epidemiological applications. *Parasitology* 104:421-6.

Wincker, P., Ravel, C., Blaineau, C., Pages, M., Jauffret, Y., Dedet, J-P. & Bastien, P. (1996) The *Leishmania* genome comprise 36 chromosomes conserved across widely divergent human pathogenic species. *Nucl.Acid Res.* 24:1688-94.

Wirth, D.F. & Pratt, D.M. (1982) Rapid identification of Leishmania species by specific hybridization of kinetoplast DNA in cutaneous lesions. Proc. Natl. Acad. Sci. USA 79:6999-7003.

World Health Organization (1990) Control of the Leishmaniases. WHO Tech.Rep.Ser. 793.

World Health Organization (1993) Control of Tropical Diseases. The Leishmaniases.

Xu, D. & Liew, F.Y. (1995) Protection against leishmaniasis by injection of DNA encoding a major surface glycoprotein, gp63 of L. Major. Immunology 84:173-6.

Xu, D., McSorley, S.J., Chatfield, S.N., Dougan, G. & Liew, F.Y. (1995) Protection against Leishmania major infection in genetically-susceptible BALB/c mice by gp63 delivered orally in attenuated Salmonella typhimurium (AroA- AroD-). Immunology. 85:1-7.

Yardley, V. & Croft, S.L. (1997) Activity of liposomal amphotericin B against experimental cutaneous leishmaniasis. Antimicrob.Agents Chemother. 41:752-6.

Yoshida, E.L.A., Correia, F.M.A., Marques, S.A., Stolf, H.O., Dillon, N.L., Momen, H. & Grimaldi, G. Jnr. (1990) Human canine and equine (*Equus caballus*) leishmaniasis due to *Leishmania braziliensis* in the South-West region of São Paulo State, Brazil. *Mem.Inst.Oswaldo Cruz* **85**:133-4.

Zeina, B., Banfield, C. & al-Assad, S. (1997) Topical glyceryl trinitrate: a possible treatment for cutaneous leishmaniasis. *Clin.Exp.Dermatol.* 22:244-5.

Zhang, Q., Tibayrenc, M. & ayala, F.J. (1988) Linkage disequilibrium in natural populations of *Trypanosoma cruzi* (flagellate), the agent of Chagas' disease. J. Protozool. 35:81-5.

Zhivotovsky, L.A. (1999) Estimating population structure in diploids with multilocus dominant DNA markers. *Mol.Ecol.* 8:907-13.

Zijlstra, E.E., Daifalla, N.S., Kager, P.A., Khalil, E.A., El-Hassan, A.M., Reed, S.G. & Ghalib, H.W. (1998) rK39 enzyme-linked immunosorbent assay for diagnosis of *Leishmania donovani* infection. *Clin.Diagn.Lab.Immunol.* 5:717-20.

Zijlstra, E.E., Siddig Ali, M., El-Hassan, A.M., El-Toum, I.A., Satti, M., Ghalib, H.W. & Kager, P.A. (1992) Kala-azar: a comparative study of parasitological methods and the direct agglutination test in diagnosis. *Trans.R.Soc.Trop.Med.Hyg.* 86:505-7.

Zimmer, E.A., Martin, S.L., Beverley, S.M., Kan, Y.W. & Wilson, A.C. (1980) Rapid duplication and loss of genes coding for the alpha-chains of hemoglobin. *Proc.Natl.Acad.Sci. USA* 77:2158-62..

APPENDIX I - CULTURE MEDIA*

*All reagents were purchased from Sigma unless otherwise stated.

LEISHMANIA SUBGENUS VIANNIA CULTURE MEDIA

4N slopes (makes approx. 25 slopes)

2g	Blood agar (Difco)
5mL	Defibrinated rabbit blood (St. Georges' Hospital)
1mL	Gentamycin (50mg/mL) (Gibco BRL)

Blood agar was added to 50mL deionized-distilled water (ddH₂O). The mix was sterilized by autoclaving (121°C for 15 minutes) before storing at room temperature until required. When 4N slopes were required, the agar was melted at low heat in a microwave. The agar was cooled to 56°C: rabbit blood and gentamycin were added and the mixture was swirled gently. 2mL aliquots were poured into bijoux tubes, sealed and left to set at an angle to produce the slope. 4N slopes were stored (up to 1 month) at +4°C until required.

2.5X α-modified MEM stock (makes 2L of 2.5X stock)

5L equivalent	aMEM powder
11.00 g	NaHCO ₃
2.95 g	L-glutamine
15.00 g	D-glucose
50.00 g	HEPES

The above reagents were dissolved in ~1900mL ddH₂O water. The pH was adjusted to 7.5 (+/- 0.5) using 10N NaOH. The solution was filter-sterilized through 2X 1L 0.22 μ M filter units (Sartorius) into sterile 1L Schott bottles.

200mg	Adenine
25mg	Folic Acid
10mg	D-biotin

The above reagents were dissolved in 2.5mL 1M NaOH. 17.5mL ddH₂O water was added: 10mL of this solution was filtered into each 1L bottle. Medium was stored (up to 3 months) at $+4^{\circ}$ C until required.

500X haemin

37.5mg	bovine haemin
15mL	50% triethanolamine

Haemin was dissolved in 50% ethanolamine by shaking vigorously. The solution was filter-sterilized through a 0.22μ M filter unit (Sartorius) and stored at $+4^{\circ}$ C.

Liquid medium (makes 1 L)

400mL	2.5X α-MEM stock
2mL	500X haemin
100mL	Heat-inactivated foetal calf serum (FCS)
1 mL	Gentamycin solution (50mg/mL stock) (Gibco BRL)

 α -MEM stock was diluted to working strength with 600mL autoclaved ddH₂O water. 100mL was decanted into a sterile container and stored at +4°C for later use. FCS, haemin and gentamycin were added. Small aliquots were removed for sterility testing: after overnight incubation at 23°C and 37°C these were examined microscopically for microbial growth Liquid medium was stored at +4°C until required (up to 1 month).

Proline-balanced salt solution (PBSS)

800mg	KCl
120mg	Na ₂ HPO ₄ .12H ₂ O
120mg	KH ₂ PO ₄
370mg	CaCl ₂ .2H ₂ O
200mg	MgSO ₄ .7H ₂ O
200mg	MgCl ₂ .6H ₂ O
16g	NaCl
2g	L-proline
2mg	Phenol red indicator

Reagents were dissolved in 1950 mL ddH₂O water. The pH was adjusted to 7.2 with Trizma base and the solution made up to 2L with ddH₂O water. The solution was aliquoted into 4 x 500mL bottles and autoclaved. PBSS was stored (up to 6 months) at $+4^{\circ}$ C until required.

BACTERIAL CULTURE MEDIA

LB medium

5g	Bacto-tryptone (Difco)
2.5g	Bacto-yeast extract (Difco)
5g	NaCl

The reagents were dissolved in 300mL ddH_2O water, made up to 500mL with ddH_2O water and autoclaved. LB medium was stored at room temperature (up to 6 months) until required.

LB agar plates

7.5g 500mL Bacto-agar (Difco) LB medium

The reagents were mixed together and autoclaved. The agar solution was cooled to approximately 50° C before immediate pouring into petri dishes. Plates were left to set, uncovered, at room temperature. Once set, lids were replaced and plates were kept at 4° C until required.

APPENDIX II - GENERAL REAGENTS

10% AMPS

2g ammonium persulphate (AMPS) 20mL distilled water

Dissolve AMPS in water. Aliquot into 20 X 1mL and store at -20°C.

1M CaCl₂

5.4g	CaCl ₂ .6H ₂ O
20mL	MilliQ water

Dissolve CaCl_{2.6}H₂O. Filter-sterilize through a 0.2μ M filter. Aliquot into 20 X 1mL and store at -20°C.

Cell lysis buffer

25 mL	IM NaCl
50 mL	0.5M EDTA
50 mL	10% SDS
25 mL	1M Tris-HCl, pH 8.0

Mix reagents thoroughly and divide into 20mL aliquots. Autoclave and store at room temperature until required.

DGGE electrophoresis buffer (MOPS - 50X)

209.3g	MOPS (3-[N-morpholino]propane-sulphonic acid)
18.6g	EDTA
36g -	NaOH

Dissolve reagents in ~800mL ddH₂O. Make up to 1L. Filter-sterilize through a $0.2\mu M$ filter. Store at room temperature.

DGGE gel mix

50X DGGE (MOPS) buffer
Ultrapure bis-acrylagel* (2% N,N'methylene bis-acrylamide solution)
Ultrapure acrylagel* (30% acrylamide solution)
glycerol
urea

Mix reagents thoroughly using heated magnetic stirrer, adding distilled water to 400mL. Store at +4°C until required. For each gel use 40mL gel mix and add 80μ L TEMED and 260 μ L 10% AMPS. *(National Diagnostics, Hull, UK).

DGGE loading buffer

10mg	xylene cyanol FF
10mg	bromophenol blue
1g	sucrose
10mL	distilled water

Mix reagents thoroughly. Aliquot into 10 x 1mL. Store at +4°C.

Appendix II - General Reagents

 PBS (10X)
 80g
 NaCl
 2g
 KCl
 11.5g
 Na2HPO4
 2g
 KH2PO4
 2g

Make up to 1L with ddH_2O , adjust to pH 7.4 and autoclave. Store at room temp. until required.

Plasmid extraction buffers

Buffer 19gglucose2.5gEDTA2mL0.5M Tris-HCl, pH 8.0

Make up to 100mL with ddH₂O and autoclave. Store at room temperature.

Buffer 2

0.2N NaOH 1% SDS

Make up fresh as required.

Buffer 3 60mL 5M potassium acetate 1.5mL glacial acetic acid

Make up to 100mL with ddH₂O. Store at room temperature.

Silver staining solutions and protocol (Vidigal et al., 1994)

Buffer A

100mL ethanol (Analar grade, BDH) 5mL glacial acetic acid 895mL distilled water

Mix reagents together thoroughly. Store at room temperature.

Buffer B

500mg silver nitrate 500mL distilled water

Mix reagents together thoroughly. Store at room temperature in a covered glass bottle. Buffer B can be reused 5-10X.

Buffer C

7.5gNaOH50mgsodium borohydride (NaBH4)2mLformaldehyde500mLdistilled water

Denhardt's solution (50X)

- 5g Ficoll (type 400, Pharmacia)
- 5g Polyvinylpyrrolidone
- 5g BSA (fraction V)

Make up to 500mL with ddH2O. Aliquot and store at -20°C.

0.5M EDTA, pH 8.0

93.05g EDTA

Dissolve EDTA in ~400mL distilled water by stirring on a magnetic stirrer. Adjust pH to 8.0 using NaOH pearls. Sterilize by autoclaving.

Electroporation buffer

46.55g	sucrose
3.5mL	1M sodium phosphate, pH 7.2
300mL	distilled water

Make up to 500mL with ddH₂O. Aliquot and store at -20°C.

Enzyme stabiliser (stock)

0.5mL	200mM EDTA, pH 7.0
15.4mg	dithithreitol (DTT).
13.1mg	ε-amino-caproic acid (□-ACA)

Dissolve DTT in EDTA solution. Add this to the ε -ACA and pipette gently to dissolve. Store at +4°C for up to 2 months (discard if turns yellow). Make up working strength solution (1/1000 in ddH₂O) fresh for preparing stabilates.

Ethidium bromide

0.1g ethidium bromide 10mL distilled water

Stir on a magnetic stirrer for 2 hours. Store in dark bottle at room temperature.

Gel electrophoresis (agarose and polyacrylamide) loading buffer

- 0.05g xylene cyanol FF
- 0.05g orange G
- 15mL glycerol
- 35mL distilled water

Mix reagents and aliquot into 5 x 10mL. Autoclave and store at room temperature.

PAGE gel mix (non-denaturing)

40mL	10X TBE buffer
42mL	Ultrapure bis-acrylagel* (2% N,N'methylene bis-acrylamide solution)
77.6mL	Ultrapure acrylagel* (30% acrylamide solution)
20mL	glycerol
220.4mL	distilled water

Mix reagents thoroughly using magnetic stirrer. Store at +4°C until required. For each gel use 40mL gel mix, filtered through a 1 μ M filter, and add 80 μ L TEMED and 260 μ L 10% AMPS. *(National Diagnostics, Hull, UK)

Appendix II - General Reagents

PBS (10X) 80g NaCl 2g KCl 11.5g Na₂HPO₄ 2g KH₂PO₄

Make up to 1L with ddH₂O, adjust to pH 7.4 and autoclave. Store at room temp. until required.

Plasmid extraction buffers

Buffer 19gglucose2.5gEDTA2mL0.5M Tris-HCl, pH 8.0

Make up to 100mL with ddH₂O and autoclave. Store at room temperature.

Buffer 2 0.2N NaOH 1% SDS

Make up fresh as required.

Buffer 3 60mL 5M potassium acetate 1.5mL glacial acetic acid

Make up to 100mL with ddH₂O. Store at room temperature.

Silver staining solutions and protocol (Vidigal et al., 1994)

Buffer A

100mL ethanol (Analar grade, BDH) 5mL glacial acetic acid 895mL distilled water

Mix reagents together thoroughly. Store at room temperature.

Buffer B

500mg silver nitrate 500mL distilled water

Mix reagents together thoroughly. Store at room temperature in a covered glass bottle. Buffer B can be reused 5-10X.

Buffer C

7.5gNaOH50mgsodium borohydride (NaBH4)2mLformaldehyde500mLdistilled water

Mix reagents together thoroughly. Buffer C must be made up immediately before use, ensuring that the NaBH₄ is dissolved.

Buffer D

3.75g sodium carbonate 500mL distilled water

Mix reagents together thoroughly. Store at room temperature.

Protocol

- 1. Incubate gel in buffer A with agitation (i.e. on rocker) for 2 X 3 mins.
- 2. Incubate gel in buffer B, as previously, for 10 mins.
- 3. Wash gel twice for 15s in distilled water.
- 4. Incubate gel in buffer C, as previously, for 20 mins or until bands are visible.
- 5. Discard buffer and rinse gel briefly in distilled water.
- 6. Soak gel in buffer D for 20 mins. Discard buffer D
- 7. Seal gel in plastic bag with a little buffer D. Store flat.

SSC (20X)

175.3g NaCl 88.2g Trizma base

Make up to 1L with ddH₂O and adjust to pH 7.0. Sterilize by autoclaving.

TAE buffer (50X)

242gTrizma base57.1mLglacial acetic acid (BDH)100mL0.5M EDTA, pH 8.0

Make up to 1L with ddH₂O.

TBE buffer (10X)

108g	Trizma base
55g	boric acid (BDH)
40mL	0.5M EDTA, pH 8.0

Make up to 1L with ddH₂O.

TE buffer

5mL	1M Tris-HCl, pH 7.2
1mL	0.5M EDTA, pH 8.0

Make up to 500mL with ddH₂O.

APPENDIX III - IEA REAGENTS AND CONDITIONS

L <u>THIN-LAYER STARCH GEL ELECTROPHORESIS</u> (TSGE)

Thin-layer starch-gel electrophoresis (TSGE) was carried out essentially as described by Godfrey & Kilgour (1976) and Harris & Hopkinson, (1976). Conditions for electrophoresis and enzyme development were based on those of Miles *et al.* (1980a, 1980b, 1981) and Evans *et al.* (1984).

Enzyme	Tank Buffer [†] (see below for recipes)	Gel buffer (tank buffer : water)	Volts	Running Time (hours)
MPI	6	1:4	300 V	2,5
NHi	1	1:4	200 V	2.0
NHd	1	1:4	200 V	2.0
ES	1	1:9	300 V	2.5
PEPD	7	1:7	250 V	2.0
PGM	2	1:4	250 V	2.5
6PGD	1	1:9	300 V	2.5
GPI	1	1:9	300 V	2.5
G6PD	9	1:4	150 V	3.5
ASAT	3	1:9	300 V	2.0
ALAT	3	1:9	300 V	2.0
MDH	4	2:1	250 V	4.0
ICD	5	1:1	250 V	3.0
SOD	8	1:9	300 V	2.5
PK	3	1:9	300 V	2.5
ME	8	1:4	200 V	3.0

TSGE electrophoresis conditions for Leishmania Viannia species

†<u>TSGE Tank Buffer Stock Recipes</u>

All TSGE buffer stocks were made up and stored at room temperature for up to 6 months. The day before use, buffers were made up to 1X strength using distilled water and stored overnight at 4°C.

1. TMEM (5X stock)

GPI, 6PGD, ES (pH 7.4); NHi/NHd (pH 8.0)

60.55g	Trizma base
58.05g	Maleic Acid
10.7g	Magnesium acetate
18.6g	EDTA

Reagents were dissolved ~800mL distilled water. The solution was adjusted to the required pH with 10N NaOH and made up to 1L.

Appendix III - IEA Reagents and Conditions

2. TBEM (5X stock)

3.

4.

PGM (pH 9.0)

151.35g	Trizma base
17.0g	Boric Acid
8.0g	Magnesium acetate
18.6g	EDTA

Reagents were dissolved ~800mL distilled water. The solution was adjusted to pH 9.0 with 10N NaOH and made up to 1L.

TRIS - CITRIC ACID (10X stock)

ASAT, ALAT, PK (pH 9.0)

90.85g	Trizma base (0.15M)
7.85g	Citric Acid (0.0075M)

Reagents were dissolved ~400mL distilled water. The solution was adjusted to pH 9.0 with 10N NaOH and made up to 500mL.

TRIS - CITRIC ACID (10X stock)

 166.4g
 Trizma base (0.25M)

 94.2g
 Citric Acid (0.09M)

Reagents were dissolved ~400mL distilled water. The solution was adjusted to pH 7.0 with 10N NaOH /conc. HCl and made up to 500mL.

5. TRIS - CITRIC ACID (10X stock)

ICD (pH 8.8)

MPI (pH 7.6)

106.2g	Trizma base (0.4M)
52.3g	Citric Acid (0.05M)

Reagents were dissolved ~400mL distilled water. The solution was adjusted to pH 7.0 with 10N NaOH and made up to 500mL.

6. TRIS-PHOSPHATE (10X stock)

 60.55g
 Trizma base (0.1M)

 78.0g
 NaH2PO4.2H2O (0.1M)

Reagents were dissolved ~400mL distilled water. The solution was adjusted to pH 7.6 with 10N NaOH and made up to 500mL.

7. TRIS- PHOSPHATE (5X stock)

 90.85g
 Trizma base (0.15M)

 78.0g
 NaH₂PO₄.2H₂O (0.1M)

Reagents were dissolved ~800mL distilled water. The solution was adjusted to pH 8.2 with 10N NaOH and made up to 1L.

PEPD (pH 8.2)

MDH (pH 7.0)

Appendix III - IEA Reagents and Conditions

8. **PHOSPHATE** (5X stock)

ME, SOD (pH 7.4)

 136.28g
 Na₂HPO₄ (0.192M)

 6.24g
 NaH₂PO₄.2H₂O (0.08M)

Reagents were dissolved ~800mL distilled water. The pH of the solution was checked and adjusted to pH 7.4 with 10N NaOH and made up to 1L.

9. PHOSPHATE (10X stock)

G6PD (PH 7.0)

57.5g	Na_2HPO_4 (0.081M)
1.48g	NaH ₂ PO ₄ .2H ₂ O (0.019M)

Reagents were dissolved ~400mL distilled water. The pH of the solution was checked and adjusted to pH 7.0 with 10N NaOH and made up to 500mL.

II. <u>CELLULOSE ACETATE ELECTROPHORESIS</u> (CAE)

CAE was carried out according to the methods described by Lanham *et al.* (1981), Evans *et al.* (1989) and according to the manufacturer's handbook (Helena Laboratories, Beaumont, Texas, USA). Conditions for electrophoresis were adapted from those used for TSGE.

CAE conditions for Leishmania Vlannia species

Enzyme	Tank Buffer ^{co} (see below for recipes)	Further dilution of 1X tank buffer : water)	Volts	Running Time (mins.)
MDH	1.1	-	160 V	45
ME	1.1	-	160 V	45
PEPD	1.2	-	275 V	20
ICD	2	-	200 V	50
NHi	3	3:37	275 V	30
SOD	4	1:2	275 V	30

²⁰CAE Tank Buffer Recipes

Tris barbital buffer stock was made up and kept refridgerated, without sucrose, for up to 1 week. All other CAE buffer stocks were made up and stored at room temperature for up to 6 months. The day before use, buffers were made up to 1X or working strength using distilled water and stored overnight at 4°C. Sucrose was added to a final concentration of 15% to all buffers immediately before electrophoresis. Sucrose was added in order to reducing the current across the plate, thus improving resolution of isoenzyme bands.

1. TRIS BARBITAL / SODIUM BARBITAL (stock, pH 9.0)

Tris-barbital buffer was obtained in preweighed powdered form from Helena Laboratories (Beaumont, Texas, USA). 1 sachet was dissolved in 500mL distilled water to make the stock buffer [ionic strength, I = 0.1] for use in making tank buffers 1.1 and 1.2.

1.1 TRIS BARBITAL / SODIUM BARBITAL / MAGNESIUM ACETATE

MDH, ME (pH 9.0 [I = 0.075])

300mL	Stock buffer	
100mL	distilled water	
86mg	magnesium acetate	

Reagents were mixed until dissolved and stored overnight at +4°C before use.

1.2 TRIS BARBITAL / SODIUM BARBITAL PEPD (pH 9.0 [I = 0.02])

100mL	Stock buffer
400mL	distilled water

Reagents were mixed and stored overnight at +4°C before use.

2. TRIS-GLYCINE (10X stock)

30gTrizma base144gglycine

Reagents were dissolved in ~400mL distilled water and made up to 500mL.

3. PHOSPHATE (5X stock)

NHi (pH 8.0)

ICD (pH 8.5)

68.13g	Na ₂ HPO ₄ (0.192M)
3.12g	NaH ₂ PO ₄ .2H ₂ O (0.08M)

Reagents were dissolved in ~400mL distilled water. The pH of the solution was adjusted, if necessary, to pH 8.0 with 10N NaOH and made up to 500mL.

4. **TBEM 2** (5X stock)

SOD (pH 9.0)

75.69g	Trizma base
8.5g	Boric Acid
4.02g	Magnesium acetate
4.65g	EDTA

Reagents were dissolved ~400mL distilled water. The solution was adjusted to pH 9.0 with 10N NaOH and made up to 500mL.

III DEVELOPMENT CONDITIONS FOR VIANNIA SPECIES (TSGE and CAE⁺)

Enzyme Code	Developer Mix		Visualization Method*
	0.3M Tris-HCl, pH 7.4	12 mL	
MPI	0.1M MgCl ₂	2 mL	1
	NADP (10 mg/mL)	0.6 mL	
	G6PD (1000 U/mL)	20 µL	
	PGI (2U/μL)	15 μL	
	M6P	10 mg	

Appendix III - IEA Reagents and Conditions

Enzyme Code	Developer Mix		Visualization Method*
NHi	0.3M Tris-HCl, pH 7.0	16 mL	
	Xanthine Oxidase	50 µL	1
	Inosine	20 mg	
NHd	0.3M Tris-HCl, pH 7.0	16 mL	
	Xanthine Oxidase	50 µL	1
	Deoxyinosine	20 mg	
ES	0.1M PO4, pH 7.4	6 mL	
	4-MUB [†]	1 mg	5
	0.3M Tris-HCl, pH 7.4	14 mL	
PEPD	0.5M MnCl ₂	0.4 mL	3
	L-leucyl L-proline	20 mg	
	Peroxidase	2 mg	
	Snake venom	2 mg	
	0.3M Tris-HCl. pH 8.0	12 mL	
PGM	0.1M MgCl ₂	2 mL	1
	NADP (10 mg/mL)	0.6 mL	
	G6PD (1000 U/mL)	10 uL	
	GIP + GI 6diP (20 mg/mL)	1.5 mL	
	0 3M Tris-HCl pH 7 4	12 mI	
6PGD	0 1M MgCla	2 mL	1
	NADP (10 mg/mI)	0.6 mL	•
	6PG(10 mg/mL)	1 mL	
	0.3M Tric HCl pH 7.0	14.5mI	
G6PD	1M McCh	0.5 mI	1
	1000000000000000000000000000000000000	1.0 ml	
	NADP (10 mg/mL) $C(D(10 mg/mL))$	1.0 mL	
		12.4mL	
GPI	U.SM THS-HCI, pH 8.0	13.4mL	,
UII	IM MgCl ₂	0.4 mL	1
	NADP (10 mg/mL)		
	G6PD (1000 U/mL)	10 μL	
_	F6P (10 mg/mL)	1.0 mL	
ASAT	0.1M PO4, pH 7.4	6 mL	
ASAI	NADH	5 mg	4
	MDH	6 μL	
	α -ketoglutaric acid	12 mg	
	Aspartic acid	20 mg	<u> </u>
	0.1M PO4, pH 7.4	6 mL	
ALAT	NADH	5 mg	4
	LDH	6 µL	
	a-ketoglutaric acid	12 mg	
	L-alanine	80 mg	
	0.3M tris-HCl, pH 7.4	6.6 mL	
MDH	water	8 mL	1
	NAD (10mg/mL)	1.0 mL	
	IM L-malate, pH 7.0	1.5 mL	
	0.1M PO4. pH 8.0	18.8mL	
SOD	200mM EDTA pH 8.0	0.2 mL	1.2
	riboflavin	0.6 mg	-, -

Appendix III - IEA Reagents and Conditions

Enzyme Code	Developer Mix		Visualization Method*
	0.3M tris-HCl, pH 7.0	14 mL	
ICD	1M MgCl ₂	1 mL	1
	NADP (10 mg/mL)	1 mL	
	dl-isocitric acid (Na3 salt)	40 mg	
	0.3M tris-HCl, pH 7.4	13.3mL	
РК	0.1M MgCl ₂	4 mL	6
	ADP(15mg/mL)	2.7 mL	
	IMKĊI	3 mL	
	LDH	10 µL	
	NADH	5 mg	
	F1.6 diP	100 mg	
	PEP	22 mg	
	0.3M tris-HCl, pH 7.4	13.4mL	
ME	1M MgCl ₂	0.4mL	1
	NADP (10 mg/mL)	0.4mL	
	1.0M L-malate, pH 7.0	1.2 mL	

§ 25 mg/mL in ethanol, filtered

[†] 4-methylumbilliferryl butyrate (4-MUB), dissolved in 200µL acetone

*Visualization Methods

 PMS (2 mg/mL) - 1.0 mL MTT (5 mg/mL) - 1.0 mL Agar (1.2%) - 20mL. Incubate at 37°C and monitor for purple/black bands.

2. NBT (5mg/mL) - 1.0 mL

Agar (1.2%) - 20mL.

Expose to light for 10 mins., incubate at 37°C and monitor for purple/black bands.

 3-amino, 9-ethyl carbazole (25mg/mL in ethanol) - 1.0mL (filtered) Agar (1.2%) - 20mL.

Incubate at 37°C and monitor for red/brown bands.

- 4. Pour developer onto Whatman No. 1 filter paper and place soaked filter paper over starch gel, avoiding bubbles between the gel and filter paper. Incubate at 37°C and monitor under UV illumination ($\lambda = 340$ nm) for dark bands on a fluorescent background.
- 5. Agar (1.2%) 20mL.

+

Incubate at 37°C and monitor under UV illumination ($\lambda = 340$ nm). for fluorescent bands on a dark background.

6. Agar (1.2%) - 20mL. Incubate at 37°C and monitorunder UV illumination ($\lambda = 340$ nm) for dark bands on a fluorescent background.

For visualization of CAE plates, the volumes and weights listed above were halved and 15% (w/v) sucrose was added (Lanham *et al.*, 1981). Staining reactions were carried out in square petri dishes. The developer was poured into the dish and allowed to set (carried out in the dark). The CAE plate was then placed face down onto the agar-based developer. Reactions were stopped by removing the plate from the developer and immersing it in distilled water.

APPENDIX IV - PCR PRIMERS AND CYCLING PARAMETERS

RAPD ANALYSIS

PCR primers

Primer	Primer sequence (5' - 3')	Primer	Primer sequence $(5' - 3')$
A1*	тса сда тдс а	НЗ	CAT CCC CCT G
A2	GAA ACG GGT G	H4	TGC CGA GCT G
A3*	AGT CAG CCA C	H5	TAG GAT CAG A
A4*	AAT CGG GCT G	H6	CAC ATG CTT C
A5	GTG ACG TAG G	D1	CAG GCC TTC
A6	GTG ATC GCA G	D2	AGG GGT CTT C
L1*	CGG CCC CTG T	D3*	TTC CGA ACC C
L2	CGG ACG TCG C	D4*	GGT CCC TGA C
L3*	CCC GCC ATC T	D5	GGG TAA CGC C
L4*	GTG GAT GCG A	D6*	CAA TCG CCG T
L5*	AAG AGC CCG T	D7*	TCG GCG ATA G
L6	AAG GAT CAG A	D8	AGC CAG CGA A
H1*	CGC GCC CGC T	D9*	AGG TGA CCG T
H2*	TTC CCC CGC T	D10*	GTT GCG ATC C

Primers indicated with the symbol " \bullet " were found to clearly distinguish between the L. V. braziliensis and L. V. guyanensis species complexes. Primers in **bold** type were found to distinguish putative parental and hybrid strains from Nicaragua.

Cycling parameters

Temp. (*C)	Time (mins)	No. of cycles
94	5	
37	1	1
72	1	
94	1	40
37	1	40
72	1	
72	10	1

Appendix IV - PCR Primers and Cycling Parameters

<u>RIBOPRINTING</u> (SSUrRNA)

(Uliana et al., 1991)

PCR primers

Primer	Primer sequence (5' - 3') and Tm
DN-SSU1	GAT CTG GTT GAT TCT GCC AG Tm = 51.6° C
DN-SSU2	GAT CCA GCT GCA GGT TCA CC Tm = 55.7° C

Cycling parameters	Temp. (°C)	Time (mins)	No. of cycles
	95	5	
	58	1	1
	72	2	
	95	1	
	58	0.5	25
	72	1.5	
	72	10	1

PCR-RFLP ANALYSIS (rRNA ITS) (Cupolillo et al., 1995; Stothard et al., 1996)

PCR primers

Primer	Primer sequence (5' - 3') and Tm
DN-IR1	GCT GTA GGT GAA CCT GCA GCA GCT GGA TCA TT
DN-IR2	IIII = 65.5 C GCG GGT AGT CCT GCC AAA CAC TCA GGT CTC $IIII = 66.9^{\circ}C$

Cycling parameters	Temp. (*C)	Time (secs.)	No. of cycles
	95	15	
	58	30	37
	72	90	
	72	10 min.	1

Appendix IV - PCR Primers and Cycling Parameters

MICROSATELLITE ANALYSIS

(Russell et al., 1999)

PCR primers

Primer	Primer sequence (5' - 3') and Tm
AC01 A	GAG AGG CCA CCA GAC ACG TCA GCA CAC $Tm = 71.4^{\circ}C$
AC01 B	CCC CCT TCC TTC GCC TTC AAC ACC TTT AC Tm = 70.1° C
AC16 A	CTT CTT CTC ATG CTG CAC GGT CTC CTC CTT $Tm = 70.2^{\circ}C$
AC16 B	CCA TGG GCG GGC TTG TTT CGT TAC TTT TTA Tm = 70.1°C
AC52 A	$CCA CCG CCG GCT TCA CTA C$ $Tm = 63.4^{\circ}C$
AC52 B	$ \begin{array}{l} \mathbf{GCG} \mathbf{GCA} \mathbf{ATC} \mathbf{GTC} \mathbf{TGG} \mathbf{CTA} \mathbf{AA} \\ \mathbf{Tm} = \mathbf{62.6^{\circ}C} \end{array} $

Cycling parameters

Temp. (*C)	Time (mins)	No. of cycles
95°C	30 s.	
62°C (60°C)	30 s.	35 (38)
72°C	60 s.	
72°C	10 min.	1

An annealing temperature of 60°C was used for amplification of the AC16 locus. This temperature was also used to amplify the AC01 and AC52 loci for isolates which gave poor yield at the higher annealing temperature; in addition, for these isolates, the number of amplification cycles was increased to 38.

STANDARD CODING SYSTEM FORMAT FOR NUCLEOTIDES AND NUCLEOTIDE COMBINATIONS:

Symbol	Meaning	Symbol	Meaning
Α	Adenine	K	T or G
C	cytosine	M	C or A
G	guanine	B	C or G or T
T	thymine	D	A or G or T
Y	C or T (pyrimidine)	Н	A or C or T
R	A or G (purine)	V	A or C or G
w	A or T	X or N	unknown
S	C or G		

APPENDIX V - POPULATION GENETIC ANALYSIS

Locus	No. alleles	Allele	Alleie	No. possible	Genotype	Gene	otype iency
	scored		frequency	genotypes		Observed	Expected
		5	0.0259		5/5	0	0.026
MPI	4	6	0.6552	10	6/5	3	1.983
					6/6	22	24.783
		7	0.2931		7/5	0	0.887
					7/6	26	22.470
					7/7	4	4.878
		9	0.0259		9/5	0	0.078
					9/6	3	1.983
					9/7	0	0.887
					9/9	0	0.026
		2	0.6983		2/2	23	28.174
NHi1	2		1	3	3/2	35	24.652
		3	0.3017	1	3/3	0	5.174
		4	0.7759		4/4	32	34.826
NHd	2			3	5/4	26	20.348
		5	0.2241	1	5/5	0	2.826
		4	0.8103		4/4	39	38.009
ES	2			3	5/4	16	17.983
		5	0.1897		5/5	3	2.009
		2	0.7845		2/2	36	35.609
PEPD	2			3	4/2	19	19.783
		4	0.2155		4/4	3	2.609
	1	2	0.0536		2/2	3	0.135
PGM	2			3	3/2	0	5.730
		3	0.9464	1	3/3	53	50.135
					null	2	-
		2	0.0517		2/2	3	0.130
GPI	2			3	4/2	0	5.722
		4	0.9483		4/4	54	51.118
					null	1	-
		2	0.9483		2/2	55	52.130
G6PD	2			3	3/2	0	5.739
		3	0.0517		3/3	3	0.130
NH12	1	7	1.0000	1	7/7	58	-
6PGD		5	1.0000	1	5/5	58	-
ASAT	1	5	1.0000	1	5/5	58	-
ALAT	1 i	4	1.0000	1	4/4	58	-

Table V-1 Allele and genotype frequency results based on IEA data from Huánuco stocks[±]

analysis excludes L. V. lainsoni stock LC2679.

*

Locus	No.	Allele	Allele	No. possible	Genotype	Gene	otype
	scored		frequency	genotypes		Observed	Expected
		225	0.0189		225/225	1	0.010
AC01	8	227	0.4057	36	227/225	0	0.819
					/ 227	0	8.600
		229	0.0377	9 scored	229/225	0	0.076
					/ 227	0	1.638
					/ 229	0	0.057
		231	0.3774		231/225	0	0.762
					/ 227	39	16.381
					/ 229	2	1.524
					/ 231	0	7.429
		233	0.0377		233/225	0	0.076
					/ 227	1	1.638
					/ 229	3	0.152
					/ 231	0	1.524
					/ 233	0	0.057
		235	0.0377	1	235/225	0	0.076
			1		/ 227	0	1.638
	i				/ 229	0	0.152
					/ 231	0	1.524
				1	/ 233	0	0.152
					/ 235	2	0.057
		237	0.0660	1	237 / 225	0	0.133
				1	/ 227	3	2.867
				1	/ 229	0	0.267
				1	/ 231	0	2.667
				1	/ 233	0	0.267
					/ 235	0	0.267
				1	/ 237	2	0.200
		241	0.0189		241/225	0	0.038
					/ 227	0	0.819
					/ 229	0	0.076
					/ 231	0	0.762
					/ 233	0	0.076
				1	/ 235	0	0.076
				-	/ 237	0	0.133
					/ 241	1	0.010
		240	0.0769		240 / 240	3	0.272
AC16	10	244	0.4231	55	244 / 240	2	3.417
			1	1	/ 244	17	9.184
		246	0.0289	14 scored	246/240	0	0.233
	1				/ 244	3	1.282
				1	/ 246	0	0.029
		248	0.2885		248 / 240	0	2.330
					/ 244	1	12.816
				1	/ 246	0	0.874
					/ 248	13	4.223
					/ 244	0	1.282

Table V-2Allele and genotype frequency results based on AC01 & AC16
microsatellite data from Huánuco stocks[±]

337

Locus	No. alleles	Allele	Allele	No. possible	Genotype	Gene	otype iency
	scored		frequency	genotypes		Observed	Expected
					248/246	0	0.087
AC16				55	/ 248	3	0.874
		250	0.0865		250 / 240	0	0.699
				14 scored	/ 244	2	3.845
					/ 246	0	0.262
					/ 248	0	2.621
					/ 250	1	0.350
		252	0.0385		252/240	0	0.311
					/ 244	0	1.709
					/ 246	0	0.117
					/ 248	0	1.165
			1		/ 250	5	0.350
					/ 252	0	0.058
		254	0.0096		254 / 240	0	0.078
					/ 244	0	0.427
					/ 246	0	0.029
				1	/ 248	0	0.291
				1	/ 250	1	0.087
				1	/ 252	0	0.039
				1	/ 254	0	0.000
		262	0.0289		262 / 240	0	0.233
					/ 244	0	1.282
				1	/ 246	0	0.087
				1	/ 248	3	0.874
				1	/ 250	0	0.262
			T	1	/ 252	0	0.117
			1	1	/ 254	0	0.029
					/ 262	0	0.029
	N 11 (11)	266	0.0096	1	266 / 240	0	0.078
				1	/ 244	1	0.427
			1	1	/ 246	0	0.029
				1	/ 248	0	0.291
	1				/ 250	0	0.087
				1	/ 252	0	0.039
			1	1	/ 254	0	0.010
					/ 262	0	0.029
				1	/ 266	0	0.000
		274	0.0096		274 / 240	0	0.078
					/ 244	1	0.427
				-	/ 246	0	0.029
			1		/ 248	0	0.291
					/ 250	0	0.087
					/ 252	0	0.039
			1		/254	0	0.010
					1262	0	0.029
				-	1262	0	0.029
				-	/ 200	0	0.010
					/ 274	0	0.000

Table V-2Allele and genotype frequency results based on AC01 & AC16microsatellite data from Huánuco stocks* (cont'd)

<u>The Index of Association</u> (I_A) (from Maynard Smith *et al.*, 1993).

n = number of individuals

 p_{ij} = frequency of the *i*th allele at *j*th locus

 $h_j =$ probability that two individuals are different at the *j*th locus



K = "distance" between two individuals (i.e. no. of loci at which they differ) = mean difference between two individuals



 V_E = expected variance of K (assuming no linkage disequilibrium)

V _E =	$\sum h_j$	(1 -	h _j)
-------------------------	------------	------	------------------

V₀ = observed variance of K (estimated, for assumptions made, see Maynard-Smith et al., 1993)



 $I_A =$ Index of association

IA	=	<u>V</u> Q -	1						
		VE							
		Allele frequencies of whole Huanuco population and subgroups therein*						For D' & r ² calculations [†]	
-------	--------	--------------------------------------------------------------------------	------------	------------	----------------	-----------	----------------	------------------------------------------------------	--
Locus	Allele	All (58)	Lb (25)	Lpe (4)	Lb/Lpe (26)	IM (3)	A ₁	A2	
	5	0.026		-	-	0.500			
MPI	6	0.655	0.940	-	0.500	0.500	0.655	0.345	
	7	0.293	-	1.000	0.500	-			
	9	0.026	0.060	-	-	-	1		
NHi1	2	0.698	0.820	-	0.500	1.000	0.698	0.302	
	3	0.302	0.180	1.000	0.500	-	1		
NHd	4	0.776	1.000	1.000	0.500	1.000	0.776	0.224	
	5	0.224	-	-	0.500	-	1		
ES	4	0.810	0.680	1.000	0.981	-	0.810	0.190	
	5	0.190	0.320	-	0.019	1.000	1		
PEPD	2	0.784	0.620	1.000	1.000	-	0.784	0.216	
	4	0.216	0.380	-	-	1.000	1		
PGMa	2	0.054	1.000	1.000	1.000	-	1		
	3	0.946	-	-	-	1.000	1		
CDIA	2	0.052	1.000	1.000	1.000	-	1		
Gri-	4	0.948				1.000	-		
G6PD	2	0.948	1,000	1 000	1.000	1.000	8 not	done	
	3	0.052	-	-	1.000	1 000	y nor	done	
NHi7	7	1,000	1 000	1 000	1 000	1 000	1		
6PCD	5	1.000	1,000	1 000	1,000	1.000	1		
ASAT	5	1.000	1.000	1.000	1,000	1.000	1		
ALAT	4	1.000	1,000	1,000	1,000	1.000	1		
	225	0.019	-	-	1.000	0.500	<u>+</u>	I	
ACOL	227	0.406	0.420		0.458	-	1		
	229	0.038	0.006	-	0.021		1		
	231	0.377	0 340		0.479	-	0.406	0.594	
	233	0.038	0.008		-	-			
	235	0.038		0 500	-	0 500	1	1000	
	237	0.066	0.100	-	0.042	-	1		
	241	0.019	-	0.500	-	-			
	240	0.077	-	-	0 174	-	<u>+</u>	<u> </u>	
AC16	244	0.423	0.229	0.250	0.700		1		
	246	0.029	0.063		-	-	1		
	248	0.289	0.604	-	0.022		1		
	250	0.087	0.021	0.500	0.065	0.500	0.423	0.577	
	252	0.039		-	0.043	0.333	1		
	254	0.010	-			0,167	1		
	262	0.029	0.063	-	-	-	1		
	266	0.010	-	0.250	-		-		
	274	0.010	0.021	0.200	1				

Table V-3 Allele frequencies of whole Huanuco population and of subgroups therein*, by locus.

* - Allele frequencies of whole Huanuco population and subgroups therein (see text).

[†] - For D' & r² calculations: A_1 = frequency of most common allele at locus; A_2 = combined frequencies of all other alleles at locus. § - not done because frequency of the most common allele greater than 0.90 (see text).

Table V-4

${f I}_A$ values calculated from isoenzyme and microsatellite data for the Huánuco stocks ${}^{\pm\dagger}$

		All data		IEA data			Microsatellite data		
	(10 polymorphic loci)		(8 polymorphic loci)			(2 polymorphic loci [†])			
	F		I _A value	I _A values		No.	I _A values		
	No. stocks	& MSTs	All	No. ETs	All	ETs	MSTs	All	MSTs
Population	examined	identified	stocks	identified	stocks	only	identified	stocks	only
All	58	23	0.9670	11*	0.5891	0.9110	16*	2.7663	5.9910
SG1	25	8	1.1210	5a	0.9581	0.7580	7+	2.2460	3.3669
SG2	4	2	1.8662*	1	0*	0	2	1.8662	2.2665
SG3	26	11	1.0437	4a	1.1355	10.7124	7\$	1.7120	2.5192
SG4	3	2	1.3585	1	0*	0	2	1.8326	1.8662

Notes to table V-4: I_A values calculated from IEA and microsatellite (MS) data for the Huanuco stocks. \pm analysis excludes L. V. lainsoni stock LC2679 and data from microsatellite locus $\dagger AC52$ (see text). Population subgroups (SG) identified using IEA: SG1 – L. V. braziliensis; SG2 – L. V. peruviana; SG3 – putative L. V. braziliensis / L. V. peruviana hybrids; SG4 – IM2832-types. ET - electrophoretic type; MST – microsatellite type. \pm Both SG2 and SG4 presented only one ET, the I_A value of which was zero. \pm One stock from both SG1 and SG3 had an identical MST. Values of I_A increased for all stocks and all subgroups when using data from ETs and MSTs only, except for SG1 (L. V. braziliensis) ET data (see text).

<u>Linkage disequilibrium calculation</u> based on maximum-likelihood estimates of gametic frequencies (taken from Hedrick, 1985)

Phenotype	Inferred	Possible	
	genotype	gametic types	
A1B1	A1A1B1B1	A ₁ B ₁	
A1B2	A1A1B2B2	A ₁ B ₂	
A2B1	A ₂ A ₂ B ₁ B ₁	A ₂ B ₁	
A2B2	A2A2B2B2	A ₂ B ₂	
A1A2B1	A1A2B1B1	A1B1 or A2B1	
A1A2B2	A1A2B2B2	A1B2 or A2B2	
A1B1B2	A1A1B1B2	A ₁ B _{1 or} A ₁ B ₂	
A2B1B2	A2A2B1B2	A ₂ B _{1 or} A ₂ B ₂	
A1A2B1B2*	A1A2B1B2	A ₁ B _{1 or A₂B₂}	
		A1B2 or A2B1	
		(repulsion)	

With 2 polymorphic loci (such as enzyme loci), the possible gametes arising from an inferred genotype (consisting of 2 loci, A and B) are shown in the table below:

Conventional analyses can be used in the case of loci with alleles showing dominance because all possible gametic types can be identified. In the case of co-dominant loci however, as here with enzyme data, it is not possible to determine the origins of gametic types in the double heterozygote^{*}. In this case the phenotype could have arisen from union between parental-type ("coupling") or recombinant-type ("repulsion") gametes.

For 2 co-dominant loci, therefore, a maximum-likelihood (ML) approach is taken to circumvent this lack of information regarding gametic types. The following table is completed and then used in subsequent analysis:

	Observe	7		
Observed phenotype at locus B	A 1 A 1	A1A2	A ₂ A ₂	Total
B ₁ B ₁	N ₁₁	N ₁₂	N ₁₃	N1
B ₁ B ₂	N ₂₁	N ₂₂	N ₂₃	N ₂
B ₂ B ₂	N ₃₁	N ₃₂	N ₃₃	N3
Total	N.1	N.2	N.3	N

where: $A_1 = most common allele of locus A$

- $A_2 =$ all other alleles of locus A
- $B_1 = most common allele of locus B$

 $B_2 =$ all other alleles of locus B

 N_{11} = number of isolates with phenotype $A_1A_1B_1B_1$

 N_{12} = number of isolates with phenotype $A_1A_2B_1B_1$ etc.

To calculate the maximum-likelihood estimation of \times_1 (\times_1 is the frequency of the A₁B₁ gamete), an arbitrary starting point is taken, where:

$$\mathbf{x}_{1} = 2N_{11} + \frac{N_{12} + N_{21}}{2(N - N_{22})}$$

This starting figure of \times_1 is then used in the equation below, which is reiterated until the value of \times_1 does not change from the previous iteration:

$\mathbf{x}_1 = 1$	$[2N_{11} + N_{12} + N_{21} +$	$N_{22} \times_1 (1 - p_1 - q_1 + \times_1) $
2N		$\mathbf{x}_{1}(1-\mathbf{p}_{1}-\mathbf{q}_{1}+\mathbf{x}_{1})+(\mathbf{p}_{1}-\mathbf{x}_{1})(\mathbf{q}_{1}-\mathbf{x}_{1})$

The maximum-likelihood estimate of \times_1 is then used to calculate the disequilibrium parameter, where:

 $\mathbf{D} = \mathbf{X}_1 - \mathbf{p}_1 \mathbf{q}_1$

D' is calculated as follows:

where D_{max} is the maximum D possible for a given set of allelic frequencies at the 2 loci. D_{max} is equal either to the lesser of p_1q_2 or p_2q_1 if D is positive, or to the lesser of p_1q_1 or p_2q_2 if D is negative.

A test of significance (Q) can be calculated for each D score from which probability (P) values are assigned:

Q	=	Nr ²
	=	<u>ND²</u>
		P1P2Q1Q2

r² is calculated as follows:

$$r^2 = \frac{D^2}{p_1 p_2 q_1 q_2}$$

2-locus combination	N	D	Q	D'	P value	r ²
MPI ⁶ x NHi1 ²	58	0.136	22.199	0.687	Sig. P < 0.005	0.383
MPI ⁶ x NHd ⁴	58	0.147	31.907	1.000	Sig. P < 0.005	0.548
MPI ⁶ x ES ⁴	58	- 0.038	2.378	- 0.579	Not Sig.	0.041
MPI ⁶ x PEPD ²	58	- 0.048	3.492	- 0.644	Not Sig.	0.060
NHil ² x NHd ⁴	58	0.156	38.512	1.000	Sig. P < 0.005	0.664
NHil ² x ES ⁴	58	- 0.056	5.607	- 0.976	Sig. P < 0.025	0.097
NHil ² x PEPD ²	58	- 0.064	6.655	- 0.981	Sig P < 0.01	0.115
NHd ⁴ x ES ⁴	58	- 0.043	3.927	- 1.000	Sig P < 0.05	0.068
NHd ⁴ x PEPD ²	58	- 0.048	4.613	- 1.000	Sig P < 0.05	0.080
ES ⁴ x PEPD ²	58	0.141	44.244	0.946	Sig. P < 0.005	0.763
AC01227 x MPI6	53	0.127	16.600	1.000	Sig. P < 0.005	0.313
AC01227 x NHi12	53	0.127	16.600	1.000	Sig. P < 0.005	0.313
AC01227 x NHd4	53	0.101	12.761	1.000	Sig. P < 0.005	0.241
AC01227 x ES4	53	0.083	9.805	1.000	Sig. P < 0.005	0.185
AC01 ²²⁷ x PEPD ²	53	0.089	10.269	1.000	Sig. P < 0.005	0.194
AC16244 x MPI6	52	- 0.061	3.662	- 0.333	Not Sig.	0.070
AC16244 x NHi12	52	- 0.087	7.566	- 0.489	Sig. P < 0.01	0.146
AC16244 x NHd4	52	- 0.074	6.685	- 0.576	Sig. P < 0.01	0.129
AC16244 x ES4	52	0.075	7.514	0.882	Sig P < 0.01	0.145
AC16244 x PEPD2	52	0.079	7.431	0.805	Sig P < 0.01	0.143
AC01227 x AC16244	51	0.044	1.652	0.191	Not Sig.	0.032

 Table V-5
 Linkage disequilibrium parameter results (complete).

N = number of stocks analyzed in calculation

